UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of The Securities and Exchange Act of 1934 Date of Report (Date of earliest event reported): November 30, 2017

HC2 HOLDINGS, INC.

Delaware (State or other jurisdiction of incorporation) 001-35210

54-1708481 (IRS Employer Identification No.)

(Commission File Number)

450 Park Avenue, 30th Floor New York, NY 10022

(Address of principal executive offices)

(212) 235-2690

(Registrant's telephone number, including area code)

Not Applicable

(Former name or former address, if changed since last report.)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

Derecommencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

Dere-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Indicate by check mark whether the registrant is an emerging growth company as defined in as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Item 2.01 Completion of Acquisition or Disposition of Assets

On November 30, 2017, Global Marine Systems Limited ("<u>GMSL</u>"), an indirect subsidiary of HC2 Holdings, Inc. (the "<u>Company</u>"), consummated the previously announced transactions contemplated by a Business Purchase Agreement (the "<u>BPA</u>") and a Warranty and Indemnity Agreement (the "<u>W&I Agreement</u>"), in each case by and among Fugro N.V., a public limited liability company incorporated in the Netherlands ("<u>Fugro</u>"), GMSL and Global Marine Holdings LLC, an indirect subsidiary of the Company and an indirect parent company of GMSL ("<u>GMHL</u>"). Pursuant to the BPA, GMSL acquired the trenching and cable laying business of Fugro (the "<u>Acquisition</u>"), consisting of, among other things, 19 employees, one vessel, two trenching systems, two work class remotely operated vehicles, and working capital.

As consideration for the Acquisition, (a) GMSL paid \$7,500,000 (the "<u>Cash Consideration</u>") to Fugro for a Q1400 Trenching System (the "<u>Trencher</u>"), and (b) GMHL issued to a subsidiary of Fugro (the "<u>Fugro Member</u>") membership units representing a 23.6% equity interest in GMHL (excluding management incentive units), valued at \$65,000,000. The limited liability company agreement of GMHL was amended and restated (the "<u>Amended and Restated GMHL LLC Agreement</u>") upon consummation of the Acquisition to reflect such issuance and to provide the Fugro Member with certain rights, including the right to designate two out of the up to seven members of its board of directors, the right to approve certain actions outside the ordinary course of business, certain "tag-along" rights to participate in sales of membership units by other members and, after five years and subject to the Fugro Member first offering its membership units to the other members at a price based upon independent valuations, the right to cause GMHL to be put up for sale in a process led by an investment banking firm.

In order to finance the Cash Consideration, GMSL incurred a loan of \$7,500,000 from a subsidiary of Fugro pursuant to a Vendor Loan Agreement, dated as of November 30, 2017, by and between Fugro Financial Resources B.V. and GMSL (the "<u>Vendor Loan Agreement</u>"). The loan bears interest, payable quarterly, at 4% per annum through December 31, 2017, and at 10% per annum thereafter, and matures 363 days following the Acquisition. The Trencher will serve as collateral security for the repayment of the loan pursuant to the terms of a lien agreement.

Pursuant to the W&I Agreement, each party has certain rights to damages or indemnification in connection with breaches of certain representations, warranties and covenants given by the other party. The W&I Agreement includes, among other things, thresholds establishing the minimum amount of losses that must be incurred before damages may be claimed and caps with respect to the maximum amount of each party's payment obligations.

In order to provide for transitional and general services between the parties after the Acquisition, Fugro and GMSL entered into a Transitional Services and Framework Services Agreement (the "<u>FSA</u>") upon consummation of the Acquisition pursuant to which, among other things, (a) GMSL will use Fugro as a preferred provider for a range of survey, marine site investigation, asset integrity, ROV (operation) and other services required by GMSL in relation to the Acquisition, (b) Fugro and GMSL will cooperate on future work prospects in the "non-trenching", "non-cable laying" market in the event that Fugro requires additional vessel capacity, and (c) Fugro will provide certain transitional services to GMSL for a period of time following the Acquisition.

The foregoing information is a summary of the agreements described above (collectively, the "<u>Transaction Agreements</u>") and, as such, is not complete, and is qualified in its entirety by reference to the full text of the Transaction Agreements. Copies of the BPA and W&I Agreement were filed as exhibits to the Company's Current Report on Form 8-K filed with the Securities and Exchange Commission on October 12, 2017. Copies of the Amended and Restated GMHL LLC Agreement, the Vendor Loan Agreement and the FSA are filed as exhibits to this Current Report on Form 8-K. Investors should review the Transaction Agreements for a complete understanding of the terms and conditions associated with this transaction.

The Transaction Agreements have been included to provide investors with information regarding their terms. They are not intended to provide any other factual information about the parties thereto or their respective subsidiaries or affiliates. The representations, warranties, and covenants contained in each Transaction Agreement were made only for purposes of such agreement and as of specific dates, were solely for the benefit of the parties thereto, may be subject to limitations agreed upon by the contracting parties, including being qualified by confidential disclosures made for the purposes of allocating contractual risk between the parties thereto instead of establishing these matters as facts, and may be subject to standards of materiality applicable to the contracting parties that differ from those applicable to investors. Investors are not third-party beneficiaries under any Transaction Agreement and should not rely on the representations, warranties, and covenants or any descriptions thereof as characterizations of the actual state of facts or condition of the parties thereto or any of their respective subsidiaries or affiliates. Information concerning the subject matter of representations and warranties may change after the date of the Transaction Agreements, which subsequent information may or may not be fully reflected in the Company's public disclosures.

Item 9.01 Financial Statements and Exhibits

(a) Financial Statement of Business Acquired

The Company intends to file the historical financial statements of Fugro for the periods specified in Rule 3-05(b) of Regulation S-X in an amendment to this report under cover of Form 8-K/A no later than 71 calendar days after the date this Current Report on Form 8-K was required to be filed.

(b) Pro Forma Financial Information

The Company intends to furnish pro forma financial information relating to the Acquisition required pursuant to Article 11 of Regulation S-X in an amendment to this report under cover of Form 8-K/A no later than 71 calendar days after the date this Current Report on Form 8-K was required to be filed.

(d) Exhibits

Exhibit No.

- 2.1 Fourth Amended and Restated Limited Liability Company Agreement of Global Marine Holdings, LLC, dated as of November 30, 2017, by and among Global Marine Holdings, LLC and the Members party thereto.*
- 2.2 Vendor Loan Agreement, dated as of November 30, 2017, by and between Fugro Financial Resources B.V. and Global Marine Systems Limited.*
- 2.3 Transitional Services and Framework Services Agreement, dated as of November 30, 2017, by and between Fugro N.V. and Global Marine Systems Limited.*
- * Certain schedules and exhibits to this agreement have been omitted pursuant to Item 601(b)(2) of Regulation S-K and the Company agrees to furnish supplementally to the Securities and Exchange Commission a copy of any omitted schedule and/or exhibit upon request.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: November 30, 2017

HC2 Holdings, Inc. (Registrant)

By: /s/ Michael J. Sena

Name: Michael J. Sena Title: Chief Financial Officer

FOURTH AMENDED AND RESTATED LIMITED LIABILITY COMPANY AGREEMENT

OF

GLOBAL MARINE HOLDINGS, LLC, A DELAWARE LIMITED LIABILITY COMPANY

Dated as of November 30, 2017

THE TRANSFER OF THE SECURITIES REPRESENTED BY THIS INSTRUMENT IS SUBJECT TO THE CONDITIONS SPECIFIED IN THIS LIMITED LIABILITY COMPANY AGREEMENT AMONG THE MEMBERS OF THE ISSUER.

THE SECURITIES REPRESENTED BY THIS INSTRUMENT HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND MAY NOT BE SOLD OR TRANSFERRED IN ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT UNDER THE ACT OR AN EXEMPTION FROM REGISTRATION THEREUNDER. THE SECURITIES REPRESENTED BY THIS INSTRUMENT ARE SUBJECT TO A SECURITYHOLDERS AGREEMENT, AS MAY BE AMENDED FROM TIME TO TIME, BY AND AMONG THE ISSUER AND CERTAIN OF THE ISSUER'S EQUITYHOLDERS. A COPY OF SUCH SECURITYHOLDERS AGREEMENT WILL BE FURNISHED WITHOUT CHARGE BY THE ISSUER TO THE HOLDER OF SECURITIES REPRESENTED BY THIS INSTRUMENT UPON WRITTEN REQUEST.

i

TABLE OF CONTENTS

38

ARTICLE	I Definitions	7
1.1 Definitions		
1.2	Other Definitional Provisions	22
ARTICLE	II Organization of the Company	23
2.1	Formation	23
2.2	Name	23
2.3	Registered Office; Agent	23
2.4	Term	23
2.5	Purposes and Powers	23
ARTICLE	III Management of the Company	24
3.1	Board of Directors	24
3.2	Committees of the Board	28
3.3	Officers	28
3.4	Fiduciary Duties	30
3.5	Performance of Duties; Liability of Directors and Officers	30
3.6	Indemnification	31
3.7	Code of Business Conduct.	31
3.8	Annual Business Plan	32
ARTICLE	IV Members; Voting Rights	32
4.1	Meetings of Members	32
4.2	Voting Rights	33
4.3	Registered Members	33
4.4	Consent of Spouse	33
4.5	Limitation of Liability	33
4.6	Withdrawal; Resignation	34
4.7	Death of a Member	34
4.8	Authority	34
4.9	Outside Activities	34
4.1	Actions Requiring Member Approval	35
4.11	Preemptive Rights	35
ARTICLE	V Units; Membership	36
5.1	Units Generally	36
5.2	Incentive Interests	37

5.3 Authorization and Issuance of Units

ii

		<u>Page</u>
5.4	Issuance of Units	39
5.5	New Members from the Issuance of Units	39
5.6	Adjustment to Initial Fugro Units	39
ARTICLE	VI Capital Contributions and Capital Accounts	39
6.1	Capital Contributions	39
6.2	Capital Accounts	40
6.3	Negative Capital Accounts	41
6.4	No Withdrawal	41
6.5	Loans From Members	41
6.6	Status of Capital Contributions	41
ARTICLE	VII Distributions	42
7.1	Generally	42
7.2	Discretionary Distributions	42
7.3	Tax Advances	43
7.4	Indemnification and Reimbursement for Payments on Behalf of a Member	43
7.5	Class A-2 Adjustments	44
ARTICLE	VIII Allocations	45
8.1	Allocations of Profits and Losses	45
8.2	Regulatory and Special Allocations	45
8.3	Curative Allocations	46
8.4	Tax Allocations	47
ARTICLE	IX Elections and Reports	47
9.1	Generally	47
9.2	Tax Status	47
9.3	Reports	48
9.4	Tax Elections	48
9.5	Tax Controversies	48
9.6	Code Section 83 Safe Harbor Election	48
9.6	Engagement in Trade or Business in the United States	49
ARTICLE	X Dissolution and Liquidation	50
10.1	Dissolution	50
10.2	Liquidation	50
ARTICLE	XI Transfer of Units	52
11.1	Restrictions on Transfer	52
11.2	Effect of Transfer	55

iii

		Page
11.3	Procedures for Transfer	55
11.4	Tag-Along Rights	56
11.5	Legend	59
11.6	Company Sale	59
11.7	Right of First Offer	62
11.8	Fugro Sale Process	63
ARTICLE >	XII Information Rights	67
12.1	Financial Statements	67
12.2	Confidential Nature of Information	68
12.3	Waiver of Section 18-305 of the Delaware Act	68
ARTICLE >	XIII Registration Rights	68
13.1	Demand Registrations of Registrable Securities	68
13.2	Piggyback Registration	70
13.3	Holdback Agreements	71
13.4	Registration Procedures	72
13.5	Shelf Registration	76
13.6	Registration Expenses	77
13.7	Indemnification; Contribution	77
13.8	Rules 144 and 144A	81
13.9	Underwritten Offerings	81
13.1	No Inconsistent Agreements	81
13.11	Initial Public Offering	82
13.12	Mergers and Consolidations	82
ARTICLE >	XIV Miscellaneous Provisions	82
14.1	Notices	82
14.2	No Action for Partition	83
14.3	Headings and Sections	83
14.4	Amendments	83
14.5	Binding Effect	85
14.6	Counterparts; Facsimile	85
14.7	Severability	85
14.8	Remedies	85
14.9	Business Days	85
14.1	Waiver of Jury Trial	85
14.11	Construction	86
14.12	Entire Agreement and Incorporation by Reference	86
14.13	Conflicting Agreements	87
14.14	GOVERNING LAW	87

iv

EXHIBITS:

Exhibit A	Form of Joinder to Limited Liability Company Agreement
Exhibit B	Form of Consent of Spouse

SCHEDULES:

Schedule A	Members Schedule
Schedule B	Business and Business Strategy of the Company

Schedule C Code of Conduct

FOURTH AMENDED AND RESTATED LIMITED LIABILITY COMPANY AGREEMENT OF GLOBAL MARINE HOLDINGS, LLC

This FOURTH AMENDED AND RESTATED LIMITED LIABILITY COMPANY AGREEMENT (this "<u>Agreement</u>"), dated as of November 30, 2017, of GLOBAL MARINE HOLDINGS, LLC, a Delaware limited liability company (the "<u>Company</u>"), is made by and among the Members (as herein defined) and the Company. Capitalized terms used herein and not otherwise defined shall have the meanings set forth in <u>Section 1.1</u>.

RECITALS

A. The Company has been formed as a limited liability company pursuant to Section 214 of the Delaware Act by filing a Certificate of Formation of the Company (the "<u>Certificate of Formation</u>") with the Office of the Secretary of State of the State of Delaware on September 12, 2014.

B. The HC2 Member entered into a Limited Liability Company Agreement of the Company dated September 12, 2014 (the "<u>Original</u> <u>Operating Agreement</u>").

C. The HC2 Member, Zencor Holdings and Commando Partners LLC entered into an Amended and Restated Limited Liability Company Agreement of the Company dated December 3, 2014 (the "<u>Amended and Restated Operating Agreement</u>").

D. The HC2 Member, Zencor Holdings, Commando Partners LLC and certain Incentive Members entered into a Second Amended and Restated Limited Liability Company Agreement of the Company dated April 27, 2015 (the "Second Amended and Restated Operating Agreement").

E. The HC2 Member, Zencor Holdings, Commando Partners LLC and certain Incentive Members entered into a Third Amended and Restated Limited Liability Company Agreement of the Company dated June 9, 2016 (the "<u>Third Amended and Restated Operating Agreement</u>").

F. In connection with the consummation of the transactions contemplated by that certain Business Purchase Agreement (the "<u>Business</u> <u>Purchase Agreement</u>"), dated as of October 11, 2017, by and among the Fugro Member, GMSL and the Company, the Members desire to amend and restate the Third Amended and Restated Operating Agreement to, among other things, reflect the issuance of 43,882,283 Class A-2 Units (representing 23.6% of the aggregate number of outstanding Class A Units, Class A-1 Units and Class A-2 Units) to the Fugro Member (such Class A-2 Units, the "<u>Initial Fugro Units</u>"), establish the rights and preferences of the Members following such consummation and provide for the governance of the Company.

NOW, THEREFORE, in consideration of the mutual covenants and agreements herein made and other good and valuable consideration, the Members, intending to be legally bound, hereby agree as follows:

#4837-3810-7979

ARTICLE I Definitions

1.1 Definitions. The following terms used in this Agreement shall have the following meanings (unless otherwise expressly provided in this Agreement):"<u>3% Holders</u>" has the meaning set forth in <u>Section 13.3(b)</u>.

"<u>Adjusted Capital Account Deficit</u>" means, with respect to any Member, the deficit balance, if any, in such Member's Capital Account as of the end of the relevant Taxable Year, after giving effect to the following adjustments:

(i) crediting to such Capital Account any amount which such Member is obligated to restore or is deemed to be obligated to restore pursuant to Treasury Regulation Sections 1.704-1(b)(2)(ii)(c), 1.704-2(g)(1), and 1.704-2(i); and

(ii) debiting to such Capital Account the items described in Treasury Regulation Section 1.704-1(b)(2)(ii)(d)(4), (5) and (6).

"<u>Adjusted Initial Fugro Units</u>" means a number of Class A-2 Units equal to (a) the number of Initial Class A/A-1 Units multiplied by (b) a fraction, the numerator of which is the Fugro BPA Percentage as of the date of determination and the denominator of which is the GMSL BPA Percentage as of the date of determination.

"<u>Adjustment Decrease Amount</u>" of each Member, means, upon the occurrence of any Class A-2 Adjustment, an amount equal to the excess, if any, of (a) the aggregate amount of distributions actually received by such Member pursuant to <u>Section 7.2(a)</u> or (b) after the Fugro Acquisition Date, over (b) the amount of distributions that such Member would have been entitled to receive pursuant to <u>Section 7.2(a)</u> or (b) if the Class A-2 Distribution Percentage and the Class A/A-1 Distribution Percentage in effect immediately following such Class A-2 Adjustment had been in effect on the Fugro Acquisition Date.

"<u>Adjustment Increase Amount</u>" of each Member, means, upon the occurrence of any Class A-2 Adjustment, an amount equal to the excess, if any, of (a) the amount of distributions that such Member would have been entitled to receive pursuant to <u>Section 7.2(a)</u> or (b) if the Class A-2 Distribution Percentage and the Class A/A-1 Distribution Percentage in effect immediately following such Class A-2 Adjustment had been in effect on the Fugro Acquisition Date, over (b) the aggregate amount of distributions actually received by such Member pursuant to <u>Section 7.2(a)</u> or (b) after the Fugro Acquisition Date.

"<u>Affiliate</u>" means, when used with reference to a specified Person, any Person that directly or indirectly controls or is controlled by or is under common control with the specified Person. As used in this definition, "control" (including, with its correlative meanings, "controlled by" and "under common control with") shall mean possession, directly or indirectly, of power to direct or cause the direction of management or policies (whether through ownership of securities or partnership or other ownership interests, by contract or otherwise). With respect to any Person who is an individual, "Affiliates" shall also include, without limitation, any member of such individual's Family Group. Notwithstanding the forgoing, in no event shall any Member or any of

its Affiliates be deemed to be an Affiliate of any other Member or any of its Affiliates (other than the Company) solely by reason of such Member's control of the Company.

"<u>Agreement</u>" has the meaning set forth in the preamble.

"Aggregate Adjustment Payment" has the meaning set forth in Section 7.5.

"Amended and Restated Operating Agreement" has the meaning set forth in the Recitals.

"Amended Sale Notice" has the meaning set forth in Section 11.4(a)(ii).

"Annual Business Plan" has the meaning set forth in Section 3.8.

"<u>Available Cash</u>" at the time of any distribution, means all cash then held by the Company to the extent not otherwise required to pay Company (or its Subsidiaries') expenses.

"Average Price" has the meaning set forth in Section 11.8(b).

"Bankruptcy" means, with respect to any Person, that (i) such Person has (A) made an assignment for the benefit of creditors; (B) filed a voluntary petition in bankruptcy; (C) been adjudged bankrupt or insolvent, or had entered against such Person an order of relief in any bankruptcy or insolvency proceeding; (D) filed a petition or an answer seeking for such Person any reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under any statute, law or regulation or filed an answer or other pleading admitting or failing to contest the material allegations of a petition filed against such Person in any proceeding of such nature; or (E) sought, consented to, or acquiesced in the appointment of a trustee, receiver or liquidator of such Person or of all or any substantial part of such Person's properties; (ii) one hundred and twenty (120) days have elapsed after the commencement of any proceeding against such Person seeking reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under any statute, law or regulation and such proceeding has not been dismissed; or (iii) ninety (90) days have elapsed since the appointment without such Person's consent or acquiescence of a trustee, receiver or liquidator of such Person or of all or any substantial part of such Person's properties and such appointment has not been vacated or stayed or the appointment is not vacated within ninety (90) days after the expiration of such stay.

"Board" has the meaning set forth in Section 3.1(a).

follows:

"Book Value" means, with respect to any Company asset, the adjusted basis of such asset for federal income tax purposes, except as

(A) The initial Book Value of any Company asset contributed (or deemed contributed) by a Member to the Company shall be the gross Fair Market Value of such Company asset as of the date of such contribution or deemed contribution;

(B) The Book Value of each Company asset shall be adjusted to equal its respective gross Fair Market Value, as of the following times: (i) the acquisition of an additional interest in the Company by any new or existing Member in exchange for more

than a *de minimis* Capital Contribution; (ii) the distribution by the Company to a Member of more than a *de minimis* amount of Company assets (other than cash) as consideration for all or part of its Units unless the Board determines that such adjustment is not necessary to reflect the relative economic interests of the Members in the Company; (iii) the liquidation of the Company within the meaning of Treasury Regulations Section 1.704-1(b)(2)(ii)(g); (iv) the acquisition of an additional interest in the Company by any new existing Member in exchange for the provisions of services to or for the benefit of the Company; and (v) in connection with the issuance by the Company of a noncompensatory option (other than an option to acquire *de minimis* interest in the Company);

(C) The Book Value of a Company asset distributed to any Member shall be the Fair Market Value of such Company asset as of the date of distribution thereof;

(D) The Book Value of each Company asset shall be increased or decreased, as the case may be, to reflect any adjustments to the adjusted basis of such Company asset pursuant to Section 734(b) or Section 743(b) of the Code, but only to the extent that such adjustments are taken into account in determining Capital Account balances pursuant to Treasury Regulations Section § 1.704-1(b)(2)(iv)(m); provided, that Book Values shall not be adjusted pursuant to this subparagraph (D) to the extent that an adjustment pursuant to subparagraph (B) above is made in conjunction with a transaction that would otherwise result in an adjustment pursuant to this subparagraph (D); and

(E) If the Book Value of a Company asset has been determined or adjusted pursuant to subparagraphs (A), (B) or (D) above, such Book Value shall thereafter be adjusted to reflect the Depreciation taken into account with respect to such Company asset for purposes of computing Profits and Losses.

"Business Day" means any day other than a Saturday, Sunday or other day on which commercial banks in New York, New York are authorized or required to close.

"Business Purchase Agreement" has the meaning set forth in the Recitals.

"Capital Account" has the meaning set forth in Section 6.2(a).

"<u>Capital Contribution</u>" means any contribution to the capital of the Company in cash or property by a Member or, in the case of a Class A-1 Member, the dollar amount allocated to the Class A-1 Units held by such Class A-1 Member in the applicable Grant Agreement with respect to the Class A-1 Units awarded thereunder, whenever made or deemed made.

"<u>Certificate</u>" means the Certificate of Formation, as such Certificate of Formation may be amended, supplemented or restated from time to time.

"<u>Certificate of Formation</u>" has the meaning set forth in the preamble.

"Class A Member" means a Member holding Class A Units.

"<u>Class A Percentage Interest</u>" of each Member means a fraction (expressed as a percentage), the numerator of which is the aggregate Class A Units held by such Member and the denominator of which is the aggregate Class A Units held by all Members.

"Class A Unit" means a Unit having the rights and obligations specified with respect to a "Class A Unit" in this Agreement.

"Class A-1 Member" means a Member holding Class A-1 Units.

"Class A-1 Unit" means a Unit having the rights and obligations specified with respect to a "Class A-1 Unit" in this Agreement.

"<u>Class A-2 Adjustment</u>" has the meaning set forth in <u>Section 5.6</u>.

"Class A-2 Member" means a Member holding Class A-2 Units.

"<u>Class A-2 Distribution Percentage</u>" means a fraction (expressed as a percentage), the numerator of which is the aggregate Class A-2 Units held by all Members and the denominator of which is the aggregate Class A Units, Class A-1 Units and Class A-2 Units held by all Members.

"<u>Class A-2 Percentage Interest</u>" of each Member means a fraction (expressed as a percentage), the numerator of which is the aggregate Class A-2 Units held by such Member and the denominator of which is the aggregate Class A-2 Units held by all Members.

"Class A-2 Unit" means a Unit having the rights and obligations specified with respect to a "Class A-2 Unit" in this Agreement.

"<u>Class A/A-1 Distribution Percentage</u>" means a fraction (expressed as a percentage), the numerator of which is the aggregate Class A Units and Class A-1 Units held by all Members and the denominator of which is the aggregate Class A Units, Class A-1 Units and Class A-2 Units held by all Members.

"<u>Class A/A-1/A-2 Percentage Interest</u>" of each Member means a fraction (expressed as a percentage), the numerator of which is the aggregate Class A Units, Class A-1 Units and Class A-2 Units held by such Member and the denominator of which is the aggregate Class A Units, Class A-1 Units and Class A-2 Units held by all Members.

"<u>Class A/A-1/A-2/B Percentage Interest</u>" of each Member means a fraction (expressed as a percentage), the numerator of which is the aggregate Class A Units, Class A-1 Units, Class A-2 Units and Class B Units held by such Member and the denominator of which is the aggregate Class A Units, Class A-1 Units, Class A-2 Units and Class B Units held by all Members.

"<u>Class A/A-1/A-2/B/C Percentage Interest</u>" of each Member means a fraction (expressed as a percentage), the numerator of which is the aggregate Class A Units, Class A-1 Units, Class A-2 Units, Class B Units and Class C Units held by such Member and the denominator of which is the aggregate Class A Units, Class A-1 Units, Class A-2 Units, Class B Units and Class C Units held by all Members.

"<u>Class A/A-1/A-2/B/C/D Percentage Interest</u>" of each Member means a fraction (expressed as a percentage), the numerator of which is the aggregate Class A Units, Class A-1 Units, Class A-2 Units, Class B Units, Class C Units and Class D Units held by such Member and the denominator of which is the aggregate Class A Units, Class A-1 Units, Class A-2 Units, Class A-2 Units, Class B Units, Class C Units and Class D Units held by all Members.

"Class B Member" means a Member holding Class B Units.

"<u>Class B Threshold Percentage</u>" of each Class B Member means a fraction (expressed as a percentage), the numerator of which is the aggregate Class B Units held by such Class B Member and the denominator of which is the aggregate Class A Units, Class A-1 Units, Class A-2 Units and Class B Units held by all Members.

"Class B Unit" means a Unit having the rights and obligations specified with respect to a "Class B Unit" in this Agreement.

"Class C Member" means a Member holding Class C Units.

"<u>Class C Threshold Percentage</u>" of each Class C Member means a fraction (expressed as a percentage), the numerator of which is the aggregate Class C Units held by such Class C Member and the denominator of which is the aggregate Class A Units, Class A-1 Units, Class A-2 Units, Class B Units and Class C Units held by all Members.

"Class C Unit" means a Unit having the rights and obligations specified with respect to a "Class C Unit" in this Agreement.

"Class D Member" means a Member holding Class D Units.

"<u>Class D Threshold Percentage</u>" of each Class D Member means a fraction (expressed as a percentage), the numerator of which is the aggregate Class D Units held by such Class D Member and the denominator of which is the aggregate Class A Units, Class A-1 Units, Class A-2 Units, Class B Units, Class C Units and Class D Units held by all Members.

"Class D Unit" means a Unit having the rights and obligations specified with respect to a "Class D Unit" in this Agreement.

"Code" means the United States Internal Revenue Code of 1986, as amended from time to time.

"<u>Code of Conduct</u>" has the meaning set forth in <u>Section 3.7(a)</u>.

"Common Stock" means the common stock or other equity securities of Newco.

"Company" has the meaning set forth in the preamble.

"<u>Company Associated Person</u>" means Zencor Holdings, Zencor Management, Dick Fagerstal or any employee of the Company or any of its Subsidiaries.

"Company Minimum Gain" has the meaning set forth for "partnership minimum gain" in Treasury Regulation Section 1.704-2(d).

"<u>Company Sale</u>" means the consummation of a transaction, whether in a single transaction or in a series of related transactions, with any Independent Third Party or group of Independent Third Parties pursuant to which such Person or Persons acquire (whether by merger, consolidation, recapitalization, reorganization, redemption, transfer or issuance of equity securities or otherwise), directly or indirectly, (a) a majority of the Voting Units then outstanding or a number of Units of the Company (or any surviving or resulting Person) possessing the voting power to elect a majority of the board of directors (or governing body of such surviving or resulting Person), or (b) assets constituting all or substantially all of the assets of the Company and its Subsidiaries (as determined on a consolidated basis); <u>provided</u> that in no event shall a Company Sale be deemed to include any transaction effected for the purpose of (i) changing, directly or indirectly, the form of organization or the organizational structure of the Company or any of its Subsidiaries or (ii) contributing equity securities to entities controlled by the Company (to the extent otherwise permitted hereunder), in either case unless following or contemporaneously with such transaction an Independent Third Party or group of Independent Third Parties makes an acquisition described in <u>clause (a) or (b)</u> above.

"Consent of Spouse" has the meaning set forth in Section 4.4.

"Covered Person" has the meaning set forth in Section 13.7(a).

"<u>Deemed Fugro Value</u>" at any time of determination means an amount equal to (a) \$65,000,000 minus (b) the aggregate amount of any Fugro Adjustment Claims settled on or prior to such date.

"<u>Deemed GMSL Value</u>" at any time of determination means an amount equal to (a) \$210,000,000 minus (b) the aggregate amount of any GMSL Adjustment Claims settled on or prior to such date.

"<u>Deemed Total Value</u>" at any time of determination means an amount equal to the sum of (a) the Deemed GMSL Value and (b) the Deemed Fugro Value.

"<u>Delaware Act</u>" means the Delaware Limited Liability Company Act, Delaware Code, Title 6, Sections 18-101, et seq., as amended from time to time.

"Demand Registrations" has the meaning set forth in Section 13.1(a).

"Depreciation" means, for each Fiscal Year, an amount equal to the depreciation, amortization, or other cost recovery deduction allowable with respect to an asset for such Fiscal Year, except that if the Book Value of an asset differs from its adjusted basis for federal income tax purposes at the beginning of such Fiscal Year, Depreciation shall be an amount which bears the same ratio to such beginning Book Value as the federal income tax depreciation, amortization, or other cost recovery deduction for such Fiscal Year bears to such beginning adjusted tax basis; provided, that if the adjusted basis for federal income tax purposes of an asset at the beginning of such Fiscal Year is zero and the Book Value of the asset is positive, Depreciation shall be determined with reference to such beginning Book Value using any permitted method selected by the Board.

"Directors" has the meaning set forth in Section 3.1(a).

"Drag Sale" has the meaning set forth in Section 11.8(c).

"<u>EBITDA</u>" means, for any period, the net income of the Company and its Subsidiaries, determined on a consolidated basis in accordance with GAAP, plus, the sum, without duplication, of amounts for (a) consolidated interest expense, (b) provisions for taxes based on income, (c) total depreciation expense, (d) total amortization expense, and (e) other non-cash charges reducing such net income.

"Electing Majority Members" has the meaning set forth in Section 11.6(a).

"Electing Person" has the meaning set forth in Section 3.1(e)(i).

"Exchange Act" has the meaning set forth in Section 13.4(b).

"<u>Fair Market Value</u>" of any asset as of any date means the purchase price which a willing buyer having all relevant knowledge would pay a willing seller for such asset in an arm's-length transaction. Except as otherwise set forth in this Agreement (including, without limitation, <u>Section 11.8</u>), the Fair Market Value of any asset shall be determined in good faith by the Board based on such factors as the Board, in the exercise of their reasonable business judgment, considers relevant.

"<u>Family Group</u>" means, with respect to any Person who is an individual, (i) such Person's spouse, former spouse, ancestors and descendants (whether natural or adopted), parents, the lineal descendants of such Person's parents (whether natural or adopted) and any spouse of the foregoing persons (collectively, "<u>relatives</u>"), (ii) the trustee, fiduciary or personal representative of such Person and any trust solely for the benefit of such Person and/or such Person's relatives or (iii) any limited partnership, limited liability company or corporation the governing instruments of which provide that such Person shall have the exclusive, nontransferable power to direct the management and policies of such entity and of which the sole owners of partnership interests, membership interests or any other equity interests are, and will remain, limited to such Person and such Person's relatives.

"<u>First Return Threshold Amount</u>" at any time of determination means the greater of (a) the excess of (i) (x) two multiplied by (y) the aggregate Capital Contributions of the Class A Members over (ii) the aggregate amount previously distributed to the Class A Members pursuant to <u>Section 7.2</u> and (b) the amount which, when taken together with all other distributions to the Class A Members, would result in an internal rate of return for the Class A Members, taken as a group (calculated taking account of the actual timing of Capital Contributions and distributions pursuant to <u>Section 7.2</u> and using the xIRR function on Microsoft Excel) equal to 15% if such amount were distributed to the Class A Members at such time.

"Fiscal Year" means the Company's Taxable Year.

"Fugro Acquisition Date" means the date of the Fourth Amended and Restated Limited Liability Company Agreement of the

Company.

"Fugro Adjustment Claim" means any claim by GMSL for indemnification under the Business Purchase Agreement or the Warranty and Indemnity Agreement which has been finally determined in favor of GMSL in accordance with the terms of the applicable agreement and which the Fugro Member has elected to settle by an adjustment to the number of Class A-2 Units issued to the Fugro Member pursuant to the Business Purchase Agreement.

"Fugro Directors" has the meaning set forth in Section 3.1(d)(i).

"<u>Fugro BPA Percentage</u>" means a fraction (expressed as a percentage), the numerator of which is the Deemed Fugro Value and the denominator of which is the Deemed Total Value.

"Fugro Interest" has the meaning set forth in Section 11.8(b).

"<u>Fugro Member</u>" means Fugro Consultants International N.V., a public limited liability corporation organized under the laws of Curacao tax resident in the Netherlands.

"Fugro Member Parent" means Fugro N.V., a Dutch public company.

"Fugro Member Sale" has the meaning set forth in Section 11.8(b).

"Fugro Offer Notice" has the meaning set forth in Section 11.8(b).

"<u>Fugro Offer Period</u>" has the meaning set forth in <u>Section 11.8(b)</u>.

"Fugro Offer Price" has the meaning set forth in Section 11.8(b).

"<u>Fugro Offerees</u>" has the meaning set forth in <u>Section 11.8(b)</u>.

"<u>Fugro Sale Process</u>" has the meaning set forth in <u>Section 11.8(a)</u>.

"GAAP" means generally accepted accounting principles in the United States of America.

"GCL" means the General Corporation Law of the State of Delaware, as the same may be amended from time to time.

"GMHL" shall mean Global Marine Holdings Limited, a company incorporated in England with registered number 04352407.

"GMSL" means Global Marine Systems Limited, a company incorporated in England with registered number 01708481.

"<u>GMSL Adjustment Claim</u>" means any claim by the Fugro Member for indemnification under the Business Purchase Agreement or the Warranty and Indemnity Agreement which has been finally determined in favor of the Fugro Member in accordance with the terms of

the applicable agreement and which GMSL has elected to settle by an adjustment to the number of Class A-2 Units issued to the Fugro Member pursuant to the Business Purchase Agreement.

"<u>GMSL BPA Percentage</u>" means a fraction (expressed as a percentage), the numerator of which is the Deemed GMSL Value and the denominator of which is the Deemed Total Value.

"<u>Grant Agreement</u>" means any agreement between the Company, on the one hand, and any of its or any of its subsidiaries' officers, directors, consultants or employees, on the other hand, pursuant to which the Company grants such individuals Class A-1 Units, Class B Units, Class C Units and/or Class D Units.

"HC2 Member" means HC2 Holdings 2, Inc., a Delaware corporation.

"<u>HC2 Member Parent</u>" means HC2 Holdings, Inc., a Delaware corporation.

"Incentive Interest" has the meaning set forth in Section 5.3(i).

"Incentive Member" means each Member holding Incentive Interests.

"Indemnified Party" has the meaning set forth in Section 13.7(c).

"Indemnifying Member" has the meaning set forth in Section 7.4.

"Indemnifying Parties" has the meaning set forth in Section 13.7(c).

"Independent Investment Bank" has the meaning set forth in Section 11.8(c).

"<u>Independent Third Party</u>" means any Person who, immediately prior to the contemplated transaction, does not own, directly or indirectly, in excess of five percent (5%) of the total Membership Interests of the Company on a fully-diluted basis (a "<u>5% Owner</u>"), who is not controlling, controlled by or under common control with, or is otherwise an Affiliate of, any such 5% Owner and who is not the spouse or descendent (by birth or adoption) of any such 5% Owner or a trust for the benefit of such 5% Owner or such other Persons.

"Initial Class A/A-1 Units" means the aggregate number of outstanding Class A Units and Class A-1 Units as of the Fugro Acquisition

Date.

"Initial Fugro Units" has the meaning set forth in the Recitals.

"Initial Public Offering" has the meaning set forth in Section 13.11(a).

"Joinder" has the meaning set forth in Section 4.7.

"<u>Leverage Ratio</u>" means the ratio as of the last day of any fiscal quarter of (a) the aggregate stated balance sheet amount of all indebtedness for borrowed money (net of all cash and cash equivalents) of the Company and its Subsidiaries as of such day, determined on a consolidated basis in accordance with GAAP, to (b) EBITDA for the four-fiscal quarter period ending on such date.

"Liquidator" has the meaning set forth in Section 10.2(a).

"Losses" means items of loss and deduction of the Company determined according to Section 6.2(b).

"<u>Majority in Voting Interest</u>" means, at any time, a Member or Members which own a majority of the votes of all of the Voting Units outstanding at such time.

"<u>Majority Member</u>" means, at any time, a Member that individually or together with its Affiliates owns a majority of the Voting Units outstanding at such time.

"<u>Majority of the Board</u>" means, at any time, a combination of any of the then Directors constituting a majority of the votes of all of the Directors who are then elected and qualified.

"<u>Majority of the Minority</u>" means, at any time, the Members holding a majority of the Voting Units held by the Members other than a Majority Member.

"<u>Member</u>" means each Person identified on the Members Schedule as of the date hereof who is a party to or is otherwise bound by this Agreement and each Person who may hereafter be admitted as a Member in accordance with the terms of this Agreement. The Members shall constitute the "members" (as that term is defined in the Delaware Act) of the Company.

"<u>Member Minimum Gain</u>" with respect to each Member Nonrecourse Debt, means the amount of Company Minimum Gain (as determined according to Treasury Regulation Section 1.704-2(d)(1)) that would result if such Member Nonrecourse Debt were treated as a nonrecourse liability, determined in accordance with Treasury Regulation Section 1.704-2(i)(3).

"<u>Member Nonrecourse Debt</u>" has the meaning set forth in Treasury Regulation Section 1.704-2(b)(4), substituting the term "Company" for the term "partnership" and the term "Member" for the term "partner" as the context requires.

"<u>Member Nonrecourse Deduction</u>" has the meaning set forth in Treasury Regulation Section 1.704-2(i), substituting the term "Member" for the term "partner" as the context requires.

"Members Schedule" has the meaning set forth in Section 5.1.

"<u>Membership Interest</u>" means the interest acquired by a Member in the Company, including such Member's right (based on the type and class and/or series of Unit or Units held by such Member), as applicable, (A) to a distributive share of Profits, Losses, and other items of income, gain, loss, deduction and credits of the Company, (B) to a distributive share of the assets of the Company, (C) to vote on, consent to or otherwise participate in any decision of the Members, and (D) to any and all other benefits to which such Member may be entitled as provided in this Agreement or the Delaware Act.

"NASDAQ" means the National Association of Securities Dealers Automated Quotations system.

"<u>Newco</u>" has the meaning set forth in <u>Section 13.11(a)</u>.

"Nonvoting Units" means the Class A-1 Units, the Class B Units, the Class C Units and the Class D Units.

"Notice" has the meaning set forth in Section 9.6(a).

"<u>Offer Notice</u>" has the meaning set forth in <u>Section 11.7(a)</u>.

"Offer Period" has the meaning set forth in Section 11.7(a).

"Offered Units" has the meaning set forth in Section 11.4(a)(i).

"Offerees" has the meaning set forth in Section 11.7(a).

"<u>officers</u>" has the meaning set forth in <u>Section 3.3(a)</u>.

"<u>Original Operating Agreement</u>" has the meaning set forth in the Recitals.

"Other Indemnitors" has the meaning set forth in Section 3.6.

"Other Registration Rights" has the meaning set forth in Section 13.1(a)(iii).

"<u>Permitted Transfer</u>" means (i) with respect to any Transfer of Units by any Member (a) in the case of an individual Member, a Transfer pursuant to applicable laws of descent and distribution or to any member of such Member's Family Group, (b) in the case of any entity, a Transfer among its Affiliates, or the partners (including limited partners) or employees of it or its Affiliates, (c) in connection with any recapitalization, reorganization or reclassification of, or any merger or consolidation involving, the Company, a Transfer in which all Members holding the same type and class or series of Units are treated the same and which transaction does not involve the Transfer of Units to an Independent Third Party, including the reorganization of the Company into Newco, (d) a Transfer to any Person that is already a Member prior to such Transfer, or (e) a Transfer to the Company pursuant to a Grant Agreement (including the incentive plan governing such Grant Agreement) or the Zencor Management Agreement and (ii) solely with respect to the HC2 Member, any pledge, hypothecation, grant of a security interest or lien, or other encumbrance on all or any portion of its Membership Interest for the benefit of the HC2 Member or any of its Affiliates.

"Permitted Transferee" means any Person to whom a Permitted Transfer was made.

"<u>Person</u>" means any individual, corporation, partnership, limited liability company, trust, joint venture, governmental entity or other unincorporated entity, association or group.

"Piggyback Registration" has the meaning set forth in Section 13.2(a).

"Preemptive Interests" has the meaning set forth in Section 4.11(a).

"<u>Preferred Return</u>" of a Class A Member or Class A-1 Member for any period means an 8% annual rate of return on the Unreturned Capital Contributions of such Class A Member or Class A-1 Member, as applicable, compounded (<u>i.e.</u>, treated as additional Unreturned Capital Contributions) annually on January 1 of each year, computed in the manner of interest and taking proper account of the timing of distributions pursuant to <u>Section 7.2</u> and Capital Contributions. For purposes of computing the Preferred Return, Capital Contributions made by the Class A Members for Class A Units issued prior to the date of the Third Amended and Restated Operating Agreement shall be deemed to have been made as of December 3, 2014, Capital Contributions made by the Class A Members for Class A Units issued after the date of the Third Amended and Restated Operating Agreement shall be deemed to have been made as of the date contributed and the Class A-1 Members shall be deemed to have made Capital Contributions in respect of their Class A-1 Units on the date such Class A-1 Units were granted.

"Proceeding" has the meaning set forth in Section 13.7(c).

"Profits" means items of income and gain of the Company determined according to Section 6.2(b).

"Prohibited Transferee" means (i) any Person with respect to which the Board has reasonably determined that such Person or any of its Affiliates, is, or is reasonably expected to become, a competitor of the Company or any of its Subsidiaries; (ii) a Person who is the subject of any ongoing investigation or sanctions by any governmental authority regarding any alleged criminal or fraudulent activity of such Person, including any alleged violation of any applicable anti-corruption, anti-money laundering, anti-terrorism and economic sanction and anti-boycott laws; (iii) a Person who is, or whose Affiliate is, then a party adverse in any pending or threatened (in writing) action, suit or proceeding (A) to the Company or any of its Subsidiaries or (B) to any Member or an Affiliate thereof in which the amount at issue exceeds \$2,000,000; or (iv) a Person who is then in Bankruptcy (or an Affiliate that is substantively consolidated with a Person that is then in Bankruptcy).

"<u>Projections</u>" means the internally-generated budget, as customarily prepared by management of the Company, including a projected consolidated balance sheet of the Company and its Subsidiaries, the related consolidated statements of projected operations or income and projected cash flow, in each case on a fiscal quarter basis.

"Proposed Tag Purchaser" has the meaning set forth in Section 11.4(a)(i).

"Public Offering" means any underwritten offering of Common Stock to the public pursuant to an effective Registration Statement under the Securities Act (or the closing of another transaction that results in any Common Stock (or the common stock or other equity securities of any successor corporation the common stock of which is exchanged for the Common Stock) being publicly traded and widely held by the public); provided that a Public Offering shall not include an offering made in connection with a business acquisition or combination or an employee benefit plan.

"Public Sale" means any sale of securities pursuant to a Public Offering or a Rule 144 Sale.

"Recommended Offer" has the meaning set forth in Section 11.8(c).

"Registrable Securities" means (i) the Units, (ii) any other securities that may be issued or distributed in respect of such Units by way of any equity split, equity dividend, exchange or other distribution, recapitalization, reclassification, merger, consolidation or similar event, including Common Stock, and (iii) the shares of Common Stock issuable upon conversion or exchange of the Units. As to any particular Registrable Securities, such securities shall cease to be Registrable Securities (x) when they have been Transferred in a Public Sale, (y) when such Registrable Securities can be sold without restriction (including as to manner of sale requirements or volume restrictions) under Rule 144 of the Securities Act and or (z) when such Registrable Securities have been otherwise Transferred if (I) new certificates not bearing the legend set forth in <u>Section 11.5</u> shall have been delivered by the Company, (II) subsequent disposition of such securities shall not require registration or qualification of such securities under the Securities Act or such state securities or blue sky laws then in force and (III) such securities can be sold without restriction (including as to manner of sale requirements or volume restrictions) under Rule 144 of the Securities Act. For purposes of this Agreement, a Person will be deemed to be a holder of Registrable Securities or otherwise, but disregarding any restrictions or limitations upon the exercise of such right), whether or not such acquisition has actually been affected.

"Registration Notice" has the meaning set forth in Section 13.1(a).

"Registration Request" has the meaning set forth in Section 13.1(a).

"<u>Registration Statement</u>" means any registration statement of the Company filed with the SEC pursuant to the Securities Act (other than a registration statement on Form S-4 or Form S-8 or any similar or successor form), including the prospectus, amendments and supplements to such registration statement, including post-effective amendments, all exhibits, and all material incorporated by reference or deemed to be incorporated by reference in such registration statement.

"<u>Regulatory Allocations</u>" has the meaning set forth in <u>Section 8.2(f)</u>.

"<u>Representatives</u>" has the meaning set forth in <u>Section 4.9(a)</u>.

"<u>Requesting Holder</u>" has the meaning set forth in <u>Section 13.1(a)</u>.

"<u>Required Registration</u>" has the meaning set forth in <u>Section 13.5</u>.

"<u>Rule 144 Sale</u>" means a sale of securities to the public through a broker, dealer or market-maker pursuant to the provisions of Rule 144 (other than Rule 144(k) prior to a Public Offering) adopted under the Securities Act (or any successor law).

"Sale Notice" has the meaning set forth in Section 11.4(a)(i).

"Sale Period" has the meaning set forth in Section 11.7(d).

"Sale Price" has the meaning set forth in Section 11.7(a).

"Sale Process" has the meaning set forth in Section 11.8(c).

"SEC" means the Securities and Exchange Commission.

"Second Amended and Restated Operating Agreement" has the meaning set forth in the Recitals.

"Second Return Threshold Amount" at any time of determination means the greater of (a) the excess of (i) (x) two multiplied by (y) the aggregate Capital Contributions of the Class A Members over (ii) the aggregate amount previously distributed to the Class A Members pursuant to Section 7.2 and (b) the amount which, when taken together with all other distributions to the Class A Members, would result in an internal rate of return for the Class A Members, taken as a group (calculated taking account of the actual timing of Capital Contributions and distributions pursuant to Section 7.2 and using the xIRR function on Microsoft Excel) equal to 20% if such amount were distributed to the Class A Members at such time.

"Securities Act" means the United States Securities Act of 1933, as amended from time to time.

"Segregated Account" has the meaning set forth in Section 5.2(b)(ii).

"Subject Interest" has the meaning set forth in Section 11.7.

"<u>Subscription Agreement</u>" means each subscription agreement entered into by the Company and a Member pursuant to which such Member has purchased Units.

"Subsidiary" means, with respect to any Person, any corporation, limited liability company, partnership, association or other business entity of which (a) if a corporation, a majority of the total voting power of shares of stock entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof, or (b) if a limited liability company, partnership, association or other business entity, a majority of the limited liability company, partnership or other similar ownership interest thereof is at the time owned or controlled, directly or indirectly, by any Person or one or more Subsidiaries of that Person or a combination thereof. For purposes hereof, a Person or Persons shall be deemed to have a majority ownership interest in a limited liability company, partnership, association or other business entity if such Person or Persons shall be allocated a majority of limited liability company, partnership, association or other business entity gains or losses or shall be or control the managing director, managing manager or general partner of such limited liability company, partnership, association or other business entity.

"<u>Subsidiary Available Cash</u>" means Available Cash calculated only with respect to the Subsidiaries of the Company on a combined basis and without giving effect to any Available Cash at the Company.

"<u>Subsidiary Leverage Ratio</u>" means the Leverage Ratio as calculated only with respect to the Subsidiaries of the Company on a combined basis and without giving effect to any indebtedness of the Company that is nonrecourse to a Subsidiary (but excluding any intercompany debt that would be eliminated upon consolidation).

"Tag Notice" has the meaning set forth in Section 11.4(a)(i).

"<u>Tag Period</u>" has the meaning set forth in <u>Section 11.4(a)(i)</u>.

"Tag Transfer" has the meaning set forth in Section 11.4(a)(i).

"Tagging Member" has the meaning set forth in Section 11.4(a)(i).

"<u>Tax Matters Partner</u>" has the meaning set forth in Code Section 6231 (as in effect immediately before the enactment of the Bipartisan Budget Act of 2015) (and any similar provision of state, local or foreign law) and, is the "partnership representative" for purposes of Code Section 6223(a) (as in effect following the enactment of the Bipartisan Budget Act of 2015) (and any similar provision of state, local or foreign law).

"<u>Taxable Year</u>" means the Company's taxable year ending on December 31 (or part thereof in the case of the Company's first and last taxable year), or such other year as is (i) required by Section 706 of the Code or (ii) determined by the Board (if no year is so required by Section 706 of the Code).

"<u>Third Amended and Restated Operating Agreement</u>" has the meaning set forth in the Recitals.

"<u>Threshold Equity Value</u>" of each Class B Unit, Class C Unit and Class D Unit means the "Threshold Equity Value" set forth in respect of such Class B Unit, Class C Unit or Class D Unit, as applicable, in the Grant Agreement pursuant to which such Class B Unit, Class C Unit or Class D Unit, as applicable, was issued.

"<u>Transfer</u>" means any gift, sale, transfer, assignment, pledge, hypothecation, grant of a security interest or lien, encumbrance or other disposition (whether with or without consideration, whether directly or indirectly, and whether voluntarily, involuntarily or by operation of law), including any derivative transaction that has the effect of changing materially the economic benefits and risks of ownership (and "<u>Transferee</u>" and any other derivation thereof shall have correlative meanings).

"Transfer Opinion" has the meaning set forth in Section 11.1(a)(iii).

"<u>Transferring Member</u>" has the meaning set forth in <u>Section 11.4(a)(i)</u>.

"<u>Treasury Regulations</u>" means the final or temporary regulations that have been issued by the U.S. Department of Treasury pursuant to its authority under the Code, and any successor regulations.

"<u>Unit</u>" means a unit representing a fractional part of a Membership Interest and shall include all types and classes or series of Units; <u>provided</u> that any type or class or series of Unit shall have the designations, preferences and rights set forth in this Agreement.

"<u>Unpaid Preferred Return Amount</u>" means, with respect to any Class A Member or Class A-1 Member as of any date of determination, the amount of the Preferred Return applicable to such Member that has not been previously distributed to such Member pursuant to <u>Section 7.2(b)</u> as of such date.

"<u>Unreturned Capital Contributions</u>" of any Class A Member or Class A-1 Member at any time of determination, means the excess of (a) the aggregate Capital Contributions made by such Class A Member or Class A-1 Member on or prior to such time over (b) the aggregate amount previously distributed to such Class A Member or Class A-1 Member pursuant to <u>Section 7.2(a)</u>.

"<u>Unvested Class A-1 Unit</u>" means any Class A-1 Unit that is not a Vested Class A-1 Unit.

"<u>Unvested Incentive Interest</u>" means any Incentive Interest that is not a Vested Incentive Interest.

"<u>Vested Class A-1 Unit</u>" means any Class A-1 Unit that has vested pursuant to the terms and conditions of any document pursuant to which such Class A-1 Unit was acquired by the initial holder thereof or any other document governing the vesting of such Class A-1 Unit.

"<u>Vested Incentive Interest</u>" means any Incentive Interest that has vested pursuant to the terms and conditions of any document pursuant to which such Incentive Interest was acquired by the initial holder thereof or any other document governing the vesting of such Incentive Interest.

"Voting Units" means the Class A Units and Class A-2 Units.

"<u>Warranty and Indemnity Agreement</u>" means that certain Warranty and Indemnity Agreement, dated as of October 11, 2017, by and among the Fugro Member, GMSL and the Company.

"Zencor Holdings" means Zencor Holdings LLC, a Delaware limited liability company.

"Zencor Management" means Zencor Management LLC, a Delaware limited liability company.

"Zencor Management Agreement" means the Management Agreement dated as of December 3, 2014 by and between Zencor Management and the Company, as amended from time to time.

1.2 <u>**Other Definitional Provisions**</u>. Capitalized terms used in this Agreement which are not defined in this <u>Article I</u> have the meanings contained elsewhere in this Agreement.

ARTICLE II Organization of the Company

2.1 Formation.(a) The Certificate of Formation was prepared, executed and filed with the Secretary of State of the State of Delaware on September 12, 2014, by an "authorized person" for such purpose within the meaning of the Delaware Act, all of which is hereby authorized and ratified in all respects. This Agreement supersedes the Original Operating Agreement, the Amended and Restated Operating Agreement, the Second Amended and Restated Operating Agreement and the Third Amended and Restated Operating Agreement and the Delaware Act) of the Company. The rights, powers, duties, obligations and liabilities of the Members shall be determined pursuant to the Delaware Act and this Agreement. To the extent that the rights, powers, duties, obligations and liabilities of any Member are different by reason of any provision of this Agreement than they would be in the absence of such provision, this Agreement shall, to the extent permitted by the Delaware Act, control.

(b) Any officer of the Company as an "authorized person" within the meaning of the Delaware Act is hereby authorized, at any time that the Board has approved an amendment to the Certificate in accordance with the terms hereof, to promptly execute, deliver and file such amendment in accordance with the Delaware Act.

(c) The Company shall, to the extent permissible, elect to be treated as a partnership for federal, state and local income tax purposes, and each Member and the Company shall file all tax returns and shall otherwise take all tax and financial reporting positions in a manner consistent with such treatment and no Member shall take any action inconsistent with such treatment. The Company shall not be deemed a partnership or joint venture for any other purpose.

2.2 <u>Name</u>. The name of the Company is "Global Marine Holdings, LLC" or such other name or names as the Board may from time to time designate; <u>provided</u>, that the name shall always contain the words "Limited Liability Company," "LLC" or "L.L.C."

2.3 <u>Registered Office; Agent</u>. The Company shall maintain a registered office in the State of Delaware at c/o Corporation Service Company at 2711 Centerville Road, Suite 400, Wilmington, Delaware 19808 or at such other place within Delaware as the Secretary of the Company may designate. The name and address of the Company's registered agent for service of process on the Company in the State of Delaware is Corporation Service Company at 2711 Centerville Road, Suite 400, Wilmington, Delaware 19808 or such other agent as the Board or the Secretary of the Company may from time to time designate.

2.4 <u>**Term**</u>. The term of existence of the Company shall be perpetual from the date the Certificate of Formation was filed with the Secretary of State of Delaware, unless the Company is dissolved in accordance with the provisions of this Agreement.

2.5 Purposes and Powers. The purposes and character of the business of the Company shall be to transact any or all lawful business for which limited liability companies may be organized under the Delaware Act. The Company shall have any and all powers which are necessary or desirable to carry out the purposes and business of the Company, including the ability to incur and guaranty indebtedness, to the extent the same may be legally exercised by limited liability companies under the Delaware Act. Notwithstanding anything herein to the contrary, nothing set forth herein shall be construed as authorizing the Company to possess any purpose or power, or to do any act or thing, forbidden by law to a limited liability company organized under the laws of the State of Delaware.

ARTICLE III Management of the Company

3.1 Board of Directors.

(a) <u>Establishment</u>. There is hereby established a committee (the "<u>Board</u>") comprised of natural persons (the "<u>Directors</u>") having the authority and duties set forth in this Agreement. Each Director shall be entitled to one vote. Except as provided in <u>Section 3.1(i)</u>, any decisions to be made by the Board shall require the approval of a Majority of the Board. Except as provided in the immediately preceding sentence, no Director acting alone, or with any other Director or Directors, shall have the power to act for or on behalf of, or to bind the Company (including as a result of each Director being a "manager" (as that term is defined in the Delaware Act) of the Company as further provided in this <u>Section 3.1(a)</u>). Directors need not be residents of the State of Delaware.

(b) <u>Powers</u>. The business and affairs of the Company shall be managed by or under the direction of the Board. Unless otherwise approved by the Board, all actions outside of the ordinary course of business of the Company or its Subsidiaries to be taken by or on behalf of the Company or its Subsidiaries shall require the approval of the Board. Except as otherwise specifically provided herein, the Board shall have all rights and powers of a "manager" under the Delaware Act, and shall have such authority, rights and powers in the management of the Company business to do any and all other acts and things necessary, proper, convenient or advisable to effectuate the purposes of this Agreement.

(c) <u>Number of Directors</u>. At any time, the Majority in Voting Interest shall have the power to determine the size of the Board, including the power to (i) decrease the size of the Board (<u>provided</u> that any decrease in size of the Board is in compliance with <u>Section 3.1(d)</u> or (ii) increase the size of the Board (<u>provided</u> that he size of the Board shall never be greater than seven (7) Directors) and to elect the Directors.

(d) <u>Appointments to the Board</u>. Subject to <u>Section 3.1(c)</u>, the Board shall be constituted as follows:

(i) for so long as the Fugro Member and its Affiliates continue to hold, in the aggregate, at least 50% of the number of Class A-2 Units acquired by the Fugro Member as of the Fugro Acquisition Date, the Fugro Member shall have the right to elect two (2) Directors (the "<u>Fugro Directors</u>"). The issuance by the Company of new Units to any Person shall not limit the Fugro Member's rights under this Section <u>3.1(d)(i)</u>;

(ii) for so long as the engagement of Zencor Management pursuant to the Zencor Management Agreement remains in effect, Dick Fagerstal shall have the right to be a Director; and

(iii) the HC2 Member shall have the right to elect all other Directors.

(e) <u>Election, Removal, Resignation and Vacancies</u>. Each Director shall hold office as a Director until his or her respective successor is elected and qualified or until his earlier death, resignation or removal.

(i) *Removal.* Upon receipt of written notice from the Persons possessing the right to elect any Director pursuant to <u>Section 3.1(d)</u> (each, an "<u>Electing Person</u>") for election or removal, each Member shall promptly vote all of his, her or its Voting Units, whether presently held or hereafter acquired, and shall take all action necessary, including without limitation, promptly calling a special meeting of the Members or promptly seeking the written consent of the Members, to elect or remove any such Director.

(ii) *Resignation*. Any Director may resign at any time by giving written notice to the Company. Such resignation shall take effect on the date of the receipt of such notice or at any later date specified therein, and, unless otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective.

(iii) *Vacancies*. At any time that a vacancy is created on the Board by the death, removal or resignation of any Director, the Electing Person shall nominate a Director to fill such vacancy in accordance with <u>Section 3.1(d)</u>.

(f) <u>Meetings of the Board</u>. The Board shall meet at least quarterly and at such other times and at such places (either within or outside of the State of Delaware) as the Board may designate. Special meetings of the Board shall be held on the call of the Majority in Voting Interest or by the Chief Executive Officer (with the prior approval of the Majority in Voting Interest (including via email from a representative of the Majority in Voting Interest)) upon at least (x) five (5) days written (including via email) notice for any quarterly meeting of the Board and (y) forty-eight (48) hours' written (including via email) notice to the Directors for any other meeting of the Board, or upon such shorter notice as may be approved by all of the Directors. A record shall be maintained by the Secretary of the Company of each meeting of the Board.

(i) <u>Conduct of Meetings</u>. Any meeting of the Directors may be held in person, telephonically or by video conference; <u>provided</u>, for the avoidance of doubt, that the

Directors shall not be required to attend any meeting of the Directors in person and may at all times attend any such a meeting telephonically.

(ii) <u>Quorum</u>. A Majority of the Board shall constitute a quorum of the Board for purposes of conducting business. At all times when the Board is conducting business at a meeting of the Board, a quorum of the Board must be present at such meeting. If a quorum shall not be present at any meeting of the Board, then the Directors present at the meeting shall adjourn the meeting from time to time, and notice of the adjournment and the time and place of the rescheduled meeting shall be given to all Directors. A Director may vote or be present at a meeting either in person or by proxy. Each proxy shall be signed by the Director giving such proxy.

(iii) <u>Attendance and Waiver of Notice</u>. Attendance of a Director at any meeting shall constitute a waiver of notice of such meeting, except where a Director attends a meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting is not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, any regular or special meeting of the Board need be specified in the notice or waiver of notice of such meeting.

(iv) <u>Actions Without a Meeting</u>. Notwithstanding any provision contained in this Agreement, any action of the Board may be taken by written consent or by electronic transmission without a meeting. Any such action taken by the Board without a meeting shall be effective only if the consent or consents set forth the actions so taken and are in writing or sent by electronic transmission by either (i) all of the Directors or (ii) a Majority of the Board if such action is taken after at least forty-eight (48) hours' prior written (including via email) notice of such proposed action having been delivered to all Directors.

(g) <u>Compensation of the Directors</u>. Unless otherwise determined by the Board, (i) any Director that is not an employee of the Company or any of its Affiliates shall be entitled to receive compensation at the rate of \$25,000 per annum for his or her services as a Director and (ii) all other Directors shall not be entitled to receive compensation for his or her services in such capacity. In addition, reimbursement for out-of-pocket expenses of attendance, if any, may be allowed for attendance at each regular or special meeting of the Board for any Director; <u>provided</u>, that nothing contained in this <u>Section 3.1(g)</u> shall be construed to preclude any Director from serving the Company or any of its Subsidiaries in any other capacity and receiving compensation for such service except as provided in <u>Section 4.9</u>.

(h) Proxy. In order to effectuate the provisions of Section 3.1(d)(ii) and 3.1(d)(iii), each Member hereby grants to the HC2 Member a proxy to vote at any annual or special meeting of Members, or to take any action by written consent in lieu of such meeting with respect to, or to otherwise take action in respect of, all of the Units owned or held of record by such Member in connection with the matters set forth in Section 3.1(d)(ii) and 3.1(d)(iii) solely to the extent necessary to comply with the provisions of Section 3.1(d)(ii) and 3.1(d)(iii). EACH OF THE PROXIES GRANTED HEREBY IS IRREVOCABLE AND IS COUPLED WITH AN INTEREST. The HC2 Member shall only exercise its proxy to effect the provisions of Section 3.1(d)(ii) and 3.1(d)(iii). To effectuate the provisions of this Section 3.1(h), the Secretary of the Company and each Subsidiary of the Company, or, if there shall be no Secretary, then such other officer or employee

of the Company or such Subsidiary as the Board or the board or comparable governing body of such Subsidiary may appoint to fulfill the duties of the Secretary, shall not record any vote or consent or other action contrary to the terms of this <u>Section 3.1(h)</u>. No Member, in such Member's capacity as such, shall grant any proxy or become party to any voting trust or other agreement which is inconsistent with, conflicts with or violates any provision of this Agreement.

(i) <u>Proxy</u>. In order to effectuate the provisions of <u>Section 3.1(d)(i)</u>, each Member hereby grants to the Fugro Member a proxy to vote at any annual or special meeting of Members, or to take any action by written consent in lieu of such meeting with respect to, or to otherwise take action in respect of, all of the Units owned or held of record by such Member in connection with the matters set forth in <u>Section 3.1(d)(i)</u> solely to the extent necessary to comply with the provisions of <u>Section 3.1(d)(i)</u>. EACH OF THE PROXIES GRANTED HEREBY IS IRREVOCABLE AND IS COUPLED WITH AN INTEREST. The Fugro Member shall only exercise its proxy to effect the provisions of <u>Section 3.1(d)(i)</u>. To effectuate the provisions of this <u>Section 3.1(i)</u>, the Secretary of the Company and each Subsidiary of the Company, or, if there shall be no Secretary, then such other officer or employee of the Company or such Subsidiary as the Board or the board or comparable governing body of such Subsidiary may appoint to fulfill the duties of the Secretary, shall not record any vote or consent or other action contrary to the terms of this <u>Section 3.1(i)</u>. No Member, in such Member's capacity as such, shall grant any proxy or become party to any voting trust or other agreement which is inconsistent with, conflicts with or violates any provision of this Agreement

(j) <u>Consent of Fugro Directors</u>. Notwithstanding any provision of this Agreement to the contrary, for so long as the Fugro Member has the right to elect the Fugro Directors pursuant to <u>Section 3.1(d</u>), the following actions shall require the approval of (i) a Majority of the Board and (ii) at least one (1) of the Fugro Directors:

(i) a material change in the business or business strategy of the Company as set forth on <u>Schedule B</u> hereto;

(ii) incurrence of any indebtedness for borrowed money to the extent that the Leverage Ratio, as of the last day of the fiscal quarter most recently ended, calculated on a pro forma basis after giving effect to such indebtedness for borrowed money, would exceed 3.75:1.00;

(iii) incurrence or making of any capital expenditure for cash, which when added to the aggregate amount of cash capital expenditures by the Company and its Subsidiaries over the then current fiscal quarter period, would exceed an amount equal to 25% of the enterprise value of the Company and its Subsidiaries as of the last day of the fiscal quarter most recently ended;

(iv) issuance or sale of any Membership Interests by the Company to the extent the price paid for such Membership Interests is below Fair Market Value;

(v) declaration of or payment of dividend or other distribution, direct or indirect, by any Subsidiary of the Company on account of the Company's equity interest in such Subsidiary to the extent that, as of any date of determination (a) the Subsidiary Leverage Ratio, as of the last day of the fiscal quarter most recently ended,

exceeds 3.75:1.00; (b) after giving effect to such dividend or distribution, Subsidiary Available Cash would be less than \$15,000,000 in the aggregate; (c) net income of the Subsidiaries of the Company (computed on a combined basis and not including the Company) for the four-fiscal quarter period most recently ended, determined on a combined basis in accordance with GAAP, is less than zero; or (d) the Projections for the Subsidiaries of the Company on a combined basis with respect to the immediately succeeding four fiscal quarters show that (1) Subsidiary Available Cash will be less than \$15,000,000 as of the last day of such period or (2) the Subsidiary Leverage Ratio, as of the last day of such period, will exceed 3.75:1.00; provided, for the avoidance of doubt, that, for so long as the Fugro Member has the right to elect the Fugro Directors pursuant to Section 3.1(d), the Members and the Company shall cause GMSL and GMHL to not, without the prior approval of at least one (1) of the Fugro Directors, declare or pay a dividend or make any other distribution, in each case, as described in this Section 3.1(j)(y);

(vi) entrance into, renewal, amendment or modification of any contract, arrangement or transaction between any Member or any Affiliate of any Member, on the one hand, and the Company or any of its Subsidiaries, on the other hand, except for (A) this Agreement (which may be amended as provided in Section 14.4), and (B) entering into any contract, arrangement or transaction (x) provided for in this Agreement or (y) on an arms' length basis; <u>provided</u> that the entrance into, renewal, amendment or modification of any such contract, arrangement or transaction relating to a loan, funding or similar instrument or equivalent thereof made by the Company or any of its Subsidiaries, on the one hand, to a Member or any Affiliate of any Member on the other shall always require the approval of (i) a Majority of the Board and (ii) at least one (1) of the Fugro Directors; and

(vii) dissolution of the Company in accordance with <u>Section 10.1</u>.

3.2 <u>Committees of the Board</u>. The Board may, by resolution, designate from among the Directors one or more committees (including, but not limited to, an Audit Committee), each of which shall be comprised of one or more Directors, and may designate one or more of the Directors as alternate members of any committee, who may, subject to any limitations imposed by the Board, replace absent or disqualified Directors at any meeting of that committee. Any members of a committee may be removed by a Majority of the Board. The initial committees of the Board shall be the Executive Committee and the Compensation Committee. As of the date hereof, the members of the Executive Committee are Dick Fagerstal, Ian Douglas and Paul Voigt, with Dick Fagerstal serving as chairman, and the members of the Compensation Committee are Michael Sena and Dick Fagerstal, with Michael Sena serving as chairman.

3.3 Officers.

(a) <u>Appointment of Officers</u>. The Board shall appoint individuals as officers ("<u>officers</u>") of the Company, which may include a Chief Executive Officer, a President, a Chief Financial Officer, a Chief Operating Officer, a General Counsel, a Secretary and such other officers (such as a Treasurer or any number of Vice Presidents) as the Board deems advisable. No officer need be a Member or a Director. An individual may be appointed to more than one office. Each officer of the Company shall be a "manager" (as that term is used in the Delaware Act) of the Company, but, notwithstanding the foregoing, no officer of the Company shall have any rights or powers beyond the rights and powers granted to such officer in this Agreement.

(b) <u>Duties of Officers Generally</u>. Under the direction of and, at all times, subject to the authority of the Board, the officers shall have full and complete discretion to manage and control the day-to-day business, operations and affairs of the Company in the ordinary course of its business, to make all decisions affecting the day-to-day business, operations and affairs of the Company in the ordinary course of its business and to take all such actions as they deem necessary or appropriate to accomplish the foregoing, in each case, unless the Board shall have previously restricted (specifically or generally) such powers. In addition, the officers shall have such other powers and duties as may be prescribed by the Board or this Agreement. The Chief Executive Officer and the President shall have the power and authority to delegate to any agents or employees of the Company rights and powers of officers of the Company to manage and control the day-to-day business, operations and affairs of the Company in the ordinary course of its business, as the Chief Executive Officer or the President may deem appropriate from time to time, in each case, unless the Board shall have previously restricted (specifically or generally) such powers.

(c) <u>Authority of Officers</u>. Subject to <u>Section 3.3(b)</u>, any officer of the Company shall have the right, power and authority to transact business in the name of the Company or to act for or on behalf of or to bind the Company. With respect to all matters within the ordinary course of business of the Company, third parties dealing with the Company may rely conclusively upon any certificate of any officer to the effect that such officer is acting on behalf of the Company.

(d) <u>Removal, Resignation and Filling of Vacancy of Officers</u>. The Board may remove any officer, for any reason or for no reason, at any time. Any officer may resign at any time by giving written notice to the Board, and such resignation shall take effect at the date of the receipt of that notice or any later time specified in that notice; <u>provided</u>, that unless otherwise specified in that notice, the acceptance of the resignation shall not be necessary to make it effective. Any such resignation shall be without prejudice to the rights, if any, of the Company or such officer under this Agreement. A vacancy in any office because of death, resignation, removal or otherwise shall be filled in the manner prescribed in this Agreement for regular appointments to that office.

(e) <u>Compensation of Officers</u>. The officers shall be entitled to receive compensation from the Company as determined by the Board.

(f) <u>Chief Executive Officer</u>. Under the direction of and, at all times, subject to the authority of the Board, the Chief Executive Officer shall have general supervision over the day-to-day business, operations and affairs of the Company and shall perform such duties and exercise such powers as are incident to the office of chief executive officer of a corporation organized under the GCL. The Chief Executive Officer shall have such other powers and perform such other duties as may from time to time be prescribed by the Board.

(g) <u>President</u>. Under the direction of and, at all times, subject to the authority of the Board, the President, if any, shall perform such duties and exercise such powers as are incident to the office of president of a corporation organized under the GCL. In the absence of the Chief Executive Officer, the President shall perform the duties of the Chief Executive Officer. The President shall have such other powers and perform such other duties as may from time to time be prescribed by the Board.

(h) <u>Chief Financial Officer</u>. The Chief Financial Officer, if any, shall keep and maintain, or cause to be kept and maintained, adequate and correct books and records of accounts of the properties and business transactions of the Company, including accounts of its assets, liabilities, receipts, disbursements, gains, losses, capital and Units, and, in general, shall perform all the duties incident to the office of the chief financial officer of a corporation organized under the GCL. The Chief Financial Officer shall have the custody of the funds and securities of the Company, and shall keep full and accurate accounts of receipts and disbursements in books belonging to the Company. The Chief Financial Officer shall have such other powers and perform such other duties as may from time to time be prescribed by the Board, the Chief Executive Officer and/or the President.

(i) <u>Secretary</u>. The Secretary shall (i) keep the minutes of the meetings of the Members and the Board in one or more books provided for that purpose; (ii) see that all notices to be given by the Company are duly given in accordance with the provisions of this Agreement and as required by law; (iii) be custodian of the company records; (iv) keep a register of the addresses of each Member (to be recorded on the Members Schedule) which shall be furnished to the Secretary by such Member; (v) have general charge of the Members Schedule; and (vi) in general perform all duties incident to the office of the secretary of a corporation organized under the GCL. The Secretary shall have such other powers and perform such other duties as may from time to time be prescribed by the Board, the Chief Executive Officer and/or the President.

(j) <u>Other Officers</u>. All other officers of the Company shall have such powers and perform such duties as may from time to time be prescribed by the Board and/or the Chief Executive Officer.

3.4 Fiduciary Duties. Subject to Section 4.9, the Directors and officers, in the performance of their duties as such, shall owe to the Members duties of loyalty and due care of the type owed by the directors and officers, respectively, of a corporation to the stockholders of such corporation under the GCL. Notwithstanding the foregoing and anything to the contrary in this Agreement, (a) no Director or officer other than a Company Associated Person shall be required to or otherwise have any fiduciary or other duty to show or offer any opportunity to the Company (and the Members waive all right and entitlement to such opportunity) and (b) no Director or officer other than a Company Associated Person, nor any of his or her respective Affiliates, will have any

liability to the Company or its Members under this Agreement for the failure to show or offer such opportunity to the Company or pursuing such opportunity, <u>provided</u> that such pursuit would not be in contravention of any other agreement between such Director or officer, as the case may be, and the Company or its Subsidiaries.

3.5 Performance of Duties; Liability of Directors and Officers. In performing his or her duties, each of the Directors and the officers shall be entitled to rely in good faith on the provisions of this Agreement and on information, opinions, reports, or statements (including financial statements and information, opinions, reports or statements as to the value or amount of the assets, liabilities, Profits or Losses of the Company or any facts pertinent to the existence and amount of assets from which distributions to Members might properly be paid), of the following other Persons or groups: (a) one or more officers or employees of the Company; (b) any attorney, independent accountant, or other Person employed or engaged by the Company; or (c) any other Person who has been selected with reasonable care by or on behalf of the Company, in each case as to matters which such relying Person reasonably believes to be within such other Person's professional or expert competence. The preceding sentence shall in no way limit any Person's right to rely on information to the extent provided in Section 18-406 of the Delaware Act. No individual who is a Director or an officer of the Company, or any combination of the foregoing, shall be personally liable under any judgment of a court, or in any other manner, for any debt, obligation, or liability of the Company, whether that liability or obligation arises in contract, tort, or otherwise, solely by reason of being a Director or an officer of the Company or any combination of the foregoing.

3.6 <u>Indemnification</u>. Notwithstanding <u>Section 3.4</u>, the Directors and officers shall not be liable, responsible or accountable for damages or otherwise to the Company, or to the Members, and, to the fullest extent allowed by the GCL (as if the Company were a corporation formed under the GCL), each Director and each officer shall be indemnified and held harmless by the Company, including advancement of reasonable attorneys' fees and other expenses, from and against all claims, liabilities, and expenses arising out of any management of the affairs of the Company and its direct and indirect subsidiaries; provided that (a) such Director's or officer's course of conduct was pursued in good faith and believed by him to be in the best interests of the Company and was reasonably believed by him to be within the scope of authority conferred on such Director or officer pursuant to this Agreement and (b) such course of conduct did not constitute a breach of the fiduciary duties set forth in Section 3.4 The rights of indemnification provided in this Section 3.6 are intended to provide indemnification of the Directors and the officers to the fullest extent permitted by the GCL regarding a corporation's indemnification of its directors and officers and will be in addition to any rights to which the Directors or officers may otherwise be entitled by contract or as a matter of law and shall extend to his heirs, personal representatives and assigns. The absence of any express provision for indemnification herein shall not limit any right of indemnification existing independently of this Section 3.6. Each Director's and each officer's right to indemnification pursuant to this Section 3.6 may be conditioned upon the delivery by such Director or such officer of a written undertaking to repay such amount if such individual is determined pursuant to this Section 3.6 or adjudicated to be ineligible for indemnification, which undertaking shall be an unlimited general obligation. The Company acknowledges that certain of the Directors and officers may have certain rights to indemnification provided by other sources with respect to such Directors' and officers' association with the Company (collectively, the "Other Indemnitors"). Notwithstanding the existence of any Other Indemnitor with respect to any such Director or officer

(i) the Company shall be the indemnitor of first resort (*i.e.*, the Company's obligations for indemnification are primary and any obligations of any Other Indemnitor to provide indemnification for the same liabilities are secondary), (ii) the Company shall be liable for the full amount of all expenses, judgments, penalties, fines and amounts paid in settlement to the greatest extent permitted by this <u>Section 3.6</u>, without regard to any rights such a Director or officer may have against any Other Indemnitors, and (iii) the Company waives, relinquishes and releases all Other Indemnitors from any and all claims against the Other Indemnitors for contribution, subrogation or any other recovery of any kind in respect thereof. No payment by any Other Indemnitor on behalf of any Director or officer with respect to any claim for which such Director or officer has sought indemnification from the Company shall affect any of the provisions of this <u>Section 3.6</u>, and the Other Indemnitors shall have a right of contribution and/or be subrogated to the extent of such payment to all of the rights of recovery of a Director or officer against the Company.

3.7 Code of Business Conduct.

(a) The Members and the Company shall, and shall cause its Subsidiaries to, comply with the code of conduct and compliance policy that is attached to this Agreement as <u>Schedule C</u> (the "**Code of Conduct**"), as amended from time to time in the manner set forth in <u>Section 3.7(b)</u>.

(b) The Code of Conduct shall be reviewed by the Board at least once a year and updated as needed. For so long as the Company is a Subsidiary of any Person subject to reporting requirements under the Exchange Act, the Code of Conduct shall be consistent with any code of conduct of such Person which is applicable to the Company. The Fugro Member shall have the right to audit the compliance of the Company and its Subsidiaries with the Code of Conduct as in effect from time to time.

3.8 Annual Business Plan. Prior to the end of each fiscal year, the Company shall prepare a business plan for the Company and its Subsidiaries for the following fiscal year, in a manner consistent, including with respect to the level of detail, with past practices (the "Annual Business Plan"). The Annual Business Plan shall require the approval of the Majority of the Board; <u>provided</u> that, prior to granting such approval, the Board shall discuss the draft Annual Business Plan during a Board meeting in which each Director shall have the opportunity to express his or her views; <u>provided</u>, <u>further</u>, that if the Annual Business Plan materially deviates from the business strategy of the Company as set forth on <u>Schedule B</u> hereto, approval of such Annual Business Plan shall require the consent of at least one (1) of the Fugro Directors pursuant to <u>Section 3.1(j)(i)</u>. The Company's management shall be expected to report to the Board on the implementation of the Annual Business Plan at meetings of the Board or otherwise as and to the extent requested by the Board.

ARTICLE IV Members; Voting Rights

4.1 Meetings of Members.

(a) <u>Generally</u>. Meetings of the Members may be called by the Board or by a Majority in Voting Interest. Only Members who hold Voting Units shall have the right to attend meetings of the Members. All meetings of the Members shall be held telephonically or at the principal office of the Company or at such other place within or without the State of Delaware as may be determined by the Board or Member(s) calling the meeting and set forth in the respective notice or waivers of notice of such meeting. A record shall be maintained by the Secretary of the Company of each meeting of the Members.

(b) <u>Notice of Meetings of Members</u>. Written (including via email) or printed notice stating the place, day and hour of the meeting and, in the case of a special meeting of the Members, describing the purposes for which the meeting is called shall be delivered not fewer than (i) five (5) days before the date of the meeting or (ii) forty-eight (48) hours before the date of the meeting if deemed reasonably necessary by the Board or a Majority in Voting Interest, to each holder of Voting Units (with a copy to the Secretary of the Company), by or at the direction of the Member(s) calling the meeting or the Board, as the case may be. Such notice may, but need not, specify the purpose or purposes of such meeting and may, but need not, limit the business to be conducted at such meeting to such purpose(s).

(c) Quorum. Except as otherwise provided herein (including Section 4.10) or by applicable law, at any time, a Majority in Voting Interest, represented in person or by proxy, shall constitute a quorum of Members for purposes of conducting business. Once a quorum is present at the meeting of the Members, the subsequent withdrawal from the meeting of any Member prior to adjournment or the refusal of any Member to vote shall not affect the presence of a quorum at the meeting. If, however, such quorum shall not be present at any meeting of the Members, the Members entitled to vote at such meeting shall have the power to adjourn the meeting from time to time, without notice other than announcement at the meeting, until Members which own a Majority in Voting Interest shall be present or represented. Except as otherwise required by applicable law, resolutions of the Members at any meeting of Members shall be adopted by the affirmative vote of a majority of the Voting Units represented and entitled to vote at such meeting at which a quorum is present.

(d) <u>Actions Without a Meeting</u>. Unless otherwise prohibited by law, any action to be taken at a meeting of the Members may be taken without a meeting if a consent or consents in writing, setting forth the action so taken, shall be signed by a Member or Members holding not less than a Majority in Voting Interest. A record shall be maintained by the Secretary of the Company of each such action taken by written consent of a Member or Members.

4.2 <u>Voting Rights</u>. Except as specifically provided herein or otherwise required by applicable law, for all purposes hereunder, each Member shall be entitled to (x) one vote per Voting Unit held by such Member and (y) no votes for any Nonvoting Units held by such Member. A Member which owns Voting Units may vote or be present at a meeting either in person or by proxy. There will be no cumulative voting in the election or removal of Directors.

4.3 <u>Registered Members</u>. The Company shall be entitled to treat the owner of record of any Units as the owner in fact of such Unit for all purposes, and accordingly shall not be bound to recognize any equitable or other claim to or interest in such Unit on the part of any other person, whether or not it shall have express or other notice of such claim or interest, except as expressly provided by this Agreement or the laws of the State of Delaware.

4.4 <u>Consent of Spouse</u>. If a Member who is a natural person is married on the date such person becomes a Member, then such Member shall obtain from his or her spouse an executed spousal consent, evidencing the spouse's acknowledgment of and consent to the existence and binding effect of all restrictions contained in this Agreement in the form of <u>Exhibit B</u> hereto ("<u>Consent of Spouse</u>") and effective as of the date such person becomes a Member. Such Consent of Spouse shall not be deemed to confer or convey to the spouse any rights in the Member's securities that do not otherwise exist by operation of law or by agreement of the parties hereto. If a Member should marry or remarry subsequent to the date such person becomes a Member, such Member shall within thirty (30) calendar days thereafter obtain a Consent of Spouse from his or her new spouse.

4.5 Limitation of Liability. No Member will be obligated personally for any debt, obligation or liability of the Company or of any of its Subsidiaries or other Members by reason of being a Member, whether arising in contract, tort or otherwise. Except as otherwise provided in the Delaware Act, by law or expressly in this Agreement, no Member will have any fiduciary or other duty to another Member with respect to the business and affairs of the Company or of any of its Subsidiaries. No Member will have any responsibility to restore any negative balance in its, his or her Capital Account or to contribute to or in respect of the liabilities or obligations of the Company or of any of its Subsidiaries or return distributions made by the Company.

4.6 Withdrawal; Resignation. A Member shall not cease to be a Member as a result of the Bankruptcy of such Member or as a result of any other events specified in § 18-304 of the Delaware Act. So long as a Member continues to own or hold any Units, such Member shall not have the ability to resign as a Member prior to the dissolution and winding up of the Company and any such resignation or attempted resignation by a Member prior to the dissolution or winding up of the Company and all other Member resigning or withdrawing in contravention of this Section 4.6 shall indemnify, defend and hold harmless the Company and all other Members from and against any losses, expenses, judgments, fines, settlements or damages suffered or incurred by the Company or any such other Member arising out of or resulting from such resignation or withdrawal. As soon as any Person who is a Member ceases to own or hold any Units, such Person shall no longer be a Member.

4.7 Death of a Member. The death of any Member shall not cause the dissolution of the Company. In such event the Company and its business shall be continued by the remaining Member or Members and the Units owned by the deceased Member shall automatically be transferred to such Member's heirs (provided that, within a reasonable time after such transfer, the applicable heirs shall sign a joinder to this Agreement substantially in the form of Exhibit A attached hereto (a "Joinder")).

4.8 <u>Authority</u>. No Member, in its capacity as a Member, shall have the power to act for or on behalf of, or to bind the Company.

4.9 Outside Activities.

(a) Subject to the terms of any written agreement by any Member to the contrary (including the non-competition agreements with employees of the Company or any of its subsidiaries), a Member, any of its Affiliates, agents, employees, officers, directors, managers, partners, members and representatives, in each case other than any Company Associated Person (collectively, its "<u>Representatives</u>"), shall have the right to, and shall have no duty (contractual or otherwise) not to, directly or indirectly (notwithstanding that one or more Affiliates, agents, employees, officers, directors, managers, partners, members and representatives are serving on the Board):

(i) have business interests and engage in business activities in addition to those relating to the Company and its Subsidiaries;

(ii) engage in the same or similar business activities or lines of business as the Company or its Subsidiaries, including business interests and activities which compete with the Company and its Subsidiaries; and

(iii) do business with any client, customer, supplier, lender or investor of, to or in the Company or any of its Subsidiaries.

(b) Subject to the terms of any written agreement by any Member to the contrary (including the non-competition agreements with employees of the Company or any of its subsidiaries), if a Member (other than a Company Associated Person) or its Representative acquire knowledge of a potential transaction or matter that may be a corporate opportunity for both (i) the Company or any of its Subsidiaries, on the one hand, and (ii) any other Person, on the other hand, neither such Member nor its Representative shall (x) have any duty (contractual or otherwise) or be required to communicate or present such corporate opportunity to the Company or its Subsidiaries, and (y) notwithstanding any provision of this Agreement to the contrary, be liable to the Company or any of its Subsidiaries or any Member for breach of any duty (contractual or otherwise) by reasons of the fact that such Member or its Representative directly or indirectly pursues or acquires such opportunity for itself, directs such opportunity to another Person, or does not present such opportunity to the Company or any of its Subsidiaries.

(c) Subject to the terms of any written agreement by any Member to the contrary, neither the Company nor any other Member shall have any rights by virtue of this Agreement in any business interests or activities of any Member.

4.10 Actions Requiring Member Approval.

(a) Notwithstanding any other provision of this Agreement, the Company shall not, without first obtaining the affirmative vote or written consent of each of (i) a Majority in Voting Interest, (ii) the Majority of the Minority and (iii) if required pursuant to <u>Section 3.1(j</u>), the consent of at least one (1) of the Fugro Directors:

(i) issue or sell any Membership Interests to the extent the price paid for such Membership Interests is below Fair Market Value; or

(ii) enter into, renew, amend or modify any contract, arrangement or transaction with any Member or any Affiliate of any Member, on the one hand, and the Company or any of its Subsidiaries, on the other hand, except for (A) this Agreement (which shall be governed by its own respective terms), and (B) entering into any contract, arrangement or transaction (x) provided for in this Agreement or (y) on an arms' length basis or otherwise fair to the Company as determined by the Board.

4.11 Preemptive Rights.

(a) Prior to the consummation of an Initial Public Offering, in the event that (x) the Company intends to sell or issue any Units or any other Membership Interests, or (y) any Subsidiary of the Company intends to sell or issue any equity interests in such Subsidiary to any Person other than to a wholly-owned Subsidiary of the Company (any such Units, Membership Interests or equity interests, the "<u>Preemptive Interests</u>"), the Fugro Member shall have the right to purchase, subject to <u>Section 4.11(b)</u> and <u>4.11(c)</u>, an amount of such Preemptive Interests up to (but not in excess of) its Class A/A-1/A-2 Percentage Interest immediately prior to such sale or issuance, on the same terms and conditions as such Preemptive Interests are being offered and sold, such subscription being conditioned upon the actual sale of such Preemptive Interests; <u>provided</u>, <u>however</u>, that such preemptive right shall not extend to any issuance of Class A-1 Units or Incentive Interests, or any other Units, Membership Interests or equity interests of the Company or any of its Subsidiaries (i) by reason of a dividend, split, split-up or other distribution on Units or Membership Interests of the Company or equity interests of such Subsidiary, (ii) to officers, employees or directors of, or consultants to, the Company or any of its Subsidiaries pursuant to any purchase plan or arrangement, option plan, or other incentive plan or agreement approved by the Board or (iv) in connection with an Initial Public Offering.

(b) At any time the Company or any of its Subsidiaries proposes to issue any Preemptive Interests, the Company shall give written notice to the Fugro Member specifying the contemplated date such Preemptive Interests are to be sold, the amount of such Preemptive Interests, the rights and limitations of such Preemptive Interests, the purchase price and other terms and conditions upon which such Preemptive Interests are to be issued, and any other material terms thereof. The Company shall deliver such notice to the Fugro Member no later than 25 Business Days prior to such contemplated purchase date, and the Fugro Member shall have until 10 Business Days prior to the contemplated purchase date specified in such notice to inform the Company of its intentions as to the exercise of the preemptive right provided under this <u>Section 4.11</u>, including the maximum number of Preemptive Interests for which it wishes to exercise its preemptive rights. If no written reply is received by the Company prior to the tenth Business Day before the contemplated purchase date specified in such notice, the Company may treat the preemptive right of the Fugro Member under this <u>Section 4.11</u> to have been waived for that, but only for that, transaction. For the avoidance of doubt, the Fugro Member shall not have any right to acquire any portion of any Preemptive Interests proposed to be issued and sold by the Company unless all of such Preemptive Interests are issued and sold concurrently.

(c) Notwithstanding the other provisions of this <u>Section 4.11</u>, if the Board determines that, in the best interests of the Company, the Company should issue or sell any Units or other Membership Interests or any Subsidiary of the Company should offer or sell any equity interests in such Subsidiary that would otherwise be Preemptive Interests required to be offered to the Fugro Member pursuant to this <u>Section 4.11</u> prior to such offer to the Fugro Member, the Company or its Subsidiary, as applicable, may effect such issuance or sale without first complying with the provisions of this <u>Section 4.11</u>; provided, however, that, within thirty (30) days after the consummation of such issuance or sale, the Company shall offer, and/or cause the Person to whom such issuance or sale was made to offer, to the Fugro Member the opportunity to purchase an amount of such securities up to (but not in excess of) its Class A/A-1/A-2 Percentage Interest immediately prior to such issuance or sale.

ARTICLE V Units; Membership

5.1 <u>Units Generally</u>. The Membership Interests of the Members shall be represented by issued and outstanding Units, which may be divided into one or more types, classes or series, with each type or class or series having the rights and privileges, including voting rights, if any, set forth in this Agreement. The Secretary of the Company shall maintain a schedule of all Members from time to time with the names and addresses of the Members, the Units held by them and the Capital Contributions made by them (as the same may be amended, modified or supplemented from time to time, the "Members Schedule"), a copy of which as of the execution of this Agreement is attached hereto as <u>Schedule A</u>. The Members Schedule shall remain strictly confidential and shall only be disclosed by the Company to the Board; <u>provided</u>, <u>however</u>, that each Member shall be entitled to receive redacted versions of the Members Schedule that only discloses the portions related to such Member as well as the aggregate number of Units outstanding. Ownership of a Unit (or fraction thereof) shall not entitle a Member to call for a partition or division of any property of the Company or for any accounting.

5.2 Incentive Interests.

(a) <u>Threshold Equity Value</u>. Notwithstanding <u>Section 7.2</u>, a holder of an Incentive Interest shall not be entitled to receive any distributions on account of such Incentive Interest unless and until the sum of the aggregate amount of distributions made by the Company in respect of all Units and the aggregate amount of proceeds paid by the Company in respect of redemptions of any Units (measured, in each case, from and after the date such Incentive Interest was issued) at least equals the Threshold Equity Value applicable to such Incentive Interest. In the event that an Incentive Interest is not entitled to participate in a distribution in which it otherwise would have participated but for the operation of the preceding sentence, then the amount as to which such Incentive Interest is not entitled to participate is not entitled to participate shall be distributed in respect of all then outstanding Units in accordance with <u>Section 7.2</u>, subject to the preceding sentence. Determinations pursuant to this <u>Section 5.2(a)</u> shall be made for each dollar of distribution made by the Company.

(b) <u>Vesting; Segregated Accounts</u>.

(i) The Incentive Interests and Class A-1 Units held by an Incentive Member or Class A-1 Member shall vest in accordance with the terms of the related Grant Agreement.

(ii) The Company shall reserve amounts otherwise distributable to an Incentive Member or Class A-1 Member pursuant to <u>Section</u> <u>7.2</u> in respect of Unvested Incentive Interests or Unvested Class A-1 Units, as applicable, held by such Member and deposit such amounts in a segregated account with respect to such Member (a "<u>Segregated Account</u>"). As promptly as practicable following the end of each calendar quarter, or such other dates as reasonably determined by the Board, the Company shall release amounts on deposit in such Member's Segregated Account related to any Unvested Incentive Interests or Unvested Class A-1 Units, as applicable, that vested on or prior to such date.

(c) <u>Termination</u>.

(i) In the event that, pursuant to any agreement between an Incentive Member or Class A-1 Member, on the one hand, and the Company or any of its Subsidiaries, on the other hand, such Member were to forfeit any Incentive Interests or Class A-1 Units, as applicable, then without further action on the part of the Company or any other Member, such Member shall (A) forfeit such Incentive Interests or Class A-1 Units, as applicable, and (B) forfeit any amounts on deposit in such Member's Segregated Account related to such forfeited Incentive Interests or Class A-1 Units, as applicable.

(ii) Upon such forfeitures in subparagraph (i) above, (A) such Member's forfeited Incentive Interests or Class A-1 Units, as applicable, shall be deemed to be no longer outstanding and (B) the forfeited amounts on deposit in such Member's Segregated Account shall be distributed to the Members pursuant to <u>Section 7.2</u>.

5.3 Authorization and Issuance of Units.

(a) <u>Class A Units</u>. The Company is hereby authorized to issue Class A Units. The number of issued and outstanding Class A Units shall be as set forth on the Members Schedule (as it may be amended from time to time by the Company to reflect future issuances).

(b) <u>Class A-1 Units</u>. The Company is hereby authorized to issue Class A-1 Units. The number of issued and outstanding Class A-1 Units shall be as set forth on the Members Schedule (as it may be amended from time to time by the Company to reflect future issuances).

(c) <u>Class A-2 Units</u>. The Company is hereby authorized to issue Class A-2 Units. The number of issued and outstanding Class A-2 Units shall be as set forth on the Members Schedule (as it may be amended from time to time by the Company to reflect future issuances).

(d) <u>Class B Units</u>. The Company is hereby authorized to issue Class B Units. The number of issued and outstanding Class B Units shall be as set forth on the Members Schedule (as it may be amended from time to time by the Company to reflect future issuances).

(e) <u>Class C Units</u>. The Company is hereby authorized to issue Class C Units. The number of issued and outstanding Class C Units shall be as set forth on the Members Schedule (as it may be amended from time to time by the Company to reflect future issuances).

(f) <u>Class D Units</u>. The Company is hereby authorized to issue Class D Units. The number of issued and outstanding Class D Units shall be as set forth on the Members Schedule (as it may be amended from time to time by the Company to reflect future issuances).

(g) <u>Other Units</u>. In addition to the Class A Units, the Class A-1 Units, the Class A-2 Units, the Class B Units, the Class C Units and the Class D Units, the Company is hereby authorized to issue other classes and series of Units. With respect to such other classes and series of Units, the Board is hereby authorized to provide for the issuance of such Units in any class or series by amending this Agreement to reflect such issuance and to establish the Units to be included in each such class or series, and to fix the relative rights, obligations, preferences and limitations of the Units of each such class or series.

(h) <u>Additional Units</u>. Except as expressly provided by this Agreement, the Company shall not authorize, issue or sell, or cause to be authorized, issued or sold, any Units.

(i) <u>Tax Matters</u>. Unless otherwise determined by the Board, it is intended that all Class B Units, Class C Units and Class D Units (each, an "<u>Incentive Interest</u>") will be treated as "profits interests" within the meaning of Rev. Proc. 93-27, 1993-2 C.B. 343.

5.4 Issuance of Units. Subject to the limitations contained in <u>Sections</u> 3.1(j), <u>4.11</u>, <u>11.1(a)(ii)</u> and <u>11.1(a)(iii)</u> hereof, the Company (with the approval of the Board) shall have the right to issue any authorized but unissued Units; <u>provided</u>, that the Company shall not issue any Units to any Person unless such Person has executed and delivered to the Secretary of the Company the documents described in <u>Section 5.5</u> hereof. Upon the issuance of Units, the Board shall adjust the Capital Accounts of the Members as necessary in accordance with <u>Section 6.2</u>.

5.5 <u>New Members from the Issuance of Units</u>. In order for a Person to be admitted as a Member pursuant to the issuance of Units to such Person, such Person shall have executed and delivered to the Secretary of the Company a Joinder. Upon the amendment of the Members Schedule by the Secretary of the Company and the satisfaction of any other applicable conditions, including, if a condition, the receipt by the Company of payment for the issuance of the applicable Units, such Person shall be admitted as a Member and deemed listed as such on the books and records of the Company and thereupon shall be issued his or its Units. The Board shall also adjust the Capital Accounts of the Members as necessary in accordance with Section 6.2.</u>

5.6 Adjustment to Initial Fugro Units. In the event that either GMSL or the Fugro Member elect, in accordance with the terms of the Business Purchase Agreement or the Warranty and Indemnity Agreement, as applicable, to settle any claim for indemnification thereunder, in whole or in part, by an adjustment to the number of Class A-2 Units issued to the Fugro Member pursuant to the Business Purchase Agreement, then the Members Schedule shall be adjusted, effective as of the date of this Agreement, to reflect the ownership by the Fugro Member of an amount of Class A-2 Units equal to the number of Adjusted Initial Fugro Units (a "<u>Class A-2 Adjustment</u>"); provided, for the avoidance of doubt, that any such Class A-2 Adjustment will apply only to the Initial Fugro Units and will not affect the number of any Units acquired by Fugro after the date of this Agreement.

ARTICLE VI Capital Contributions and Capital Accounts

6.1 Capital Contributions.

(a) On or prior to the date hereof, as set forth in the applicable Subscription Agreement and on the Members Schedule, the Members have made (or are deemed to have made) the respective Capital Contributions giving rise to their respective Capital Accounts and are deemed to own the number, type, series and class of Units, in each case, in the amounts set forth opposite such Member's name on the Members Schedule as in effect on the date hereof.

(b) No Member shall make or be required to make any additional contributions to the Company with respect to such Member's Units. Except as expressly provided herein, no Member, in its capacity as a Member, shall have the right to receive any cash or any other property of the Company.

6.2 Capital Accounts.

(a) <u>Maintenance Rules</u>. The Company shall maintain for each type of Member a separate capital account (a "<u>Capital Account</u>") in accordance with this <u>Section 6.2(a)</u>. Each Capital Account shall be maintained in accordance with the following provisions:

(i) Such Capital Account shall be increased by the cash amount or Book Value of any property contributed by such Member to the Company pursuant to this Agreement, such Member's allocable share of Profits and any items in the nature of income or gains which are specially allocated to such Member pursuant to <u>Section 8.2</u> or <u>Section 8.3</u>, and the amount of any liabilities of the Company assumed by such Member or which are secured by any property distributed to such Member.

(ii) Such Capital Account shall be decreased by the cash amount or Book Value of any property distributed to such Member pursuant to this Agreement, such Member's allocable share of Losses and any items in the nature of deductions or losses which are specially allocated to such Member pursuant to <u>Section 8.2</u> or <u>Section 8.3</u> and the amount of any liabilities of such Member assumed by the Company or which are secured by any property contributed by such Member to the Company.

(iii) If all or any portion of a Unit is transferred in accordance with the terms of this Agreement, the transferee shall succeed to the Capital Account of the transferor to the extent it relates to the transferred Unit (or portion thereof).

(iv) If a new or existing Member contributes money or property to the Company (other than a *de minimis* amount as determined by the Board) as consideration for the issuance by the Company of any Units after the date hereof, the Capital Accounts of the Members shall be adjusted in accordance with Treasury Regulation Section 1.704-1(b)(2)(iv)(f).

The foregoing provisions and the other provisions of this Agreement relating to the maintenance of Capital Accounts are intended to comply with Section 1.704-1(b) of the Treasury Regulations and shall be interpreted and applied in a manner consistent with such Treasury Regulations. If the Board determines that it is prudent to modify the manner in which the Capital Accounts, or any increases or decreases to the Capital Accounts, are computed in order to comply with such Treasury Regulations, the Board may authorize such modifications.

(b) <u>Definition of Profits and Losses</u>. "<u>Profits</u>" and "<u>Losses</u>" mean, for each Taxable Year or other period, an amount equal to the Company's taxable income or loss, respectively, for such Taxable Year or other period, determined in accordance with Code Section 703(a) (for this purpose, all items of income, gain, loss or deduction required to be stated separately pursuant to Code Section 703(a)(1) shall be included in taxable income or loss), with the following adjustments:

(i) The computation of all items of income, gain, loss and deduction shall include tax-exempt income and those items described in Treasury Regulation Section 1.704-1(b)(2)(iv)(i), without regard to the fact that such items are not includable in gross income or are not deductible for federal income tax purposes.

(ii) If the Book Value of any Company property is adjusted pursuant to Treasury Regulation Section 1.704-1(b)(2)(iv)(e) or (*f*), the amount of such adjustment shall be taken into account as gain or loss from the disposition of such property (provided that if the Book Value of any Company property is adjusted pursuant to Treasury Regulation Section 1.704-1(b)(2)(iv)(f)(5)(i), the allocation of gain or loss shall be made immediately prior to the related acquisition of the interest in the Company).

(iii) Items of income, gain, loss or deduction attributable to the disposition of Company property having a Book Value that differs from its adjusted basis for tax purposes shall be computed by reference to the Book Value of such property.

(iv) Items of depreciation, amortization and other cost recovery deductions with respect to Company property having a Book Value that differs from its adjusted basis for tax purposes shall be computed by reference to the property's Book Value in accordance with Treasury Regulation Section 1.704-1(b)(2)(iv)(g).

(v) To the extent an adjustment to the adjusted tax basis of any Company property pursuant to Code Sections 732(d), 734(b) or 743(b) is required, pursuant to Treasury Regulation Section 1.704-1(b)(2)(iv)(m), to be taken into account in determining Capital Accounts, the amount of such adjustment to the Capital Accounts shall be treated as an item of gain (if the adjustment increases the basis of the asset) or loss (if the adjustment decreases such basis).

6.3 <u>Negative Capital Accounts</u>. If any Member has a deficit balance in its Capital Account, such Member shall have no obligation to restore such negative balance or to make any Capital Contributions to the Company by reason thereof, and such negative balance shall not be considered an asset of the Company or of any Member.

6.4 <u>No Withdrawal</u>. No Member will be entitled to withdraw any part of his or its Capital Contribution or Capital Account or to receive any distribution from the Company, except as expressly provided in this Agreement.

6.5 Loans From Members. Loans by Members to the Company shall not be considered Capital Contributions.

6.6 Status of Capital Contributions.

(a) No Member shall receive any interest, salary or drawing with respect to its Capital Contributions or its Capital Account, except as otherwise specifically provided in this Agreement.

(b) Except as otherwise provided herein, no Member shall be required to lend any funds to the Company or to make any additional Capital Contributions to the Company. No Member shall have any personal liability for the repayment of any Capital Contribution of any other Member.

ARTICLE VII Distributions

7.1 <u>Generally</u>.

(a) Subject to the provisions of Section 7.3, the Board shall have sole discretion regarding the amounts and timing of distributions to Members, in each case subject to the retention and establishment of reserves of, or payment to third parties of, such funds as it deems necessary with respect to the reasonable business needs of the Company and its Subsidiaries which shall include the payment or the making of provision for the payment when due of the Company's and its Subsidiaries' obligations, including the payment of any management or administrative fees and expenses or any other obligations. Once the Board has determined to make a distribution to the Members, such distribution shall be made pursuant to the terms of Section 7.2 and Section 7.3. For the avoidance of doubt, subject to Section 5.2, all distributions under this Article VII with respect to any class of Units shall be made *pro rata* among the Members holding such class of Units based on their ownership of such class of Units.

(b) Notwithstanding any provision to the contrary contained in this Agreement, the Company shall not make any distribution to Members if such distribution would violate Section 18-607 of the Delaware Act or other applicable law.

7.2 <u>**Discretionary Distributions**</u>. Subject to the provisions of <u>Section 5.2(b)</u>, <u>Section 7.3</u> and <u>Section 7.5</u>, the Company shall distribute Available Cash at the times and in amounts determined by the Board. Any distribution made to Members pursuant to this <u>Section 7.2</u> shall be made as follows:

(a) First, (i) in the Class A/A-1 Distribution Percentage to the Class A Members and Class A-1 Members in proportion to their respective Unreturned Capital Contributions at the time of such distribution and simultaneously (ii) in the Class A-2 Distribution Percentage to the Class A-2 Members in proportion to their respective Class A-2 Percentage Interests, until the Class A Members and Class A-1 Members shall have received aggregate distributions pursuant to this <u>paragraph (a)</u> equal to the aggregate Unreturned Capital Contributions of the Class A Members and Class A-1 Members and Class A-1 Members at the time of such distribution;

(b) Second, (i) in the Class A/A-1 Distribution Percentage to the Class A Members and Class A-1 Members in proportion to their respective Unpaid Preferred Return Amounts at the time of such distribution and simultaneously (ii) in the Class A-2 Distribution Percentage to the Class A-2 Members in proportion to their respective Class A-2 Percentage Interests, until the Class A Members and Class A-1 Members shall have received aggregate distributions pursuant to this <u>paragraph (b)</u> equal to the aggregate Unpaid Preferred Return Amounts of the Class A Members and Class A-1 Members and Class A-1 Members at the time of such distribution;

(c) Third, 100% to the Class B Members in proportion to their respective Class B Units until the Class B Members have received aggregate distributions pursuant to this <u>paragraph (c)</u> equal to the product of (x) the aggregate Class B Threshold Percentages of all Class B Members and (y) the aggregate distributions made pursuant to <u>paragraph (b)</u> above and this <u>paragraph (c)</u>;

(d) Fourth, 100% to the Class A Members, Class A-1 Members, Class A-2 Members and Class B Members in proportion to their respective Class A/A-1/A-2/B Percentage Interests until the Class A Members have received aggregate distributions pursuant to this <u>Section 7.2</u> equal to the First Return Threshold Amount;

(e) Fifth, 100% to the Class C Members in proportion to their respective Class C Units until the Class C Members have received aggregate distributions pursuant to this <u>paragraph (e)</u> equal to the product of (x) the aggregate Class C Threshold Percentages of all Class C Members and (y) the aggregate distributions made pursuant to <u>paragraphs (b)</u> through (d) above and this <u>paragraph (e)</u>;

(f) Sixth, 100% to the Class A Members, Class A-1 Members, Class A-2 Members, Class B Members and Class C Members in proportion to their respective Class A/A-1/A-2/B/C Percentage Interests until the Class A Members have received aggregate distributions pursuant to this <u>Section 7.2</u> equal to the Second Return Threshold Amount;

(g) Seventh, 100% to the Class D Members in proportion to their respective Class D Units until the Class D Members have received aggregate distributions pursuant to this <u>paragraph (g)</u> equal to the product of (x) the aggregate Class D Threshold Percentages of all Class D Members and (y) the aggregate distributions made pursuant to <u>paragraphs (b)</u> through (f) above and this <u>paragraph (g)</u>; and

(h) Thereafter, to the Members in proportion to their respective Class A/A-1/A-2/B/C/D Percentage Interests.

7.3 Tax Advances. The Board shall have the discretion to enter into any agreement or other arrangement with any Member with respect to periodic distributions of cash to such Member, which distributions may be made prior to distributions that would otherwise be made pursuant to Section 7.2, to assist the Member in paying the Member's income taxes attributable to Company income allocated to the Member; provided, however, any distributions made to any Member pursuant to this Section 7.3 shall be treated as an advance on future distributions that would otherwise be made to such Member pursuant to Section 7.2 and shall reduce any such distributions.

7.4 Indemnification and Reimbursement for Payments on Behalf of a Member. Except as otherwise provided in this Agreement, if the Company is required by law (as determined by the Tax Matters Partner based on the advice of legal or tax counsel to the Company) to make any payment on behalf of a Member in its capacity as such (including in respect of withholding taxes, personal property taxes, and unincorporated business taxes, etc.), then such Member (the "Indemnifying Member") will indemnify the Company in full for the entire amount paid, including interest, penalties and expenses associated with such payment. At the option of the Board, the amount to be indemnified may be charged against a Capital Account of the Indemnifying Member, and, at the option of the Board, either:

(a) promptly upon notification of an obligation to indemnify the Company, the Indemnifying Member will make a cash payment to the Company in an amount equal to the full amount to be indemnified (and the amount paid will be added to the Indemnifying Member's Capital Account but will not be deemed to be a Capital Contribution), or

(b) the Company will reduce distributions which would otherwise be made to the Indemnifying Member until the Company has recovered the amount to be indemnified (and the amount of such reduction will be deemed to have been distributed for all purposes, but such deemed distribution will not further reduce the Indemnifying Member's Capital Account).

A Member's obligation to make contributions to the Company under this <u>Section 7.4</u> will survive the termination, dissolution, liquidation and winding up of the Company, and for purposes of this <u>Section 7.4</u>, the Company will be treated as continuing in existence. The Company may pursue and enforce all rights and remedies it may have against each Member under this <u>Section 7.4</u>, including instituting a lawsuit to collect such contribution with interest calculated at a rate equal to the Company's and its subsidiaries' effective cost of borrowed funds.

7.5 Class A-2 Adjustments. In the event that the Company has made distributions pursuant to Section 7.2(a) or (b) after the date of this Agreement but prior to any Class A-2 Adjustment, the Company shall promptly provide written notice of such Class A-2 Adjustment to the Members, and (i) each Class A Member, Class A-1 Member and Class A-2 Member shall, within ten (10) Business Days after the furnishing of such notice, pay to the Company an amount in cash equal to such Member's Adjustment Decrease Amount, if any (all such payments received by the Company, the "Aggregate Adjustment Payment"), and (ii) as soon as practicable after the Company has received any such payments, the Company shall distribute, in cash, the Aggregate Adjustment Payment to the Members in proportion to their respective Adjustment Increase Amounts, if any; provided, however, that, in the event that any Member does not, within the time period set forth in clause (i) above, make payment to the Company in the full amount of its Adjustment Decrease Amount, if any, (x) such Member's unpaid Adjustment Decrease Amount shall bear interest until repaid in full, calculated at a rate equal to fifteen percent (15%) per annum compounded semi-annually and, (y) in the event that the Company makes any distribution pursuant to Section 7.2 prior to the repayment in full of such Member's Adjustment Decrease Amount (including any accrued interest), the Company shall withhold such unpaid Adjustment Decrease Amount from the amount of distributions such Member would otherwise have been entitled to receive, and the Company shall distribute such withheld amount to the other Members in proportion to their respective unpaid Adjustment Increase Amounts, if any; and provided, further, that any payments made, or distributions received, by any Member in respect of its Adjustment Decrease Amount or Adjustment

Increase Amount, as applicable (not including the amount of interest accrued on any Adjustment Decrease Amount which is not timely repaid), shall be deemed to be an adjustment to the aggregate amount of distributions received by such Member pursuant to <u>Section 7.2(a)</u> or <u>(b)</u>, as applicable, and the amount of such Member's Unreturned Capital Contributions and Unpaid Preferred Return Amount shall be adjusted accordingly.

ARTICLE VIII Allocations

8.1 <u>Allocations of Profits and Losses</u>. After giving effect to the allocations set forth in <u>Section 8.2</u>, the Company's Profit and Loss (and items of income, gain, credit, deduction, loss and expense) for any Fiscal Year (or other applicable period) shall be allocated among the Members so as, to the extent possible, to cause each Member's Capital Account balance, as increased by the amount of such Member's share of Company Minimum Gain, the amount of such Member's share of Member nonrecourse debt minimum gain (as defined in Treasury Regulation Section 1.704-(i)(5)) and any other amount such Member is deemed obligated to restore pursuant to Treasury Regulation Section 1.704-2, to equal, as nearly as possible, the amount that would be distributed to such Member if: (a) the Company were dissolved and terminated, (b) the assets of the Company were sold for an amount equal to their respective Book Values, (c) all Company liabilities were satisfied (limited with respect to each nonrecourse liability to the Book Value of the assets(s) securing such liability) and (d) the net proceeds were distributed in accordance with <u>Section 10.2</u>.

8.2 <u>Regulatory and Special Allocations</u>. Notwithstanding the provisions of <u>Section 8.1</u>:

(a) To the extent an adjustment to the adjusted tax basis of any Company asset pursuant to Code Section 734(b) or 743(b) is required to be taken into account in determining Capital Accounts, the amount of such adjustment to the Capital Accounts shall be treated, as provided in Treasury Regulation Section 1.704-1(b)(2)(iv)(m), as an item of Profit (if the adjustment increases the basis of the asset) or Loss (if the adjustment decreases such basis) and such Profit or Loss shall be specially allocated to the Members in a manner consistent with the manner in which their Capital Accounts are required to be adjusted pursuant to such Section of the Treasury Regulations.

(b) If there is a net decrease in Company Minimum Gain (determined according to Treasury Regulation Section 1.704-2(d)(1)) during any Taxable Year, each Member shall be specially allocated Profits for such Taxable Year (and, if necessary, subsequent Taxable Years) in an amount equal to such Member's share of the net decrease in Company Minimum Gain, determined in accordance with Treasury Regulation Section 1.704-2(g). The items to be so allocated shall be determined in accordance with Treasury Regulation Section 1.704-2(f)(6) and 1.704-2(j)(2). This paragraph is intended to comply with the minimum gain chargeback requirement in Treasury Regulation Section 1.704-2(f) and shall be interpreted consistently therewith.

(c) Member Nonrecourse Deductions shall be allocated in the manner required by Treasury Regulation Section 1.704-2(i). Except as otherwise provided in Treasury Regulation Section 1.704-2(i)(4), if there is a net decrease in Member Minimum Gain during any Taxable Year, each Member that has a share of such Member Minimum Gain shall be specially allocated Profits for such Taxable Year (and, if necessary, subsequent Taxable Years) in an amount equal to that Member's share of the net decrease in Member Minimum Gain. Items to be allocated pursuant to this paragraph shall be determined in accordance with Treasury Regulation Section 1.704-2(i)(4) and 1.704-2(j)(2). This paragraph is intended to comply with the minimum gain chargeback requirements in Treasury Regulation Section 1.704-2(i)(4) and shall be interpreted consistently therewith.

(d) In the event any Member unexpectedly receives any adjustments, allocations or distributions described in Treasury Regulation Section 1.704-1(b)(2)(ii)(d)(4), (5) or (6), Profits shall be specially allocated to such Member in an amount and manner sufficient to eliminate the Adjusted Capital Account Deficit created by such adjustments, allocations or distributions as quickly as possible, <u>provided</u> that an allocation pursuant to this <u>Section 8.2(d)</u> shall be made if and only to the extent that such Member would have an Adjusted Capital Account Deficit after all other allocations provided for in this <u>Article VIII</u> have been tentatively made as if this <u>Section 8.2(d)</u> were not a term of this Agreement. This paragraph is intended to comply with the qualified income offset requirement in Treasury Regulation Section 1.704-1(b)(2)(ii)(d) and shall be interpreted consistently therewith.

(e) If any Member has an Adjusted Capital Account Deficit at the end of any Taxable Year, each such Member shall be specially allocated Profits in the amount of such excess as quickly as possible, <u>provided</u> that an allocation pursuant to this <u>Section 8.2(e)</u> shall be made only if and to the extent that such Member would have an Adjusted Capital Account Deficit in excess of such sum after all other allocations provided for in this <u>Article VIII</u> have been made as if <u>Section 8.2(d)</u> above and this <u>Section 8.2(e)</u> were not in this Agreement.

(f) The allocations set forth in <u>paragraphs (a)</u>, (b), (c), (d) and (e) above (the "<u>Regulatory Allocations</u>") are intended to comply with certain requirements of the Treasury Regulations under Code Section 704. Notwithstanding any other provisions of this <u>Article VIII</u> (other than the Regulatory Allocations), the Regulatory Allocations shall be taken into account in allocating Profits and Losses among Members so that, to the extent possible, the net amount of such allocations of Profits and Losses and other items and the Regulatory Allocations to each Member shall be equal to the net amount that would have been allocated to such Member if the Regulatory Allocations had not occurred.

8.3 <u>Curative Allocations</u>. If the Tax Matters Partner determines, after consultation with counsel experienced in income tax matters, that the allocation of any item of Company income, gain, loss, deduction or credit is not specified in this <u>Article VIII</u> (an "<u>unallocated item</u>"), or that the allocation of any item of Company income, gain, loss, deduction or credit hereunder is clearly inconsistent with the Members' economic interests in the Company (determined by reference to the general principles of Treasury Regulation Section 1.704-1(b)(3)(ii)) (a "<u>misallocated item</u>"), then the Board may allocate such unallocated items, or reallocate such misallocated items, to reflect such economic interests; provided that no such allocation will be made without the prior consent of each Member that would be affected

thereby (which consent no such Member may unreasonably withhold) and <u>provided further</u> that no such allocation shall have any material effect on the amounts distributable to any Member, including the amounts to be distributed upon the complete liquidation of the Company.

8.4 Tax Allocations.

(a) All income, gains, losses, deductions and credits of the Company shall be allocated, for federal, state and local income tax purposes, among the Members in accordance with the allocation of such income, gains, losses, deductions and credits among the Members for computing their Capital Accounts, except that if any such allocation for tax purposes is not permitted by the Code or other applicable law, the Company's subsequent income, gains, losses, deductions and credits shall be allocated among the Members for tax purposes, to the extent permitted by the Code and other applicable law, so as to reflect as nearly as possible the allocation set forth herein in computing their Capital Accounts. Each item of income, gain, loss, deduction and credit realized by the Company in any Taxable Year shall be allocated *pro rata* to the Members according to the amount of Profit or Loss, as the case may be, allocated to them in such year.

(b) Items of Company taxable income, gain, loss and deduction with respect to any property contributed to the capital of the Company shall be allocated among the Members in accordance with Code Section 704(c) and the traditional method of Treasury Regulation Section 1.704-3(b), or such other method elected by the Tax Matters Partner, so as to take account of any variation between the adjusted basis of such property to the Company for federal income tax purposes and its Book Value.

(c) If the Book Value of any Company property is adjusted pursuant to <u>Section 6.2(a)(iv</u>), subsequent allocations of items of taxable income, gain, loss and deduction with respect to such property shall take account of any variation between the adjusted basis of such property for federal income tax purposes and its Book Value in the same manner as under Code Section 704(c).

(d) Allocations of tax credit, tax credit recapture, and any items related thereto shall be allocated to the Members according to their interests in such items as determined by the Board taking into account the principles of Treasury Regulation Section 1.704-1(b)(4)(ii).

(e) Allocations pursuant to this <u>Section 8.4</u> are solely for purposes of federal, state and local taxes and shall not affect, or in any way be taken into account in computing, any Member's Capital Account or share of Profits, Losses, distributions or other items pursuant to any provisions of this Agreement.

ARTICLE IX Elections and Reports

9.1 <u>**Generally**</u>. The Company will keep appropriate books and records with respect to the Company's business, including all books and records necessary to provide any information, lists and copies of documents required to be provided pursuant to <u>Section 9.3</u>.

9.2 <u>**Tax Status**</u>. The Members intend that the Company be treated as a partnership for federal, state and local income tax purposes and the Company and each Member shall file all tax returns on the basis consistent therewith.

9.3 Reports. The Company will use reasonable efforts to deliver or cause to be delivered, by March 15 of each year or as soon as practicably possible, to each person who was a Member at any time during the previous Taxable Year, all information (including a Schedule K-1) reasonably necessary for the preparation of such person's United States federal income tax returns and any state, local and foreign income tax returns which such person is required to file as a result of the Company being engaged in a trade or business within such state, local or foreign jurisdiction, including a statement showing such person's share of income, gains, losses, deductions and credits for such year for United States federal income tax purposes (and, if applicable, state, local or foreign income tax purposes).

9.4 <u>**Tax Elections**</u>. The Tax Matters Partner will determine whether to make or revoke any available election pursuant to the Code. Each Member will upon request supply the information necessary to give proper effect to any such election.

9.5 Tax Controversies. The initial Tax Matters Partner shall be the HC2 Member. A Majority in Voting Interest may replace the Tax Matters Partner at any time and from time to time. The Tax Matters Partner is authorized and required to represent the Company (at the Company's expense) in connection with all examinations of the Company's affairs by tax authorities, including resulting administrative and judicial proceedings, and to expend Company funds for professional services and costs associated therewith. Each Member agrees to cooperate with the Tax Matters Partner and to do or refrain from doing any or all things reasonably requested by the Tax Matters Partner with respect to the conduct of such proceedings. The Tax Matters Partner will have sole discretion to determine whether the Company (either in its own behalf or on behalf of the Members) will contest or continue to contest any tax deficiencies assessed or proposed to be assessed by any taxing authority. Any deficiency for taxes imposed on or attributable to any Member (including penalties, additions to tax or interest imposed with respect to such taxes) will be paid by such Member, and if required to be paid (and actually paid) by the Company, will be recoverable from such Member as provided in <u>Section 7.4</u>.

9.6 Code Section 83 Safe Harbor Election.

(a) <u>Safe Harbor Election</u>. Notwithstanding <u>Section 9.4</u>, by executing this Agreement, each Member authorizes and directs the Company to elect to have the "Safe Harbor" described in the proposed Revenue Procedure set forth in Internal Revenue Service Notice 2005-43 (the "<u>Notice</u>") apply to any interest in the Company transferred to a service provider or a Person who provides services for the benefit of the Company (should such an election be applicable to a Person who provides services for the benefit of the Company) by the Company on or after the effective date of such Revenue Procedure (or any substantially similar Revenue Procedure or other guidance issued by the Internal Revenue Service) in connection with services provided to (or, as applicable, the benefit of) the Company; <u>provided</u> that all Members shall be provided with written notice of any such Safe Harbor election. For purposes of making such Safe Harbor election, the Tax Matters Partner is hereby designated as the "partner who has responsibility for federal income tax reporting" by the Company and, accordingly, that execution of such Safe Harbor election by the Tax Matters Partner constitutes execution of a "Safe Harbor Election" in accordance with Section 3.03(1) of the Notice. The Company and each Member hereby agree to comply with all requirements of the Safe Harbor described in the Notice (and any substantially similar Revenue Service), including, without limitation, the requirement that each Member shall prepare and file all federal income tax refects of each Safe Harbor Partnership Interest issued by the Internal Revenue Service).

Failure to Comply. Any Member or former Member that fails to comply with requirements set forth in Section 9.6(a) shall (b) indemnify and hold harmless the Company and each adversely affected Member and former Member from and against any and all losses, liabilities, taxes, damages, judgments, fines, costs, penalties, amounts paid in settlement and reasonable out-of-pocket costs and expenses incurred in connection therewith (including, without limitation, costs and expenses of suits and proceedings, and reasonable fees and disbursements of counsel), in each case resulting from such Member's or former Member's failure to comply with such requirements. The Board may offset distributions to which a Person is otherwise entitled under this Agreement against such Person's obligation to indemnify the Company and any other Person under this Section 9.6(b) (and any amount so offset with respect to such Person's obligation to indemnify a Person other than the Company shall be paid over to such other Person by the Company). A Member's obligations to comply with the requirements of <u>Section 9.6(a)</u> and to indemnify the Company and any Member or former Member under this <u>Section 9.6(b)</u> shall survive such Member's ceasing to be a Member of the Company and/or the termination, dissolution, liquidation and winding up of the Company, and, for purposes of this Section 9.6, the Company shall be treated as continuing in existence. The Company and any Member or former Member may pursue and enforce all rights and remedies it may have against each Member or former Member under this Section 9.6(b), including (i) instituting a lawsuit to collect such indemnification and contribution, with interest calculated at a rate equal to the prime rate plus three (3) percentage points per annum (but not in excess of the highest rate per annum permitted by law), compounded on the last day of each fiscal guarter and (ii) specific performance and/or immediate injunctive or other equitable relief from any court of competent jurisdiction (without the necessity of showing actual money damages, or posting

any bond or other security) in order to enforce or prevent any violation of the provisions of Section 9.6(a).

(c) <u>Certain Amendments</u>. Each Member authorizes the Board to amend <u>Section 9.6(a)</u> and <u>Section 9.6(b)</u> to the extent necessary to achieve substantially the same tax treatment with respect to any interest in the Company transferred by the Company to a service provider in connection with services provided to the Company or for the benefit of the Company (should such an election be applicable to a person who provides services for the benefit of the Company) as set forth in Section 4 of the Notice (*e.g.*, to reflect changes from the rules set forth in the Notice in subsequent Internal Revenue Service guidance), <u>provided</u> that such amendment is not materially adverse to any Member (as compared with the after-tax consequences that would result if the provisions of the Notice applied to all interests in the Company transferred to a service provider by the Company in connection with services provided to or for the benefit of the Company).

9.7 Engagement in Trade or Business in the United States. The Company shall conduct any future activity that would constitute a United States trade or business for tax purposes through GMSL or another Subsidiary of GMSL.

ARTICLE X DISSOLUTION AND LIQUIDATION

10.1 <u>Dissolution</u>. The Company shall be dissolved and its affairs wound up only upon the happening of any of the following events:

(a) Upon the election to dissolve the Company by action of a Majority in Voting Interest; or

(b) The entry of a decree of judicial dissolution under § 18-802 of the Delaware Act; <u>provided</u>, that, notwithstanding anything contained herein to the contrary, no Member shall make an application for the dissolution of the Company pursuant to § 18-802 of the Delaware Act without the unanimous approval of the Members.

Dissolution of the Company shall be effective on the day on which the event occurs giving rise to the dissolution, but the Company shall not terminate until the winding up of the Company has been completed, the assets of the Company have been distributed as provided in <u>Section 10.2</u> and the Certificate shall have been canceled.

10.2 Liquidation.

(a) Liquidator. Upon dissolution of the Company, the Board will appoint a Person to act as the "Liquidator," and such Person shall act as the Liquidator unless and until a successor Liquidator is appointed as provided in this <u>Section 10.2</u>. The Liquidator will agree not to resign at any time without thirty (30) days' prior written notice to the Board. The Liquidator may be removed at any time, with or without cause, by notice of removal and appointment of a successor Liquidator approved by the Board. Any successor Liquidator will succeed to all rights, powers and duties of the former Liquidator. The right to appoint a successor or substitute Liquidator in the manner provided in this <u>Section 10.2</u> will be recurring and continuing for so long as the functions and services of the Liquidator are authorized to continue under the provisions of this Agreement, and every reference in this <u>Section 10.2</u>. The Liquidator will be deemed to refer also to any such successor or substitute Liquidator appointed in the manner provided in this <u>Section 10.2</u>. The Liquidator will receive as compensation for its services (1) no additional compensation, if the Liquidator is an employee of the Company or any of its subsidiaries, or (2) if the Liquidator is not such an employee, such compensation as the Board may approve, plus, in either case, reimbursement of the Liquidator's out-of-pocket expenses in performing its duties.

(b) <u>Liquidating Actions</u>. The Liquidator will liquidate the assets of the Company and apply and distribute the proceeds of such liquidation, in the following order of priority, unless otherwise required by mandatory provisions of applicable law:

(i) *First*, to the payment of the Company's debts and obligations to its creditors (including Members), including sales commissions and other expenses incident to any sale of the assets of the Company, in order of the priority provided by law.

- (ii) <u>Second</u>, to the establishment of and additions to such reserves as the Board deems necessary or appropriate.
- (iii) *Third*, to the Members, in accordance with <u>Section 7.2</u>.

The reserves established pursuant to <u>clause (ii)</u> above will be paid over by the Liquidator to a bank or other financial institution, to be held in escrow for the purpose of paying any such contingent or unforeseen liabilities or obligations and, at the expiration of such period as the Board deems advisable, such reserves will be distributed to the Members in accordance with <u>Section 7.2</u> in the manner provided above in this <u>Section 10.2(b)</u>. The allocations and distributions provided for in this Agreement are intended to result in the Capital Account of each Member immediately prior to the distribution of the Company's assets pursuant to this <u>Section 10.2(b)</u> being equal to the amount distributable to such Member pursuant to this <u>Section 10.2(b)</u>.

(c) <u>Distribution in Kind</u>. Notwithstanding the provisions of <u>Section 10.2(b)</u> which require the liquidation of the assets of the Company, but subject to the order of priorities set forth in <u>Section 10.2(b)</u>, if upon dissolution of the Company the Board determines that an immediate sale of part or all of the Company's assets would be impractical or could cause undue loss to the Members, the Board may, in its sole discretion, defer the liquidation of any assets except those necessary to satisfy Company liabilities and reserves, and may, in its absolute discretion, distribute to the Members, in lieu of cash, as tenants in common and in accordance with the provisions of

<u>Section 10.2(b)</u>, undivided interests in such Company assets as the Liquidator deems not suitable for liquidation on a *pro rata* basis among the Members of each class based on their ownership of such class of Units. Any such distribution in kind will be subject to such conditions relating to the disposition and management of such properties as the Liquidator deems reasonable and equitable and to any agreements governing the operating of such properties at such time. For purposes of any such distribution, any property to be distributed will be valued at its Fair Market Value.

(d) <u>Reasonable Time for Winding Up</u>. A reasonable time will be allowed for the orderly winding up of the business and affairs of the Company and the liquidation of its assets pursuant to <u>Section 10.2(b)</u> in order to minimize any losses otherwise attendant upon such winding up. Distributions upon liquidation of the Company (or any Member's interest in the Company) and related adjustments will be made by the end of the Fiscal Year of the liquidation (or, if later, within ninety (90) days after the date of such liquidation) or as otherwise permitted by Treasury Regulation Section 1.704-1(b)(2)(ii)(b).

(e) <u>Termination</u>. Upon completion of the distribution of the assets of the Company as provided in <u>Section 10.2(b)</u> hereof, the Company shall be terminated and the Liquidator shall cause the cancellation of the Certificate in the State of Delaware and of all qualifications and registrations of the Company as a foreign limited liability company in jurisdictions other than the State of Delaware and shall take such other actions as may be necessary to terminate the Company.

ARTICLE XI Transfer of Units

11.1 <u>Restrictions on Transfer</u>.

(a) <u>General Restrictions</u>.

(i) Each Member acknowledges and agrees that such Member shall not Transfer any Unit(s) except in accordance with the provisions of this <u>Article XI</u>, <u>Article XIII</u> and, to the extent applicable to such Member, any Grant Agreement or any Subscription Agreement. Any attempted Transfer in violation of the preceding sentence shall be deemed null and void for all purposes, and the Company will not record any such Transfer on its books or treat any purported Transfere as the owner of such Unit(s) or a Member for any purpose.

(ii) Notwithstanding anything to the contrary in this Agreement, except in connection with a Company Sale or a Drag Sale, no Unit may be Transferred and the Company shall not issue any Units unless (A) such Transfer or issuance, as the case may be, shall not affect the Company's existence or qualification as a limited liability company under the Delaware Act, and (B) such Transfer or issuance, as the case may be, shall not cause the Company to be classified as a publicly traded partnership treated as a corporation or otherwise as other than a partnership for United States federal income tax purposes.

(iii) The Company shall have the right to require, as a condition to any Transfer (other than in connection with a Company Sale or a Drag Sale), receipt of an opinion of counsel, which opinion of counsel shall be reasonably satisfactory to the Company, to the effect that such Transfer is not required to be registered under the Securities Act and is not in violation of any other applicable securities law, provided, however, that in the case of a Transfer to an accredited investor (as defined below), the Transferee may provide, in lieu of such opinion, evidence, reasonably satisfactory to the Board, that the Transferee is an accredited investor under applicable securities laws and that such Transfer is not required to be registered under the Securities Act and is not in violation of any other applicable securities laws and that such Transfer is not required to be registered under the Securities Act and is not in violation of any other applicable securities law (such opinion of counsel or evidence in lieu of such opinion, the "Transfer Opinion").

(b) <u>Additional Restrictions</u>. From and after the date hereof, in addition and subject to the restrictions set forth in <u>Section 11.1(a)</u>, no Member or other holder of Units may Transfer any direct or indirect interest in such Member's or holder's Units other than:

- (i) the HC2 Member:
 - (A) to any Person pursuant to a Permitted Transfer;
 - (B) as a Transferring Member to any Person subject to the obligations set forth in <u>Section 11.4;</u>
 - (C) to any Person pursuant to Section 11.6 in connection with a Company Sale; or
 - (D) to the applicable purchaser pursuant to <u>Section 11.8</u> in connection with a Drag Sale;
- (ii) the Fugro Member:
 - (A) to any Person pursuant to a Permitted Transfer;
 - (B) as a Tagging Member to the Proposed Tag Purchaser pursuant to <u>Section 11.4;</u>
 - (C) to the applicable purchaser pursuant to <u>Section 11.6</u> in connection with a Company Sale;

(D) at any time on or after the second anniversary of the Fugro Acquisition Date, to any Person subject to the obligations set forth in <u>Section 11.7</u>; or

(E) to any Person pursuant to Section 11.8 in connection with a Fugro Member Sale or a Drag Sale;

(iii) any other Member or holder of Units (other than the HC2 Member, the Fugro Member or their respective Permitted Transferees):

(A) to any Person pursuant to a Permitted Transfer;

- (B) as a Tagging Member to the Proposed Tag Purchaser pursuant to <u>Section 11.4</u>;
- (C) to the applicable purchaser pursuant to <u>Section 11.6</u> in connection with a Company Sale;
- (D) to the applicable purchaser pursuant to <u>Section 11.8</u> in connection with a Drag Sale;

(E) in the case of any Class A Units or Class A-2 Units, any Transfer in accordance with the provisions of any arrangement approved by the Board to the Company in connection with any repurchase of such Units; or

(F) in the case of any Incentive Interests or Class A-1 Units, any Transfer in accordance with the provisions of any Grant Agreement or arrangement approved by the Board to the Company in connection with any repurchase of such Units in connection with the termination of his or her employment or any consulting relationship with the Company or any of its Subsidiaries.

(c) No Circumvention of Transfer Restrictions.

(i) Each Member agrees that the Transfer restrictions in this Agreement may not be avoided by the holding of Units directly or indirectly through a Person that can itself be sold in order to dispose of an interest in Units free of such restrictions. Any Transfer of any securities (or other interest) resulting in any change in the control, directly or indirectly, of a Member or of any other Person having control, directly or indirectly, over any such Member shall be treated as being a Transfer of the Units held by such Member, and the provisions of this Agreement that apply in respect of the Transfer of Units shall thereupon apply in respect of the Units so held; <u>provided</u>, that a change in control of either the HC2 Member Parent or the Fugro Member Parent shall not constitute a Transfer of Units.

(ii) Zencor Holdings hereby acknowledges and agrees that:

(A) as of the date hereof, Dick Fagerstal owns, directly or indirectly, 100% of the economic, voting and all other rights relating to Zencor Holdings (and each of its parent entities (other than a natural person)), and Zencor Holdings shall not permit any, direct or indirect, Transfer of any securities (or other interest) issued by it (or any of its parent entities (other than a natural person)) without the prior written consent of the Majority in Voting Interest, and that it shall treat any attempted Transfer of any such securities (or other interest) in violation of this <u>clause (A)</u> as null and void for all purposes, and shall not record any such Transfer on its books or treat any purported Transferee as the owner of such securities (or other interest) for any purpose; provided however, that the prior written consent of the

Majority in Voting Interest shall not be unreasonably withheld in connection with any Transfer of any such securities (or other interest) that would otherwise be in violation of this <u>clause (A)</u> (1) pursuant to applicable laws of descent and distribution as they relate to Dick Fagerstal, as applicable, or (2) for bona fide estate planning purposes (which bona fide estate planning purposes, if requested by the Majority in Voting Interest, shall be reasonably verified by the Majority in Voting Interest);

(B) neither it nor any of its parent entities (other than a natural person) has incurred nor will any of them incur any obligations or liabilities (whether contingent or absolute) other than obligations or liabilities pursuant to this Agreement or any other agreement contemplated thereby and hereby, contracts for the maintenance of Zencor Holdings' registered agent in the state of its incorporation or formation and obligations with respect to the payment of franchise (or similar) tax in the state of its formation;

(C) neither it nor any of its parent entities (other than a natural person) has nor will any of them have any other asset or interest other than the Units (or the equity interests of its direct Subsidiary) and the rights pursuant to this Agreement or any other agreement contemplated thereby and hereby;

(D) neither it nor any of its parent entities (other than a natural person) has ever conducted, nor will any of them conduct any business other than ownership of the Units; and

(E) neither it nor any of its parent entities (other than a natural person), nor will any of them become, a party to or bound by, any contract other than contracts for the maintenance of Zencor Holdings' registered agent in the state of its incorporation or formation, and contracts with an accounting firm or other tax professional for the preparation of tax returns an account applications with the bank(s) for bank account(s).

11.2 Effect of Transfer.

(a) Notwithstanding anything to the contrary in this Agreement, no Transferee of any Unit(s) received pursuant to a Transfer (but excluding (i) Transferees that were Members immediately prior to such a Transfer, who shall automatically become a Member with respect to any additional Units they so acquire, and (ii) any Transferee that is an Affiliate of the Member Transferring Units to such Transferee, who shall automatically become a Member upon such Transferee executing and delivering to the Secretary of the Company a Joinder) shall become a Member in respect of or be deemed to have any ownership rights in the Unit(s) so Transferred unless the purported Transferee is admitted as a Member as set forth in Section 11.3.

(b) Following a Transfer of any Unit(s) that is permitted under this <u>Article XI</u>, the Transferee of such Unit(s) shall succeed to the Capital Account associated with such Unit(s) and shall receive allocations and distributions under <u>Articles VI</u>, <u>VII</u>, <u>VIII</u> and <u>X</u> in respect of such Unit(s). Notwithstanding the foregoing, Profits, Losses and other items will be allocated between the transferor and the Transferee according to Code Section 706.

(c) Any Member who Transfers all of his or its Units (i) shall automatically and without any further action cease to be a Member upon such Transfer, and (ii) shall no longer possess or have the power to exercise any rights or powers of a Member of the Company.

(d) In the case of any Permitted Transfer to an Affiliate, if such Transferee ceases to be an Affiliate of such Transferring entity, such Transferee shall re-Transfer the Units Transferred by such Member back to the Transferring entity or to another Affiliate thereof and, pending such re-Transfer, such Transferee shall not have any rights under this Agreement in respect of such Units Transferred by such Member.

11.3 <u>Procedures for Transfer</u>.

(a) Subject in all events to the general restrictions on Transfers contained in this Agreement and, to the extent applicable, any Grant Agreement and any Subscription Agreement, no Transfer of Unit(s) (other than a Company Sale or a Drag Sale) may be completed, nor shall the Company be liable or responsible in respect of any Transfer or Transferee, until the prospective Transferee is admitted as a Member.

(b) The Company shall not admit any prospective Transferee as a Member unless (x) such Person executes and delivers to the Secretary of the Company a Joinder (unless such Transferee is already a Member immediately prior to such Transfer) and (y) the Company receives (or waives in writing its right to receive) the Transfer Opinion.

(c) If a Transfer is consummated in accordance with the terms of this Agreement, the Secretary of the Company shall amend the Members Schedule to reflect such Transfer. Upon the amendment of the Members Schedule by the Secretary of the Company, such prospective Transferee shall be admitted as a Member and deemed listed as such on the books and records of the Company and thereupon the Company shall reissue the applicable Units in the name of such prospective Transferee.

11.4 Tag-Along Rights.

(a) <u>Tag-Along Process</u>.

(i) *Sale Notice*. If any of (x) the HC2 Member, (y) Zencor Holdings or (z) any Class A Member or group of Class A Members that hold, individually or in the aggregate, Class A Units representing a Class A/A-1/A-2 Percentage Interest of at least four percent (4%) (each, a "<u>Transferring Member</u>"), proposes to Transfer, in a single transaction or series of related transactions (other than pursuant to a Permitted Transfer (that does not constitute a Transfer to any Person that is already a Member prior to such Transfer) or a Company Sale), any Class A Units, the Transferring Member(s) shall deliver a written notice (the "<u>Sale Notice</u>") to the Company, each Class A Member and each Class A-2 Member at least thirty

(30) calendar days prior to making such Transfer (the "<u>Tag Transfer</u>"), specifying in reasonable detail the identity of the prospective Transferee(s) (the "<u>Proposed Tag Purchaser</u>"), the number of Class A Units to be Transferred (the "<u>Offered Units</u>") and the terms and conditions of the Tag Transfer. Each Class A Member and Class A-2 Member may elect (each such Member that so elects, a "<u>Tagging Member</u>") to participate in the Tag Transfer upon the terms and conditions set forth in the Sale Notice by delivering written notice (each, a "<u>Tag Notice</u>") to the Transferring Member(s) within ten (10) calendar days after delivery of the Sale Notice (the "<u>Tag Period</u>"), which Tag Notice shall specify the number of Class A Units or Class A-2 Units that such Tagging Member desires to include in such Tag Transfer; <u>provided</u>, <u>however</u>, that (A) each Tagging Member may not elect to sell any Units other than Class A Units or Class A-2 Units and (B) no Class A Unit that is subject to vesting shall be entitled to be sold pursuant to this <u>Section 11.4</u> unless such Unit has fully vested. Delivery of a Tag Notice shall constitute an irrevocable and unconditional offer by such Tagging Member to sell to the Proposed Tag Purchasers the number of Class A Units or Class A-2 Units set forth therein on the same terms and conditions as set forth in the Sale Notice; <u>provided</u> that such Tagging Member shall not be obligated to consummate such Transfer if the Transferring Member(s) either fail to sell their Offered Units or otherwise deviate from the terms set forth in such Sale Notice.

(ii) *Amended Sale Notice*. Notwithstanding anything to the contrary herein, the Transferring Member(s) shall not consummate the Tag Transfer contemplated by the Sale Notice on material terms more favorable in the aggregate to the Transferring Member(s) than those set forth in the Sale Notice (including as to price per Unit, amount of Units or form of consideration to be received) unless (x) each Class A Member and Class A-2 Member has elected to be a Tagging Member for the maximum number of Class A Units and Class A-2 Units that can be included in the Offered Units or (y) the Transferring Member(s) shall first have delivered a second notice setting forth such more favorable material terms (the "<u>Amended Sale Notice</u>") to each Class A Member and Class A-2 Member. Each Class A Member and Class A-2 Member receiving an Amended Sale Notice may elect to participate in the contemplated Tag Transfer on such amended terms by delivering written notice (also, a "<u>Tag Notice</u>") to the Transferring Member(s) not later than five (5) calendar days after delivery of the Amended Sale Notice.

(iii) *Closing of Tag Transfer*. The closing of any Tag Transfer shall be held at the principal office of the Company at 11:00 a.m. local time on a Business Day chosen by the Transferring Member(s) (upon at least five (5) days' notice to the Tagging Members); <u>provided</u> that such closing may be held at such other time and place as the parties to the Tag Transfer may agree.

(b) <u>Tagging Members' Rights and Obligations</u>.

(i) Transfer Units.

(A) If one or more Class A Members or Class A-2 Members elect to be Tagging Members and give the Transferring Member(s) a timely Tag Notice, then each such Tagging Member shall be entitled to request to sell, subject to the other provisions of this <u>Section 11.4</u>, in the contemplated Tag Transfer a percentage of Class A Units and Class A-2 Units held by such Tagging Member that is equal to (1) the total number of Class A Units and Class A-2 Units which the Transferee is willing to purchase multiplied by (2) a fraction (A) the numerator of which is the number of Class A Units and Class A-2 Units held by such Tagging Member, and (B) the denominator of which is the aggregate number of Class A Units and Class A-2 Units owned by the Transferring Member(s) and all Tagging Members (including such Tagging Member) who are participating in the applicable Tag Transfer.

(B) The Transferring Member(s) shall use commercially reasonable efforts to cause the Proposed Tag Purchasers to agree to acquire all of the Class A Units and Class A-2 Units desired to be sold by such Tagging Members set forth in each such Tag Notice.

(C) In the event that the Proposed Tag Purchasers are unwilling or unable to acquire all Class A Units and Class A-2 Units proposed to be included in the Tag Transfer upon such terms, the Transferring Member(s) shall have the option to either (1) cancel such Tag Transfer (in which case all Class A Units and Class A-2 Units shall remain subject to all of the restrictions of this Agreement, including the provisions of this <u>Section 11.4</u>), or (2) allocate the total number of Class A Units and Class A-2 Units that the Proposed Tag Purchasers are willing to purchase among the Transferring Member(s) and the Tagging Members on a *pro rata* basis in proportion to the Class A Units and Class A-2 Units proposed to be sold in such Tag Transfer.

(ii) Consideration.

(A) If the applicable Tag Transfer is actually consummated, then (1) each Tagging Member shall receive in exchange for or in respect of, as applicable, the Class A Units and Class A-2 Units held by such Tagging Member the same consideration per Unit from such Tag Transfer as received by the Transferring Member(s), and (2) each Tagging Member shall pay its own costs of any sale and a *pro rata* share on a several and not joint basis (based on the relative consideration to be received in respect of the Units to be sold) of the expenses incurred by the Transferring Member(s) (to the extent such expenses are incurred for the benefit of all Tagging Members and are not otherwise paid by the Company or the Proposed Tag Purchasers).

(B) In the event that any non-cash consideration is paid by the Proposed Tag Purchaser, each Tagging Member shall be entitled to receive, on a *pro rata* basis, the same proportionate amount of cash and non-cash consideration as received by the Transferring Member(s) (the amount and value of such consideration shall be determined in good faith by the Board).

(iii) Purchase Agreement Obligations. Each Tagging Member that sells Class A Units or Class A-2 Units in a Tag Transfer shall (A) take all reasonably necessary actions as reasonably directed by the Transferring Member(s) in connection with the consummation of the Tag Transfer, including executing the applicable purchase agreement and all documents and instruments which are necessary or desirable to effectuate such Tag Transfer, (B) be obligated to provide the same representations, warranties, covenants and agreements (other than any non-compete or non-solicit covenant or agreement) with respect to such Tagging Member as provided by the Transferring Member(s), and (C) join on a *pro rata* several and not joint basis (based on the relative consideration to be received in respect of the Units to be sold) in any indemnification obligations (including participating in any escrow arrangements) that the Transferring Member(s) agree to provide in connection with such Tag Transfer; <u>provided</u> that, other than in the case of fraud, no Tagging Member's obligations for indemnification and similar obligations shall exceed the aggregate cash proceeds received by, and any amount deposited into escrow on behalf of, such Tagging Member on account of the Class A Units or Class A-2 Units sold in such Tag Transfer.

(c) <u>Effect of Sale Notice</u>. If none of the Class A Members or Class A-2 Members give a Tag Notice prior to the expiration of the Tag Period for giving such notice, then the Transferring Member(s) may Transfer Class A Units to any Person at a price no greater, and on other material terms and conditions that are not more favorable in the aggregate to the Transferring Member(s), than those set forth in the Sale Notice at any time within ninety (90) calendar days after expiration of the Tag Period. Any such Class A Units not Transferred by the Transferring Member(s) during such ninety (90)-calendar day period shall again be subject to the provisions of this <u>Section 11.4</u> upon any subsequent proposed Transfer.

(d) <u>Termination of Restrictions</u>. The restrictions set forth in this <u>Section 11.4</u> shall continue with respect to each Class A Unit until the earlier of (i) the Transfer of such Class A Unit in a Public Sale, a Company Sale or a Drag Sale, or (ii) the consummation of an Initial Public Offering.

11.5 Legend. Any certificates or instruments representing the Units will bear the following legend: "THE OFFER AND SALE OF THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), AND MAY NOT BE SOLD OR TRANSFERRED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT UNDER THE ACT OR AN EXEMPTION FROM REGISTRATION THEREUNDER. THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE FOURTH AMENDED AND RESTATED LIMITED LIABILITY COMPANY AGREEMENT OF GLOBAL MARINE HOLDINGS, LLC (THE "ISSUER"), AS MAY BE AMENDED FROM TIME TO TIME, BY AND AMONG THE ISSUER'S MEMBERS. A COPY OF SUCH AGREEMENT WILL BE FURNISHED WITHOUT CHARGE BY THE ISSUER TO THE HOLDER HEREOF UPON WRITTEN REQUEST."

11.6 Company Sale.

Obligations of the Members. If, at any time, the Majority of Voting Interests elect to consummate, or to cause the Company to (a) consummate, a Company Sale (the "Electing Majority Members"), then such Electing Majority Members shall notify the Company and the other Members in writing at least thirty (30) calendar days prior to the consummation of such Company Sale of the Electing Majority Members' election to exercise its rights under this Section 11.6. If the Electing Majority Members deliver such notice, then, subject to this Section 11.6, (i) the Company shall (x) authorize the Electing Majority Members to initiate a process to seek a Company Sale and direct and control all decisions in connection therewith (including the hiring or termination of any investment bank or professional adviser and making all decisions regarding valuation and consideration), (y) participate in, and cooperate in good faith with, such process, in each case as requested by the Electing Majority Members and (z) take all other necessary and desirable actions as are reasonably directed by the Electing Majority Members in connection with the consummation of any such Company Sale and (ii) the Members, to the extent they have any right or standing to do so, shall vote for, consent to, and raise no objections to the proposed Company Sale, and the Members and the Company shall take all other actions necessary or reasonably required to cause the consummation of such Company Sale on the terms proposed by the Electing Majority Members. Without limiting the foregoing, (A) if the proposed Company Sale is structured as a sale of assets or a merger or consolidation, then each Member shall vote or cause to be voted all Units that such Member holds or with respect to which such Member has the power to direct the voting and which are entitled to vote on such Company Sale in favor of such Company Sale and shall waive any dissenter's rights, appraisal rights or similar rights which such Member may have in connection therewith, (B) if the proposed Company Sale is structured as or involves a sale or redemption of Units, then each Member shall agree to sell such Member's pro rata share of Units being sold in such Company Sale on the same terms and conditions approved by the Electing Majority Members, and applicable to the Electing Majority Members (other than with respect to price per Unit, which will be determined in accordance with the distribution provisions set forth in this Agreement), and such Members shall execute all documents necessary or reasonably required to effectuate such Company Sale approved by the Electing Majority Members in connection with such Company Sale, (C) each Member shall be obligated to provide with respect to such Member the same representations, warranties, covenants and agreements (other than any non-compete or non-solicit covenant or agreement) that the Electing Majority Member agree to provide with respect to their ownership of the Units in connection with such Company Sale, and (D) each Member shall be obligated to join on a pro rata several and not joint basis (based on the relative consideration to be received in respect of the Units to be sold) in any indemnification obligations (including participating in any escrow arrangements) that the Electing Majority Members agree to provide or undertake with respect to itself and/or the Company in connection with such Company Sale; provided that, other than in the case of fraud, no Member's obligations for indemnification and similar obligations shall exceed the aggregate cash proceeds received by, and any amount deposited into escrow on behalf of, such Member on account of the Units sold in such Company Sale; and provided, further, that no Member shall be obligated to enter into indemnification obligations with respect to any representations, warranties or covenants in the nature of those described in <u>clause (C)</u> to the extent relating to or in respect of any other Member or any other Person's Membership Interests. Notwithstanding anything herein to the contrary, in the event that a Member would not be receiving any consideration upon the consummation of a Company Sale with respect to certain of the Units

then held by such Member (as determined based on a deemed distribution in accordance with the distribution provisions set forth in this Agreement of the full equity value of the Company based on the aggregate sale proceeds to be received in such Company Sale), then such Units shall automatically be forfeited and cancelled without any further action or consideration required by any of the parties hereto.

(b) Expenses. If the Company Sale is consummated, then each Member shall bear such Person's *pro rata* share on a several and not joint basis (based upon the relative amount of consideration received by such Member) of the reasonable costs of any Company Sale to the extent such costs are incurred for the benefit of all Members and are not otherwise paid by the Company or the acquiring party. Costs incurred by or on behalf of a Member for such Person's sole benefit shall not be considered costs of the transaction hereunder. Notwithstanding any other provision herein to the contrary, the Electing Majority Members shall have the right at any time prior to the consummation of the Company Sale to abandon the Company Sale, and none of the actions taken or requested to be taken by any Member or the Company prior to such abandonment shall in any way obligate the Company or the Electing Majority Members to consummate the Company Sale.

(c) <u>Consideration</u>. In connection with any Company Sale to which this <u>Section 11.6</u> applies, each Member shall receive in exchange for or in respect of, as applicable, the Units held by such Member, the same portion of the aggregate consideration per Unit from such Company Sale (the amount and value of such consideration shall be determined in good faith by the Board) received by the Electing Majority Members, but giving effect to the rights and preferences of the Units set forth in this Agreement, including <u>Section 7.2</u>, as in effect immediately prior to such Company Sale. Each Member shall take all necessary or reasonably required actions in connection with the distribution or allocation among the Company's Members of the aggregate consideration from such Company Sale as necessary or otherwise reasonably requested by the Company for the purpose of effecting such distribution or allocation; <u>provided</u> that, in connection with any such Company Sale that is structured as an exchange by the Members of all or substantially all of the Units held by the Members, any Person holding rights to acquire Units prior to such Company Sale shall thereafter have the right to only receive the securities that were exchanged for the Units such Person previously had the right to acquire (and if the exchange was in part, a proportional amount thereof). In the event that any non-cash consideration is paid to the Electing Majority Members, each such Member shall be entitled to receive, on a *pro rata* basis, (i) the same proportionate amount of cash and non-cash consideration as received by the right to participate, *pro rata*, in any registration rights received by the Electing Majority Members and (iii) the right to participate, *pro rata*, in any registration rights received by the Electing Majority Members.

(d) <u>Representative</u>. If the Company Sale involves the negotiation or a transaction for which Rule 506 (or any similar rule then in effect) promulgated by the SEC may be available, each Member who is not an "accredited investor," as that term is defined in Regulation D as promulgated under the Securities Act, will, at the request of the Company, appoint either a purchaser representative (as such term is defined in Rule 501 under the Securities Act) designated by the Company, in which event the Company will pay the fees of such purchaser representative, or another purchaser representative (reasonably acceptable to the Company), in which event such holder will be responsible for the fees of the purchaser representative so appointed.

(e) <u>Proxy</u>. Each Member hereby grants the Company such Member's perpetual and irrevocable power of attorney with full right, power and authority to take all actions necessary and/or desirable to sell, transfer or otherwise dispose of all Units (and any other equity of the Company) held by such Member, in connection with the consummation of a Company Sale. Pursuant to such power of attorney, the Company shall have the right to execute any and all documents related to a Company Sale (including documents granting customary indemnities to a buyer of assets or securities consistent with this Agreement) on behalf of such Member. The Company shall exercise such power of attorney in good faith solely in compliance with the terms of this Section 11.6.

(f) <u>Termination</u>. This <u>Section 11.6</u> shall automatically terminate upon the earlier of the consummation of a Company Sale, a Drag Sale or an Initial Public Offering.

11.7 <u>Right of First Offer</u>. Prior to any Transfer by the Fugro Member of all or any of its Class A-2 Units (the "<u>Subject Interest</u>") other than to a Permitted Transferee, the Fugro Member must first comply with the provisions of this <u>Section 11.7</u>:

(a) The Fugro Member shall first deliver to each Class A Member (the "<u>Offerees</u>") a written notice (the "<u>Offer Notice</u>") that sets forth the number of its Class A-2 Units represented by the Subject Interest, the amount that the Fugro Member proposes to be paid for the Subject Interest (the "<u>Sale Price</u>"), the manner of payment and the material terms of such sale. The Offer Notice shall constitute an irrevocable offer by the Fugro Member to sell to the Offerees the Subject Interest for cash at the Sale Price on the terms set forth in the Offer Notice. Each of the Offerees shall have until the 20th Business Day following the delivery of the Offer Notice (the "<u>Offer Period</u>") in which to notify the Fugro Member that it accepts such offer as to all or any portion of the Subject Interest offered to such Offeree for the Sale Price (or the applicable portion thereof based on the portion of the Subject Interest it wishes to purchase) and on the payment terms set forth in the Offer Notice, which notice shall specify the maximum portion of the Subject Interest it wishes to purchase.

(b) If more than one Offeree elects to purchase all or any portion of the Subject Interest prior to the expiration of the Offer Period and such Offerees in the aggregate elect to purchase an amount that exceeds the Subject Interest, the Subject Interest shall be allocated among such Offerees pro rata according to their respective Class A Percentage Interests up to the maximum portion of the Subject Interest specified by each such Offeree in its notice until the entire Subject Interest shall have been so allocated or each Offeree shall have been allocated the maximum portion of the Subject Interest specified by each such Offeree in its notice.

(c) If one or more Offerees accept such offer with respect to all but not less than all of the Subject Interest, a closing of the purchase of such Subject Interest shall take place at the principal office of the Company at 10:00 a.m. on the 15th Business Day after the date on which the Offer Notice was delivered unless the parties agree on a different place or time. The Sale Price shall be payable in accordance with the payment terms of the Offer Notice. The definitive agreements providing for the purchase of the Subject Interest by the Offerees shall include customary representations and warranties by the Fugro Member regarding its valid title to and ownership of the Subject Interest, free and clear of all liens, claims and encumbrances (excluding those arising under securities laws and this Agreement) and its authority, power and right to enter into and consummate the purchase of the Subject Interest.

(d) If the Offerees do not elect to purchase all but not less than all of the Subject Interest for the Sale Price prior to expiration of the Offer Period, the Fugro Member shall have the right, subject to the other provisions of this <u>Article XI</u>, to sell the Subject Interest for a period of 120 calendar days (the "<u>Sale Period</u>") at a price per share no less than the Sale Price and on other terms no more favorable to the Transferees thereof than offered to the Offerees in the Offer Notice; <u>provided</u>, <u>however</u>, that the Fugro Member shall not be permitted to sell the Subject Interest to a Prohibited Transferee. The Fugro Member shall be permitted to engage a qualified investment banking firm or professional adviser in connection with such proposed Transfer; <u>provided</u> that any fees and expenses payable to such investment bank or professional advisor shall be borne solely by the Fugro Member; <u>provided</u>, <u>further</u>, that the Fugro Member shall not be charged any fees or expenses for the management time of the officers of the Company and its Subsidiaries or for any of the other assistance set out in <u>Section 11.7(e)</u> below.

(e) During the Sale Period, the Company shall reasonably cooperate with the Fugro Member in connection with such proposed Transfer, including, subject to the potential Transferee executing and delivering to the Company a confidentiality agreement in form and substance reasonably acceptable to the Company, by (i) permitting any potential Transferee to conduct a due diligence review of the Company, its Subsidiaries and their respective businesses, operations, prospects, assets, liabilities, financial condition and results of operations and (ii) making available the officers of the Company and its Subsidiaries for the purpose of making presentations to any such potential Transferee and answering questions posed by them, which, in any case, shall be during normal business hours, upon reasonable advance notice; provided that the Company will not be subject to a due diligence process which imposes an unreasonable burden to the Company or its operations. If the Fugro Member does not Transfer the Subject Interest before the end of the Sale Period, it may not sell any Subject Interest without repeating the foregoing procedures. Upon the consummation by the Fugro Member of any Transfer pursuant to this Section 11.7(d) of a Subject Interest representing a Class A/A-1/A-2 Percentage Interest of at least fifteen percent (15%), the Fugro Member and its Transferee may, by written notice to the Company given contemporaneously with the consummation of such Transfer, elect to have the Transferee succeed to the specific rights and obligations of the Fugro Member under this Agreement in place of the Fugro Member, and such substitution of the Transferee for the Fugro Member shall take effect without any further action by the Company or the Members.

11.8 Fugro Sale Process.

(a) The process for the sale of the Membership Interests held by the Fugro Member or a sale of the Company as described in this <u>Section 11.8</u> (a "Fugro Sale Process") may be initiated by the Fugro Member by providing written notice to each other Members and the Company at any time on or after the fifth anniversary of the Fugro Acquisition Date; <u>provided</u>, <u>however</u>, that the Fugro Member shall only be permitted to initiate a Fugro Sale Process three (3) times; <u>provided</u>, <u>further</u>, that the Fugro Member shall only be permitted to initiate a Fugro Sale Process for so long as the Fugro Member and its Affiliates continue to hold, in the aggregate, at least 50% of the number of Class A-2 Units acquired by the Fugro Member as of the Fugro Acquisition Date; and <u>provided</u>, <u>further</u>, that, in the event that a Fugro Sale Process does not result in the consummation of a Fugro Member Sale or a Drag Sale, the Fugro Member shall be prohibited from initiating a subsequent Fugro Sale Process until the date which is twelve (12) months from the termination of such prior Fugro Sale Process.

(b) In the event that the Fugro Member initiates a Fugro Sale Process, the HC2 Member and the Fugro Member shall each, at its own cost and expense, select and appoint a qualified investment banking firm (which firm shall be reasonably acceptable to the other Member). Each such investment banking firm shall determine the Fair Market Value of the Company as of such date and, based on such determination with respect to the Company, the Fair Market Value of all of the Membership Interests held by the Fugro Member (the "Fugro Interest") as soon as practicable, but not later than twenty (20) Business Days after the date of its appointment. Within twenty (20) Business Days following the receipt of such Fair Market Value determination of both investment banking firms, the Fugro Member may, in its sole discretion, deliver a written notice (the "Fugro Offer Notice") to the Class A Members (the "Fugro Offerees"), containing an irrevocable offer to sell the Fugro Interest to the Fugro Offerees for cash at the price equal to the average of the Fair Market Value of such Fugro Interest as determined by the investment banking firms (the "Average Price"), or at such lower price as the Fugro Member may determine in its sole discretion (the Average Price or such lower price, the "Fugro Offer Price"). Each of the Fugro Offerees shall have until the 20th Business Day following the delivery of the Fugro Offer Notice (the "Fugro Offer Period") in which to notify the Fugro Member that it accepts such offer as to all or any portion of the Fugro Interest offered to such Fugro Offeree for the Fugro Offer Price (or the applicable portion thereof based on the portion of the Fugro Interest such Fugro Offeree elects to purchase) and on such payment terms as set forth in the Fugro Offer Notice, which notice shall specify the maximum portion of the Fugro Interest it wishes to purchase. If more than one Fugro Offeree elects to purchase all or any portion of the Fugro Interest prior to the expiration of the Fugro Offer Period and such Fugro Offerees in the aggregate elect to purchase an amount that exceeds the Fugro Interest, the Fugro Interest shall be allocated among such Fugro Offerees pro rata according to their respective Class A Percentage Interests up to the maximum portion of the Fugro Interest specified by each such Fugro Offeree in its notice until the entire Fugro Interest shall have been so allocated or each Fugro Offeree shall have been allocated the maximum portion of the Fugro Interest specified by each such Fugro Offeree in its notice. If one or more Fugro Offerees accept such offer with respect to all but not less than all of the Fugro Interest, a closing of the purchase of such Fugro Interest (a "Fugro Member Sale") shall take place at the principal office of the Company at 10:00 a.m. on the 15th Business Day after the date on which the Fugro Offer Notice was delivered unless the parties agree on a different place or time; provided, however, that the Fugro Member Sale must be

consummated within three (3) months of the date of the Fugro Offer Notice. The Fugro Offer Price shall be payable in accordance with the payment terms of the Fugro Offer Notice. The definitive agreements providing for the purchase of the Fugro Interest by the Fugro Offerees shall include customary representations and warranties by the Fugro Member regarding its valid title to and ownership of the Fugro Interest, free and clear of all liens, claims and encumbrances (excluding those arising under securities laws and this Agreement) and its authority, power and right to enter into and consummate the purchase of the Fugro Interest.

(c) In the event that the Fugro Member delivers a Fugro Sale Notice, but a Fugro Member Sale is not consummated in accordance with <u>Section 11.8(b)</u>, the Fugro Member may cause a sale of the Company in accordance with the remainder of this <u>Section 11.8</u> (a "<u>Drag Sale</u>"). Upon the initiation of a Drag Sale by written notice from the Fugro Member to the Company, the HC2 Member and Zencor Holdings, HC2 Member and the Fugro Member shall jointly select and appoint a qualified investment banking firm to manage and oversee the process of the Drag Sale (the "<u>Sale Process</u>"); <u>provided</u>, <u>however</u>, that, if the HC2 Member and the Fugro Member fail or are unable to agree upon a mutually acceptable investment banking firm within fifteen (15) Business Days after the receipt of such notice, the HC2 Member and the Fugro Member shall each, at its own cost and expense, promptly designate a qualified investment banking firm which two firms shall, in turn, select a third a qualified investment banking firm (such jointly selected firm or the firm so selected, the "<u>Independent Investment Bank</u>"). The Company shall retain the Independent Investment Bank to manage and oversee the Sale Process, including offering the Company for sale, whether by means of a sale of all of the Membership Interests, merger, sale of all or substantially of the assets or otherwise. In addition to third parties, each Member (and any of their respective Affiliates) shall be entitled to bid for and purchase the Company pursuant to the Sale Process.

(d) The Company shall (i) participate in, and cooperate in good faith with, the Sale Process, in each case as reasonably requested by the Independent Investment Bank and (ii) take all other necessary and desirable actions as are reasonably directed by the Independent Investment Bank or the Fugro Member in connection with the Sale Process, including, subject to the potential Transferee executing and delivering to the Company a confidentiality agreement in form and substance reasonably acceptable to the Company, by (i) permitting any qualified potential acquiror (as determined by the Independent Investment Bank) to conduct a due diligence review of the Company, its Subsidiaries and their respective businesses, operations, prospects, assets, liabilities, financial condition and results of operations and (ii) making available the officers of the Company and its Subsidiaries for the purpose of making presentations to any such potential acquirors and answering questions posed by them, which, in any case, shall be during normal business hours, upon reasonable advance notice; provided that the Company will not be subject to a due diligence process which imposes an unreasonable burden to the Company or its operations.

(e) Upon completion of the Sale Process, the Independent Investment Bank shall advise the Board and the Members as to its recommendation (the "<u>Recommended Offer</u>") of which bid represents the best offer for the Company, taking into account the type and amount of consideration offered as well as certainty of closing. Unless otherwise unanimously agreed by the HC2 Member, the Fugro Member and Zencor Holdings, the Company and the Members shall not be required to consummate a Drag Sale unless the price per Unit pursuant to such Recommended Offer is greater than the Average Price.

(f) Subject to Section 11.8(e), the Members, to the extent they have any right or standing to do so, shall vote for, consent to, and raise no objections to the Recommended Offer, and the Members and the Company shall take all other actions necessary or reasonably required to cause the consummation of a Drag Sale on the terms set forth in the Recommended Offer. Without limiting the foregoing, (A) if the Recommended Offer is structured as a sale of assets or a merger or consolidation, then each Member shall vote or cause to be voted all Units that such Member holds or with respect to which such Member has the power to direct the voting and which are entitled to vote on such Recommended Offer in favor of such Recommended Offer and shall waive any dissenter's rights, appraisal rights or similar rights which such Member may have in connection therewith, (B) if the Recommended Offer is structured as or involves a sale or redemption of Units, then each Member shall agree to sell such Member's pro rata share of Units being sold in such Recommended Offer, each on the same terms and conditions as set forth in the Recommended Offer (other than with respect to price per Unit, which will be determined in accordance with the distribution provisions set forth in this Agreement), and such Members shall execute all documents necessary or reasonably required to effectuate a Drag Sale in accordance with the Recommended Offer, (C) each Member shall be obligated to provide with respect to such Member customary representations, warranties, covenants and agreements (other than any non-compete or non-solicit covenant or agreement) with respect to their ownership of the Units in connection with such Drag Sale, and (D) each Member shall be obligated to join on a pro rata several and not joint basis (based on the relative consideration to be received in respect of the Units to be sold) in any indemnification obligations (including participating in any escrow arrangements) of the Company or the Members in connection with such Drag Sale, as set forth in the Recommended Offer; provided that, other than in the case of fraud, no Member's obligations for indemnification and similar obligations shall exceed the aggregate cash proceeds received by, and any amount deposited into escrow on behalf of, such Member on account of the Units sold in such Drag Sale; and provided, further, that no Member shall be obligated to enter into indemnification obligations with respect to any representations, warranties or covenants in the nature of those described in <u>clause (C)</u> to the extent relating to or in respect of any other Member or any other Person's Membership Interests. Notwithstanding anything herein to the contrary, in the event that a Member would not be receiving any consideration upon the consummation of a Drag Sale with respect to certain of the Units then held by such Member (as determined based on a deemed distribution in accordance with the distribution provisions set forth in this Agreement of the full equity value of the Company based on the aggregate sale proceeds to be received in such Drag Sale), then such Units shall automatically be forfeited and cancelled without any further action or consideration required by any of the parties hereto.

(g) <u>Expenses</u>. If a Drag Sale is consummated, then the Company shall bear the reasonable costs of any Drag Sale to the extent such costs are incurred for the benefit of all Members and are not otherwise paid by the acquiring party. Costs incurred by or on behalf of a Member for such Person's sole benefit shall not be considered costs of the transaction hereunder. Notwithstanding any other provision herein to the contrary, the Fugro Member shall have the right at any time prior to the consummation of the Drag Sale to abandon the Drag Sale, and none of the actions taken or requested to be taken by any Member or the Company prior to such abandonment shall in any way obligate the Company or the Members to consummate the Drag Sale.

(h) <u>Consideration</u>. In connection with any Drag Sale to which this <u>Section</u> 11.8 applies, each Member shall receive in exchange for or in respect of, as applicable, the Units held by such Member, the same portion of the aggregate consideration per Unit from such Drag Sale (the amount and value of such consideration shall be determined in good faith by the Board), but giving effect to the rights and preferences of the Units set forth in this Agreement, including <u>Section 7.2</u>, as in effect immediately prior to such Drag Sale. Each Member shall take all necessary or reasonably required actions in connection with the distribution or allocation among the Company's Members of the aggregate consideration from such Drag Sale as necessary or otherwise reasonably requested by the Company for the purpose of effecting such distribution or allocation; <u>provided</u>, <u>however</u>, that, in connection with any such Drag Sale that is structured as an exchange by the Members of all or substantially all of the Units held by the Members, any Person holding rights to acquire Units prior to such Drag Sale shall thereafter have the right to only receive the securities that were exchanged for the Units such Person previously had the right to acquire (and if the exchange was in part, a proportional amount thereof). In the event that any non-cash consideration is paid in connection with any Drag Sale, each such Member shall be entitled to receive, on a *pro rata* basis, (i) the same proportionate amount of cash and non-cash consideration, (ii) substantially the same information and transfer rights and (iii) the right to participate, *pro rata*, in any registration rights provided in connection with such Drag Sale.

(i) <u>Representative</u>. If the Drag Sale involves the negotiation or a transaction for which Rule 506 (or any similar rule then in effect) promulgated by the SEC may be available, each Member who is not an "accredited investor," as that term is defined in Regulation D as promulgated under the Securities Act, will, at the request of the Company, appoint either a purchaser representative (as such term is defined in Rule 501 under the Securities Act) designated by the Company, in which event the Company will pay the fees of such purchaser representative, or another purchaser representative (reasonably acceptable to the Company), in which event such holder will be responsible for the fees of the purchaser representative so appointed.

(j) <u>Proxy</u>. Each Member hereby grants the Company such Member's perpetual and irrevocable power of attorney with full right, power and authority to take all actions necessary and/or desirable to sell, transfer or otherwise dispose of all Units (and any other equity of the Company) held by such Member, in connection with the consummation of a Drag Sale. Pursuant to such power of attorney, the Company shall have the right to execute any and all documents related to a Drag Sale (including documents granting customary indemnities to a buyer of assets or securities consistent with this Agreement) on behalf of such Member. The Company shall exercise such power of attorney in good faith solely in compliance with the terms of this <u>Section 11.8</u>.

(k) This <u>Section 11.8</u> shall automatically terminate upon the earlier of the consummation of a Company Sale, a Drag Sale or an Initial Public Offering.

ARTICLE XII Information Rights

12.1 <u>Financial Statements</u>. The Company shall, upon request, deliver to each Member:

(a) within forty-five (45) days after the end of each quarterly accounting period in each Fiscal Year of the Company (other than any quarterly accounting period ending on the last day of a Fiscal Year of the Company), unaudited consolidated statements of income and cash flows of the Company and its Subsidiaries for such quarterly period (as well as unaudited consolidated statements of income of the Company and its Subsidiaries for the beginning of the Fiscal Year to the end of such quarter) and unaudited consolidated balance sheets of the Company and its Subsidiaries as of the end of such quarterly period. Such financial statements shall be prepared in accordance with generally accepted accounting principles, consistently applied, subject to the absence of footnote disclosures and to normal year-end adjustments;

(b) within one hundred and twenty (120) days after the end of each Fiscal Year of the Company, unaudited consolidated statements of income and cash flows of the Company and its Subsidiaries for such Fiscal Year, and unaudited consolidated balance sheets of the Company and its Subsidiaries as of the end of such Fiscal Year; <u>provided</u> that if any such financial statements are prepared for any Person in a manner to include customary managements' discussion and analysis, the financial statements delivered to all Members shall include such discussion and analysis. Such financial statements shall be prepared in accordance with generally accepted accounting principles, consistently applied; and

(c) such other statements of financial information as the Board may reasonably request or approve.

12.2 <u>Confidential Nature of Information</u>. Each Member hereby agrees that any and all information delivered to, obtained by or that otherwise becomes in the possession of such Member or any of its Representatives or advisors pursuant to the provisions of this <u>Section 12.2</u> shall be kept in strict confidence by such Member and/or its Representatives and/or advisors, except any of the same that (a) was, is now, or becomes generally available to the public (but not as a result of a breach of any duty of confidentiality by which such Member or its Representatives or advisors are bound), (b) was disclosed to such Member by a third party not, to the knowledge of such Member, subject to any duty of confidentiality to the Company or any of its Affiliates, (c) is required to be disclosed pursuant to applicable law or an applicable stock exchange, (d) in the case of any such Member that is an investment fund, is disclosed in a customary manner to such Member's investors or prospective investors, or (e) as otherwise approved by the Board.

12.3 <u>Waiver of Section 18-305 of the Delaware Act</u>. Except as otherwise required by any written agreement between the Company and any Member, each Member hereby irrevocably waives any and all rights that such Member may have to receive information from the Company pursuant to Section 18-305 of the Delaware Act.

ARTICLE XIII Registration Rights

13.1 Demand Registrations of Registrable Securities.

(a) Subject to the provisions of this <u>Article XIII</u> (including the restrictions set forth in <u>Section 13.1(c)</u>), (x) at any time, the HC2 Member shall have the right to request, by delivering a written notice to the Company, registration under the Securities Act of all or any portion of the Registrable Securities held by the HC2 Member and, (y) at any time after the Company has consummated an Initial Public Offering, beginning the date that is one hundred eighty (180) days after the completion of such Initial Public Offering, the Fugro Member shall have the right to request, by delivering a written notice to the Company and in all other respects compliant with Section 13.5 below, a "shelf registration" on Form S-3 under the Securities Act of all or any portion of the Registrable Securities held by the Fugro Member. All registrations requested pursuant to this <u>Section 13.1(a)</u> are referred to herein as "<u>Demand Registrations</u>," and either the HC2 Member or the Fugro Member, as applicable, is referred to herein as the "<u>Requesting Holder</u>" with respect to such registration. Each request for a Demand Registration shall specify the number of Registration Request in accordance with <u>Section 13.2(a)</u> (the "<u>Registration Notice</u>") to all other holders of Registrable Securities and shall thereupon use reasonable best efforts to effect the Demand Registration under the Securities Act on any form available to the Company (<u>provided</u> that the Requesting Holder (if such Requesting Holder is the HC2 Member) may require such registration to be effected on Form S-1 (or any comparable or successor long form registration form) if the Requesting Holder concludes that the use of such Form S-1 (or any comparable or successor long form registration form) will be beneficial to the distribution of the Registrable Securities to be sold) of:

(i) the Registrable Securities requested to be registered by the Requesting Holder, and all other Registrable Securities which the Company has received a written request to register within fifteen (15) calendar days after the Registration Notice is given;

(ii) any securities of the Company proposed to be included in such registration by the Company for its own account; and

(iii) any securities proposed to be included in such registration by the holders of any registration rights granted other than pursuant to this Agreement ("<u>Other Registration Rights</u>").

(b) If the sole or managing underwriter(s) of a Demand Registration advises the Company in writing that in its opinion the number of Registrable Securities and, if permitted hereunder, other securities requested to be included exceeds the number of Registrable Securities and other securities, if any, which can be sold in such offering without adversely affecting the distribution of the securities being offered, the price that will be paid in such offering, or the marketability of such securities, then the Company shall include in such registration in the following order of priority:

(i) <u>first</u>, the greatest number of Registrable Securities proposed to be registered by Members which in the opinion of such underwriter(s) can be so sold, such amount to be allocated ratably among such Members based on the amount of Registrable Securities held by each such Member (or, if any Member requests to include less than its ratable share, then such excess shall be allocated ratably among those Members requesting to include more than their ratable share);

(ii) <u>second</u>, after all Registrable Securities that the Members propose to register, the greatest number of securities proposed to be registered by Persons with Other Registration Rights which in the opinion of such underwriter(s) can be so sold, such amount to be allocated ratably among the respective holders thereof based on the amount of securities held by each such holder (or, if any holder requests to include less than its ratable share, then such excess shall be allocated ratably among those holders requesting to include more than their ratable share); and

(iii) <u>third</u>, after all securities that the Members and the Persons with Other Registration Rights propose to register, the greatest number of securities proposed to be registered by the Company for its own account, which in the opinion of such underwriter(s) can be so sold;

<u>provided</u> that, notwithstanding the foregoing, if such registration is the Company's Initial Public Offering, then (1) the Company may elect not to include securities proposed to be registered by Members who are (or their Affiliates are) employees of the Company or its Subsidiaries or holders of Class A-1 Units, Class B Units, Class C Units or Class D Units and (2) subject to the prior written consent of the Requesting Holder, such Initial Public Offering may be solely a primary Public Offering by the Company if the underwriter(s) believe(s) that any secondary Public Offering would be detrimental to the Initial Public Offering.

(c) The Requesting Holder may request an unlimited number of Demand Registrations. Unless otherwise agreed by the Requesting Holder, any Demand Registration requested must be for a firmly underwritten Public Offering (to be managed by an underwriter(s) of recognized national standing selected by the Requesting Holder and reasonably satisfactory to the Company). If the Requesting Holder had the right to register and sell Registrable Securities in a Demand Registration, then the Company shall not be obligated to effect a Demand Registration within six (6) months after the effective date of such previous Demand Registration unless such Demand Registration was effected pursuant to a Registration Statement on Form S-3 (or any comparable or successor short-form registration form), in which case the Company shall not be obligated to effect a Demand Registration form). The Company may defer but, unless it receives the consent of the Requesting Holder, not more than two times for a period not to exceed ninety (90) calendar days in the aggregate during any twelve (12)-month period from each receipt of the request to file a Registration Statement for a Demand Registration if the Board in good faith determines that such Demand Registration might reasonably be expected to have a materially adverse effect on any proposal or plan by the Company or any of its Subsidiaries to engage in any acquisition of assets (other than in the ordinary course of business) or any merger, consolidation, tender offer or other material transactions; provided that in such event, the Requesting Holder shall be entitled to withdraw such request.

(d) Any holder of Registrable Securities shall be entitled to withdraw such holder's request to participate in any Demand Registration that is an underwritten offering at any time prior to the execution and delivery of the related underwriting agreement.

13.2 Piggyback Registration.

(a) If at any time the Company proposes or is required to register any securities under the Securities Act (other than in connection with a business acquisition or combination or in connection with an employee benefit plan), whether in connection with a primary or secondary offering, then the Company shall give written notice to each holder of Registrable Securities at least twenty (20) calendar days prior to the initial filing of such Registration Statement with the SEC of the Company's intent to file such Registration Statement and of such holder's rights under this Section 13.2. Upon the written request of any holder of Registrable Securities made within fifteen (15) calendar days after any such notice is given (which request shall specify the Registrable Securities intended to be disposed of by such holder), the Company shall use all reasonable efforts to effect the registration (a "Piggyback Registration") under the Securities Act of all Registrable Securities which the Company has been so requested to register by the holders thereof; provided, however, that, if, at any time after giving written notice of its intention to register any securities and prior to the effective date of the Registration Statement filed in connection with such Piggyback Registration, the Company shall determine for any reason not to register or to delay registration of such securities, then the Company may, at its election, give written notice of such determination to each holder of Registrable Securities and, thereupon, (i) in the case of a determination not to register, the Company shall be relieved of its obligation to register any Registrable Securities under this Section 13.2 in connection with such registration (but not from its obligation to pay the expenses incurred in connection therewith), and (ii) in the case of a determination to delay registration, the Company shall be permitted to delay registering any Registrable Securities under this Section 13.2 during the period that the registration of such other securities is delayed. If any Piggyback Registration is a firmly underwritten Public Offering, then the underwriter(s) in such Piggyback Registration shall be of recognized national standing and be selected by the Company with the prior written consent of the HC2 Member and the Fugro Member.

(b) If the sole or managing underwriter(s) of a Piggyback Registration advises the Company in writing that in its opinion the number of Registrable Securities, if permitted hereunder, and other securities requested to be included exceeds the number of Registrable Securities and other securities, if any, which can be sold in such offering without adversely affecting the distribution of the securities being offered, the price that will be paid in such offering or the marketability of such securities, then the Company shall include in such Piggyback Registration the Registrable Securities and other securities of the Company in the following order of priority:

(i) <u>first</u>, the greatest number of securities of the Company proposed to be included in such registration by the Company for its own account, which in the opinion of such underwriter(s) can be so sold;

(ii) <u>second</u>, after all of the securities that the Company proposes to register, the greatest number of Registrable Securities proposed to be registered by the Members which in the opinion of such underwriter(s) can be so sold, such amount to be allocated ratably among the Members based on the amount of Registrable Securities held by each such

Members (or, if any Member requests to include less than its ratable share, then such excess shall be allocated ratably among those Members requesting to include more than their ratable share); and

(iii) <u>third</u>, after all securities that the Company and the Members propose to register, the greatest number of securities held by Persons with Other Registration Rights requested to be registered by the holders thereof which in the opinion of such underwriter(s) can be so sold, such amount to be allocated ratably among the respective holders thereof based on the amount of securities held by each such holder (or, if any holder requests to include less than its ratable share, then such excess shall be allocated ratably among those holders requesting to include more than their ratable share).

(c) Any holder of Registrable Securities shall be entitled to withdraw such holder's request to participate in any Piggyback Registration that is an underwritten offering at any time prior to the execution and delivery of the related underwriting agreement.

13.3 Holdback Agreements.

(a) Each Member agrees not to effect any Public Sale of any Units or of any other equity securities of the Company, or any securities convertible into or exchangeable or exercisable for such securities, during the period beginning seven (7) calendar days prior to the date of the underwriting agreement of each underwritten offering made pursuant to a Registration Statement and ending on the earlier to occur of, (A)(1) in the case of the Company's Initial Public Offering, the date that is one hundred and eighty (180) calendar days after the effective date of such Registration Statement, or (2) in the case of any other Public Offering, the date that is ninety (90) calendar days after the effective date of such Registration Statement, and (B) such earlier date as the sole or managing underwriter otherwise may permit (or such later date as the sole or managing underwriter may require with the prior written consent of the HC2 Member and the Fugro Member), other than, in each case, any Registrable Securities sold pursuant to such underwritten Public Offering.

(b) The Company agrees (i) not to effect any Public Sale or distribution of its equity securities (or any securities convertible into or exchangeable or exercisable for such securities) during the period beginning seven calendar days prior to the date of the underwriting agreement of each underwritten offering made pursuant to a Registration Statement and ending on the later to occur of, (1)(a) in the case of the Company's Initial Public Offering, the date that is one hundred and eighty (180) calendar days after the effective date of such Registration Statement, or, (b) in the case of any other Public Offering, the date that is ninety (90) calendar days after the effective date of such Registration Statement, and (2) such earlier date as the sole or managing underwriter otherwise may permit (or such later date as the sole or managing underwriter may require with the prior written consent of the HC2 Member and the Fugro Member), other than, in each case, as part of such Demand Registration or in connection with any employee benefit or similar plan, any dividend reinvestment plan, or a business acquisition or combination, and (ii) to use all reasonable efforts to cause each holder of at least three percent (3%) (on a fully diluted basis) of its equity securities (or any securities convertible into or exchangeable or exercisable for such securities) (collectively, the "<u>3% Holders</u>") which are or may be purchased from the Company at any time after the date of this Agreement (other than in a registered offering) to agree not to effect

any sale or distribution of any such securities during such period (except as part of such underwritten offering, if otherwise permitted); <u>provided</u> that if all 3% Holders agree to such restrictions and the Company thereafter releases any one or more of the 3% Holders from such restrictions, then all of the 3% Holders shall be immediately released from such restrictions.

13.4 <u>Registration Procedures</u>. Whenever required under this <u>Article XIII</u> to effect the registration of any Registrable Securities, the Company shall effect such registrations to permit the sale of such Registrable Securities in accordance with the intended method or methods of disposition thereof, and, pursuant thereto, the Company shall as expeditiously as possible:

(a) prepare and file with the SEC a Registration Statement or Registration Statements on a form available for the sale of the Registrable Securities by the holders thereof in accordance with the intended method of distribution thereof, and use reasonable best efforts to cause each such Registration Statement to become effective;

(b) prepare and file with the SEC such amendments and post-effective amendments to each Registration Statement as may be necessary to keep such Registration Statement continuously effective for a period ending on the earlier of (i) ninety (90) calendar days from the effective date, unless such Registration Statement is filed pursuant to Rule 415 (or any similar provisions then in force) under the Securities Act, in which case such Registration Statement shall be subject to the provisions of <u>Section 13.5</u>, and (ii) such time as all of such securities have been disposed of in accordance with the intended method of disposition thereof; cause the related prospectus to be supplemented by any required prospectus supplement and, as so supplemented, to be filed pursuant to Rule 424 (or any similar provisions then in force) under the Securities Act; and comply with the provisions of the Securities Act, the United States Securities Exchange Act of 1934, as amended from time to time (the "<u>Exchange Act</u>") and the rules and regulations of the SEC promulgated thereunder applicable to it with respect to the disposition of all securities covered by such Registration Statement as so amended or in such prospectus as so supplemented;

(c) notify each holder of Registrable Securities included in the applicable registration as promptly as reasonably practicable (but in any event within two Business Days), and confirm such notice in writing, (i) when a prospectus or any prospectus supplement or post-effective amendment has been filed, and, with respect to a Registration Statement or any post-effective amendment, when the same has become effective, (ii) of the issuance by the SEC of any stop order (or threat of such issuance of a stop order) suspending the effectiveness of a Registration Statement or of any order preventing or suspending the use of any preliminary prospectus, (iii) if at any time when a prospectus is required by the Securities Act to be delivered in connection with sales of Registrable Securities the Company becomes aware that the representations and warranties of the Company contained in any agreement (including any underwriting agreement) contemplated by <u>Section 13.4(j)</u> cease to be true and correct in all material respects, (iv) of the receipt by the Company of any notification with respect to the suspension of the qualification or exemption from qualification of a Registration Statement or any of the Registrable Securities for offer or sale in any jurisdiction, and (v) if the Company becomes aware of the happening of any event that makes any statement made in such Registration Statement or related prospectus or any document incorporated to be incorporated therein by reference untrue in any material respect or that requires the making of any changes in such Registration Statement, it will not contain any untrue statement of a material fact or omit to state

any material fact required to be stated therein or necessary to make the statements therein not misleading, and that in the case of the prospectus, it will not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading;

(d) use all reasonable efforts to prevent the issuance of any order suspending the effectiveness of a Registration Statement or of any order preventing or suspending the use of a prospectus or suspending the qualification (or exemption from qualification) of any of the Registrable Securities for sale in any jurisdiction, and, if any such order is issued, then to obtain the withdrawal of any such order at the earliest possible moment;

(e) furnish, upon request, to each holder of Registrable Securities to be included in such registration and the underwriter(s), if any, without charge, one original copy and such number of conformed copies of the Registration Statement and any post-effective amendment thereto, and such number of copies of the prospectus (including each preliminary prospectus and each prospectus filed under Rule 424 under the Securities Act), any amendments or supplements thereto and any documents incorporated by reference therein, as such holder or underwriter(s) may reasonably request in order to facilitate the disposition of the Registrable Securities being sold by such holder (it being understood that the Company consents to the use of the prospectus and any amendment or supplement thereto by each holder of Registrable Securities covered by such Registration Statement and the underwriter(s), if any, in connection with the Public Offering and sale of the Registrable Securities covered by the prospectus or any amendment or supplement thereto);

(f) if requested by the sole or managing underwriter(s) if any, or any holder of Registrable Securities to be included in such registration in connection with any sale pursuant to a Registration Statement, promptly incorporate into a Registration Statement, or if such Registration Statement is already effective, in a prospectus supplement or post-effective amendment such information relating to the Company and/or such underwriting as the sole or managing underwriter(s) or such holder reasonably requests to be included therein; and make all required filings of such Registration Statement, prospectus supplement or post-effective amendment as soon as practicable after being notified of the matters incorporated in such Registration Statement, prospectus supplement or post-effective amendment;

(g) in connection with any sale pursuant to a registration, cooperate with the holders of Registrable Securities to be included in such registration and the sole or managing underwriter(s), if any, to facilitate the timely preparation and delivery of certificates (not bearing any restrictive legends including those set forth in <u>Section 11.5</u>) representing securities to be sold under such registration, and enable such securities to be in such denominations and registered in such names as the sole or managing underwriter(s), if any, or such holders may request;

(h) prior to any Public Offering of Registrable Securities, to use all reasonable efforts to (i) register or qualify, and cooperate with each holder of Registrable Securities, the underwriter(s), if any, the sales agents, and their respective counsel in connection with the registration or qualification (or exemption from such registration or qualification) of such Registrable Securities for offer and sale under the securities or "blue sky" Laws of such jurisdictions within the United States of America as any holder of Registrable Securities or the sole or managing underwriter(s), if any, reasonably request in writing and (ii) to cause such Registrable Securities to

be registered with or approved by such other governmental agencies or authorities as may be necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be reasonably necessary or advisable to enable any holder of Registrable Securities to consummate the disposition of the Registrable Securities owned by such holder; <u>provided</u>, <u>however</u>, that the Company shall not be required to (1) qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify but for this <u>Section 13.4(h)</u>, (2) subject itself to taxation in any jurisdiction or (3) consent to general service of process in any such jurisdiction where it is not then so subject;

(i) upon the occurrence of any event contemplated by Section 13.4(c)(v), as promptly as practicable prepare a supplement or posteffective amendment to the Registration Statement or a supplement to the related prospectus or any document incorporated or deemed to be incorporated therein by reference, or file any other required document so that, as thereafter delivered to the purchasers of the Registrable Securities being sold thereunder, such prospectus will not contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading;

(j) in the case of Registrable Securities distributed by means of an underwritten offering, enter into an underwriting agreement in such form, scope and substance as is customary in underwritten offerings and take all such other actions as are reasonably requested by the managing or sole underwriter in order to expedite or facilitate the registration or the disposition of such Registrable Securities, and in such connection, (i) make such representations and warranties to the underwriter(s), with respect to the business of the Company and its Subsidiaries, and the Registration Statement, prospectus and documents, if any, incorporated or deemed to be incorporated by reference therein, in each case, in form, substance and scope as are customarily made by issuers to underwriter(s) in underwritten offerings, and confirm the same if and when requested; (ii) obtain opinions of counsel to the Company and updates thereof (which counsel and opinions (in form, scope and substance) shall be reasonably satisfactory to the sole or managing underwriter(s)), addressed to the underwriter(s) covering the matters customarily covered in opinions requested in underwritten offerings and such other matters as may be reasonably requested by underwriter(s); (iii) obtain "comfort" letters and updates thereof from the independent certified public accountants of the Company (and, if necessary, any other independent certified public accountants of any Subsidiary of the Company or of any business acquired by the Company for which financial statements and financial data are, or are required to be, included in the Registration Statement), addressed to each of the underwriter(s), such letters to be in customary form and covering matters of the type customarily covered in "comfort" letters in connection with underwritten offerings; and (iv) if an underwriting agreement is entered into, the same shall contain indemnification provisions and procedures no less favorable to the holders of Registrable Securities than those set forth in Section 13.7 (or such other provisions and procedures acceptable to holders of a majority of the Registrable Securities covered by such Registration Statement and the sole or managing underwriter(s) or agents) with respect to all parties to be indemnified pursuant to Section 13.7 (and each of the foregoing shall be done at each closing under such underwriting agreement, or as and to the extent required thereunder);

(k) comply with all applicable rules and regulations of the SEC and make generally available to its stockholders an earnings statement satisfying the provisions of

Section 11(a) of the Securities Act and Rule 158 thereunder (or any similar rule promulgated under the Securities Act) no later than the time prescribed under Regulation S-X (A) commencing at the end of any fiscal quarter in which Registrable Securities are sold to underwriter(s) in a firm commitment or best efforts underwritten offering and (B) if not sold to underwriter(s) in such an offering, commencing on the first day of the first fiscal quarter of the Company after the effectiveness of a Registration Statement, which statements shall cover such twelve (12)-month periods;

(1) use commercially reasonable efforts to cause the appropriate officers of the Company to cooperate fully in any offering of Registrable Securities hereunder on a customary basis and upon reasonable notice, including (A) participation in meetings with potential investors, analysts and rating agencies and preparation of all materials for such investors, analysts and rating agencies, (B) taking other actions to obtain ratings for any Registrable Securities and (C) cooperating as reasonably requested by the underwriters in the offering, marketing or selling of the Registrable Securities; and

(m) (A) use reasonable best efforts to cause all such Registrable Securities covered by such Registration Statement to be listed on the principal securities exchange on which Common Stock is then listed, if any, if the listing of such Registrable Securities is then permitted under the rules of such exchange, or (B) if no Common Stock is then so listed, use reasonable best efforts to, either, at the Company's election, (1) cause all such Registrable Securities to be listed on a national securities exchange or (2) secure designation of all such Registrable Securities as a NASDAQ "national market system security" within the meaning of Rule 11Aa2-1 or, failing that, to secure NASDAQ authorization for such shares and, without limiting the generality of the foregoing, to arrange for at least two market makers to register as such with respect to such shares with the Financial Industry Regulatory Authority.

The Company may require each holder of Registrable Securities as to which any registration is being effected to furnish to the Company such information regarding such holder and the distribution of such Registrable Securities as the Company may from time to time reasonably request in writing; <u>provided</u> that such information shall be used only in connection with such registration. The Company may exclude from such registration the Registrable Securities of any holder who unreasonably fails to furnish such information promptly after receiving such request. The Company shall permit any holder of Registrable Securities that, in such holder's judgment, may be deemed to be an underwriter or a controlling person of the Company, to participate in the preparation of the Registration Statement and to require the insertion therein of material, furnished to the Company in writing, which in the reasonable judgment of such holder and its counsel should be included. Each holder agrees that, upon receipt of any notice from the Company of the happening of any event of the kind described in <u>Section 13.4(c)(ii)</u>, <u>13.4(c)(v)</u>, such holder's receipt of the copies of the supplemented or amended prospectus contemplated by this <u>Section 13.4</u>, or until it is advised in writing by the Company that the use of the applicable prospectus may be resumed, and has received copies of any amendments or supplements thereto.

13.5 Shelf Registration. Subject to the provisions set forth in <u>Section 13.4</u>, if the Requesting Holders specify in the Registration Request that they desire the Company to undertake a shelf registration of any portion or all of the Registrable Securities held by the Requesting Holders,

then the Company shall file with the SEC a Registration Statement under the Securities Act on the appropriate form pursuant to Rule 415 under the Securities Act (the "<u>Required Registration</u>"). The Company shall use reasonable best efforts to cause the Required Registration to be declared effective under the Securities Act as soon as practical after filing, and once effective, the Company shall use reasonable best efforts to cause such Required Registration to remain effective for a period ending on the earlier of (a) the second anniversary of the effectiveness thereof, (b) the date on which all Registrable Securities have been sold pursuant to the Required Registration and (c) the date as of which there are no longer any Registrable Securities in existence.

13.6 <u>Registration Expenses</u>.

(a) All fees and expenses incident to the performance of or compliance with this Agreement by the Company shall be borne by the Company, whether or not any Registration Statement is filed or becomes effective, including all (i) registration and filing fees, including fees payable in connection with the listing of securities on any securities exchange or automated interdealer quotation system, (ii) fees and expenses of compliance with state securities or "blue sky" laws (including reasonable fees and disbursements of counsel in connection with "blue sky" qualifications of the securities registered), (iii) expenses in connection with the preparation, printing, mailing and delivery of any Registration Statements, prospectuses and other documents in connection therewith and any amendments or supplements thereto, (iv) security engraving and printing expenses, (v) internal expenses of the Company (including all salaries and expenses of its officers and employees performing legal or accounting duties), (vi) fees and expenses of counsel for the Company, (vii) fees and expenses for all independent certified public accountants (including the expenses relating to any comfort letters or costs associated with the delivery by independent certified public accountants of any comfort letters), (viii) fees and expenses of any special experts retained by the Company in connection with any registration, (ix) fees and expenses of custodians, transfer agents and registrars appointed in connection with any registration, (x) fees and expenses of underwriters, including the fees and expenses of any counsel thereto and the fees and expenses of any "qualified independent underwriter," but in any event excluding any discounts, commissions or fees of underwriter(s), placement agents, selling brokers, dealer managers or similar securities industry professionals relating to the distribution of the Registrable Securities, (xi) expenses incurred in connection with any Securities Act liability insurance, (xii) expenses of any annual audit or interim review, (xiii) fees and expenses incurred in connection with the listing of the securities to be registered on any securities exchange, (xiv) fees and expenses of any Person, including special experts, retained by the Company (except for those fees and expenses excluded pursuant to the preceding <u>clause (x)</u>, (xv) costs of printing and producing any agreements among underwriters, underwriting agreements, any "blue sky" or legal investment memoranda and any selling agreements and other documents in connection with the offering, sale or delivery of the Registrable Securities, (xvi) expenses relating to any analyst or investor presentations or any "road shows" undertaken in connection with the registration, marketing or selling of the Registrable Securities, (xvii) fees and expenses payable in connection with any ratings of the Registrable Securities, including expenses relating to any presentations to rating agencies, and (xviii) costs and expenses incurred by the Company or its appropriate officers in connection with their compliance with Section 13.4(l).

(b) In connection with any registration hereunder, the Company shall reimburse the holders of the Registrable Securities being registered in such registration for the reasonable and documented fees and disbursements of not more than one counsel (together with appropriate local

counsel) chosen by, (A) in the case of any Demand Registration, the Requesting Holder, and (B) in the case of any Piggyback Registration, the holders of a majority of the Registrable Securities to be included in such Piggyback Registration.

13.7 Indemnification; Contribution.

The Company shall, and shall cause each of its Subsidiaries to, jointly and severally, without limitation as to time, indemnify, (a) defend and hold harmless, to the fullest extent permitted by law, each holder of Registrable Securities, the partners, members, officers, directors, managers, agents, employees and Affiliates of each of them, each Person who controls each such holder (within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act), the partners, members, officers, directors, managers, agents and employees of each such controlling person and any financial or investment adviser (each, a "Covered Person"), to the fullest extent permitted by law, from and against any and all losses, claims, damages, liabilities, actions or proceedings (whether commenced or threatened), costs (including reasonable and documented costs of preparation and attorneys' fees) and expenses (including reasonable and documented expenses of investigation) (as used in this Section 13.7, collectively, "Losses"), as incurred, arising out of or based upon (A) any untrue or alleged untrue statement of a material fact contained in any Registration Statement, prospectus or form of prospectus under which any Registrable Securities were registered or in any amendment or supplements thereto or in any preliminary prospectus (if used prior to the effective date of such Registration Statement), or arising out of or based upon any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein (in the case of a prospectus or preliminary prospectus, in light of the circumstances under which they were made) not misleading, except to the extent that the same arise out of or are based upon information furnished in writing to the Company by such Covered Person or the related holder of Registrable Securities expressly for use therein or (B) any violation by the Company of any federal, state or common law applicable to the Company and relating to action required of or inaction by the Company in connection with any such registration; provided that the Company shall not be liable to any Person who participates as an underwriter in the offering or sale of Registrable Securities or any other Person, if any, who controls such underwriter(s) within the meaning of the Securities Act to the extent that any such Losses arise out of or are based upon an untrue statement or alleged untrue statement or omission or alleged omission made in any preliminary prospectus if (1) such Person failed to send or deliver a copy of the prospectus with or prior to the delivery of written confirmation of the sale by such Person to the Person asserting the claim from which such Losses arise, (2) the prospectus would have corrected such untrue statement or alleged untrue statement or such omission or alleged omission, and (3) the Company has complied with its obligations under Section 13.4(c). Each indemnity and reimbursement of costs and expenses shall remain in full force and effect regardless of any investigation made by or on behalf of such Covered Person. If the Public Offering pursuant to any Registration Statement provided for under this Section 13.7 is made through underwriter(s), no action or failure to act on the part of such underwriter(s) (whether or not such underwriter is an Affiliate of any holder of Registrable Securities) shall affect the obligations of the Company to indemnify any holder of Registrable Securities or any other Person pursuant to this Section 13.7. If the Public Offering pursuant to any Registration Statement provided for under this Article XIII is made through underwriter(s), the Company agrees to enter into an underwriting agreement in customary form with such underwriter(s) and the Company agrees to indemnify such underwriter(s), their officers, directors, managers, employees and agents, if any, and each Person,

if any, who controls such underwriter(s) within the meaning of Section 15 of the Securities Act to the same extent as provided in this <u>Section 13.7</u> with respect to the indemnification of the holders of Registrable Securities; <u>provided</u> that the Company shall not be required to indemnify any such underwriter, or any officer, director, manager or employee of such underwriter or any Person who controls such underwriter within the meaning of Section 15 of the Securities Act, to the extent that the loss, claim, damage, liability (or proceedings in respect thereof) or expense for which indemnification is claimed results from such underwriter's failure to send or give a copy of an amended or supplemented final prospectus to the Person asserting an untrue statement or alleged untrue statement or omission or alleged omission at or prior to the written confirmation of the sale of Registrable Securities to such Person if such statement or omission was corrected in such amended or supplemented final prospectus prior to such written confirmation and the underwriter was provided with such amended or supplemented final prospectus.

In connection with any Registration Statement in which a holder of Registrable Securities is participating, such holder, or an (b) authorized officer of such holder, shall furnish to the Company in writing such information regarding such holder as the Company reasonably requests for use in connection with any Registration Statement or prospectus and agrees, severally and not jointly, to indemnify, defend and hold harmless to the fullest extent permitted by law, the Company, its directors, managers, officers, agents and employees, each Person who controls the Company (within the meaning of Section 15 of the Securities Act and Section 20 of the Exchange Act), and the partners, members, directors, managers, officers, agents or employees of such controlling persons, from and against all Losses arising out of or based upon any untrue or alleged untrue statement of a material fact contained in any Registration Statement, prospectus, or form of prospectus, or arising out of or based upon any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, to the extent, but only to the extent, that such untrue or alleged untrue statement is contained in, or such omission or alleged omission is required to be contained in, any information regarding such holder so furnished in writing by such holder to the Company expressly for use in such Registration Statement or prospectus and that such statement or omission was relied upon by the Company in preparation of such Registration Statement, prospectus or form of prospectus; provided that such holder of Registrable Securities shall not be liable in any such case to the extent that the holder has furnished in writing to the Company within a reasonable period of time prior to the filing of any such Registration Statement or prospectus or amendment or supplement thereto information expressly for use in such Registration Statement or prospectus or any amendment or supplement thereto which corrected or made not misleading, information previously furnished to the Company, and the Company failed to include such information therein. In no event shall the liability of any holder of Registrable Securities hereunder be greater in amount than the dollar amount of the proceeds (net of payment of all taxes, fees and expenses incurred in connection therewith) received by such holder upon the sale of the Registrable Securities giving rise to such indemnification obligation. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of such Indemnified Party.

(c) If any Person shall be entitled to indemnification or contribution pursuant to this <u>Section 13.7</u> (an "<u>Indemnified Party</u>"), then such Indemnified Party shall give prompt notice to the party or parties from which such indemnity is sought (the "<u>Indemnifying Parties</u>") of the commencement of any action, suit, proceeding or investigation or written threat thereof (each, a

"Proceeding") with respect to which such Indemnified Party seeks indemnification or contribution pursuant hereto; provided that the failure to so notify the Indemnifying Parties shall not relieve the Indemnifying Parties from any obligation or liability except to the extent that the Indemnifying Parties have been prejudiced by such failure. The Indemnifying Parties shall have the right, exercisable by giving written notice to an Indemnified Party promptly after the receipt of written notice from such Indemnified Party of such Proceeding, to assume, at the Indemnifying Parties' expense, the defense of any such Proceeding, with counsel reasonably satisfactory to such Indemnified Party; provided that an Indemnified Party or parties (if more than one such Indemnified Party is named in any Proceeding) shall have the right to employ separate counsel in any such Proceeding and to participate in the defense thereof, but the fees and expenses of such counsel shall be at the expense of such Indemnified Party or parties unless: (A) the Indemnifying Parties agree to pay such fees and expenses; (B) the Indemnifying Parties fail promptly to assume the defense of such Proceeding or fail to employ counsel reasonably satisfactory to such Indemnified Party or parties; or (C) the named parties to any such Proceeding (including any impleaded parties) include both such Indemnified Party or parties and the Indemnifying Parties or an Affiliate of the Indemnifying Parties or such Indemnified Parties, and there may be one or more defenses available to such Indemnified Party or parties that are different from or additional to those available to the Indemnifying Parties, in which case, if such Indemnified Party or parties notifies the Indemnifying Parties in writing that it elects to employ separate counsel at the expense of the Indemnifying Parties, the Indemnifying Parties shall not have the right to assume the defense thereof and such counsel shall be at the expense of the Indemnifying Parties, it being understood, however, that, unless there exists a conflict among Indemnified Parties, the Indemnifying Parties shall not, in connection with any one such Proceeding or separate but substantially similar or related Proceedings in the same jurisdiction, arising out of the same general allegations or circumstances, be liable for the fees and expenses of more than one separate firm of attorneys (together with appropriate local counsel) at any time for such Indemnified Party or parties. Whether or not such defense is assumed by the Indemnifying Parties, such Indemnifying Parties or Indemnified Party or parties shall not be subject to any liability for any settlement made without its or their consent (but such consent shall not be unreasonably withheld, conditioned or delayed). The Indemnifying Parties shall not consent to entry of any judgment or enter into any settlement which (1) provides for other than monetary damages without the consent of the Indemnified Party or parties (which consent shall not be unreasonably withheld, conditioned or delayed) or (2) does not include as an unconditional term thereof the giving by the claimant or plaintiff to such Indemnified Party or parties of a release, in form and substance satisfactory to the Indemnified Party or parties, from all liability in respect of such Proceeding for which such Indemnified Party would be entitled to indemnification hereunder.

(d) If the indemnification provided for in this <u>Section 13.7</u> is unavailable to an Indemnified Party or is insufficient to hold such Indemnified Party harmless for any Losses in respect of which this <u>Section 13.7</u> would otherwise apply by its terms, then each applicable Indemnifying Party, in lieu of indemnifying such Indemnified Party, shall have an obligation to contribute to the amount paid or payable by such Indemnified Party as a result of such Losses, in such proportion as is appropriate to reflect the relative fault of the Indemnifying Party, on the one hand, and such Indemnified Party, on the other hand, in connection with the actions, statements or omissions that resulted in such Losses as well as any other relevant equitable considerations. The relative fault of such Indemnifying Party, on the other hand, shall be determined by reference to, among other things, whether any action in question, including

any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, has been taken by, or relates to information supplied by, such Indemnifying Party or Indemnified Party, and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent any such action, statement or omission. The amount paid or payable by a party as a result of any Losses shall be deemed to include any legal or other fees or expenses incurred by such party in connection with any Proceeding, to the extent such party would have been indemnified for such expenses if the indemnification provided for in <u>Section 13.7(a)</u> or <u>13.7(b)</u> was available to such party. The parties hereto agree that it would not be just and equitable if contribution pursuant to this <u>Section 13.7(d)</u> were determined by *pro rata* allocation or by any other method of allocation that does not take account of the equitable considerations referred to in this <u>Section 13.7(d)</u>. Notwithstanding the provisions of this <u>Section 13.7(d)</u>, an Indemnifying Party that is a holder of Registrable Securities shall not be required to contribute any amount in excess of the amount by which the net proceeds received by such Indemnifying Party exceeds the amount of any damages that such Indemnifying Party has otherwise been required to pay by reasons of such untrue or alleged untrue statement or omission or alleged omission. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation.

13.8 Rules 144 and 144A. At all times after the Company effects its Initial Public Offering, the Company shall file the reports required to be filed by it under the Securities Act and the Exchange Act and the rules and regulations promulgated thereunder (or, if the Company is not required to file such reports, it shall, upon the request of any holder of Registrable Securities, make publicly available other information so long as such information is necessary to permit sales under Rule 144A), and shall take such further action as any holder of Registrable Securities may reasonably request, all to the extent required from time to time to enable such holder to sell Registrable Securities in connection with such holder's sale pursuant to Rule 144A without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 and Rule 144A. Upon the request of any holder of Registrable Securities, the Company shall deliver to such holder a written statement as to whether it has complied with such requirements and cause its counsel to prepare and deliver any needed legal opinions for sales under Rule 144 and 144A and pay for the cases thereof.

13.9 <u>Underwritten Offerings</u>. No holder of Registrable Securities may participate in any underwritten offering effected pursuant hereto unless such holder (i) agrees to sell such holder's Registrable Securities on the basis provided in any underwriting arrangements approved by the Persons entitled hereunder to approve such arrangements and (ii) completes and executes all questionnaires, powers of attorney, indemnities, underwriting agreements and other documents required under the terms of such underwriting arrangements. Subject to <u>Section 13.2(a)</u>, the underwriters of any underwritten offering shall be chosen by the Company.

13.10 No Inconsistent Agreements. The Company has not and shall not enter into any agreement with respect to the Company's securities that is inconsistent with the rights granted to the holders of Registrable Securities in this <u>Section 13.10</u> or otherwise conflicts with the provisions hereof without the written consent of the HC2 Member and the Fugro Member. The Company represents and warrants that it is not a party to, or otherwise subject to, any other agreement granting registration rights to any other Person with respect to any Common Stock or Common Stock

equivalents. Except as provided in this Agreement, the Company shall not grant Other Registration Rights to any Persons without the prior written consent of the HC2 Member and the Fugro Member.

13.11 Initial Public Offering.

(a) If at any time the Board desires to cause (i) a transfer of all or a substantial portion of (x) the assets of the Company or any of its Subsidiaries or (y) the Units to a newly organized corporation or other business entity ("<u>Newco</u>"), (ii) a merger or consolidation of the Company or any of its subsidiaries into or with a Newco as provided under Section 18-209 of the Delaware Act or otherwise, or (iii) another restructuring of all or substantially all of the assets or Units of the Company into a Newco, including by way of the conversion of the Company into a Delaware corporation (any such corporation, also "<u>Newco</u>"), in any case in anticipation of or otherwise in connection with a registered initial public offering of securities of a Newco or any of its Affiliates (an "<u>Initial Public Offering</u>"), each Member shall take such steps to effect such transfer, merger, consolidation, distribution or other restructuring as may be requested by the Board, including, without limitation, transferring or tendering such Member's Units to a Newco in exchange or consideration for shares of capital stock or other equity interests of Newco, determined in accordance with the valuation procedures set forth in <u>Section 13.11(b</u>).

(b) In connection with a transaction described in <u>Section 13.11(a)</u> Board shall, in good faith, determine the Fair Market Value of the assets and/or Units transferred to or merged into Newco, the aggregate Fair Market Value of Newco and the number of shares of capital stock or other equity interests to be issued to each Member in exchange or consideration therefor. Notwithstanding the foregoing, in determining the Fair Market Value of the Units, (i) the offering price of the applicable Initial Public Offering will be used by the Board to determine such Fair Market Value, and (ii) the impact of the provisions of <u>Section 7.2</u> and <u>Section 10.2(b)</u> will be taken into account.

(c) Each Member hereby makes, constitutes and appoints the Company, with full power of substitution and resubstitution, its true and lawful attorney, for it and in its name, place and stead and for its use and benefit, to act as its proxy in respect of any vote or approval of Members required to give effect to this <u>Section 13.11</u>, including any vote or approval required under Section 18-209 of the Delaware Act. The proxy granted pursuant to this <u>Section 13.11(c)</u> is a special proxy coupled with an interest and is irrevocable.

13.12 <u>Mergers and Consolidations</u>. Any merger or consolidation of the Company with or into another entity shall require the approval of only a Majority in Voting Interest, except in connection with an Initial Public Offering, in which case, subject to the provisions of <u>Section 13.11</u>, the approval of the Board. The approval of any such merger or consolidation as provided in the immediately preceding sentence shall be deemed to meet all of the requirements of Member approval of a merger or consolidation, as the case may be, for purposes of the Delaware Act, including Section 18-209 of the Delaware Act.

ARTICLE XIV Miscellaneous Provisions

14.1 Notices.

(a) All notices, requests and other communications hereunder must be in writing and will be deemed to have been duly given only if delivered personally against written receipt, by facsimile transmission against facsimile confirmation, by electronic mail against written receipt by return electronic mail or mailed by internationally recognized overnight courier prepaid, to (i) any Member, at such Member's address set forth on the Members Schedule, and (ii) the Company, to the Company's Secretary at the Company's principal place of business (or in any case to such other address as the addressee may from time to time designate in writing to the sender).

(b) All such notices, requests and other communications will (i) if delivered personally to the address as provided in <u>Section 14.1(a)</u>, be deemed given upon delivery, (ii) if delivered by facsimile transmission to the facsimile number as provided in <u>Section 14.1(a)</u>, be deemed given upon facsimile confirmation, (iii) if delivered by electronic mail to the electronic mail address as provided in <u>Section 14.1(a)</u>, and (iv) if delivered by overnight courier to the address as provided in <u>Section 14.1(a)</u>, be deemed given on the earlier of the first Business Day following the date sent by such overnight courier or upon receipt (in each case regardless of whether such notice, request or other communication is received by any other person to whom a copy of such notice is to be delivered pursuant to this <u>Section 14.1</u>).

14.2 <u>No Action for Partition; Waiver of Judicial Dissolution</u>. No Member shall have any right to maintain any action for partition with respect to the property of the Company. Each Member agrees that irreparable damage would occur if any Member should bring or have brought on its behalf an action for judicial dissolution of the Company. Accordingly, each Member accepts the provisions under this Agreement as such Member's sole entitlement on dissolution of the Company and waives and renounces all rights to seek or have sought for such Member a court decree of dissolution or to seek the appointment by a court of a liquidator for the Company.

14.3 <u>Headings and Sections</u>. The headings in this Agreement are inserted for convenience only and are in no way intended to describe, interpret, define, or limit the scope, extent or intent of this Agreement or any provision of this Agreement. Unless the context requires otherwise, all references in this Agreement to Sections, Articles, Exhibits or Schedules shall be deemed to mean and refer to Sections, Articles, Exhibits or Schedules of or to this Agreement.

14.4 Amendments.

(a) Except as otherwise expressly set forth in this Agreement, the Certificate and this Agreement may be amended or restated only upon the written consent of a Majority in Voting Interest and any such amendment or restatement to which such written consent is obtained will be binding upon the Company and each Member.

(b) Notwithstanding anything in this Agreement to the contrary:

(i) in addition to any approval required by <u>clause (a)</u> above, no amendment or restatement to <u>Sections 3.1(c)</u>, <u>3.1(d)</u>, <u>3.1(e)</u>, <u>3.1(f)</u>, <u>3.1(g)</u>, <u>3.2</u>, <u>4.1</u>, <u>4.2</u>, <u>4.10</u>, <u>11.1</u>, <u>11.4</u>, <u>11.6</u>, <u>12.1</u>, <u>13.1</u>, <u>13.2</u>, <u>13.10</u>, this <u>Section 14.4</u>, <u>Section 14.15</u> or any definition contained in <u>Article I</u> that is utilized in any of the aforementioned provisions may be made unless approved

in advance in writing by a Majority of the Minority which approval shall not be unreasonably withheld or delayed, and which approval shall be deemed to have been given ten (10) Business Days after notice of the request for such approval was deemed to be delivered by the Company to the Members constituting at least a Majority of the Minority in accordance with the notice provisions of <u>Section 14.1</u>, unless Members constituting a Majority of the Minority shall provide the Company a written denial to such approval within such ten (10) Business Days period;

(ii) in addition to any approval required by clause (a) and (b)(i) above, no amendment or restatement to <u>Sections 3.1(i)</u>, <u>3.1(j)</u>, <u>4.11</u>, <u>6.1(b)</u>, <u>11.7</u>, <u>11.8</u> or <u>14.4</u> or any definition contained in <u>Article I</u> that is utilized in any of the aforementioned provisions may be made unless approved in advance in writing by the Fugro Member which approval shall not be unreasonably withheld or delayed, and which approval shall be deemed to have been given ten (10) Business Days after notice of the request for such approval was deemed to be delivered by the Company to the Fugro Member in accordance with the notice provisions of <u>Section 14.1</u>, unless the Fugro Member shall provide the Company a written denial to such approval within such ten (10) Business Days period; and

(iii) no amendment or restatement that amends (A) in any adverse manner the rights and obligations of the holders of Class A-1 Units, Class B Units, Class C Units or Class D Units, as applicable, with respect to the Class A-1 Units, Class B Units, Class C Units or Class D Units, as applicable, vis-à-vis the other Members, or (B) the manner in which distributions on account of Class A Units, Class A-1 Units, Class B Units, Class C Units and Class D Units are allocated among the holders of Class A Units, Class A-1 Units, Class B Units, Class B Units, Class C Units are allocated among the holders of Class A Units, Class A-1 Units, Class B Units, Class B Units, Class C Units are allocated among the holders of Class A Units, Class A-1 Units, Class B Units, Class C Units and Class D Units are allocated among the holders of Class A Units, Class B Units, Class B Units, Class C Units and Class D Units are allocated among the holders of Class A Units, Class B Units, Class B Units, Class C Units and Class D Units are allocated among the holders of Class A Units, Class B Units, Class B Units, Class C Units and Class D Units are allocated among the holders of Class A Units, Class B Units, Class C Units and Class D Units are allocated among Class A-1 Units, Class B Units, Class C Units and Class D Units.

(c) Notwithstanding anything to the contrary in Section 14.4(b), this Agreement (other than Section 4.10) may be amended or restated with only the written consent of a Majority in Voting Interest (and any such amendment or restatement to which such written consent is obtained will be binding upon the Company and each Member) in order (i) solely to permit the creation or issuance of Units or any other equity interests that were issued in compliance Section 4.10 and Section 4.11, and the rights and preferences thereof; provided, however, that with respect to any newly created class or series of Units, any such amendment or restatement with respect to such Units shall only be made at or prior to the time of the creation or issuance of such Units (and not thereafter); (ii) to the extent permitted by Section 3.1(c), to provide any new Member with rights with respect to the election of Directors; (iii) to provide for rights and restrictions relating to the Transfer or registration of Units held by any Person that becomes a Member after the date hereof (other than a Permitted Transferee); and (iv) to provide any Person that becomes a Member after the date hereof is consent to certain actions to be taken by the Company or its Subsidiaries. For purposes of clarity, in the event that there is any conflict between the provisions in Section 14.4(b) and this Section 14.4(c), the provisions of this Section 14.4(c) shall govern.

14.5 <u>Binding Effect</u>. Except as otherwise provided to the contrary in this Agreement, this Agreement shall be binding upon and inure to the benefit of the Members, their distributees, heirs, legal representatives, executors, administrators, successors and permitted assigns.

14.6 Counterparts; Facsimile. This Agreement, the agreements referred to herein, and each other agreement or instrument entered into in connection herewith or therewith or contemplated hereby or thereby, and any amendments hereto or thereto, to the extent signed and delivered by means of a facsimile machine or email with scan or facsimile attachment, shall be treated in all manner and respects as an original agreement or instrument and shall be considered to have the same binding legal effect as if it were the original signed version thereof delivered in person. At the request of any party hereto or to any such agreement or instrument, each other party hereto or thereto shall re-execute original forms thereof and deliver them to all other parties. No party hereto or to any such agreement or instrument was transmitted or communicated through the use of a facsimile machine or email to deliver a signature or the fact that any signature or agreement or instrument was transmitted or communicated through the use of a facsimile machine or email as a defense to the formation or enforceability of a contract, and each such party forever waives any such defense.

14.7 <u>Severability</u>. Whenever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under law, but if any provision of this Agreement is held to be invalid, illegal or unenforceable in any respect under any law in any jurisdiction, such invalidity, illegality or unenforceability shall not affect the validity, legality or enforceability of any other provision of this Agreement in such jurisdiction or affect the validity, legality of any provision in any other jurisdiction, but this Agreement shall be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision had never been contained herein.

14.8 Remedies. Each of the parties to this Agreement shall be entitled to enforce its rights under this Agreement specifically, to recover damages and costs (including reasonable attorney's fees) caused by any breach of any provision of this Agreement and to exercise all other rights existing in its favor. The Members agree and acknowledge that money damages may not be an adequate remedy for any breach of the provisions of this Agreement and that any party may in its sole discretion apply to any court of law or equity of competent jurisdiction (without posting any bond or deposit) for specific performance and/or other injunctive relief in order to enforce or prevent any violations of the provisions of this Agreement.

14.9 Business Days. If any time period for giving notice or taking action hereunder expires on a day which is not a Business Day, the time period shall automatically be extended to the Business Day immediately following such day.

14.10 Waiver of Jury Trial. EACH PARTY TO THIS AGREEMENT HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION, OR CAUSE OF ACTION (a) ARISING UNDER THIS AGREEMENT OR (b) IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO IN RESPECT OF THIS AGREEMENT OR ANY OF THE TRANSACTIONS RELATED HERETO, IN EACH CASE, WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER IN CONTRACT, TORT, EQUITY OR OTHERWISE.

14.11 <u>Construction</u>. In this Agreement, unless otherwise specified or where the context otherwise requires:

(a) the headings of particular provisions of this Agreement are inserted for convenience only and shall not be construed as a part of this Agreement or serve as a limitation or expansion on the scope of any term or provision of this Agreement;

- (b) words importing any gender shall include other genders;
- (c) words importing the singular only shall include the plural and vice versa;
- (d) the words "include," "includes" or "including" shall be deemed to be followed by the words "without limitation";

(e) the words "hereof," "herein" and "herewith" and words of similar import shall, unless otherwise stated, be construed to refer to this Agreement as a whole and not to any particular provision of this Agreement;

- (f) references to any Person include the successors and permitted assigns of such Person;
- (g) the use of the words "or," "either" and "any" shall not be exclusive;
- (h) the phrase "to the extent" means the degree to which a thing extends, and does not simply mean "if";

(i) wherever a conflict exists between this Agreement and any other agreement, this Agreement shall control but solely to the extent of such conflict;

(j) references to any agreement, contract, schedule statute or regulation, unless otherwise stated, are to such agreement, contract, schedule statute or regulation as amended, modified or supplemented from time to time (subject to any restrictions on such amendments, modifications or supplements set forth herein); and

(k) the parties hereto have participated jointly in the negotiation and drafting of this Agreement; accordingly, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties hereto, and no presumption or burden of proof shall arise favoring or disfavoring any party hereto by virtue of the authorship of any provisions of this Agreement.

14.12 Entire Agreement and Incorporation by Reference. Except as otherwise expressly set forth in this Agreement, this Agreement and the other agreements referred to in this Agreement (including, without limitation, any Subscription Agreement and any Grant Agreement) embody the complete agreement and understanding among the parties to this Agreement with respect to the subject matter of this Agreement and supersedes and preempts any prior understandings, agreements or representations by or among the parties, written or oral, which may have related to the subject matter of this Agreement in any way.

14.13 <u>**Conflicting Agreements**</u>. Each Member represents that such Member has not granted and is not a party to any proxy, voting trust or other agreement which is inconsistent with or conflicts with the provisions of this Agreement, and no Member shall grant any proxy or become party to any voting trust or other agreement which is inconsistent with or conflicts with the provisions of this Agreement.

14.14 GOVERNING LAW. ALL ISSUES AND QUESTIONS CONCERNING THE APPLICATION, CONSTRUCTION, VALIDITY, INTERPRETATION AND ENFORCEMENT OF THIS AGREEMENT AND THE EXHIBITS AND SCHEDULES TO THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF DELAWARE, AND SPECIFICALLY THE DELAWARE ACT, WITHOUT GIVING EFFECT TO ANY CHOICE OF LAW OR CONFLICT OF LAW RULES OR PROVISIONS (WHETHER OF THE STATE OF DELAWARE OR ANY OTHER JURISDICTION) THAT WOULD CAUSE THE APPLICATION OF THE LAWS OF ANY JURISDICTION OTHER THAN THE STATE OF DELAWARE.

14.15 <u>Venue and Submission to Jurisdiction</u>. ANY AND ALL SUITS, LEGAL ACTIONS OR PROCEEDINGS ARISING OUT OF THIS AGREEMENT (INCLUDING AGAINST ANY DIRECTOR OR OFFICER OF THE COMPANY) SHALL BE BROUGHT SOLELY IN THE COURT OF CHANCERY OF THE STATE OF DELAWARE AND EACH MEMBER HEREBY SUBMITS TO AND ACCEPTS THE EXCLUSIVE JURISDICTION OF SUCH COURT FOR THE PURPOSE OF SUCH SUITS, LEGAL ACTIONS OR PROCEEDINGS. IN ANY SUCH SUIT, LEGAL ACTION OR PROCEEDING, EACH MEMBER WAIVES PERSONAL SERVICE OF ANY SUMMONS, COMPLAINT OR OTHER PROCESS AND AGREES THAT SERVICE THEREOF MAY BE MADE BY CERTIFIED OR REGISTERED MAIL DIRECTED TO IT AT ITS ADDRESS SET FORTH IN THE BOOKS AND RECORDS OF THE COMPANY. TO THE FULLEST EXTENT PERMITTED BY LAW, EACH PARTY HERETO HEREBY IRREVOCABLY WAIVES ANY OBJECTION WHICH IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OR ANY SUCH SUIT, LEGAL ACTION OR PROCEEDING BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM.

IN WITNESS WHEREOF, the undersigned, have executed this Fourth Amended and Restated Limited Liability Company Agreement of Global Marine Holdings, LLC as of the date first written above.

GLOBAL MARINE HOLDINGS, LLC

By: <u>/s/ Dick Fagerstal</u> Name: Dick Fagerstal Title: Chairman & CEO

HC2 HOLDINGS 2, INC.

By: <u>/s/ Michael Sena</u> Name: Michael Sena Title: CFO

FUGRO CONSULTANTS INTERNATIONAL N.V.

By: <u>/s/ Paul Verhagen</u> Name: Paul Verhagen Title: Director

ZENCOR HOLDINGS LLC

By: <u>/s/ Dick Fagerstal</u> Name: Dick Fagerstal Title: Chairman & CEO

> [Signature Page to Fourth Amended and Restated Limited Liability Company Agreement of Global Marine Holdings, LLC]

COMMANDO PARTNERS LLC

By: <u>/s/ Fred C. Farkouh</u> Name: Fred C. Farkouh Title: Manager

<u>/s/ Ian Douglas</u> IAN DOUGLAS

<u>/s/ Richard Fraser-Smith</u> RICHARD FRASER-SMITH

<u>/s/ Bruce Neilson-Watts</u> BRUCE NEILSON-WATTS

> [Signature Page to Fourth Amended and Restated Limited Liability Company Agreement of Global Marine Holdings, LLC]

FORM OF JOINDER TO <u>FOURTH AMENDED AND RESTATED</u> LIMITED LIABILITY COMPANY AGREEMENT

THIS JOINDER to the Fourth Amended and Restated Limited Liability Company Agreement of Global Marine Holdings, LLC, a Delaware limited liability company (the "<u>Company</u>"), dated as of November 30, 2017, as amended or restated from time to time, by and among the Members of the Company (the "<u>Agreement</u>"), is made and entered into as of _____ by and between the Company and _____ ("<u>Holder</u>"). Capitalized terms used herein but not otherwise defined shall have the meanings set forth in the Agreement.

WHEREAS, on the date hereof, Holder has acquired _____ **[Class [A/A-1/B/C/D] Units]** and the Agreement and the Company require Holder, as a holder of such Units, to become a party to the Agreement, and Holder agrees to do so in accordance with the terms hereof.

NOW, THEREFORE, in consideration of the mutual covenants contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties to this Joinder hereby agree as follows:

1. <u>Agreement to be Bound</u>. Holder hereby (i) acknowledges that it has received and reviewed a complete copy of the Agreement and (ii) agrees that upon execution of this Joinder, it shall become a party to the Agreement and shall be fully bound by, and subject to, all of the covenants, terms and conditions of the Agreement as though an original party thereto and shall be deemed, and is hereby admitted as, a Member for all purposes thereof and entitled to all the rights incidental thereto.

2. <u>Members Schedule</u>. For purposes of the Members Schedule, the address of the Holder is as follows:

[Name] [Address]

3. <u>Governing Law</u>. This Agreement and the rights of the parties hereunder shall be interpreted in accordance with the laws of the State of Delaware, and all rights and remedies shall be governed by such laws without regard to principles of conflicts of laws.

4. <u>Counterparts</u>. This Joinder may be executed in separate counterparts each of which shall be an original and all of which taken together shall constitute one and the same agreement.

5. <u>Descriptive Headings</u>. The descriptive headings of this Joinder are inserted for convenience only and do not constitute a part of this Joinder.E

IN WITNESS WHEREOF, the parties hereto have executed this Joinder to the Fourth Amended and Restated Limited Liability Company Agreement of Global Marine Holdings, LLC as of the date set forth in the introductory paragraph hereof.

GLOBAL MARINE HOLDINGS, LLC

By:

Name: Title:

[HOLDER]

By:

Name: Title:

CONSENT OF SPOUSE

I, ______, spouse of ______, acknowledge that I have read the Fourth Amended and Restated Limited Liability Company Agreement of Global Marine Holdings, LLC, a Delaware limited liability company, dated as of November 30, 2017, to which this Consent of Spouse is attached as <u>Exhibit B</u> (the "<u>Agreement</u>") and that I know the contents of the Agreement. I am aware its provisions provide that certain rights are granted to certain other holders of equity securities of the Company upon the sale or other disposition of Class A-1 Units, Class B Units, Class C Units and Class D Units of the Company (or any other equity securities of the Company) which my spouse owns including any interest I might have therein. Capitalized terms used but not otherwise defined herein shall have the meanings set forth in the Agreement.

I hereby agree that my interest, if any, in the Class A-1 Units, Class B Units, Class C Units and Class D Units (or any other equity securities of the Company) owned by my spouse subject to the Agreement shall be irrevocably bound by the Agreement and further understand and agree that any community property interest I may have in the Class A-1 Units, Class B Units, Class C Units and Class D Units (or any other equity securities of the Company) owned by my spouse shall be similarly bound by the Agreement.

I am aware that the legal, financial and related matters contained in the Agreement are complex and that I am free to seek independent professional guidance or counsel with respect to this Consent of Spouse. I have either sought such guidance or counsel or determined after reviewing the Agreement carefully that I will waive such right.

Dated as of _____, 20___.

(Signature)

(Print Name)

#4853-2431-2109

#4837-3810-7979

BUSINESS AND BUSINESS STRATEGY OF THE COMPANY

For purposes of this Schedule B, Global Marine Holdings, LLC together with its subsidiaries and joint ventures is referred to collectively as the "Global Group").

The Global Group is a global offshore engineering business focused on specialist subsea services across three market sectors, namely telecommunications, oil and gas, and offshore power. It is a leading independent operator in the subsea cable installation and maintenance markets. It aims to maintain its leading market position in the telecommunications maintenance segment and is seeking opportunities to grow its installation activities in the three main segments of the market while capitalizing on high market growth in the offshore power sector through expansion of its installation and maintenance services in that sector. In order to accomplish these goals, The Global Group has developed a comprehensive strategy which includes:

- Developing opportunities in the offshore power market;
- Diversifying the business by pursuing growth within its three market segments (telecommunications, oil and gas, and offshore power), which it believes will strengthen its quality of earnings and reduce exposure to one particular market segment;
- Retaining and building its leading position in telecommunications maintenance and installation;
- Working to develop convergence of its maintenance services across all three market segments; and
- Pursuing targeted mergers and acquisitions, joint ventures, partnerships and opportunities to build a larger operating platform that can benefit from increased operating efficiencies.

Telecommunications: The Global Group provides maintenance and installation services to its global telecommunications customers. Additionally, the Global Group provides installation of cable systems, including route planning, mapping, route engineering, cable-laying, trenching and burial.

Oil and Gas: The Global Group provides installation, maintenance and repair of fiber optic communication and power infrastructure to offshore platforms. Its primary activities include providing power from shore, enabling fiber-based communication between platforms and shorebased systems and installing permanent reservoir monitoring systems that allow customers to monitor subsea seismic data.

Offshore Power: Global Marine Energy ("GME") a former subsidiary of Global Marine Systems Limited ("GMSL") was established in 2011 as the vehicle for The Global Group's

significant offshore power activities, which included installing inter-array power cables for use in offshore wind farms and in the offshore wind market. GME was sold to Prysmian UK Group Ltd. ("Prysmian") in November 2012. As part of this sale, the Global Group entered into a noncompete agreement regarding offshore power operations with Prysmian (the "Prysmian Non-Compete Agreement") but retained certain key personnel and assets to ensure that The Global Group maintained its core capabilities and experience in the offshore power sector. Following entry into the Prysmian Non-Compete Agreement, The Global Group continued to install offshore power cables on behalf of Prysmian with chartered vessels through June 2014.

As a result of the expiration of the Prysmian Non-Compete Agreement in November 2015, The Global Group has been able to recommence bidding for projects in the high-growth, high-margin offshore power market. Given that renewable energy production is predicted to grow over the next decade, with a substantial proportion of that energy to be harvested offshore, The Global Group believes it is well positioned to capitalize on this anticipated growth of the offshore alternative energy market in construction as well as operations and maintenance. In 2016, The Global Group acquired 100% of CWind Limited ("CWind"), a UK-based specialist service provider in the offshore wind farm industry in the UK and northern Europe. The Global Group's management believes the offshore wind farm operations and maintenance sub-sector represents a significant opportunity for The Global Group and, through its acquisition of CWind in 2016, The Global Group is developing strategies to realize that opportunity.

Product Research and Development

Over the years the Global Group has provided many important innovations to the subsea cable market. One such innovation was GEOCABLE, the Global Group's proprietary Geographical Information System (GIS), which the Global Group believes to be the largest cable database in the market and was developed specifically to meet the needs of the cable industry.

In addition to GEOCABLE, the Global Group also develops and owns (in a consortium with other industry participants) intellectual property associated with the Universal Joint, a product which easily and effectively links together cables from different manufacturers.

The Global Group intends to conduct product research and development relating to its business on an ongoing basis.

CODE OF CONDUCT



HC2 HOLDINGS, INC.

CODE OF CONDUCT AND WHISTLEBLOWER POLICY AND PROCEDURES

I. INTRODUCTION

A. Purpose and Scope

This Code of Conduct and Whistleblower Policy and Procedures (the "*Code*") of HC2 Holdings, Inc. ("*HC2*" and collectively with any successor thereto and the consolidated subsidiaries as defined in <u>Schedule A</u>, the "*Company*") is a global document with uniform application across all of HC2's operating geographies and job levels, full-time and part-time employees, executives and directors. <u>Schedule A</u> may be amended from time to time as necessary to reflect the acquisition, merger, sale or dissolution of entities. It is also expected that everyone working on the Company's behalf, including consultants, agents, suppliers and business partners, will comply with the Code. The Code provides such persons with general standards of conduct and high-level guidance that are intended to allow many compliance situations to be addressed with a minimum of disruption to the Company's business. The Code will help ensure that, as we work toward our key goals and objectives, we do so in a manner that is above reproach.

Obeying the law is the foundation on which this Company's ethical standards are built. Accordingly, the Company requires its employees, officers and directors to comply with all applicable laws, rules and regulations of the United States and the countries where the Company does business. Violation of domestic or foreign laws, rules and regulations may subject an individual, as well as the Company, to civil and/or criminal penalties. To ensure compliance with applicable laws, rules and regulations, the Company has established various policies and procedures, including those referred to in this Code. Employees, officers and directors have an obligation to comply with these policies and procedures and to promptly alert HC2's Legal Department or your entity's legal department (the "*Legal Department*") of any deviation from them.

The Board of Directors of HC2 (the "Board") has adopted this Code primarily to:

- Promote honest and ethical conduct, including the ethical handling of actual or apparent conflicts of interest between personal and professional relationships;
- Promote full, fair, accurate, timely and understandable disclosure in reports and documents that the Company files with, or submits to, the United States Securities

and Exchange Commission (the "SEC") and in other public communications made by or on behalf of the Company;

- Promote compliance with applicable governmental laws, rules and regulations;
- Promote the protection of Company assets, including intellectual property and confidential information;
- Ensure the prompt internal reporting of violations of the Code (sometimes known as whistleblower complaints) to an appropriate person or persons identified herein;
- Provide guidance and procedures for the receipt, investigation and retention of whistleblower complaints; and
- Ensure accountability for adherence to the Code.

As a condition of your directorship of and/or employment with the Company, you are required to read the Code, become familiar with its content, and acknowledge your compliance with the Code. The Code will allow you to have a better understanding of our compliance obligations and should be used as a resource for guidance on issues as they arise. In cases where a clear governing law, rule, regulation or other Company policy does not exist, the Code can help navigate through gray areas by allowing you to apply informed judgment and common sense to find a solution.

While the Code touches on many subjects, it cannot be expected to directly address every issue. It is a complex world and the ethical aspects or legality of many day-to-day decisions may not always be clear. For this reason, it is important that we seek advice and guidance when the best path is unclear. Moreover, each director and employee of the Company shares the common obligation to promptly speak up when we suspect violations of laws, rules, regulations, the Code or any other Company policy. The procedures for requesting guidance as to the application of, and reporting suspected violations of, laws, rules, regulations, the Code or any other Company policy are outlined in Section III of the Code.

B. Expectations for Directors and Employees (Including Executives and Managers)

Every director and employee, regardless of position in the Company, is accountable for complying with applicable laws, rules, regulations and Company policies and observing high ethical standards when conducting business on the Company's behalf. Compliance with the Code is a key director and employee responsibility and will be included in the regular performance evaluation process. All directors and employees are expected to:

- Familiarize themselves with the Code as well as any policies, laws, rules and regulations that may be applicable to their job function;
- Be alert for situations and promptly report any known or suspected business conduct that may violate laws, rules, regulations, the Code or any other Company policy;______
- Seek appropriate guidance when faced with an ethical or compliance-related challenge and when unsure how to proceed;

- Never let the needs of the business justify an improper action that violates laws, rules, regulations, the Code or any other Company policy;
- Cooperate fully in any investigation of any suspected violation of laws, rules, regulations, the Code or any other Company policy; and
- Certify their commitment to the principles and standards of the Code.

In addition to the obligations shared by all directors and employees, executives and managers of the Company must:

- Lead by example and practice the highest standards of ethical business conduct;
- Set a tone reinforcing the importance of integrity and compliance, embracing the Company's open door and non-retaliation policies, and encouraging frank, honest communications within their business unit or functional area;
- Engage employees in conversations about the Code, highlighting sections and policies with particular application to their business unit or functional area;
- Support a positive work environment in which employees feel comfortable raising questions and concerns about the Code, other Company policies and business misconduct;
- Ensure employees who raise an issue or report a suspected violation of laws, rules, regulations, the Code or any other Company policy are protected from any reprisal or retaliation; and
- Regularly reinforce that business performance is never more important than ethical business conduct.

II. THE CODE OF CONDUCT

A. Behavior in the Marketplace

1. Competition Law and Business Conduct

The Company operates in some of the most competitive industries and environments in the world. Free and open competition on a level playing field is in the Company's best interest at all times. Because our products and services can compete on their own merits, the Company is committed to complying with the letter and spirit of competition and antitrust laws wherever we do business. The United States and other countries in which we operate have enacted antitrust and fair competition laws intended to encourage competition among businesses while protecting consumers from anti-competitive actions. While it is beyond the scope of the Code to cover these laws in detail, compliance with them by all Company directors and employees is essential, as violations can lead to significant reputational damage as well as penalties for both the Company and individual directors and employees, up to and including imprisonment. While competition laws and business conduct regulations are often complex and can be confusing, there are a few common

sense guidelines to help ensure compliance with these laws, rules and regulations, including avoiding the following practices:

- Discussing the allocation of markets, territories or customers with competitors;
- · Discussing price, credit terms, costs, or marketing information with competitors; and
- Discussing boycotts of third parties (*e.g.*, suppliers, distributors, etc.) with competitors, suppliers or customers.

Employees should be particularly careful at industry association or other professional trade meetings to avoid even the appearance of questionable interactions with competitors. If a competitor discusses any of these topics, no matter how casually, employees must stop the conversation immediately and report the incident to the Legal Department.

2. Trade Practices, Consumer Laws and Responsible Sales and Marketing

Employees must endeavor to deal fairly with their colleagues as well as the Company's customers, suppliers, competitors or others with whom they may come into contact. Honesty and truthfulness should be hallmarks of our reputation in the marketplace. As a company, we should never take advantage of anyone through manipulation, concealment, abuse of confidential information, misrepresentation of material facts or any other unfair or dishonest business practice, including with respect to our products and services and the products and services of our competitors.

3. Confidential and Competitive Information

Gathering information about competitors and alternative product offerings in our marketplace is a key element of allowing us to effectively compete. The Company respects the confidential information and trade secrets of others, so it is essential that we obtain information in a legal and ethical manner. Legitimate sources for competitive information include news reports, company internet sites, industry/association surveys, and conference or trade show presentations. Employees are prohibited from actively seeking or acquiring proprietary information, trade secrets or other sensitive data without the owner's consent. If an employee is in possession of confidential information in error, it should be reported to their manager and/or the Legal Department, as applicable, and returned to the rightful owner. Managers and executives should be particularly sensitive to the confidentiality obligations that employees may have with regard to former employers who may be our competitors. It is never permissible to pressure or otherwise attempt to coerce an employee into disclosing confidential information about a former employer in violation of their obligations.

4. Vendors, Suppliers and Procurement

Due to the impact that suppliers, vendors, and subcontractors can have on our business, the Company is committed to working with people and companies whose standards and practices are consistent with the Code and our procurement policies. To ensure these expectations are clearly understood, any party signing an agreement with the Company should be provided with a link to the Code on our website.

Employees with responsibilities that include sourcing, evaluating and acquiring materials or services on behalf of the Company must be consistent and objective with their analysis and any procurement decision should be based on obtaining the best overall value for the Company. Purchasing decisions must never be made on the basis of personal relationships and friendships or the opportunity for personal gain. Please refer to the "Conflicts of Interest" section of the Code for additional guidance.

5. Export/Import Laws and Trade Restrictions

The United Nations, the United States and other countries and jurisdictions often impose international trade controls and restrictions to promote security around the world and advance foreign policy agendas. The Company is committed to trading lawfully and in compliance with all export and import laws, trade regulations and restrictions imposed by recognized national and international authorities. Employees engaged in international business are responsible for knowing and complying with both the laws, rules and regulations of the countries in which they operate as well as any U.S. laws, rules and regulations that apply outside U.S. borders. Laws and regulations regarding international trade are complex and change frequently, so it is essential that directors and employees contact the Legal Department for guidance when engaging in these types of transactions.

6. Transactions with Prohibited Persons or Sanctioned Countries

The Company is prohibited from doing business with certain countries and individuals because of U.S. trade sanctions. The U.S. Department of the Treasury's Office of Foreign Assets Control ("*OFAC*") administers economic sanctions and trade embargoes against targeted non-U.S. countries, regimes, terrorism-sponsoring organizations, and individuals (known as Specially Designated Nationals). All U.S. persons and corporations must comply with OFAC regulations. Therefore, it is essential that the appropriate level of due diligence is conducted on any transactions with an international element. For detailed guidance on these issues, directors and employees should contact the Legal Department.

7. Corporate Responsibility and the Environment

The Company is committed to being a responsible corporate citizen. We comply with all environmental laws, rules, and regulations wherever we do business and expect our suppliers, vendors and contractors to do the same. Beyond environmental compliance, the Company is committed to being a responsible environmental steward, and directors and employees are asked to be aware of and actively participate in reducing the environmental burdens related to our business. All Company facilities will be operated in an environmentally responsible and safe manner. Employees whose work involves the management of potentially hazardous materials (*e.g.*, fire suppression chemicals) should be particularly mindful of proper handling and reporting obligations.

B. Conflicts of Interest

The Company respects the rights of its directors and employees to manage their personal affairs and investments. At the same time, directors and employees must avoid conflicts of interest that occur when their personal interests (or the interests of an immediate family member) may interfere or could reasonably appear to interfere with the performance of their jobs or the best interests of the Company. Perceptions of a conflict of interest can often be just as damaging as an actual conflict. Any decision taken by a director or employee on behalf of the Company should be impartial and objective. A conflict of interest arises when a director or employee competes with the Company, takes advantage of their position at the Company, or misuses confidential or proprietary information for personal gain (or the gain of an immediate family member). Examples of conflicts of interest include, but are not limited to the following:

- **Corporate Opportunities.** Directors and employees are generally prohibited from taking personal gain or other advantage from opportunities that are discovered through the use of Company property, information or position. This prohibition includes, but is not limited to, business or investment opportunities, discounts on personal purchases of a supplier's or customer's products or services, as well as preferential allocations of stock or offers to participate in initial public offerings from companies that either conduct or could be expected to conduct business with the Company. However, the Company may from time to time renounce its interest in certain corporate opportunities pursuant to applicable law, in which case no conflict of interest will exist with respect thereto.
- **Investments in other Companies.** Directors and employees should disclose any significant financial interests that they or their immediate family may have in a customer, supplier or competitor of the Company. Investments in companies that are customers, suppliers or competitors of the Company may create an improper conflict of interest, depending on the circumstances, and must be evaluated to determine appropriate steps to mitigate or eliminate the conflict or its potential effects. Investments that are made through mutual funds or other investment instruments where a director or employee has no influence over investment decisions in individual companies are not prohibited by the Code.
- **Related-Party Transactions.** Directors and executives of the Company must familiarize themselves and comply with all rules regarding related-party transactions established and enforced by the SEC and Company policies regarding related-party transactions. All material related-party transactions involving any director or employee must be approved by the Board or a duly authorized committee thereof and will be publicly disclosed as required by applicable law, rule and regulation.
- **Family and Friends.** Mixing family or personal relationships with work relationships can often create special challenges. Directors and employees should be aware of the complications that can arise when a member of their family or a close personal friend is involved with the Company or in another activity that could reasonably appear to impact a director's or employee's objectivity managing Company business. Directors and employees are not permitted to directly supervise or indirectly influence the job activities of another employee with whom they have

a close personal relationship. In addition, directors and employees may not circumvent the Code by having a family member or others with whom they have a close personal relationship do something indirectly that the director or employee is prohibited from doing under the Code.

- **Outside Employment.** Directors and employees should generally not accept a position with or run an outside business that has the potential to interfere with their responsibilities to the Company. Employment by a customer, supplier or others with whom the Company does business is generally discouraged, but outside employment or business ventures may be permissible, depending on the circumstances, including if the entity does not compete with the Company or provide services to any Company competitor.
- **Board Memberships.** Employees are generally prohibited from serving on the board of directors of a company or organization that has the potential to create a significant conflict of interest such as customers, suppliers and competitors. Subject to the foregoing standard, employees may serve on the board or an advisory committee of a for-profit or non-profit entity if it is determined that the service will not negatively impact the fulfillment of their responsibilities to the Company.
- **Company Loans.** Even if permitted under applicable law, loans to directors, employees or their respective immediate family members may create improper conflicts of interest.

Conflicts of interest are generally prohibited as a matter of Company policy, unless authorized, mitigated or eliminated, by action of the Legal Department, the Chief Compliance Officer or HC2's Audit Committee (the "*Audit Committee*"), and/or the Board or a duly authorized committee thereof, as appropriate and to the extent permissible pursuant to applicable law, rule or regulation.

Conflicts of interest are not always clear-cut. As a result, when dealing with actual, apparent or potential conflicts of interest, disclosure is an essential first step in managing the situation and limiting disruption to the business, as described below in Section III. As part of the disclosure and analysis process with respect to conflicts of interest, directors and employees will be asked to summarize the details of the transaction, which will help to identify the issues presented and provide a framework for discussions to analyze and authorize, mitigate or eliminate such issues. Directors and employees are expected to fulfill any remediation actions requested by the Company.

C. Gifts, Entertainment and Travel

Business gifts, entertainment and travel are often an integral part of building relationships with our customers, suppliers and others with whom the Company does business and all are appropriate to the extent they are not provided with the intention of unduly influencing a business decision. Infrequent exchanging of logoed promotional items, hosting an occasional lunch or dinner at a moderately priced restaurant, and attending a local athletic or cultural event are all generally considered acceptable forms of business gifts and entertainment.

Directors and employees are accountable for the exercise of good judgment and common sense when managing gift, entertainment and travel issues. Gifts, entertainment and travel that directors and employees either offer or receive in connection with their work should always be reasonable and proportionate for the Company's relationship with the other party, and in terms of value and frequency. It is also good practice to check with a customer prior to offering a gift or extending an invitation, as many companies have rules that restrict their receipt. In general, directors and employees must:

- Never solicit gifts, travel or entertainment;
- Only offer or accept gifts of a nominal value;
- Never offer or accept gifts of cash or cash equivalents (*e.g.*, gift cards);
- Never offer or accept gifts, entertainment or travel during a proposal or bidding process with another involved party if the employees are involved in the procurement decision process;
- Only offer or accept gifts, entertainment or travel that do not impair objectivity, improperly influence a decision, or have the potential to be misconstrued or misinterpreted by third parties; and
- Never offer or accept gifts, entertainment or travel that would be illegal.

Additionally, directors and employees must always obtain pre-approval from the Legal Department prior to offering gifts, entertainment or travel, of any kind or value, to government or public officials.

Directors and employees are responsible for ensuring that any expense reimbursement reports accurately reflect the relevant details for any gifts, entertainment and travel that may have been provided. In addition to the standard expense report data, directors and employees must be sure to include the names, titles, and employers of the recipients as well as the purpose for the expenditures.

D. Political and Charitable Activity/Contributions

1. Charitable Activity

It is in the Company's best interest to have a positive and beneficial impact upon the communities in which we operate, so directors and employees are encouraged to actively participate in their communities and charitable organizations of their choice. However, directors and employees should make it clear that they are acting in a personal capacity and ensure that the activities do not interfere with their responsibilities to the Company. Additionally, directors and employees are prohibited from pressuring other employees or Company business partners to either participate in or make a contribution to any charitable organization. Charitable contributions on behalf of the Company are permitted with the approval of the Board or a duly authorized executive, with appropriate consultation of the Legal Department. Any contribution should not be made with the intent or expectation that the Company will either obtain or retain business as a result of such contribution.

2. Political Activity

Directors and employees are free to support candidates and causes and make political contributions on an individual basis as long as these activities are separate and distinct from their work at the Company and do not create any real or perceived conflicts of interest. Political contributions or expenditures on behalf of the Company are permitted only when allowed under local law and with the approval of the Board or a duly authorized executive, with appropriate consultation of the Legal Department. Any contribution or expenditure should not be made with the intent or expectation that the Company will either obtain or retain business as a result of the contribution or expenditure.

3. Political Office

To ensure that any potential conflicts of interest are addressed in a timely manner, directors and employees should promptly notify the Legal Department if they are either seeking or have been elected/appointed to any political office or other government/public position.

4. Lobbying

The Company will not attempt to exert improper influence on any governmental agency, representative or official to produce a beneficial outcome. Lobbying is a legitimate pursuit and the Company will comply with any applicable local, state, and/or federal laws, rules and regulations governing this activity in the jurisdictions in which we do business. Directors and employees should not, directly or indirectly, contact any government official in an attempt to influence legislation or other government policy on behalf of the Company without the written pre-approval of the Legal Department.

E. Intellectual Property

Intellectual property can mean many things, ranging from concepts and inventions to business processes and software. While some intellectual property is protected by law, such as copyrights and trademarks, it is the responsibility of employees to safeguard the Company's intellectual property and to respect the intellectual property rights of others. To the extent permitted by law, the Company's intellectual property includes everything that is either created or modified by employees in the course of their work for the Company.

Compliance with these standards can take many forms, including without limitation the appropriate use of authorized and fully licensed software and refraining from hiring someone in an attempt to obtain a competitor's confidential information. The laws, rules and regulations around intellectual property can be complex and confusing, so employees are encouraged to seek guidance on these issues as appropriate, from their the Legal Department.

F. Books and Records

The Company is committed to creating and maintaining transparent, accurate and reliable business records, which allow us to make informed business decisions, properly prepare financial statements and timely make required SEC reports and other public disclosures. All business information and records, financial or otherwise, along with Company disclosures and communications, must be accurate, consistent, and conform to required accounting and reporting principles as well as the Company's internal policies. Employees are never permitted to make false or misleading entries into any books, records or accounts of the business and Company funds may never be used for any purpose other than as detailed on the documentation supporting the payment.

1. Financial Personnel

Employees responsible for any aspect of the Company's internal controls over financial reporting must be particularly careful to ensure that entries are honest, accurate and consistent with all applicable legal and accounting requirements. Senior executives (*e.g.*, CEO, CFO, etc.) and other employees with these responsibilities have strict obligations to ensure that the principles laid out in this section of the Code are adhered to by Company employees and that any deviation from the internal control policies or financial reporting requirements are resolved on a timely basis. In addition, senior financial staff will regularly assess the effectiveness of the Company's internal controls, implement improvements and remediate any detected weaknesses that could affect our ability to make full, fair and understandable filings with the SEC.

2. Audits and Investigations

Employees are prohibited from attempting, directly or indirectly, to influence, coerce, manipulate or otherwise mislead any internal or external party engaged in the performance of an audit. Employees are required to cooperate and be forthcoming with any auditors or investigators retained by the Company to evaluate any financial matter.

3. Records and Information Management

The Company's management of business records and destruction practices reflect legal requirements and internal control procedures. Business records can include documents, emails, and spreadsheets, regardless of their physical form or storage format. Employees should never tamper with, remove or destroy business records except in accordance with the Company's records management guidelines. Any records that have been placed on a "legal hold" by the Legal Department should never be destroyed until the Legal Department provides notice that the hold has been lifted. If an employee is notified that business records in their possession may be relevant to an investigation, audit or litigation, care must be taken to ensure that the records are not copied, altered, overwritten, or otherwise changed from their current form.

G. Anti-Corruption

The Company is subject to a number of anti-corruption laws around the world (*e.g.*, the U.S. Foreign Corrupt Practices Act, the U.K. Bribery Act, and the Organization of Economic Cooperation and Development Convention on Combating Bribery of Foreign Officials) that generally prohibit companies from providing anything of value (including gifts and entertainment) to foreign government officials and other parties in an attempt to improperly influence behavior, maintain or

secure business, or obtain some other business advantage or benefit. Bribes are against the law and a violation of the Code, regardless of what "local custom" may be. Directors and employees may not provide or accept from any public official, private individual or other entity, a bribe or other exchange of value.

It is important that directors and employees seek guidance when contemplating an offer of gifts, entertainment and/or travel for another party, but it is mandatory to seek pre-approval from the Legal Department before extending these offers to any foreign officials (including government employees or employees of government owned entities) well in advance of the anticipated action.

On occasion, it may be necessary for the Company to engage agents or other third party representatives ("*Agents*") to assist the Company in countries where we have no staff or limited operational experience. Agents must operate at all times in accordance with our Code, so it is important that these relationships be established with a clear understanding of both the business objectives involved and the Company's expectations for compliance and integrity. Regardless of local custom or cultural norm, Agents are never authorized to do indirectly what the Company and its directors and employees are prohibited from doing directly.

Agents in countries with traditionally high corruption rates should be subject to an appropriate level of due diligence as part of any preagreement screening process. While each due diligence exercise is different based on the potential risks involved, the objective is always the same – ensure that the Company only does business with qualified and reputable parties. Beyond verifying business qualifications, effective due diligence can include looking at an Agent's personal and professional relationships, such as its reputation and that of its clients in the marketplace as well as any ties with local business, banks and governments.

When establishing engagements with Agents, employees must ensure that the written agreement governing the relationship is in full compliance with applicable laws, rules and regulations of the United States and any other countries involved and provides for a commission, fee or other form of compensation that is reasonable and commensurate with the functions or services to be provided. Additionally, the agreement must require the Agent to affirmatively agree to abide by our standards for integrity and compliance with anti-corruption laws. Once a relationship has been established, it is the responsibility of the applicable engagement manager to confirm that an Agent's ongoing behavior is consistent with the Code and the terms of the engagement agreement. Ignorance of an Agent's actions on our behalf will not be tolerated, and any detected violation must be brought to the attention of the Legal Department.

H. Securities Compliance

Many of the countries in which we operate have securities laws, rules and regulations that potentially govern transactions in Company securities as well as those of many of our customers, suppliers and other companies with whom we do business. In the course of working at the Company, directors and employees may be exposed to material non-public information about the Company or third parties. Material non-public information is any information that has not been publicly disclosed that could affect an investor's decision to buy or sell securities. Examples include, but are not limited to, changes in senior management, potential mergers and acquisitions, financial results, major changes in strategy, or new product and service offerings. Using material non-public information in connection with a decision to sell or purchase securities (also known as "insider")

trading") is unethical and prohibited by law. Additionally, it is also a violation to share this information (also known as "tipping") with other parties who may in turn act on it or otherwise use it for personal advantage. In order to assist with compliance with laws against insider trading, the Company has adopted the HC2 Holdings, Inc. and Subsidiaries Amended and Restated Insider Trading Policy (the "*Insider Trading Policy*") governing trading in securities of the Company by directors and officers. If you have any questions regarding the Insider Trading Policy, please consult HC2's Chief Legal Officer.

I. Use of Company Assets, Proprietary and Confidential Information

All employees should protect the Company's assets and seek to ensure their efficient use. Theft, carelessness and waste have a direct impact on the Company's profitability. Company assets should be used for legitimate business purposes only. Any suspected incident of fraud or theft should be immediately reported for investigation. Company equipment should not be used for non-Company business, though incidental personal use may, in some circumstances, be permitted.

The obligation of employees to protect the Company's assets includes its proprietary information. The Company's proprietary and confidential information is an important asset upon which our continued success depends. Directors and employees must properly protect and use the proprietary and confidential information of the Company and its customers, suppliers, other business partners and fellow employees. Proprietary and confidential information includes any information that if disclosed might be of use to competitors of the Company or harmful to the Company, its customers, suppliers or other business partners. Examples include, but are not limited to, financial documents, vendor contracts, non-public presentations, consumer personally identifiable information, employee personnel files, and internal reports and memoranda. Proprietary and confidential information should only be shared internally with employees who have a "need to know" – when in doubt, it is always best to treat information as confidential and to only share it when specifically authorized to do so. Directors and employees should also take care to refrain from discussing proprietary and confidential information with family and friends or when in public spaces such as elevators and restaurants.

Employees must return all Company documents and other materials containing confidential information upon conclusion of their employment with the Company. In addition, employees are reminded that obligations related to the protection of confidential information may continue even after employment by the Company concludes. The Company is committed to compliance with all applicable laws related to the storage, access and use of confidential and/or personal information. Many countries in which the Company operates have their own legal requirements, so it is essential that employees with questions about local laws, rules and regulations regarding the management of confidential information, privacy or data protection contact the Legal Department for guidance.

The Company respects the privacy of its directors and employees and is committed to protecting personal and confidential information entrusted to us by directors and employees as part of their directorship and/or employment, respectively. Director and employee personal and confidential information and related data is strictly confidential and should only be accessed and used by authorized individuals for legitimate business purposes.

Directors and employees should have no expectation of privacy when using technologies provided by the Company such as computers, mobile phones and voicemail and email services. The Company may monitor director and employee activity to ensure company resources are being utilized appropriately and may at its discretion inspect messages, software or other data and information stored on these devices or transmitted over the Company's electronic networks.

J. Behavior in the Workplace

1. Respect

The Company and its employees are committed to treating each other as well as our customers, suppliers and others with whom we do business with the utmost dignity and respect. Without exception, every Company employee should experience a work environment that is safe, productive and free from discrimination, bullying or harassment of any type, whether visual, verbal, written or physical. Definitions of such terms may vary from country to country, but what is important is the effect of the action, not the intent of the offender. Employees should report any known or suspected incidents of discrimination, bullying or harassment as provided herein.

2. Equal Opportunity

The Company is an equal opportunity employer. Employment decisions, including, but not limited to, recruiting, hiring, compensation, promotions, discipline and termination, will be made based on consistent, objective criteria in full compliance with any applicable law or regulation. In addition, the Company will make reasonable accommodations for any qualified applicants or employees in compliance with applicable law or regulation.

3. Diversity

The diversity of the Company's employees is a tremendous asset. We are firmly committed to providing equal opportunity in all aspects of employment and will not tolerate any illegal discrimination or harassment of any kind. We comply with laws concerning discrimination and equal opportunity that specifically prohibit discrimination on the basis of certain differences. In particular, we must not use race, color, religion, gender, national origin or any other characteristic protected by law as a factor in hiring, firing or promotion decisions or when determining terms or conditions of employment, or in retaliating against an employee because he or she has made a complaint of discrimination in good faith, or is or has been a participant in a related investigation. The Company's employees are expected to treat others with respect in the workplace and not engage in any harassment, including making derogatory comments based on racial or ethnic characteristics or unwelcome sexual advances.

4. Safe and Secure Workplace

The Company is committed to providing a safe, healthy and nonthreatening workplace for our employees and those who may work for us or visit our facilities. To support this commitment, employees are expected to observe all safety and health related work rules, and report any work-related accident or injury promptly. Unsafe working conditions or other workplace safety issues should be immediately reported as provided herein.

5. Workplace Violence

An important aspect of the Company's commitment to providing a safe and secure workplace is a zero tolerance for workplace violence. Violent or threatening behavior, even if made in a seemingly joking manner, will not be tolerated and should be reported immediately as provided herein. "Violence" can include, but is not limited to, threats or acts of violence, intimidation to instill fear of physical harm, pushing, harassing, brandishing weapons, and threatening to engage in these activities. Employees are not permitted to possess weapons of any kind when on the job or at Company facilities. For purposes of this policy, "weapons" will be interpreted broadly to include guns, knives, explosives or any other item with the potential to inflict harm or destruction. As detailed above, employees have a responsibility to report any unsafe behavior or conditions; however, if an employee believes that they or someone they know are in immediate danger, local law enforcement should be contacted before reporting the matter internally. Employees who violate any of these provisions of the Code will be subject to disciplinary action, up to and including termination of employment, and when appropriate, referral for criminal prosecution.

6. Drugs and Alcohol

Alcohol and drug abuse are serious issues in our society and can endanger the health and safety of Company employees as well as our customers, suppliers and the companies with whom we do business. Company employees are required to work free from the influence of any substance, including drugs and alcohol, which have the potential to impact their ability to work safely and effectively. This includes both prescription and non-prescription drugs that could impair an employee's alertness or judgment while on the job and therefore jeopardize their safety and the safety of others. With the exception of moderate, responsible consumption of alcohol during Company-sponsored events or when attending business entertainment functions, employees may not purchase, manufacture, sell, possess, consume or distribute illegal drugs and alcohol when on or using Company property or conducting Company business.

7. Electronic Systems and Technology

The Company's computer systems and networks are provided to enable employees to conduct our internal business operations and to serve our customers. The information transmitted, received and stored using these systems are valuable Company assets that must be protected. Employees must maintain compliance with all applicable security and use policies designed to protect the confidentiality, integrity and availability of our systems, and the information they manage. Fundamental to these protections are safeguarding user IDs, passwords and access to our computer and network facilities.

Employees must use the Company's computer systems and networks appropriately at all times. In general, employee use of electronic systems (*e.g.*, email, Internet, telephone, etc.) should be limited to Company business and any non-business use should be incidental, infrequent, and not distract from work obligations. Under no circumstances are employees permitted to use these systems to access or download inappropriate online content or distribute communications that others may find to be harassing, threatening, offensive or discriminatory. Given the spectrum of possible uses for these systems, the Company relies upon the common sense and good judgment of employees to ensure their appropriate use.

8. External Communications

As a public company, the Company is subject to close scrutiny from many sources, including news organizations, government regulators and the investment community. As such, it is essential that all Company external communications and disclosures be consistent, accurate and delivered on a timely basis. To ensure that the Company complies with applicable disclosure laws, rules and regulations, external communications should only be made by directors and employees or third parties authorized to speak on behalf of the Company. Any director, employee or third party who is not so authorized and receives an inquiry from the media, government regulators or the investment community regarding the Company should refer the party to Andrew Backman, at abackman@hc2.com or (212) 339-5836.

9. External Organizations and Presentations

Occasionally, directors and/or employees may be invited to participate in professional groups or deliver presentations at industry conferences. In these circumstances, it is important that directors and employees make it clear that the opinions expressed are their own and not necessarily representative of the Company's position on a particular issue or subject. Additionally, such directors and employees should submit any presentation decks to the Legal Department for pre-approval.

10. Social Media

While the format may be different, the foregoing rules also apply when directors and employees use online venues (*e.g.*, blogs, chat rooms, and social media) to communicate with others. In addition to not disclosing any confidential or proprietary information, directors and employees must be sure to choose their words carefully in these settings, as even a seemingly harmless comment or blog posting could be misconstrued as an official Company statement and viewed by potentially thousands of individuals and organizations.

11. Government Inquiries

The Company cooperates with appropriate requests from governmental authorities. All information provided by directors and employees must be truthful and accurate. It is never permissible to mislead an investigator, alter or destroy records, or withhold information in response to an investigation. Directors and employees who receive a subpoena, court order or other request for information or inquiry from a law enforcement or other governmental agency should contact the Legal Department for guidance on how to proceed.

K. Waivers

The Code is approved by the Board or a duly authorized committee thereof and the Board or such committee must approve any subsequent amendment to the Code. The Company reserves the right to amend, alter or terminate the Code at any time for any reason. The most current version of the Code can be found on the Company's website. In extremely limited circumstances, a director or employee may make a request for a waiver to the Code in writing to HC2's Chief Legal Officer. Any waivers for executive officers and directors can only be approved by the Board or a duly authorized committee thereof and will be promptly disclosed as required by applicable law, regulation or stock exchange rule.

L. Non-Exclusivity

The Code is not a comprehensive document that addresses every compliance issue that may arise or covers every law or policy that could apply to the Company's businesses around the world. Many of the topics included in the Code are addressed in greater detail in other Company materials, including other Company policies, any applicable handbooks, country-specific policies and departmental operating procedures. Directors and employees have an ongoing obligation to familiarize themselves with any laws, rules, regulations and Company policies that may be applicable to their positions within the Company. The Code has Company-wide application and a global reach; therefore, any direct conflict between a non-Board approved policy and the letter or spirit of the Code is to be resolved in favor of the Code. Individual business units and departments are permitted to adopt policies that are more restrictive than those found in the Code or other specific policies. If a particular country's local laws, rules or regulations conflict with the Code, then the Company will follow the local law.

M. Enforcement and Discipline

The consequences of non-compliance with the Code can include damage to our reputation in the marketplace, significant fines and penalties, and even criminal liability for individuals or the Company. Failure to comply with the standards detailed by the Code will result in disciplinary action. All discipline will be applied in a manner consistent with the Company's policies, procedures and applicable law. Certain violations may require the Company to refer the matter to an appropriate governmental or regulatory authority for additional investigation or prosecution.

III. WHISTLEBLOWER PROCEDURES

A. Purpose

In recognition of the value that an empowered and informed workforce can offer, the Company has established these whistleblower procedures (the "*Whistleblower Policy*") to provide guidance and procedures for the receipt, investigation and retention of reports of suspected violations of laws, rules, regulations, the Code or other Company policies, sometimes known as whistleblower complaints. These procedures reinforces the protections afforded to individuals who come forward, including anonymously, to make reports or voice concerns about violations of the Code, Company policy or suspected violations of law or regulations that govern the Company's operations. The Company appreciates that individuals will raise concerns only if they know that doing so will make a difference; therefore, all requests for guidance and reports of misconduct will be treated seriously

and the Company is committed to taking corrective action whenever necessary to ensure the appropriate and timely resolution and remediation of matters.

B. Non-Retaliation

An empowered workforce, free to ask the tough questions or report suspected wrongdoing, is integral to an open environment of integrity and compliance. The Company prohibits any form of retaliation or adverse action against an employee for, in good faith, either raising an issue or reporting a suspected violation of laws, rules, regulations, the Code or any other Company policy or procedure or assisting with an investigation. Violators will be subject to disciplinary action up to and including termination of employment.

C. Seeking Guidance and Reporting

Employees, other than executives, that have a concern are encouraged to discuss such matters first with their supervisors, managers or other appropriate personnel such as local Human Resources, or local legal counsel. Concerns relating to the employment terms or matters at the local workplace such as conflicts with colleagues and managers, disciplinary sanctions, correct and timely payment of wages and benefits should normally be resolved locally, unless the matter is exceptionally serious as described below or cannot be reported through the normal reporting line.

Complaints that involve or may have the potential to involve either HC2's Chief Legal Officer or HC2's Chief Administrative Officer should be submitted directly to the Chairman of the Audit Committee.

This Policy is intended for serious and sensitive concerns that could have an adverse impact on the operations and performance of the business of the Company, and which due to the nature of the concern, cannot be reported through your normal reporting line, such as:

- unlawful activity;
- financial fraud (for example accounting manipulation, non-compliance with internal controls procedures, misappropriation of assets or fraudulent statements);
- bribery or corruption (for example conflicts of interest, bribery, sponsorships and donations, gifts or facilitation payments);
- serious endangerment to environment, health and safety; and
- activities which otherwise by law, treaty or agreement, amount to serious improper conduct.

Individuals seeking to report serious and sensitive concerns have a number of reporting options available that allow for confidential, and if they so choose, anonymous reports:

The Compliance Hotline

- Toll-free (+1-888-475-8376)
- Available 24x7, 365 days a year
- Managed by Navex Global
- · Able to take reports on an anonymous basis

<u>The Audit Committee</u>

Regular mail marked "Confidential" HC2 Holdings, Inc. Attn.: Audit Committee Chairman 450 Park Ave, 30th Floor New York, NY 10022

Compliance Email

compliance@hc2.com

D. Contents of Report

In order for the Company to properly respond and investigate a report under this Policy, the report should set forth facts and evidence, rather than speculation and innuendo. Preferably, the report should include, but is not limited to, the following:

- The event or issue that is the subject of the report;
- To the extent applicable, the approximate date, time and location of any specific event(s);
- The name(s) of the person(s) involved; and
- Any documentary or other evidence to support the report.

E. Treatment of Complaints

All reports to HC2's Chief Legal Officer and HC2's Chief Administrative Officer (the "*Compliance Officers*") or otherwise will be treated confidentially, to the extent practicable and legally permissible. Directors and employees have the option of remaining anonymous, although there may be instances where the investigation of a report may be negatively impacted if the Compliance Officers or other personnel are unable to follow up with an anonymous caller to obtain additional information. All investigations will be conducted in a fair and impartial manner, consistent with Company policies and any applicable law.

Individuals making an anonymous report to the Compliance Hotline will be given a reference number to allow for follow-up on the matter including the opportunity to respond to additional questions from those responsible for investigating the matter.

Reports of misconduct must always be made in good faith, which means that individuals must believe the information they are providing is complete and truthful.

Directors and employees are expected to cooperate fully with an investigation and any failure to do so may lead to disciplinary action. When the circumstances and findings of a completed investigation permit, the Compliance Officers or other personnel will contact the party making the original report to close out the matter.

The Compliance Officers will conduct a preliminary review of the facts and circumstances of reports captured by internal resources and contact the Audit Committee Chairman to provide a high-level overview of the allegation and to seek approval for the anticipated plan of investigation. At its discretion, the Audit Committee can remove an allegation from the internal investigative process and assign the matter to an independent third-party for investigation. If a third-party investigator is chosen, the Company will provide the Audit Committee with funding to compensate the third-party investigator.

The conclusions of the investigation shall be submitted to the Audit Committee. The Audit Committee may decide on placing any investigation report before the Board depending on the gravity and magnitude of the violation.

F. Retention of the Records

The Company shall retain a copy of all complaints or concerns, investigation reports and all relevant documentation thereof. The Audit Committee shall decide the period of retention of all these records by the Company, subject to limitations in applicable legislation.

G. Policy Updates and Exceptions

The Compliance Officers, while working with appropriate internal or external parties, is required to review and propose amendments to the Whistleblower Policy as may be necessary to ensure continuous alignment with all applicable laws, rules and regulations. Any updates to the Whistleblower Policy must be approved by the Audit Committee. Exceptions to the Whistleblower Policy can only be approved by the Audit Committee to the extent that such exceptions may be permitted by law, regulation and exchange listing standard.

H. Other Majority-Owned Subsidiaries

The Company will share this Policy with other majority-owned subsidiaries who have installed similar Whistleblower Policies and Procedures, see <u>Schedule B</u>. All majority-owned subsidiaries are required to report to the Company any Whistle Blower activity and results of investigations on a quarterly basis, and may request to present to the Board at its next regularly scheduled Board meeting.

IV. CERTIFICATION

By signing this statement, I acknowledge that I have reviewed, understand, and agree to adhere to the Company's Code of Conduct and Whistleblower Policy and Procedures. I am not personally aware of any violations of the Code and furthermore, I understand that if the Company determines my conduct warrants it, I may be subject to disciplinary action.

Signature	
Name	
	(Please print)

Date

Approved: May 8, 2014 Amended and Restated: November 3, 2016 Amended and Restated: May 4, 2017 Schedule A Updated: November 27, 2017

2	n
4	v

Schedule A

Subsidiaries Included in Policy (Updated as of 11/27/17)

704Games Company ANG Holdings, Inc. Benevir Biopharm, Inc. Continental Insurance Group, LTD Continental LTC, Inc. Continental General Insurance Company DBM Global Inc. Genovel Orthopedics, Inc. Global Marine Holdings, LLC HC2 Broadcasting Holdings, Ic. Nervve Technologies, Inc. PTGi International Carrier Services, Inc. PTGi International Carrier Services, LTD Pansend Life Sciences, LLC R2 Dermatology Incorporated

Schedule B

Subsidiaries with Own Policy

Global Marine Systems Limited



HC2 GLOBAL ANTI-CORRUPTION, TRADE AND ANTI-BOYCOTT COMPLIANCE PROGRAM

TABLE OF CONTENTS

INTRODUCTION 1 ANTI-CORRUPTION COMPLIANCE POLICY 2 Appendix 1 Questionnaire for Prospective Non-U.S. Business Partners 8 Appendix 2 HC2 Internal Form for Engaging a Non-U.S. Business Partner 12 Appendix 3 Anti-Corruption Provision to Appear in Contracts with Non-U.S. Business Partners 15 Appendix 4 HC2 Anti-Corruption Compliance Certification for Non-U.S. Business Partners 17 EXPORT CONTROLS AND SANCTIONS COMPLIANCE POLICY 18 Appendix 1 OFAC Sanctions Compliance Matrix 22 U.S. ANTI-BOYCOTT REGULATIONS COMPLIANCE POLICY

HC2 PERSONNEL COMPLIANCE CERTIFICATION	35

INTRODUCTION

HC2 Holdings, Inc. and all of its domestic and international subsidiaries, divisions and controlled affiliates (collectively, "HC2" or "the Company") are committed to complying with all applicable laws, including but not limited to the anti-corruption, economic sanctions, export controls, and anti-boycott laws of the United States and other countries where the Company conducts business. As part of this company-wide commitment, it is important that all HC2 directors, officers, and employees (collectively, "HC2 Personnel") familiarize themselves with the policies contained in this Global Anti-Corruption, Trade and Anti-Boycott Compliance Program. The Company also expects and requires its agents, contractors (to the extent they act on behalf of the Company), consultants, joint venture partners and other persons acting for or on behalf of the Company (collectively, "Business Partners") to comply with all such policies where applicable.

Non-compliance with the requirements enunciated in the policies and procedures set forth in the Company's Global Anti-Corruption, Trade and Anti-Boycott Compliance Program will not be tolerated, and failure by any HC2 Personnel or Business Partners to adhere to these requirements may result in disciplinary action up to and including termination of employment or other business relationships. Please consult with your business unit's compliance officer or the office of the Chief Legal Officer if you have any questions or concerns regarding these policies.

1. PURPOSE OF POLICY

HC2 Holdings, Inc. and all of its domestic and international subsidiaries, divisions and controlled affiliates (collectively, "HC2" or the "Company") are committed to complying with all applicable anti-corruption laws, including but not limited to the U.S. Foreign Corrupt Practices Act ("FCPA") and U.K. Bribery Act. It is a long-standing policy of the Company to maintain company-wide awareness of the importance of anti-corruption compliance.

HC2 strictly prohibits both bribery of public officials and commercial bribery among private parties. HC2 does not condone direct or indirect provision or receipt of any improper payment or other benefits for the purpose of gaining any commercial advantage.

2. APPLICABILITY OF POLICY

This Policy applies to all directors, officers, and employees of the Company (collectively, "HC2 Personnel"), regardless of where they are located. The Company also expects and requires its agents, contractors (to the extent they act on behalf of the Company), consultants, joint venture partners and other persons acting for or on behalf of the Company (collectively, "Business Partners") to comply with all applicable anti-corruption laws and adhere to the principles enunciated in this Policy. Failure to comply with or adhere to this Policy may result in sanctions, including termination.

3. OVERVIEW OF THE APPLICABLE LAWS

All HC2 Personnel and Business Partners must adhere to the following principles:

- General Requirements to Prevent Bribery and Corruption: HC2 Personnel and Business Partners may not directly or indirectly give, offer, promise, approve, or authorize to give to anyone or accept or solicit from anyone anything of value (including without limitation a bribe, kickback or other unlawful incentive) to induce the recipient to take improper action that would bestow a commercial benefit or advantage on HC2 or any other party. Such behavior is prohibited by the anti-corruption laws in virtually all jurisdictions.
 - **Dealing with Public Officials:** While it is both the policy of HC2 and the laws of most jurisdictions to prohibit bribery of any kind, interactions with Public Officials (defined below) must be handled with particular care. Bribery of Public Officials is subject to heightened scrutiny in every country and often carries severe criminal penalties for both the HC2 Personnel and the Company. In addition, a number of countries, including the United States, penalize the bribery of Public Officials in countries other than their own. For purposes of this Policy, a "Public Official" includes:
 - any officer or employee of a government or any department, agency, or instrumentality thereof (including, for example, a government-owned or government-controlled state enterprise) or of a public international organization (such as the World Bank, the International Monetary Fund, the World Intellectual Property Organization, the World Trade Organization, the OECD, and the Organization of American States);
 - any person acting in an official capacity for or on behalf of any of the foregoing (including, for example, a civil servant, administrative or judicial personnel, a member of the military, or a consultant who holds a government position); or
 - any political party or official thereof, or any candidate for political office.

Though they are not themselves Public Officials, interactions with family members of, and others closely associated with, Public Officials are subject to the same restrictions as interactions with Public Officials.

It is not uncommon for HC2 Personnel or Business Partners to come into contact with employees of state-owned or state-controlled enterprises, who are considered Public Officials under the laws of most jurisdictions and this Policy. HC2 Personnel and Business Partners should exercise extra care in dealing with employees of state-owned or state-controlled enterprises, particularly if such employees are in a position to purchase, administer, recommend, or arrange for the purchase of HC2 products or services.

- **Dealing with Persons Who May Engage in Bribery:** HC2 Personnel or Business Partners may not give, offer, promise, approve, or authorize to give to any person anything of value if circumstances reasonably suggest a high probability that such person has offered or given, or will offer or give, anything of value for the purpose of inducing improper action.
- **Books and Records:** Certain anti-corruption and other laws require that HC2's books and records accurately and completely reflect the Company's transactions and activities, including the nature and purpose of the transaction or activity. Accordingly, HC2 Personnel and Business Partners must not make any false, inaccurate, incomplete, or misleading entry in Company books or records for any reason.
 - **Engagement of Business Partners:** Anti-corruption laws generally prohibit improper payments made indirectly or otherwise through a third party. Under certain circumstances, HC2 may be held liable for violations of applicable anti-corruption laws committed by its Business Partners. Accordingly, HC2 Personnel must exercise particular care when engaging Business Partners to act on behalf, or in the interest, of the Company.

4. PROCEDURES

While HC2 Personnel must comply with the general requirements of this Policy at all times, detailed procedures regarding certain activities are important for anticorruption compliance. Accordingly, all HC2 Personnel must understand and comply with the following procedures and related requirements.

Gifts, Entertainment/Meals, and Hosted Travel for Public Officials: HC2 Personnel must consult and comply with the following chart before providing any gift, entertainment, meal, or hosted travel. Please note that the requirements here apply to all types of gifts, entertainment, meal, or hosted travel provided to Public Officials (or their family members and/or other persons closely associated with the Public Official) regardless of what they are called (e.g. compensation for the Public Officials' time, consulting fee, promotional rebates or rewards, etc.).

	Allowed	<u>Not Allowed Without Prior</u> <u>Approval from the office of</u> <u>Chief Legal Officer</u>
Gifts	Gifts that (1) are not cash or cash equivalents, (2) have market value not exceeding \$50 at any one time or \$100 in total in any calendar year, (3) are incidental to promoting, demonstrating or explaining HC2's products/services and bear the Company's name or logo, (4) are not a kickback, reward, or rebate conditioned on the sale of HC2's services or products, (5) are presented openly with complete transparency, and (6) clearly are allowed under local laws.	Cash or cash equivalents (e.g., gift cards, travelers' checks, gift certificates, pre- paid telephone or shopping cards). Any gift that functions as a kickback, reward, or rebate linked to the sale of HC2's services or products. Any gift with a market value greater than \$50; or gifts to one Public Official and/or persons closely associated with the Public Official with an aggregate value exceeding \$100 in a calendar year. Jewelry, electronics, art work, home furnishings, wine or alcoholic products or other items that are personal in nature and have no connection to HC2's business. Any gift that is not clearly allowed under local laws.
Entertainment/Meals	Meals that (1) are modest by local standards and in any case do not exceed \$50 per person or \$200 in total, (2) are incidental to legitimate informational discussions in which HC2's services or products are being promoted, explained or demonstrated, (3) involve the presence of HC2 Personnel, (4) do not involve any family member or guest of the Public Official, and (5) clearly are allowed under local laws.	Any form of entertainment (e.g., theater or sports events, vacation trips, etc.). Any meal that exceeds \$50 per person or \$200 in total. Any meal that is not incidental to legitimate business discussions. Examples include (1) meals that are considered luxurious by local standards; (2) meals at which no HC2 Personnel are present; (3) meals at which family members or guests of the Public Official are also present. Any meal for which the payment is made through reimbursement to the Public Official. Any meal that is not clearly legal under local laws.
Hosted Travel	Payment of reasonable travel and accommodation expenses that (1) are not considered luxurious by local standards and in any case do not exceed \$50 per person or \$200 in total, (2) are associated solely with inviting the Public Official to attend legitimate business/informational meetings for the purpose of promoting, explaining, demonstrating HC2's products or services, (3) do not cover the expenses of the Public Official's family members or guests, (4) do not include any cash per diems or "pocket money," and (5) clearly are allowed under local laws.	Paying for expenses that exceed \$50 per person or \$200 in total. Paying for travel for family members or guests. Paying for "side trips" or sight-seeing trips or hotel stays longer than necessary to conduct legitimate business meetings for the purpose of promoting or demonstrating company products and services. Cash per diems or "pocket money."

	Allowed	<u>Not Allowed Without Prior Approval from the</u> <u>office of Chief Legal Officer</u>
Other benefits	Not Applicable	Internship (paid or unpaid)
		Employment opportunities
		Any benefit that functions as an incentive or reward linked to the sale of HC2's services or products

- Facilitating Payments to Public Officials: HC2 does not allow facilitating payments (i.e., small-sum payments to low-level Public Officials to facilitate routine, non-discretionary government functions). Before making or authorizing any forms of payment that may be perceived as facilitating payments, HC2 Personnel, Business Partners or any person acting on behalf of the Company must contact your business unit's compliance officer or the office of the Chief Legal Officer for prior approval. Please note that legally mandated fees for expedited administrative services are not prohibited facilitating payments for the purpose of this provision if such fees are set in an open and transparent manner and if proper receipt is provided.
 - **Political Contributions and Charitable Donations:** Political contributions and charitable donations on behalf of HC2 require prior approval from the Chairman of the Company. While HC2 Personnel may make personal political or charitable contributions allowed by applicable laws, any such payments must be made in a way that makes clear to an outside observer that they are purely personal in nature and are unrelated to any business of HC2. In this regard, HC2 Personnel or Business Partners should exercise extra caution when the personal political or charitable donation is suggested or recommended by a party that has an existing or potential business relationship with HC2.
 - Accounting: HC2 Personnel must not under any circumstance record any entry in Company books or records in a way that may conceal the true nature of a transaction. All Company operations and record keeping shall be conducted in accordance with the following principles:
 - Transactions are executed in accordance with management's general or specific authorization;
 - Transactions are accurately recorded in sufficient detail as necessary to: (i) permit preparation of financial statements in conformity with generally accepted accounting principles or any other criteria applicable to such statements; and (ii) maintain accountability for assets;
 - Access to assets is permitted only in accordance with management's general or specific authorization;
 - The recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences;
 - No undisclosed or unrecorded fund or asset of the Company shall be established or maintained for any purpose.

Engagement of Non-U.S. Business Partners: HC2 Personnel may not engage a non-U.S. Business Partner on behalf of HC2 without consulting with your business unit's compliance officer or the office of the Chief Legal Officer. HC2 Personnel who wish to retain or engage a non-U.S. Business Partner must conduct pre-engagement due diligence, obtain written compliance commitments and certifications, and undertake appropriate post-engagement supervision and oversight of non-U.S. Business Partners' activities on behalf of the Company.

Before engaging a non-U.S. Business Partner, HC2 Personnel must:

§Determine that there is a legitimate business-related reason for engaging the non-U.S. Business Partner;

§Conduct appropriate due diligence to ensure that the non-U.S. Business Partner is reputable, such as through an in-person interview, a visit to the Business Partner's premises, reviewing the credentials of the Business Partner, and obtaining independent business references;

§Confirm that the non-U.S. Business Partner is qualified to perform the required service and has sufficient capacity to do so;

§Ensure that the non-U.S. Business Partner's proposed compensation is based on prevailing industry standards and is commensurate with his/her experience and the services to be rendered;

Sobtain answers to the Questionnaire for Prospective Non-U.S. Business Partner attached as Appendix 1; and

§Fill out the HC2 Internal Form for Engaging a Non-U.S. Business Partner attached as Appendix 2.

When finalizing the engagement of a non-U.S. Business Partner, HC2 Personnel must:

§Discuss the Company's anti-corruption compliance expectations, as reflected in this Policy and applicable anti-corruption laws, with the non-U.S. Business Partner, and bring to the attention of your business unit's compliance officer or the office of the Chief Legal Officer any concerns or suspicions that the non-U.S. Business Partner might not meet those expectations;

§Using the information provided by a non-U.S. Business Partner in Appendix 1:

- o Screen the non-U.S. Business Partner and its key personnel against the U.S. Government's Consolidated Screening List of Restricted Parties, which is available at http://2016.export.gov/ecr/eg_main_023148.asp (click on "CSL Search Engine" in the "Tools" section); and
- Contact your business unit's compliance officer or the office of the Chief Legal Officer to perform a detailed screening through the Company's web-based subscription with Dow Jones Risk & Compliance;¹

§Include in all contracts with non-U.S. Business Partners provisions requiring compliance with applicable anti-corruption laws and providing indemnification for any costs or penalties stemming from the Business Partner's violations of anti-corruption laws. A sample <u>Anti-Corruption Provision to Appear in Contracts with Non-U.S. Business Partners</u> is attached as **Appendix 3**.

- After engaging a non-U.S. Business Partner, HC2 Personnel must:
 - §Actively monitor the Non-U.S. Business Partner's actions and activities on behalf of HC2 and report any suspicious activities to your business unit's compliance officer or the office of the Chief Legal Officer;

§Collect the HC2 Anti-Corruption Compliance Certification for Non-U.S. Business Partners attached as Appendix 4.

§In appropriate circumstances, or as directed by your business unit's compliance officer or the office of the Chief Legal Officer, provide, or arrange for, compliance training of the non-U.S. Business Partner and auditing of the non-U.S. Business Partner's business operations.

- Personnel Training and Certification: In consultation with the office of the Chief Legal Officer, each business unit will determine which of its respective employees should receive anti-corruption compliance training and how often such training will be provided. The compliance officers for each business unit shall submit an annual report and proposal for training for the following year to the office of the Chief Legal Officer <u>by no later than</u> <u>December 15 of each year</u>.
- Compliance with This Policy: Compliance with the requirements and procedures set forth in this Policy, whether or not involving any adverse consequences to the Company, will be an important consideration in HC2 Personnel's annual evaluation and decisions concerning HC2 Personnel's promotion and continuing employment.
 - **Reporting Violations:** HC2 Personnel must report all known or suspected violations of this Policy and/or applicable anti-corruption laws. The Company has established procedures for submitting complaints, reports, and concerns by HC2 Personnel regarding violations of this Policy as well as other questionable accounting or auditing matters. These procedures are published on the Company's website, at www.HC2.com, under the link chain "Investors Corporate Governance Whistleblower Reporting". Such complaints, reports or concerns may be communicated to the Company's office of the Chief Legal Officer or the Chairman of the Audit Committee through a toll-free hotline at 1-888-475-8376 (available 24x7, 365 days a year) or by email at compliance@hc2.com. In accordance with the Company's policy, no person who in good faith reports violations or suspected violations of this Policy shall be subject to retaliation.
 - **Contact Information for Additional Guidance:** Any questions regarding this Policy should be directed to the office of the Chief Legal Officer at compliance@hc2.com.

Approved by the Board of Directors of HC2 Holdings, Inc. on May 4, 2017.

¹ Each business unit with international operations, must obtain a separate license to perform these searches. For information on obtaining a seat, the business unit's compliance officer should contact Jeanne E. Rouleau at jrouleau@hc2.com.

Appendix 1 Questionnaire for Prospective Non-U.S. Business Partners

Full Legal Name of Company:

Please attach separate sheets to this form if additional space is needed.

Basic Company Information

1. Your company's location, website, and contact information (please provide both a phone number and an email address)

2. Full legal names of owners, principals, and key personnel of company

- a. Owners/major shareholders _____
- b. Chairman/president
- c. Managing director _____
- d. Other <u>senior</u> company officers

3. Please provide one or more general references (persons/firms familiar with your company).

Interactions with Government Officials

4. Are any governmental interactions, approvals or licenses required or needed for this project?

Yes _____ No _____

If the answer is yes, please describe.

5. a. Does any current or former government official, political party official, candidate for political office, or relative of such a person have a direct or indirect ownership interest in your company?

Yes ____ No ____

b.Is any current or former government official, political party official, candidate for political office, or relative of such a person an employee, officer or director of the Company?

Yes _____ No _____

If the answer to either (a) or (b) above is "yes", please state the name(s) and official position(s) of such person(s).

6.

Is your company in any way affiliated with one or more state-owned entities?

Yes _____ No _____

If the answer is yes, please provide the full legal name(s) of such state-owned entit(ies).

7. Does your company or any employee of your company have any business, familial, or other association(s) with any current or former government department, government official, political party official, candidate for political office, or relative of such a person?

Yes ____ No ____

If the answer is yes, please describe.

Contemplated	Business	Relationship	with HC2

8.	Description of the work to be performed for HC2
9.	Countries and territories where the work at issue will be performed
	Full legal names of individuals who will perform the work at issue (if more than five individuals, please list the names of individuals who will supervise the work)
10.	Description of the background of individuals performing the work at issue and explanation on why you believe they are qualified to perform the work
11.	Bank account into which your company will be paid:
	a. Bank name and location
	b. Account number
	C. Name on account
<u>Anti-C</u>	Corruption Compliance
12.	Has your company, any associated entity, any prior entity, any prior associated entity, any present or former employee of your entity been suspended from doing business

Has your company, any associated entity, any prior entity, any prior associated entity, any present or former employee of your entity been suspended from doing business in any capacity, been charged with any criminal act, or been the subject of any allegation of fraud, misrepresentation, bribery, or other related activities?

Yes ____ No ____

If	the	answer	is	yes,	please	explain.

13.	Have you reviewed the attached HC2 Anti-Corruption Compliance Policy and HC2 Anti-Corruption Compliance Certification for Non-U.S. Business Partners?
	Yes No
14.	Will your company comply with applicable anti-corruption laws and the compliance expectations in the attached HC2 Anti-Corruption Compliance Policy?
	Yes No
15.	Will your company be willing to cooperate with HC2 if HC2 has any inquiries regarding the work your company performs for HC2 (including allowing HC2 to audit your company's books and records related to the work performed for HC2) and provide to HC2 the signed <u>HC2 Anti-Corruption Compliance Certification for Non-U.S.</u> <u>Business Partners</u> annually during the terms of the work at issue?
	Yes No
I certify	that, to my best knowledge and belief, the above answers are complete and accurate.
Author	ized Representative Signature:
Printed	Name of Representative:
Repres	entative Title:
Compa	ny:
Date: _	

Appendix 2 <u>HC2 Internal Form for Engaging a Non-U.S. Business Partner</u>

Individual and Business Unit requesting approval:

Please fill out this form after you review the completed <u>HC2 Questionnaire for Prospective Non-U.S. Business Partners</u> and conduct due diligence on the prospective Non-U.S. Business Partner. Please attach separate sheets to this form if additional space is needed.

Contemplated Business Relationship with HC2

- 1. How were you referred to the prospective Non-U.S. Business Partner? If applicable, please provide name and contact information of the person/entity that provides the referral.
- 2. Who at HC2 will supervise the work done by the prospective Non-U.S. Business Partner?

3. Please describe the business reason for engaging the prospective Non-U.S. Business Partner. If there are other parties offering similar services, please explain why HC2 should engage the prospective Non-U.S. Business Partner rather than such other parties.

4. Please describe the compensation to be provided to the prospective Non-U.S. Business Partner.

5. Please explain why you believe the compensation to the prospective Non-U.S. Business Partner is commercially reasonable.

Due Diligence

6. Please describe the due diligence conducted regarding the prospective Non-U.S. Business Partner (e.g. interview of company officers, visits to company officers, review of company books and records, internet search, third-party references, etc.).

	prospective Non-U.S. Business Partner's answers to the <u>HC2 Questionnaire for Prospective Non-U.S. Business Partners</u> and the due dilig firm that the answers are both complete and, to your best knowledge after conducting due diligence, accurate?
Yes No	
If the answer is no, please	xplain.
	ed by the prospective Non-U.S. Business Partner in the prospective Non-U.S. Business Partner's name and lawfully organized in the co U.S. Business Partner does business?
Yes No	
If the answer is no, please	xplain.
Do you believe that the p <u>Corruption Compliance F</u>	spective Non-U.S. Business Partner will strictly comply with applicable anti-corruption laws and the compliance expectations in the <u>HC2 /</u>
Yes No	
If the answer is no, please	xplain.
	pective Non-U.S. Business Partner will be willing to (1) cooperate with HC2 if HC2 has any inquiries regarding the work performed for

Do you expect that the prospective Non-U.S. Business Partner will be willing to (1) cooperate with HC2 if HC2 has any inquiries regarding the work performed for HC2 (including allowing HC2 to audit the prospective Non-U.S. Business Partner's books and records related to the work performed for HC2) and (2) provide to HC2 the signed HC2 Anti-Corruption Compliance Certification for Non-U.S. Business Partners annually during the terms of the work at issue?

If the answer is no, please explain.

11. Please identify any potential areas of risks associated with the prospective Non-U.S. Business Partner that would cause the prospective Non-U.S. Business Partner to fail to strictly comply with applicable anti-corruption laws and the <u>HC2 Anti-Corruption Compliance Policy</u>. You are encouraged to be as forthcoming and as specific as possible in answering this question (attaching additional pages if needed). Identification of potential risks will not necessarily mean that the prospective Non-U.S. Business Partner will not be engaged, and failure to report risk areas that HC2 Personnel reasonably should have known after due inquiries could adversely impact HC2 Personnel's evaluations and continuing employment at HC2).

I certify that, to my best knowledge after due inquiries, the above answers are complete and accurate.

HC2 Personnel Signature:

Printed Name of HC2 Personnel: _____

HC2 Personnel Title:

Division/Office:

Date: ____

APPROVAL BY HC2 DIVISIONAL SUPERVISOR/OFFICE OF THE CHIEF LEGAL OFFICER

Date: _____

Appendix 3 <u>Anti-Corruption Provision to Appear in Contracts with Non-U.S. Business Partners</u>

<u>Compliance with Laws</u>: **{NON-U.S. BUSINESS PARTNER}** agrees that it will adhere to and comply with all applicable laws, including but not limited to all applicable anticorruption laws. **{NON-U.S. BUSINESS PARTNER}** hereby, as appropriate, represents, warrants, affirms and agrees that:

1. None of the principals, officers or employees of **{NON-U.S. BUSINESS PARTNER}** is a government official, an official of any public international organization, a political party official, a candidate for political office, or an officer, director or employee of an enterprise owned or controlled by a government (referred to collectively herein as "Public Official").

2. In connection with the performance of **{NON-U.S. BUSINESS PARTNER'S}** obligations under this Agreement, **{NON-U.S. BUSINESS PARTNER}** and any person acting on **{NON-U.S. BUSINESS PARTNER'S}** behalf shall comply with all applicable anti-corruption laws and regulations, including but not limited to the U.S. Foreign Corrupt Practices Act (the "FCPA") and the U.K. Bribery Act (the "UKBA"). In particular, **{NON-U.S. BUSINESS PARTNER}** and any person acting on **{NON-U.S. BUSINESS PARTNER'S}** behalf shall not, directly or indirectly, make, promise, approve, authorize or offer to give to anyone – or accept or solicit from anyone – anything of value (including but not limited to cash payments) to induce the recipient (whether or not such person is a Public Official) to take improper action that would bestow a commercial benefit or advantage on HC2 or any other party.

3. If there are any additional owners in, or changes in the ownership of, **{NON-U.S. BUSINESS PARTNER}**, **{NON-U.S. BUSINESS PARTNER}** shall notify HC2 of the identity of the new owner(s) within fifteen (15) business days after such ownership change. HC2 may terminate this Agreement if HC2 disapproves of such ownership change.

4. Before retaining or engaging any other party to perform the Services under this Agreement, **{NON-U.S. BUSINESS PARTNER}** shall obtain written approval from HC2 and require the other party to agree in writing to the same anti-corruption compliance representations and warranties provided herein.

5. **{NON-U.S. BUSINESS PARTNER}** shall, upon request of HC2, sign and provide a certification, attached as Annex [___], to HC2 confirming **{NON-U.S. BUSINESS PARTNER'S}** compliance with the aforementioned compliance commitments.

6. **{NON-U.S. BUSINESS PARTNER}** shall fully cooperate with HC2's request for internal anti-corruption compliance investigations, and shall allow HC2 (or its representatives) to audit the **{NON-U.S. BUSINESS PARTNER'S}** books and records with respect to the work performed on behalf of HC2 at any time.

7. **{NON-U.S. BUSINESS PARTNER}** shall provide periodic training on applicable anti-corruptions laws and regulations, including without limitation the FCPA and the UKBA, to its employees who are or may be involved in performing the Services under this Agreement.

8. If HC2 believes that a breach of any of the anti-corruption compliance representations and warranties described herein has occurred or will occur. HC2 may withhold further commission payments until it has received confirmation to its satisfaction that no breach has occurred or will occur. HC2 shall not be liable to **{NON-U.S.**

BUSINESS PARTNER} for any claims, losses, interests, or damages related to its decision to withhold commission payments under this provision.

9. If **{NON-U.S. BUSINESS PARTNER}** or any person acting on behalf of **{NON-U.S. BUSINESS PARTNER}** fails to comply with the anti-corruption compliance representations and warranties described herein, (i) HC2 has the right to terminate this Agreement, notwithstanding any other provision of the Agreement to the contrary, (ii) **{NON-U.S. BUSINESS PARTNER}** shall forfeit any commissions owed to it by HC2, (iii) **{NON-U.S. BUSINESS PARTNER}** shall immediately return to HC2 any commissions paid by HC2 arising from any transaction in violation of the anti-corruption compliance representations and warranties described herein, (iv) **{NON-U.S. BUSINESS PARTNER}** shall keep HC2 fully and effectively indemnified against any and all losses, claims, damages, charges, expenses, liabilities, demands, proceedings, and actions which HC2 or HC2's affiliates may sustain or incur by reason of **{NON-U.S. BUSINESS PARTNER'S}** failure to comply with the anti-corruption compliance representations and warranties described herein.

10. All payments due to **{NON-U.S. BUSINESS PARTNER}** under the Agreement shall be made by bank wire transfer to a lawful business account of **{NON-U.S. BUSINESS PARTNER}** at a designated bank in the country/countries where performances of **{NON-U.S. BUSINESS PARTNER'S}** obligations under this Agreement will take place.

11. In no event shall HC2 be obligated under this Agreement to take any action that HC2 believes in good faith would cause it to be in violation of any anti-corruption laws of any country, including, without limitation, the FCPA and the UKBA.

12. All of the information and certifications **{NON-U.S. BUSINESS PARTNER}** has provided to HC2 was and remains accurate and complete. If subsequent developments cause such information or the certifications provided herein to no longer be accurate or complete, **{NON-U.S. BUSINESS PARTNER}** shall immediately notify HC2 of such change in circumstances.

Appendix 4

HC2 Anti-Corruption Compliance Certification for Non-U.S. Business Partners

My company, ______, which has a business relationship with HC2 or its affiliates (hereinafter "HC2" or the "Company"), hereby as appropriate represents, warrants, affirms, acknowledges and agrees that:

1. Throughout the duration of my company's business relationship with HC2, my company has complied and will comply with all of the requirements set forth in the "Compliance with Laws" provision in the agreement between my company and HC2.

2. In particular, my company and any person acting on my company's behalf have complied and will comply with all applicable laws and regulations, including but not limited to the U.S. Foreign Corrupt Practices Act and the U.K. Bribery Act. In this regard, those who represent or perform work for HC2 have reviewed and complied with, and will comply with, the <u>HC2 Anti-Corruption Compliance Policy</u> throughout the duration of my company's business relationship with HC2.

3. All of the information and certifications my company has provided to HC2 were and remain accurate and complete. If subsequent developments cause such information or the certifications provided herein to no longer be accurate or complete, my company shall immediately notify HC2 of such change in circumstances.

4. My company authorizes HC2 to use any information contained in this Certification in any litigation, arbitration, or other proceeding in connection with my company's representation and/or performance of work for HC2.

Authorized Representative Signature:		
Printed Name of Representative:		
Representative Title:		
Company:		
Date:		

HC2 HOLDINGS, INC. EXPORT CONTROLS AND SANCTIONS COMPLIANCE POLICY

1. PURPOSE OF POLICY

HC2 Holdings, Inc. and all of its domestic and international subsidiaries, divisions and controlled affiliates (collectively, "HC2" or "the Company") are committed to complying with all applicable U.S. economic sanctions and export control laws and regulations (collectively, "U.S. Sanctions" and "Export Controls").

2. APPLICABILITY OF POLICY

This policy applies to all directors, officers, and employees of the Company (collectively, "HC2 Personnel"), regardless of where they are located. The Company also expects and requires its agent, contractors (to the extent they act on behalf of the Company), consultants, joint venture partners and other persons acting for or on behalf of the Company (collectively "Business Partners") to comply with all applicable U.S. Sanctions and Export Controls and adhere to the principles enunciated in this Policy. Failure to comply with or adhere to this Policy may result in sanctions, including termination.

3. OVERVIEW OF THE APPLICABLE LAWS

All HC2 Personnel and Business Partners must adhere to the following principles:

- General Requirements of U.S. Sanctions and Export Controls: U.S. Sanctions and Export Controls are primarily administered by the U.S. Department of the Treasury, Office of Foreign Assets Control ("OFAC") and the U.S. Department of Commerce, Bureau of Industry and Security ("BIS"). U.S. Sanctions are intended to protect and promote the national security, foreign policy and economy of the United States and are composed of various distinct sanctions programs, consisting of comprehensive economic sanctions against certain countries and territories and targeted economic sanctions against individuals and entities worldwide. With limited exceptions, economic sanctions programs are generally applicable only to U.S. persons and do not directly restrict the activities of foreign persons when such activities involve no U.S. nexus (*e.g.*, no involvement of goods, software or technology subject to U.S. law, or payments in U.S. Dollars that have to clear through the U.S. financial system). U.S. Export Control restrictions "follow the goods," meaning that non-U.S. persons may also have obligations under Export Control laws when dealing in goods, software or technology (collectively, "Items") subject to U.S. law (either due to its origin or level of controlled U.S. content in Items made outside the United States).
 - O Exceptions to General Rules: U.S. Sanctions contain several important exceptions that have extraterritorial applications. Specifically, U.S. Sanctions against Cuba and Iran contain provisions that require non-U.S. entities that are "owned or controlled" by U.S. entities to comply with the same restrictions as their U.S. parent company. Because HC2 portfolio companies are "owned or controlled" by HC2, a U.S. Company, all HC2 portfolio companies and all HC2 Personnel and Business Partners acting for or on behalf of the Company must comply with U.S. Sanctions against Cuba and Iran.
 - O Joint Ventures: For purposes of U.S. Sanctions, joint ventures in which an HC2 portfolio company holds a 50% or greater stake are subject to the same restrictions as any HC2 portfolio company. For JVs in which an HC2 portfolio company holds less than a 50% stake, restrictions may still apply depending upon the specific facts and circumstances. Consultation with the Chief Legal Officer is

required before any dealings by HC2 Personnel detailed or seconded to, or otherwise working with, such a JV in connection with any JV activities involving a sanctioned country.

- **Restricted Parties**: U.S. Sanctions and Export Controls impose restrictions on dealings with a number of individuals and entities (collectively, "Restricted Parties"). These restrictions can range from limitations on specific kinds of activities to a total prohibition on nearly all dealings involving a Restricted Party in any capacity. There are a number of different lists of Restricted Parties, the most important of which include the following:
 - O Specially Designated Nationals: Individuals and entities embargoed or "blocked" pursuant to U.S. Sanctions are generally included on OFAC's Specially Designated and Nationals and Blocked Persons List ("SDN List"). The SDN List identifies individuals and entities that have engaged in sanctionable conduct or have been determined to be owned or controlled by, or to be acting on behalf of, OFAC-sanctioned governments, entities or individuals. The SDN List also identifies other individuals and entities that have been designated as sanctions targets pursuant to sanctions programs that are not country-specific (*e.g.*, terrorists, narco-traffickers, arms dealers, etc.). U.S. persons are generally prohibited from transferring, paying, exporting, withdrawing or otherwise dealing in the property or interests in property of any Specially Designated National ("SDN"), as well as any entity in which an SDN owns, or two or more SDNs own collectively, directly or indirectly, an interest of 50% or more, even if such entity is not separately identified on the SDN List.
 - O Sectoral Sanctions: OFAC has designated a number of entities, primarily in Russia, on the Sectoral Sanctions Identification List ("SSIL"). SSIL entities are subject to narrower restrictions than SDNs, but certain activities (*e.g.*, providing services for certain deepwater, Arctic offshore, or shale oil projects, extensions of credit beyond certain specified time periods) are prohibited.
 - O Foreign Sanctions Evaders: OFAC also maintains the Foreign Sanctions Evaders List ("FSE List"), which identifies foreign individuals and entities determined to have violated, attempted to violate or caused a violation of U.S. Sanctions against Syria or Iran or facilitated deceptive transactions for or on behalf of persons subject to U.S. Sanctions (each such individual or entity, an "FSE"). Transactions involving an FSE are generally prohibited when the activity has any U.S. nexus.
 - Entity List: BIS maintains a number of lists, including the Entity List ("Entity List"), which identifies foreign individuals and entities that have engaged in activities contrary to U.S. national security and/or foreign policy interests. Transactions with entities on the Entity List that involve U.S.-origin Items, or foreign-origin items containing greater than a *de minimis* quantity of U.S.-origin controlled content, generally require a license from BIS.
- Due Diligence Required: OFAC, BIS, and other U.S. Government agencies maintain additional lists of sanctioned entities and individuals that prescribe more limited
 restrictions on dealings, which may also be relevant depending on the circumstances. Therefore, consulting the SDN List, the FSE List, the SSIL, the Entity List, and other
 U.S. Government restricted party lists (collectively, the "Consolidated Screening List") and conducting adequate background checks with respect to all counterparties,
 including agents, suppliers, and customers, is critical to ensuring compliance with U.S. Sanctions and Export Controls.
- Export Control Licensing Requirements: U.S. Export Controls impose compliance requirements beyond merely avoiding certain dealings with Restricted Parties, and may impose a licensing requirement for the transfer of U.S.-origin Items or foreign-origin Items containing greater than a *de minimis* quantity of U.S.-origin controlled content, even to persons and countries that are not subject to U.S. Sanctions, including many European countries, or for certain end users (for example, for use in targeted oil/gas exploration projects in Russia, even if no designated parties are involved). Accordingly, all exports of such Items must be screened

by the your business unit's compliance officer or the office of the Chief Legal Officer to determine any potential licensing obligations before they are exported.

4. PROCEDURES

While HC2 Personnel must comply with the general requirements of this Policy at all times, detailed procedures regarding certain activities are important for U.S. Sanctions and Export Controls compliance. Accordingly, all HC2 Personnel must understand and comply with the following procedures and related requirements.

- **New Business Intake**: Prior to pursuing or engaging in any dealings with a potential customer, agent, consultant, third-party service provider, intermediary or business partner (each a "Counterparty"), all HC2 Personnel must conduct due diligence in accordance with the following procedures in furtherance of ensuring full compliance with U.S. Sanctions:
 - O Step 1 Determine the location, domicile, and nationality of the Counterparty: U.S. Sanctions prescribe broad prohibitions on dealings in, with and involving certain countries and territories, as well as persons owned or controlled by, or acting on behalf of, persons who are domiciled, or who are nationals of, such countries and territories.
 - § If the Counterparty is located or domiciled in, or is a national of or ordinarily resident in, a country/territory categorized as "red" or "orange" in the below table of Sanctioned Countries and Territories, <u>do not</u> engage in any further communications with the Counterparty without first contacting your business unit's compliance officer or the office of the Chief Legal Officer and obtaining guidance and approval with regard to any dealings with such Counterparty.
 - It is highly likely that dealings with a Counterparty that is located or domiciled in, or that is the national of, a "red" country/territory will be prohibited under U.S. Sanctions.
 - It is likely that dealings with a Counterparty that is located or domiciled in, or that is the national of, an "orange" country or territory will also be prohibited under U.S. Sanctions if they involve certain industries or projects (*e.g.*, oil and gas) or a certain part of the territory (*i.e.*, the Crimea region of Ukraine/Russia please note that Crimea is itself listed as a "red" territory but when dealing with Ukraine or Russia, you need to ensure that a transaction does not involve the Crimea region itself).
 - § If the Counterparty is located or domiciled in, or is a national of, a country/territory categorized as "yellow" or "green" in the below table of Sanctioned Countries and Territories, proceed to Step 2 below.
 - O Step 2 Obtain the legal name and the address of the Counterparty: Provide such legal name and address to your business unit's compliance officer or the office of the Chief Legal Officer for screening against the Consolidated Screening List. Do not proceed with any further communications or dealings with such Counterparty until your business unit's compliance officer or the office of the Chief Legal Officer has confirmed that such Counterparty is not designated or that any designation does not give rise to a prohibition under U.S. Sanctions on relevant dealings with such Counterparty.
 - o Step 3- Obtain additional information regarding the Counterparty: Obtain any additional information needed to confirm that the Counterparty is not owned or controlled by, or acting on behalf



of, any entities or individuals (i) that are located or domiciled in, or nationals of or ordinarily resident in, sanctioned countries or territories or (ii) are designated persons or entities. Important questions to ask a Counterparty in this regard include:

- § Who are the officers and directors of the Counterparty? Where are they located or domiciled? Of which countries are they nationals?
- § Who are the owners (including ultimate beneficial owners, not just immediate corporate owners in the case of multi-layered structures) of the Counterparty and what are the levels of their respective ownership interests? Where are they located or domiciled?
- § Does the Counterparty maintain any significant connections with sanctioned countries/territories under U.S. Sanctions? Does the Counterparty engage, or has it engaged in the past, in dealings with entities or individuals that are the subject or target of U.S. Sanctions?
- Once such information is obtained, it should be provided to your business unit's compliance officer or the office of the Chief Legal Officer, who will assess the compliance risks associated with dealings with the Counterparty.

Sanctioned Countries and Territories [*]	
Country or Territory	Protocol
Cuba, Crimea, Iran, Sudan (North Sudan)**, Syria, North Korea	Avoid. Do not engage in any communications with the Counterparty without first contacting your business unit's compliance officer or the office of the Chief Legal Officer.
Russia, Ukraine	Proceed with extreme caution. Do not engage in any communications with the Counterparty without first contacting your business unit's compliance officer or the office of the Chief Legal Officer.
Balkans, Belarus, Burundi, Central African Republic, Democratic Republic of Congo, Iraq, Lebanon, Libya, Somalia, South Sudan, Venezuela, Yemen, Zimbabwe	
All other countries/territories	Proceed to STEP 2

*For further information regarding the various countries and territories sanctioned under U.S. Sanctions, please refer to the OFAC Sanctions Compliance Matrix, attached hereto at Appendix 1, and contact your business unit's compliance officer or the office of the Chief Legal Officer.

**Sudan sanctions are currently subject to an OFAC general license authorizing most transactions that are otherwise prohibited under the Sudanese Sanctions Regulations – however, this general license is scheduled to be reviewed by July 2017. Unless and until such time as the U.S. Government revokes the Sudan sanctions, it is HC2's policy to continue treating Sudan as a "red" country for compliance purposes.

Approved by the Board of Directors of HC2 Holdings, Inc. on May 4, 2017

Appendix 1

OFAC Sanctions Compliance Matrix

The following matrix is intended as a reference guide to be used to facilitate compliance with U.S. Sanctions and is not comprehensive as to the full scope of all relevant laws and regulations, nor does it address sanctions imposed by the EU and other jurisdictions. For example, certain non-geographic sanctions programs, such as counter narcotics trafficking or counterterrorism sanctions, are not included in this matrix. Therefore, all transactions must be screened against the Consolidated Screening List to ensure that any SDNs designated under non-geographic sanctions programs (or other restricted parties) will be captured; while the countries listed below are particularly likely to raise compliance concerns, SDNs or other restricted parties may be found in any country. Please consult your business unit's compliance officer or the office of the Chief Legal Officer for further guidance.

OFAC Sanctions Program by Country	Description of Sanctions Program	Recommended Action
Balkans-Related Sanctions	 Prohibits dealings with individuals and entities determined to be involved in extremist violence in the former Yugoslav Republic of Macedonia, southern Serbia, the former Federal Republic of Yugoslavia or elsewhere in the Western Balkans region. These sanctions are not territorial in nature (they do not apply to all of the countries in the Western Balkans region); instead, these are list-based restrictions that apply to designated parties/SDNs. 	 Before engaging in any dealings: (i) screen all counterparties, including agents, suppliers and customers, against the Consolidated Screening List; and (ii) conduct adequate background checks regarding such counterparties to ensure that they are not owned, controlled or acting on behalf of any sanctioned parties.
		 If any counterparty is designated on the Consolidated Screening List or suspected to be owned, controlled or acting on behalf of a sanctioned party, cease contact with such counterparty, do not proceed with the contemplated dealings and consult the office of your business unit's compliance officer or the office of the Chief Legal Officer immediately.
Belarus Sanctions	 Prohibits dealings with individuals and entities determined to be involved in, or responsible for, activity that has undermined the democratic process in Belarus, human rights abuses related to political oppression in Belarus or public corruption in Belarus. The entire government of Belarus is not designated but several high-ranking officials, including the President, are SDNs, so dealings with the government can raise issues if such SDNs are involved, including as a signatory for agreements. 	 Before engaging in any dealings: (i) screen all counterparties, including agents, suppliers and customers, against the Consolidated Screening List; and (ii) conduct adequate background checks regarding such counterparties to ensure that they are not owned, controlled or acting on behalf of any sanctioned parties.
		 If any counterparty is designated on the Consolidated Screening List or suspected to be owned, controlled or acting on behalf of a sanctioned party, cease contact with such counterparty, do not proceed with the contemplated dealings and consult your business unit's compliance officer or the office of the Chief Legal Officer immediately.

OFAC Sanctions Program by Country	Description of Sanctions Program	Recommended Action
Burundi-Related Sanctions	 Prohibits dealings with individuals and entities determined to be involved in large-scale human rights abuses in Burundi. These sanctions are not territorial in nature (they do not apply to the whole country); instead, these are list-based restrictions that apply to designated parties/SDNs. 	counterparties, including agents, suppliers and customers, against the Consolidated Screening List; and (ii) conduct adequate background checks regarding such counterparties to ensure that they are
		 If any counterparty is designated on the Consolidated Screening List or suspected to be owned, controlled or acting on behalf of a sanctioned party, cease contact with such counterparty, do not proceed with the contemplated dealings and consult your business unit's compliance officer or the office of the Chief Legal Officer immediately.
Central African Republic Sanctions	 Prohibits dealings with individuals and entities determined to be involved in large-scale human rights abuses in the Central African Republic. These sanctions are not territorial in nature (they do not apply to the whole country); instead, these are list-based restrictions that apply to designated parties/SDNs. 	counterparties, including agents, suppliers and customers, against the Consolidated Screening List; and (ii) conduct adequate background checks regarding such counterparties to ensure that they are not owned, controlled or acting on behalf of any
		Screening List or suspected to be owned, controlled or acting on behalf of a sanctioned party, cease contact with such counterparty, do not proceed with the contemplated dealings and consult your business unit's compliance officer or the office of the Chief Legal Officer immediately.

OFAC Sanctions Program by Country	Description of Sanctions Program	Recommended Action
Cuba Sanctions	 Comprehensive country-based sanctions program that prohibits virtually all dealings in, with or involving Cuba and Cuban nationals. 	 DO NOT ENGAGE IN ANY ACTIVITIES RELATED TO DEALINGS IN, WITH OR INVOLVING CUBA OR NATIONALS OF CUBA WITHOUT FIRST CONSULTING WITH YOUR BUSINESS UNIT'S COMPLIANCE OFFICER OR THE OFFICE OF THE CHIEF LEGAL OFFICER.
Democratic Republic of Congo-Related Sanctions	 Prohibits dealings with individuals and entities determined to be involved in large scale human rights abuses in the Democratic Republic of Congo. These sanctions are not territorial in nature (they do not apply to the whole country); instead, these are list-based restrictions that apply to designated parties/SDNs. 	 Before engaging in any dealings: (i) screen all counterparties, including agents, suppliers and customers, against the Consolidated Screening List; and (ii) conduct adequate background checks regarding such counterparties to ensure that they are not owned, controlled or acting on behalf of any sanctioned parties. If any counterparty is designated on the Consolidated Screening List or suspected to be owned, controlled or acting on behalf of a sanctioned party, cease contact with such counterparty, do not proceed with the contemplated dealings and consult your business unit's compliance officer or the office of the Chief Legal Officer immediately.
Iran Sanctions	 Comprehensive country-based sanctions program that prohibits most dealings in, with or involving Iran and Iranian nationals. Includes certain extraterritorial, "secondary sanctions" applicable to non-U.S. persons and directs the imposition of penalties, such as restrictions on accessing the U.S. financial system and ineligibility for contracting with U.S. government entities (among others), on foreign persons determined to have engaged in certain dealings in, with or involving Iran or Iranian nationals. 	• DO NOT ENGAGE IN ANY ACTIVITIES RELATED TO DEALINGS IN, WITH OR INVOLVING IRAN OR NATIONALS OF IRAN WITHOUT FIRST CONSULTING WITH YOUR BUSINESS UNIT'S COMPLIANCE OFFICER OR THE OFFICE OF THE CHIEF LEGAL OFFICER.

OFAC Sanctions Program by Country	Description of Sanctions Program	Recommended Action
Iraq-Related Sanctions	 Prohibits dealings with individuals and entities determined to be associated with the former regime of Saddam Hussein. 	 Before engaging in any dealings: (i) screen all counterparties, including agents, suppliers and customers, against the Consolidated Screening List; and (ii) conduct adequate background checks regarding such counterparties to ensure that they are not owned, controlled or acting on behalf of any sanctioned parties. If any counterparty is designated on the Consolidated Screening List or suspected to be owned, controlled or acting on behalf of a sanctioned party, cease contact with such counterparty, do not proceed with the contemplated dealings and consult your business unit's compliance officer or the office of the Chief Legal Officer immediately.
Lebanon-Related Sanctions	 Prohibits dealings with individuals and entities determined to be involved in actions designed to undermine Lebanon's legitimate and democratically elected government. These sanctions are not territorial in nature (they do not apply to the whole country); instead, these are list-based restrictions that apply to designated parties/SDNs. 	 Before engaging in any dealings: (i) screen all counterparties, including agents, suppliers and customers, against the Consolidated Screening List; and (ii) conduct adequate background checks regarding such counterparties to ensure that they are not owned, controlled or acting on behalf of any sanctioned parties. If any counterparty is designated on the Consolidated Screening List or suspected to be owned, controlled or acting on behalf of a sanctioned party, cease contact with such counterparty, do not proceed with the contemplated dealings and consult your business unit's compliance officer or the office of the Chief Legal Officer immediately.

OFAC Sanctions Program by Country	Description of Sanctions Program	Recommended Action
Libya Sanctions	 Prohibits dealings with individuals and entities determined to be associated with the former regime of Muammar al-Gaddafi. These sanctions are no longer territorial in nature (they do not apply to the whole country and do not target the entire Government of Libya any more); instead, these are list-based restrictions that apply to designated parties/SDNs, and continue to restrict access to certain assets of the Libyan Investment Authority (and its owned/controlled entities) that were blocked as of September 2011 in the United States, or within possession/control of U.S. persons. 	screen all counterparties, including agents, suppliers and customers, against the Consolidated Screening List; and (ii) conduct adequate background checks regarding such counterparties to ensure that they are not owned, controlled or acting
		 If any counterparty is designated on the Consolidated Screening List or suspected to be owned, controlled or acting on behalf of a sanctioned party, cease contact with such counterparty, do not proceed with the contemplated dealings and consult your business unit's compliance officer or the office of the Chief Legal Officer immediately.
North Korea Sanctions	 Broad, country-based program that prohibits certain dealings in, with or involving North Korea and certain North Korean entities and individuals, including the Government of North Korea. 	ACTIVITIES RELATED TO
Somalia Sanctions	 Prohibits dealings with individuals and entities determined to be involved in acts that directly or indirectly threaten the peace, security, or stability of Somalia. These sanctions are not territorial in nature (they do not apply to the whole country); instead, these are list-based restrictions that apply to designated parties/SDNs. 	screen all counterparties, including agents, suppliers and customers, against the Consolidated Screening
		 If any counterparty is designated on the Consolidated Screening List or suspected to be owned, controlled or acting on behalf of a sanctioned party, cease contact with such counterparty, do not proceed with the contemplated dealings and consult your business unit's compliance officer or the office of the Chief Legal Officer immediately.

OFAC Sanctions Program by Country	Description of Sanctions Program	Recommended Action
Sudan Sanctions	 Broad, country-based program that prohibits certain dealings in, with or involving Sudan (North Sudan) and certain Sudanese entities and individuals, including the Government of Sudan. 	
		 Note: On January 17, 2017, OFAC issued a general license authorizing all transactions prohibited by the Sudanese Sanctions Regulations. This license is currently scheduled for review by July 2017. Unless and until such time as the U.S. Government revokes the Sudan sanctions, it is HC2's policy to continue treating Sudan as a "red" country for compliance purposes.
South Sudan-Related Sanctions	 Prohibits dealings with individuals and entities determined to be involved in acts that directly or indirectly threaten the peace, security, or stability of South Sudan. These sanctions are not territorial in nature (they do not apply to the whole country); instead, these are list- based restrictions that apply to designated parties/SDNs. 	counterparties, including agents, suppliers and
	partes of the	 If any counterparty is designated on the Consolidated Screening List or suspected to be owned, controlled or acting on behalf of a sanctioned party, cease contact with such counterparty, do not proceed with the contemplated dealings and consult your business unit's compliance officer or the office of the Chief Legal Officer immediately.
Syria Sanctions	 Broad, country-based sanctions program that prohibits certain dealings with respect to Syria and certain entities and individuals associated with the Syrian government. 	• DO NOT ENGAGE IN ANY ACTIVITIES RELATED TO DEALINGS IN, WITH OR INVOLVING SYRIA OR NATIONALS OF SYRIA WITHOUT FIRST CONSULTING WITH YOUR BUSINESS UNIT'S COMPLIANCE OFFICER OR THE OFFICE OF THE CHIEF LEGAL OFFICER.

OFAC Sanctions Program by Country	Description of Sanctions Program	Recommended Action
Ukraine/Russia- Related Sanctions including Comprehensive Sanctions on the Crimea region	 Prescribes broad restrictions on dealings in, with or involving Crimea region of Ukraine/Russia. Targets certain individuals and entities – most of which are domiciled or located in Ukraine or Russia – determined to be involved in activities that undermine democratic processes and institutions in Ukraine, threaten the peace, security, stability, sovereignty and territorial integrity of Ukraine or contribute to the misappropriation of the assets of Ukraine. Parties sanctioned under so-called "sectoral" sanctions and subject to more limited restrictions (compared to SDNs) may be designated under the Sectoral Sanctions Identifications List, which carries restrictions regarding certain dealings by U.S. persons with such designated parties. Sectoral sanctions also target certain projects in Russia's oil and gas sector, not just parties in the financial and defense sectors. Parties subject to asset freeze restrictions are independently designated under the SDN List. 	 DEALINGS IN, WITH OR INVOLVING CRIMEA OR CRIMEAN PERSONS WITHOUT FIRST CONSULTING WITH YOUR BUSINESS UNIT'S COMPLIANCE OFFICER OR THE OFFICE OF THE CHIEF LEGAL OFFICER. With respect to any other dealings in, with or involving Ukraine or Russia, before engaging in any such dealings: (i) screen all counterparties, including agents, suppliers and customers, against the Consolidated Screening List; and (ii) conduct adequate background checks regarding such counterparties to ensure that they are not owned, controlled or acting on behalf of any sanctioned parties.
<u>Venezuela-</u> <u>Related</u> <u>Sanctions</u>	 Prohibits dealings with individuals or entities determined to be involved in activities that undermine democratic processes and institutions in Venezuela, or who have participated in serious human rights abuses or the corruption of public officials in the Government of Venezuela. These sanctions are not territorial in nature (they do not apply to the whole country); instead, these are list-based restrictions that apply to designated parties/SDNs. 	including agents, suppliers and customers, against the Consolidated Screening List; and (ii) conduct adequate background checks regarding such counterparties to ensure that

OFAC Sanctions Program by Country	Description of Sanctions Program	Recommended Action
Yemen-Related Sanctions	 Prohibits dealings with individuals and entities determined to be involved in acts that directly or indirectly threaten the peace, security or stability of Yemen. These sanctions are not territorial in nature (they do not apply to the whole country); instead, these are list-based restrictions that apply to designated parties/SDNs. 	 Before engaging in any dealings: (i) screen all counterparties, including agents, suppliers and customers, against the Consolidated Screening List; and (ii) conduct adequate background checks regarding such counterparties to ensure that they are not owned, controlled or acting on behalf of any sanctioned parties. If any counterparty is designated on the Consolidated Screening List or suspected to be owned, controlled or acting on behalf of a sanctioned party, cease contact with such counterparty, do not proceed with the contemplated dealings and consult your business unit's compliance officer or the office of the Chief Legal Officer immediately.
Zimbabwe Sanctions	 Prohibits dealings with individuals and entities determined to be involved in undermining democratic processes or institutions in Zimbabwe. These sanctions are not territorial in nature (they do not apply to the whole country); instead, these are list-based restrictions that apply to designated parties/SDNs. 	 Before engaging in any dealings: (i) screen all counterparties, including agents, suppliers and customers, against the Consolidated Screening List; and (ii) conduct adequate background checks regarding such counterparties to ensure that they are not owned, controlled or acting on behalf of any sanctioned parties. If any counterparty is designated on the Consolidated Screening List or suspected to be owned, controlled or acting on behalf of a sanctioned party, cease contact with such counterparty, do not proceed with the contemplated dealings and consult your business unit's compliance officer or the office of the Chief Legal Officer immediately.

HC2 HOLDINGS, INC. U.S. ANTI-BOYCOTT REGULATIONS COMPLIANCE POLICY

1. PURPOSE OF POLICY

HC2 Holdings, Inc. and all of its domestic and international subsidiaries, divisions and controlled affiliates (collectively, "HC2" or the "Company") are committed to complying with U.S. anti-boycott laws and regulations. It is the policy of the Company to maintain company-wide awareness of the importance of anti-boycott compliance.

2. APPLICABILITY OF POLICY

This Policy applies to all directors, officers, and employees of the Company (collectively, "HC2 Personnel"), regardless of where they are located. The Company also expects and requires its agents, contractors (to the extent they act on behalf of the Company), consultants, joint venture partners and other persons acting for or on behalf of the Company (collectively, "Business Partners") to comply with all applicable anti-boycott laws and adhere to the principles enunciated in this Policy. Failure to comply with or adhere to this Policy may result in sanctions, including termination.

3. OVERVIEW OF THE APPLICABLE LAWS

All HC2 Personnel and Business Partners must adhere to the following principles:

- **General Requirements:** U.S. Anti-boycott Regulations prohibit HC2, including its non-U.S. controlled subsidiaries, from engaging in activities that directly or indirectly support international boycotts opposed by the United States, such as (but not limited to) the Arab League boycott of Israel. This includes activities such as agreeing to comply with all laws of a boycotting country, refusing to do business with an entity in a boycotted country, responding to a request to provide information about business relationships with boycotted countries or blacklisted persons, or entering into contracts or agreements with prohibited boycott-related language.
- **Reporting** Requirements: The applicable regulations administered by the U.S. Department of Commerce's Bureau of Industry and Security ("BIS") and the U.S. Department of the Treasury's Internal Revenue Service ("IRS") also require reporting to the U.S. Government the receipt of a request to engage in boycott-related activities, even if the Company does not respond or agree to the request.

4. PROCEDURES

- **Review:** In reviewing transaction documentation (e.g. letters of credit, requests for proposals, bid invitations, proposed contracts, purchase orders, shipping documents), HC2 Personnel should flag any suspected boycott-related language or requests and must report such language to your business unit's compliance officer or the office of the Chief Legal Officer. Boycott-related language or requests can come in any form, written or oral, and from any party, regardless of location. Consult Appendix 1 for examples of boycott-related language or requests. The following countries are currently listed by the U.S. Government as Boycotting Countries:
 - Iraq Libya Saudi
 - Kuwait Qatar Syria
 - Lebanon United Arab Emirates Yemen

Boycott-related requests and language can, however, be received from any country. In particular, transactions involving the following countries may generate boycott requests:

- Bahrain Oman
- Bangladesh Pakistan
- Malaysia
- **Contact Legal**: HC2 Personnel who receive a boycott-related request or language <u>should not respond to the request in any way</u>, and should immediately contact your business unit's compliance officer or the office of the Chief Legal Officer with a general explanation of the circumstances of the request/language and a copy of the request/language. Your business unit's compliance officer or the office of the Chief Legal Officer will review the suspected boycott-related language or requests to determine whether reporting is required under the relevant Anti-boycott Regulations and will provide guidance to employees as to whether the language must be removed, amended or other action taken.
- Quarterly Reporting: HC2 must report to BIS quarterly whether it received any boycott-related requests and what actions were taken in response to any request.
- Annual Reporting: HC2 must report to the IRS, on Form 5713 filed in conjunction with its annual federal corporate tax return, any operations HC2 has in or related to Boycotting Countries, as well as whether it has received requests and what actions were taken in response to any request.
- Training: HC2 shall maintain and disseminate training materials for appropriate personnel to increase awareness of Anti-boycott Regulations compliance.
- **Recordkeeping**: HC2 shall maintain records of all Anti-boycott-related documents, including boycott requests, related correspondence, legal advice, and the company's response to any boycott requests. Such records must be maintained for a minimum of five years.

Approved by the Board of Directors of HC2 Holdings, Inc. on May 4, 2017

Appendix 1

Anti-boycott Regulations Compliance: Examples and Tips

The following examples will help identify when a potential boycott request has been received and must be reported to your business unit's compliance officer or the office of the Chief Legal Officer.

1. Boycott-related Language or Requests

2.

Boycott-related language or requests can include:

- Request to certify that the company's goods are not Israeli
- Request to certify that the company will "comply with" local laws of a Boycotting Country
- Request to certify that the company is not blacklisted
- Requirement in an RFP that no "blacklisted" suppliers are eligible
- Provisions or sections of contracts or other documents titled "Israel Boycott" or "Import Embargo"
- Agreements that seek confirmation that no company employees are nationals of Israel
- Requests to certify that vessels carrying goods are "eligible to enter Arab ports"
- Agreements which require that packaging used on the company's products not include any "six pointed stars"
- Requests not to provide service or instructions in Hebrew
- Requests to provide a boycott certification letter during product registration

Examples of Common Boycott-related Language or Requests

- The Contractor shall comply in all respects with the requirements of the law of Country X relating to the boycott of Israel.
- The Contractor shall comply in all respects with the laws of Country X.
- Company shall at its own expense and at all times comply with all laws, rules, regulations or requirements of the Government of State X and Country X and any bodies having jurisdiction over the site and the access thereto and therefrom including, but not limited to, the Boycott of Israel Regulations.
- All relevant laws, rules and regulation of all duly constituted government authorities of State X of Country X, including laws with respect to boycott of Israel, shall apply in the performance of this purchase order.
- Supplies of our purchase order should never be consigned or shipped by steamers included on Israel Boycott list.
- A signed statement from the shipping company, or its agent, stating the name, flag, and nationality of the carrying vessel and confirming...that it is permitted to enter the ports of Country X in accordance with the laws and regulations of Country X.
- The following questions should be answered on behalf of the Company and all its branch companies:
 - 0 Do you have or ever have had a branch or main company, factory or assembly plant in Israel or have sold to an Israeli?
 - 0 Do you have or ever have had general agencies or offices in Israel for your Middle Eastern or international operations?
- In order for Company to be paid under the letter of credit, Company agrees to provide a certificate at port of entry into Country X that the goods do not contain any parts manufactured by blacklisted companies.



- Requirement for the registration of pharmaceutical companies: certification letter regarding the boycott of Israel (i.e. do not comprise any parts, raw materials, labor
 or capital or Israeli origin).
- Company agrees that no goods provided under this contract shall be manufactured by companies located in Israel, or by subsidiaries or affiliates of such companies.

3. Key "Issue Spotting" Phrases

- "Israel" or "Israeli"
- "Boycott"
- "Blacklist" or "Blacklisted"
- "Whitelist" or "Whitelisted"
- References to ethnicity or religious affiliation
- "Comply with" vs "Apply" local law
- Destination restrictions
- Eligible source country restrictions
- Eligible to enter ports of boycotting country
- Negative certificates of origin (i.e. goods are not of Israeli origin)

4. What To Do Next

- If you have received a possible boycott-related request, immediately transmit the document to your business unit's compliance officer or the office of the Chief Legal Officer for review.
- Do not respond to the request until you have received guidance from your business unit's compliance officer or the office of the Chief Legal Officer
- You may be asked by your business unit's compliance officer or the office of the Chief Legal Officer to take certain actions in the company's response, such as removing language, implementing revisions or making other modifications to an RFP, bid, offer, contract, or purchase order.
- Retain copies of the original request, related correspondence, legal guidance, and the company's response for a minimum of five years.

5. Questions or Concerns?

Please contact Jeanne E. Rouleau at jrouleau@hc2.com or at 954-663-1147 with any questions or concerns regarding this document.

HC2 Holdings, Inc.

HC2 Global Anti-Corruption, Trade and Anti-Boycott Compliance Program

HC2 Personnel Compliance Certification

Please complete, sign, date and return this form to your Human Resources Department with a copy to the *Legal Department*, *HC2 Holdings*, *Inc.*, 450 *Park Avenue*, 30th *Floor*, *New York*, *NY* 10022.

I, the undersigned:

- (i) acknowledge that I have received and reviewed the HC2 Global Anti-Corruption, Trade and Anti-Boycott Compliance Program effective May 4, 2017 (the "*Policy*");
- (ii) fully understand that, in connection with my work for HC2, I have an obligation to fully adhere to the Policy and all applicable (i) anticorruption laws and regulations, including but not limited to the U.S. Foreign Corrupt Practices Act and the U.K. Bribery Act, (ii) U.S. economic sanctions and export control laws and regulations, and (iii) U.S. anti-boycott laws and regulations;
- (iii) agree to comply fully with the Policy in the future;
- (iv) understand that, in order to ensure that the Policy remains in compliance with current laws and regulations, among other reasons, HC2 may, at its option, change, delete, suspend, or discontinue any part or parts of the Policy at any time and without prior notice; and
- (v) understand any violation of this Policy or applicable laws or regulations may be considered grounds for termination of my employment or other disciplinary action by HC2.

HC2 Personnel Signature

Name (Please Print)

Business Unit/Office:

Date:



HC2 HOLDINGS, INC.

2017 INSIDER TRADING POLICY

Purpose

The purpose of this Insider Trading Policy (the **Policy**") is to promote compliance with applicable securities laws by HC2 Holdings, Inc. (the "**Company**") and all directors, officers and employees thereof, in order to preserve the reputation and integrity of the Company as well as that of all persons affiliated with it.

Scope

This Policy applies to (i) all directors, officers and employees of the Company, its subsidiaries and controlled affiliates (collectively, the "*Subsidiaries*") and their respective immediate family members living in their households ("*Immediate Family Members*"), (ii) other persons as may be designated from time to time by the Legal Department in consultation with the Chief Executive Officer and Chief Financial Officer, and (iii) any entities controlled by individuals subject to the Policy, including any corporations, partnerships or trusts (collectively referred to as "*Covered Persons*"). Transactions by the entities described in the foregoing clause (iii) should be treated for the purposes of this Policy and applicable securities laws as if they were for the individual's own account.

The Policy applies to any and all transactions in the Company's securities. For purposes of this Policy, the Company's securities include without limitation its common stock, options to purchase or sell its common stock, preferred stock, convertible and non-convertible debentures, warrants and other similar instruments, and exchange-traded options or other derivative securities related thereto (collectively, "*Company Securities*"). The Policy also governs similar transactions in the securities of other entities, including without limitation (i) Subsidiaries, (ii) any other company in which the Company has a direct or indirect beneficial ownership interest, and (iii) for Restricted Security Persons (as defined below) only, companies included on the prohibited trading list maintained by the Company's Legal Department (as such list may be amended from time to time, the "*Prohibited Trading List*").

For the purposes of this Policy, the term "*HC2 Entity*" means (a) for all Covered Persons who are Restricted Security Persons, any company meeting the definition of the foregoing clauses (i), (ii) or (iii), and (b) for all Covered Persons who are not Restricted Security Persons, any company meeting the definition of the foregoing clauses (i) or (ii). The Prohibited Trading List shall contain a list of companies with which the Company has entered into a non-disclosure or other confidentiality agreement with respect to a potential investment or acquisition or any other potential investment or aquisition. Companies shall be added to and removed from the Prohibited Trading List at the

discretion of the Chief Executive Officer in consultation with the Legal Department. Before entering into a transaction potentially covered by this Policy with respect to a company, a Restricted Security Person shall first be required to confirm with the Legal Department or his or her designee that such company is not on the Prohibited Trading List, as more fully set forth in the "Transaction Pre-approvals" section of this Policy.

For the purposes of this Policy, "Restricted Security Persons" include (i) all directors of the Company, (ii) those officers and other employees of the Company and its Subsidiaries who regularly work on mergers and acquisitions and related investment opportunities, (iii) other employees of the Company and its Subsidiaries as may be designated from time to time by the Chief Executive Officer in consultation with the Chief Financial Officer and the Legal Department, and (iv) the respective Immediate Family Members of any person covered by clauses (i), (ii) or (iii) of this paragraph. The Legal Department shall maintain a list of all persons covered by this paragraph (the "*Restricted Security Person List*").

Prohibited Activities in Company Securities

No Covered Person may trade, buy, sell, transfer, dispose of or acquire Company Securities while in possession of material, non-public information concerning the Company and for one (1) full trading day after the date such material information has been publicly released. Notwithstanding the foregoing, this Policy shall not be deemed to prohibit:

- purchases of Company Securities by a Covered Person from the Company or sales of Company Securities by a Covered Person to the Company;
- the exercise of employee stock options or other equity awards or the surrender of shares to the Company in payment of the exercise price
 or in satisfaction of any tax withholding obligations in a manner permitted by the applicable equity award agreement, or vesting of equitybased awards, that in each case do not involve a market sale of Company Securities (the "cashless exercise" of a Company stock option
 through a broker does involve a market sale of Company Securities, and therefore would not qualify under this exception);
- bona fide gifts of Company Securities that do not satisfy a legal obligation; or
- purchases or sales of Company Securities made pursuant to a properly established 10b5-1 Plan (as defined below under "Rule 10b5-1 Trading Plans") that has been approved in writing by the Chief Legal Officer (or in the case of the Chief Legal Officer, by the Chief Executive Officer or Chief Financial Officer) in accordance with this Policy and has not been modified in any respect after such initial approval without such modification being approved in writing in accordance with this Policy.

No Covered Person may "tip" or disclose material, non-public information to any outside person (including spouses, parents, children, siblings or other family members, investors, or members of the investment community and/or news media), unless in accordance with the Company's policies regarding the protection or authorized external disclosure of Company information. Furthermore, no Covered Person may give trading advice of any kind about the

Company or Company Securities to anyone, whether or not such Covered Person is aware of material, non-public information.

Prohibited Activities in HC2 Entity Securities

Acquisitions

No Covered Person may acquire any securities (including without limitation common stock, options to purchase or sell common stock, preferred stock, convertible and non-convertible debentures, warrants and other similar instruments, and exchange-traded options or other derivative securities related thereto) of an HC2 Entity ("*HC2 Entity Securities*") at any time, whether or not such securities are publicly traded and whether or not the Covered Person is in possession of material non-public information about the applicable HC2 Entity.

Notwithstanding the foregoing, this Policy shall not be deemed to prohibit grants of HC2 Entity Securities to a Covered Person pursuant to an incentive compensation plan, employee benefit plan, employment agreement or director compensation plan.

Sales, Dispositions and Tipping

No Covered Person may sell, transfer or dispose of HC2 Entity Securities as to which the Covered Person is in possession of material, non-public information. In addition, no Restricted Security Person may sell, transfer, or dispose of the securities of any company included on the Prohibited Trading List.

No Covered Person may "tip" or disclose material, non-public information regarding any HC2 Entity to any outside person, unless in accordance with the Company's policies regarding the protection or authorized external disclosure of such information. Furthermore, no Covered Person may give trading advice of any kind about any HC2 Entity or HC2 Entity Securities to anyone, whether or not such Covered Person is aware of material, non-public information about that company.

Prohibited Activities in Other Companies' Securities

No Covered Person may, while in possession of material, non-public information about any other public company gained in the course of his or her relationship with the Company, trade in the securities of the other public company while aware of such material, non-public information concerning that company, "tip" or improperly disclose such material, non-public information concerning that company to anyone or give trading advice of any kind to anyone concerning the other public company while aware of such material, non-public information about that company.

No Covered Person shall engage in a transaction intended to circumvent or otherwise achieve a result inconsistent with the purpose and intent of this Policy. For example, a Covered Person should not transact in an index investment in which Company Securities are a meaningful component if the basis for such transaction is material, non-public information concerning the Company Securities.

Material, Non-Public Information

Information about the Company or any other company is "material" when a reasonable investor would consider it important in making an investment decision to buy or sell such company's securities. In simple terms, material information is any type of information that could reasonably be expected to affect the price of a company's securities. While it is not possible to identify all information that would be deemed material, examples of information that would generally be considered material include the following:

- financial performance;
- potential mergers, acquisitions, dispositions or recapitalizations;
- new major contracts, orders, suppliers, customers, or finance sources, or the loss thereof;
- significant changes in volume, market share or product pricing;
- stock splits, public or private securities/debt offerings, or changes in a company's dividend policies or amounts;
- significant changes in senior management;
- actual, threatened or potential exposure to major litigation, or the resolution of such litigation;
- approvals or denials of requests for regulatory approval by government agencies of significant products, patents or trademarks;
- the contents of forthcoming publications that may affect the market price of a company's securities;
- significant changes in accounting treatment, write-offs or effective tax rate; and
- impending bankruptcy or financial liquidity problems of the company.

Material, non-public information is material information that has not been disclosed generally to the marketplace. Information is considered "non-public" until it has been widely disseminated to the public through a Securities and Exchange Commission (the "*SEC*") filing, press release, or other non-exclusionary method of disclosure reasonably designed to provide public access, and there has been sufficient time for the market to digest that information.

Any Covered Person who is unsure whether the information that he or she possesses constitutes material, non-public information should consult the Chief Executive Officer or the Legal Department for guidance before trading in any Company Securities.

Pledges and Margin Loans

Covered Persons are prohibited from pledging Company Securities or HC2 Entity Securities as collateral to secure loans or purchasing Company Securities on margin. Purchasing on margin means borrowing from a brokerage firm, bank or other entity in order to purchase Company Securities (other than in connection with a cashless exercise of stock options through a broker under the Company's equity plans).

Hedging

Covered Persons are prohibited from purchasing financial instruments (including prepaid variable forward contracts, equity swaps, collars, and exchange funds) that are designed to hedge or offset any decrease in the market value of Company Securities or HC2 Entity Securities and from engaging in any other type of transaction in Company Securities or HC2 Entity Securities that would have similar economic consequences.

Short Sales

Covered Persons are prohibited from engaging in short sales (and uncovered short derivative positions) of Company Securities and HC2 Entity Securities.

Other Transactions

Covered Persons are encouraged to consult with the Legal Department **before** entering into arrangements involving any Company Securities or HC2 Entity Securities under circumstances that are not expressly contemplated in this Policy ("*Other Transactions*"), which may also be subject to the foregoing restrictions. Other Transactions by Restricted Security Persons and Blackout Persons (as defined below) are subject to the "Transaction Pre-approvals" section of this Policy.

Blackout Periods

In addition to the foregoing restrictions, Blackout Persons are presumed to be in possession of material, non-public information in the ordinary course of their duties with the Company, and therefore are subject to more restrictive limitations on when they may buy or sell Company Securities, as follows:

Quarterly Restrictions.

Blackout Persons are prohibited from buying, selling or otherwise transferring or trading in any Company Securities beginning from twenty (20) calendar days before the end of a fiscal quarter and ending upon the completion of the first full trading day after the Company has publicly disseminated its financial results for such fiscal quarter (any such period, a "*Blackout Period*"). For example, see below for an illustration of when a Blackout Period begins and ends:



Exceptions to the Blackout Period policy may be approved only by the Chief Legal Officer (or, in the case of an exception for the Chief Legal Officer or persons or entities subject to this Policy as a result of their relationship with the Chief Legal Officer, the Chief Executive Officer or Chief Financial Officer or, in the case of exceptions for directors or persons or entities subject to this Policy as a result of their relationship with the Board of Directors).

Trades by Blackout Persons outside of Blackout Periods may take place provided that (a) the Blackout Person does not then have knowledge of material, non-public information and (b) the proposed trade would not otherwise violate federal securities laws or the terms of this Policy.

For purposes of this Policy, "Blackout Persons" include (i) all Restricted Security Persons, (ii) all directors and officers of the Company and its Subsidiaries, (iii) other employees of the Company and its Subsidiaries as may be designated by the Chief Executive Officer, Chief Financial Officer or the Legal Department and (iv) the respective Immediate Family Members of any person covered by clauses (i), (ii) or (iii) of this paragraph.

Additional Restrictions in Specific Circumstances.

From time to time, the Chief Executive Officer, the Chief Financial Officer or the Legal Department may recommend, or require, that directors, officers, designated employees and others refrain from trading because of developments known to the Company and not yet disclosed to the public. In such a case, the persons so advised shall not engage in any transaction involving the purchase, sale or other transfer of Company Securities or HC2 Entity Securities until advised that the restriction has been terminated and should not disclose to others inside or outside of the Company the fact that such a trading restriction has been imposed. Restricted Security Persons are at all times prohibited from trading in the securities of any company included on the Prohibited Trading List.

Transaction Pre-Approvals

Restricted Security Persons and Blackout Persons are required to obtain the written approval of the Chief Legal Officer (or in the case of the Chief Legal Officer, the written approval of the Chief Executive Officer or the Chief Financial Officer) before buying, selling, transferring or otherwise trading any Company Securities or selling, transferring or otherwise disposing of HC2 Entity Securities, regardless of whether they are subject to a Blackout Period at the time they intend to transact. Furthermore, Restricted Security Persons and Blackout Persons also are required to obtain the written approval of the Chief Legal Officer (or in the case of the Chief Legal Officer, the

Chief Executive Officer or Chief Financial Officer) before entering into Other Transactions and entering into any arrangement with respect to Company Securities or HC2 Entity Securities that is not otherwise expressly contemplated in this Policy. A request for approval of a transaction should be made at least three (3) business days in advance of the proposed transaction. In addition, the Restricted Security Person or Blackout Persons must execute a certification (in a form approved by the Legal Department) that he, she or it is not aware of material, nonpublic information about the company whose securities he, she or it wishes to trade. Should the Company, at some time in the future, determine to automate this preapproval process through the use of financial tracking software, all Restricted Security Persons and Blackout Persons will be required to register with the software vendor and follow instructions provided for proper compliance through such software program.

All trades that are approved by the Chief Legal Officer (or in the case of the Chief Legal Officer, the Chief Executive Officer or Chief Financial Officer) must be effected within five (5) business days of receipt of such approval unless a specific exception has been granted by the Chief Legal Officer, Chief Executive Officer or Chief Financial Officer, as applicable. A pre-approved trade (or any portion of a pre-approved trade) that has not been effected during the five (5) business day period must be approved again prior to execution. Notwithstanding receipt of approval from the Chief Legal Officer, Chief Executive Officer or Chief Financial Officer, as applicable, if the Restricted Security Person or Blackout Person becomes aware of material, non-public information or becomes subject to a Blackout Period or other trading suspension before the transaction is effected, the transaction may not be completed.

Rule 10b5-1 Trading Plans

Rule 10b5-l under the Exchange Act was adopted by the SEC to protect persons from insider trading liability for transactions under a written contract, instruction or plan previously established at a time when the insider did not possess material, non-public information and entered into in good faith and in accordance with all applicable federal and state laws. Under a properly established 10b5-1 plan (a "10b5-1 Plan"), Covered Persons may complete transactions in Company Securities at any time, including during Blackout Periods and other times when the Covered Person possesses material, non-public information. Thus a 10b5-1 Plan offers an opportunity for Covered Persons to establish a systematic program of transactions in Company Securities over periods of time that might include periods in which such transactions would otherwise be prohibited under the federal securities laws or this Policy. A variety of arrangements can be structured to meet the requirements of Rule 10b5-1. In particular, a 10b5-1 Plan can take the form of a blind trust, other trust, pre-scheduled stock option exercises and sales, pre-arranged trading instructions and other brokerage and third-party arrangements over which the Covered Person has no control once the plan takes effect. Covered Persons should note that Rule 10b5-1 only provides an affirmative defense in the event there is an insider trading lawsuit. It does not prevent someone from bringing a lawsuit.

Covered Persons who desire to implement a 10b5-1 Plan must first obtain approval of the plan by the Chief Legal Officer (or in the case of the Chief Legal Officer, the Chief Executive Officer or the Chief Financial Officer). In order to be eligible for approval, the 10b5-1 Plan (a) must be established only when a Covered Person is not in possession of material, non-public information about the Company and outside of a Blackout Period; (b) must be in writing; (c) must either irrevocably set forth the future date or dates on which purchases or sales of securities are to

be made and the prices at which the securities are to be purchased or sold, or provide a formula for determining the price of the securities to be purchased or sold and the date or dates on which the transactions are to be completed (provided that no aspect of the formula may permit the direct or indirect exercise of any influence over the timing or terms of the purchase or sale by the Covered Person); (d) must set forth the broker who will be responsible for effecting the transactions (or method of transaction if not through a broker); and (e) may not take effect until at least thirty (30) days after the plan is established. The Legal Department will maintain a copy of all 10b5-1 Plans.

The Covered Person must provide the Chief Legal Officer (or in the case of the Chief Legal Officer, the Chief Executive Officer or Chief Financial Officer) written notice of any proposed modification or termination of a 10b5-1 Plan. In the case of a proposed modification, such modification must be approved in writing by the Chief Legal Officer (or in the case of the Chief Legal Officer, the Chief Executive Officer or Chief Financial Officer), following consultation with outside counsel, and may not take effect until at least thirty (30) days after the modification is made. Any modification must be made in good faith, outside of a Blackout Period and at a time when the Covered Person does not possess material, non-public information.

Termination of a 10b5-1 Plan should occur only in unusual circumstances, and any proposed termination of a 10b5-1 Plan must be approved in writing by the Chief Legal Officer (or in the case of the Chief Legal Officer, the Chief Executive Officer or Chief Financial Officer). Once a 10b5-1 Plan has been terminated, the Covered Person must wait at least thirty (30) days before trading outside of a 10b5-1 Plan and one (1) year before establishing a new 10b5-1 Plan.

The foregoing provisions relating to the implementation, modification and termination of a 10b5-1 Plan also apply to any 10b5-1 Plan entered into or proposed to be entered into for the purpose of selling HC2 Entity Securities.

Section 16 Insider Reporting Obligations

All transactions in Company Securities, including transactions effected pursuant to a 10b5-1 Plan and transactions involving restricted stock and other equity awards granted by the Company, by directors and officers that are subject to Section 16 of the Exchange Act ("*Section 16 Insiders*") must be executed only through brokers that agree to confirm approval of the transaction pursuant to this Policy before executing the transaction. At least three (3) business days prior to executing the transaction, Section 16 Insiders shall provide written notice to, and obtain the written approval of, the Chief Legal Officer (or in the case of the Chief Legal Officer, the Chief Executive Officer or Chief Financial Officer) for each transaction in Company Securities that may be reportable on SEC Form 4 or Form 5. The requisite notice shall include all information necessary to accurately report the transaction on SEC Form 4 or 5 (with the exception of the final price and date of execution, which shall be confirmed by either the Section 16 Insider or the broker as soon as the transaction is executed).

In addition to transactions for the Section 16 Insider's own account, Section 16 of the Exchange Act requires Section 16 Insiders to report transactions in Company Securities in which the Section 16 Insider beneficially owns a direct or indirect "pecuniary interest" (i.e., where the Section 16 Insider has the opportunity, directly or indirectly, to profit or share in any profit from a

transaction in Company Securities). Section 16 Insiders may be deemed to beneficially own Company Securities held or acquired by:

- the Section 16 Insider's immediate family sharing the same household;
- trusts (including living or family trusts) in which the Section 16 Insider is a settlor or has or shares investment control;
- trusts in which the Section 16 Insider, or any member of his or her immediate family, is a beneficiary;
- general partnerships in which the Section 16 Insider is a general partner;
- limited partnerships where the Section 16 Insider is a general partner or has or shares investment control over the limited partnership's portfolio securities;
- corporations where the Section 16 Insider is a controlling shareholder or has or shares investment control over the corporation's portfolio securities; and
- limited liability companies in which the Section 16 Insider is a member, if structured like a general partnership, or where the Section 16 Insider has control or has or shares investment control over the limited liability company's portfolio securities, if structured like a limited partnership or corporation.

Under Section 16 of the Exchange Act, Section 16 Insiders face strict liability for effecting non-exempt purchases and sales (or sales and purchases) in Company Securities (including Company Securities deemed to be beneficially owned by Section 16 Insiders) within a six-month period that result in a "short swing profit" (whether this profit is actual or imputed). The statute compels the Section 16 Insider to disgorge all profits gained in the transactions to the Company.

All Section 16 Insiders shall execute a special power of attorney in form and substance satisfactory to the Legal Department empowering each of the Chief Executive Officer, the Chief Financial Officer, the Chief Operating Officer, the Chief Legal Officer or such other person as may be designated by the Chief Legal Officer, to execute and timely file with the SEC the required reports with respect to any transaction in Company Securities on behalf of the Section 16 Insider. Although the Legal Department will assist Section 16 Insiders in preparing and filing the required reports with the SEC, Section 16 Insiders retain responsibility for filing the required reports under the federal securities laws.

Any person who is unsure whether they are subject to Section 16 of the Exchange Act, whether they may be deemed to beneficially own Company Securities held or acquired by members of their immediate family or trusts, partnerships, corporations or limited liability companies over which they exercise control or in which they have an interest, whether a contemplated transaction in Company Securities is reportable on Form 4 or Form 5 or whether a contemplated transaction in Company Securities may result in liability for short-swing profits should consult the Legal Department for guidance.

Reporting of Violations

Any Covered Person who violates this Policy or any federal, state or exchange rule or law governing insider trading, including tipping, or knows of any such violation by any other Covered Person, must report the violation immediately to the Legal Department. Upon learning of any such violation, the Legal Department, in consultation with the Company's legal counsel, will determine whether the Company should release any material nonpublic information, or whether the Company should report the violation to the SEC or other appropriate governmental authority.

Compliance with the Insider Trading Policy

The Chief Legal Officer is responsible for supervising compliance with this Policy by Covered Persons. The Chief Legal Officer may designate one or more individuals who may perform the Chief Legal Officer's duties in the event that the Chief Legal Officer is unable or unavailable to perform such duties.

As a condition to present and continued employment with the Company or any Subsidiary of the Company, as applicable, each Covered Person, shall acknowledge their receipt, understanding and compliance with, the Policy on an annual basis through (i) the use of the Company's online tracking software or (ii) in the event the online tracking software is unavailable to the Covered Person, by executing an Acknowledgement, substantially in the form attached hereto as <u>Appendix A</u>. In the case of a Covered Person's Immediate Family Members, the Covered Person shall make a representation that his or her Immediate Family Members are in compliance with the Policy and will continue to comply with the Policy in the future.

Potential Civil, Criminal and Disciplinary Sanctions

A. Civil and Criminal Penalties

The consequences of prohibited insider trading or tipping can be severe. Persons violating insider trading or tipping rules may be required to disgorge the profit made or the loss avoided by the trading, pay the loss suffered by the person who purchased securities from or sold securities to the insider tippee, pay civil penalties up to three times the profit made or loss avoided, pay a criminal penalty of up to \$1 million, and serve a jail term of up to ten (10) years. The Company and/or the supervisors of the person violating the rules may also be required to pay major civil or criminal penalties.

B. Company Discipline

Violation of this policy or federal or state insider trading or tipping laws by any Director, Senior Officer, or Employee, or their family members, may subject the Director to removal proceedings and the Senior Officer or Employee to disciplinary action by the Company, up to and including termination for cause.

<u>Inquiries</u>

Please direct all inquiries regarding any of the provisions or procedures of this Policy to the Legal Department.

* * * *

Approved: November 9, 2011 Amended and Restated: August 4, 2016 Amended and Restated: March 2, 2017

APPENDIX A

HC2 Holdings, Inc.

Insider Trading Policy

ACKNOWLEDGEMENT FORM

Please complete, sign, date and return this form to your Human Resources Department with a copy to the *Legal Department*, HC2 Holdings, Inc., 450 Park Ave 30th Floor, New York, NY 10022.

I, the undersigned:

- (i) acknowledge receipt of the HC2 Holdings, Inc. ("HC2") 2017 Insider Trading Policy, effective as of March 2, 2017 (the "Policy");
- (ii) certify that, if *I* have been designated as a Blackout Person or a Restricted Security Person, I (a) will comply with the additional restrictions for Blackout Persons and Restricted Persons, and (b) acknowledge that I cannot trade in securities listed on the Prohibited Trading List maintained by the Company's Legal Department;
- (iii) certify that, to the best of my knowledge, I and my Immediate Family Members complied fully with the Policy during the prior year;
- *(iv)* agree to comply fully with the Policy in the future;
- (v) understand that, in order to ensure that the Policy remains in compliance with current laws and regulations, among other reasons, HC2 may, at its option, change, delete, suspend, or discontinue any part or parts of the Policy at any time and without prior notice; and
- (vi) understand any violation of this Policy or applicable laws or regulations may be considered grounds for termination of my employment or other disciplinary action by HC2.

Covered Person Signature

Name (Please Print)

Date: _____

VENDOR LOAN AGREEMENT

relating to

Project Furrow

between

Fugro Financial Resources B.V. (as the lender)

and

Global Marine Systems Ltd. (as the borrower)

Dated 30 November 2017

CONTENTS

1	DEFIN	ITIONS	4
2	THE LO	DAN	4
3	REPAY	MENT	5
	3.1	Ranking	5
	3.2	Repayment	5
	3.3	Prepayment	5
4	INTER	EST	5
	4.1	Interest rate	5
	4.2	Payment of interest	5
	4.3	Default interest	6
	4.4	Day count convention	6
5	LIENS	AND WAIVERS	6
6	COSTS		6
	6.1	Costs	6
	6.2	Enforcement costs	6
7	PAYMENTS AND SET-OFF		7
	7.1	Payments	7
	7.2	Partial payments	7
	7.3	Set-off and deductions	8
8	EVENT	TS OF DEFAULT	8
	8.1	Events of Default	8
	8.2	Acceleration	9
9	TERMINATION, SUSPENSION, AMENDMENTS AND WAIVERS		9
	9.1	No rescission; errors	9
	9.2	No suspension	9
	9.3	Amendments and waivers	9
10	CHAN	GES TO PARTIES	10
	10.1	Changes to Fugro	10
	10.2	Transfer	10
	10.3	Changes to GMSL	10
	10.4	No increased cost	11
11	NOTIC	ES	11
12	GOVE	RNING LAW AND ENFORCEMENT	11
	12.1	Governing law	11
	12.2	Jurisdiction	12

2 /14

Schedules

Schedule 1	Definitions
Schedule 2	Trencher Lien Agreement
Schedule 3	Parties' details for Notices

VENDOR LOAN AGREEMENT

THIS AGREEMENT IS DATED 30 NOVEMBER 2017 AND MADE BETWEEN:

(1) Fugro Financial Resources B.V., a private limited liability company incorporated in the Netherlands, with corporate seat in Leidschendam, and registered address at Veurse Achterweg 10, 2264 SG Leidschendam, the Netherlands, registered with the trade register of the Dutch Chamber of Commerce with number 27163112 ("Fugro");

and

(2) Global Marine Systems Limited, a private limited liability company incorporated and registered in England and Wales with company number 01708481 whose registered address is at Ocean House, 1 Winsford Way, Boreham Interchange, Chelmsford CM2 5PD, United Kingdom ("GMSL"),

WHEREAS:

- (A) Fugro, GMSL and Global entered into a Business Purchase Agreement on 11 October 2017 (the "BPA"), pursuant to which GMSL has acquired, subject to the conditions set out in the BPA, the 'Trenching Business' (as defined in the BPA) from Fugro (the "Transaction");
- (B) In order to facilitate the Transaction, GMSL and Fugro have agreed to enter into the Loan whereby the consideration payable to Fugro by GMSL for the Trencher pursuant to the BPA will be advanced by Fugro to GMSL pursuant to the Loan); and
- (C) The Loan shall be secured by a first ranking lien on the Trencher by means of the Trencher Lien Agreement.

THE PARTIES AGREE AS FOLLOWS:

1 DEFINITIONS

1 Certain terms used in this agreement are defined in <u>Schedule 1</u> (*Definitions*).

2 THE LOAN

Subject to the terms and conditions of this Agreement, and in accordance with clause 3 of the BPA, Fugro and GMSL agree by executing this Agreement that the consideration for the Trencher that GMSL is holding to Fugro's order shall be advanced to GMSL as a US dollar term loan in an amount of USD 7,500,000 (seven million five hundred thousand US dollars) (the "Loan").

3 REPAYMENT

3.1 Ranking

GMSL's payment obligations under this Agreement rank at least *pari passu* with all its other present and future unsecured and unsubordinated payment obligations, except for the obligations mandatorily preferred by law applying to companies generally.

3.2 Repayment

GMSL shall repay the Loan (including all interest accrued thereon) in full no later than on the Maturity Date.

3.3 Prepayment

- 3.3.1 GMSL may, if it gives Fugro no less than three (3) Business Days prior Notice, voluntarily prepay the whole or any part of the Loan.
- 3.3.2 If this Agreement or the Trencher Lien Agreement is not or ceases to be legal, valid, binding or enforceable (in any material respect) and GMSL fails to remedy such situation or enter into a replacement agreement within five (5) Business Days of being notified of such event or circumstance, GMSL shall promptly prepay the Loan.
- 3.3. Any prepayment shall be made together with accrued interest on the amount prepaid, but without premium or penalty.
- 3.3.4 GMSL may not redraw any part of the Loan which is repaid or prepaid.

4 INTEREST

4.1 Interest rate

The Loan shall bear interest at a rate of four percent (4%) per annum as from the date of this Agreement until the date three (3) months after the date of the BPA and ten percent (10%) per annum thereafter, provided that the Loan shall solely bear interest at a rate of ten percent (10%) per annum if this Agreement is entered into at or after the date three (3) months after the date of the BPA.

4.2 Payment of interest

GMSL shall pay interest on the Loan in arrears on the last Business Day of each calendar quarter and, if applicable, on the Maturity Date.

4.3 Default interest

- 4.3.1 If GMSL fails to pay any amount payable by it under this Agreement on its due date, default interest shall accrue on the overdue amount from the due date up to the date of actual payment (both before and after judgment) at the rate of an additional 2% over the interest rate set out in Clause 4.1 as applicable at the date of the default.
- 4.3.2 Any default interest accruing under this Clause 4.3 shall be immediately payable by GMSL. Any unpaid default interest will be compounded with the overdue amount on which it arises at the end of each month (but will remain immediately due and payable).

4.4 Day count convention

Any interest accruing under this Agreement shall accrue from day to day and shall be calculated on the basis of the actual number of days elapsed and a year of three hundred and sixty (360) days.

5 LIEN AND WAIVERS

- 5.1.1 At the date of this Agreement, GMSL shall grant a lien to Fugro attached to the Trencher in the form as set out in <u>Schedule 2</u> (*Trencher Lien Agreement*) (the "Trencher Lien Agreement").
- 5.1.2 GMSL represents and warrants that no waivers or consents are required for it to accept the Loan and grant the lien described in Clause 5.1.1 to Fugro.

6 COSTS

6.1 Costs

Subject to Clause 6.2 (Enforcement costs), each Party shall bear its own costs in relation to this Agreement.

6.2 Enforcement costs

GMSL shall pay and, within three (3) Business Days of demand, indemnify Fugro against all reasonable costs, expenses and losses (including legal fees) incurred by Fugro (at its discretion) in connection with:

- (a) the occurrence of an Event of Default;
- (b) investigating any event or circumstance which it reasonably believes is an Event of Default; or

(c) enforcing or preserving its rights under or in connection with this Agreement.

7 PAYMENTS AND SET-OFF

7.1 Payments

7.1.1 GMSL shall make each payment to be made by it under this Agreement on, and for value on, the due date to the following bank account:

Bank: Bank Mendes Gans

Account name: FUGRO FINANCIAL RESOURCES BV

Currency: USD

IBAN: NL39BKMG0261329685

SWIFT: BKMGNL2A

Bank address:

Bank Mendes Gans

Herengracht 619

1017 CE Amsterdam

The Netherlands

or to such other account with such bank as Fugro specifies, provided that Fugro shall have specified such other account and such bank no later than five (5) Business Days prior to the due date for payment of the relevant amount.

- 7.1.2 If a payment is due on a day that is not a Business Day, the payment shall be due on the preceding Business Day.
- 7.1.3 GMSL shall make each payment to be made by it under this Agreement in US dollars.

7.2 Partial payments

If GMSL makes a payment to Fugro that is insufficient to discharge all matured obligations then due under this Agreement by GMSL to Fugro, that payment shall be applied (irrespective of any appropriation made by GMSL) towards the obligations of GMSL in such order as Fugro may determine.

7.3 Set-off and deductions

- 7.3.1 Fugro may set off any obligation due from GMSL under this Agreement against any obligation owed by Fugro to GMSL under the W&I Agreement or the BPA, with the exception of the obligation of Fugro to pay or procure payment of the Aggregate WC Amount to GMSL (as defined in the BPA).
- 7.3.2 All payments to be made by GMSL under this Agreement shall be calculated and be made:

without, and clear of any deduction for, set-off or counterclaim, including under the BPA; and

clear of any deduction or withholding for, or on account of, any tax, levy, impost, duty or other charge of a similar nature (a "**Tax Deduction**"), other than any such deduction or withholding required by law.

7.3.3 If GMSL is required to make a Tax Deduction, the amount of the payment due from it shall be increased to an amount which, after making the Tax Deduction, leaves an amount equal to the payment which would have been due if no Tax Deduction had been required.

8 EVENTS OF DEFAULT

8.1 Events of Default

Each of the events or circumstances set out in this Clause 8.1 is an Event of Default.

8.1.1 Breach of Agreement

- (a) GMSL does not pay (or evidences an intention not to pay) any amount payable under this Agreement at the time and in the manner provided for in this Agreement and such failure continues unremedied for a period of five (5) Business Days after written notice of such failure has been given to GMSL.
- (b) GMSL does not comply (or evidences an intention not to comply) with any of its other obligations under this Agreement and such failure is not remedied within fifteen (15) Business Days of written notice from Fugro stating non-compliance to GMSL.

- 8.1.2 Any of the following events or circumstances that affects GMSL:
 - (a) It is unable or admits its inability to pay its debts as they fall due or, by reason of actual or anticipated financial difficulties, it suspends making payments on any of its debts.
 - (b) It becomes subject to a winding-up.
 - (c) It proposes or takes any corporate action, or any person commences any litigation or administrative proceedings or other formal procedure, for its windingup.

8.2 Acceleration

As long as an Event of Default is continuing, Fugro may, by notice to GMSL, declare that all or part of the Loan, together with accrued interest and all other amounts outstanding under this Agreement, be immediately due and payable, whereupon they shall become immediately due and payable.

9 TERMINATION, SUSPENSION, AMENDMENTS AND WAIVERS

9.1 No rescission; errors

- 9.1.1 No Party can rescind (ontbinden) this Agreement.
- 9.1.2 Each Party shall bear the risk of any error (*dwaling*) made by it in relation to this Agreement.

9.2 No suspension

No Party may suspend (opschorten) compliance with its obligations under or in connection with this Agreement on whatever grounds.

9.3 Amendments and waivers

- 9.3.1 Subject to Clause 10.1 (Changes to Fugro), this Agreement may be amended, supplemented or waived only by a written agreement between the Parties.
- 9.3.2 No failure to exercise, nor any delay in exercising, by Fugro, any right or remedy under this Agreement shall operate as a waiver, nor shall any single or partial exercise of any right or remedy prevent any further or other exercise or the exercise of any other right or remedy. The rights and remedies of Fugro provided in this Agreement are cumulative and not exclusive of any rights or remedies provided by law.

10 CHANGES TO PARTIES

10.1 Changes to Fugro

10.1.1 Fugro may:

(a) assign (overdragen) its rights under this Agreement (in whole or in part); or

(b) transfer its legal relationship (rechtsverhouding) under this Agreement (in whole or in part);

to:

(i) any bank or other reputable entity that provides financing as part of its ordinary course business; or

(ii) to any member of the Fugro Group (provided that if such transferee or assignee subsequently ceases to be a member of the Fugro Group, Fugro shall procure that prior to its ceasing to be so, such assignee or transferee reassigns or transfers to Fugro or another member of the Fugro Group any rights or legal relationship as have been assigned or transferred to it).

10.2 Transfer

- 10.2.1 Any transfer as referred to in Clause 10.1.1(b) shall be made by a deed of transfer governed by Dutch law.
- 10.2.2 GMSL hereby cooperates with any transfer in anticipation.
- 10.2.3 Fugro shall use reasonable efforts to provide notice in writing to GMSL of any assignment or transfer as referred to in Clause 10.1.1 at least three (3) Business Days before such assignment or transfer is carried out. Such notice shall set out the identity of the assignee or transferee and the date that the assignment or transfer will take effect, provided that the assignment or transfer shall in any event not take effect until three (3) Business Days after the date of such notice.

10.3 Changes to GMSL

GMSL may not assign, transfer or encumber any of its rights or obligations under or in connection with this Agreement.

10.4 No increased cost

10.4.1 If:

- (a) Fugro undertakes a transfer or assignment pursuant to this Clause 10; and
- (b) as a result of circumstances existing at the date of the assignment or the transfer, GMSL would be obliged to make a payment to the assignee or transferee under Clause 7.3.3 (gross-up),

then the assignee or transferee is only entitled to receive payment under that Clause 7.3.3 to the same extent as Fugro would have been if the assignment or transfer had not occurred.

11 NOTICES

- 11.1.1 Any notice, request, consent, claim, demand and other communication between the Parties in connection with this Agreement (a "Notice") shall be in writing and shall be given and shall be deemed to have been duly given if written in the English language and:
 - (a) delivered personally (Notice deemed given upon receipt);
 - (b) delivered by registered post (Notice deemed given upon confirmation of receipt); or
 - (c) sent by an internationally recognized overnight courier service such as Federal Express (Notice deemed given upon receipt),

in each case with a copy by email, which copy shall not constitute a Notice.

11.1.2 A Notice to a Party shall be sent to such Party at the address set out in <u>Schedule 3</u> (*Parties' details for Notices*) or such other person or address as such Party may notify to the other Parties from time to time.

12 GOVERNING LAW AND ENFORCEMENT

12.1 Governing law

This Agreement (including Clause 12.2 (*Jurisdiction*) but excluding the Trencher Lien Agreement) and any non-contractual obligations arising out of or in connection with it are governed by Dutch law.

12.2 Jurisdiction

- 12.2.1 The Courts of Amsterdam, the Netherlands, have exclusive jurisdiction to settle any dispute arising out of or in connection with this Agreement (including any dispute regarding the existence, validity, termination or amendment of this Agreement).
- 12.2.2 This Clause 12.2 (*Jurisdiction*) is for the benefit of Fugro only. As a result, Fugro shall not be prevented from initiating proceedings relating to a dispute as referred to in Clause 12.2.1 in any other court with jurisdiction. To the extent allowed by law, Fugro may initiate concurrent proceedings in any number of jurisdictions.

[SIGNATURES TO FOLLOW ON THE NEXT PAGE]

12 /14

THIS AGREEMENT HAS BEEN SIGNED ON THE DATE STATED AT THE BEGINNING OF THIS AGREEMENT BY:

<u>Js/ Paul Verhagen</u> **Fugro Financial Resources B.V.** By: Paul Verhagen Title: Director THIS AGREEMENT HAS BEEN SIGNED ON THE DATE STATED AT THE BEGINNING OF THIS AGREEMENT BY:

<u>/s/ Ian Douglas</u> Global Marine Systems Limited By: Ian Douglas Title: Chief Executive Officer <u>/s/ Richard Fraser-Smith</u> Global Marine Systems Limited By: Richard Fraser-Smith Title: Chief Financial Officer

[Signature Page to Project Furrow Vendor Loan Agreement]

PROJECT FURROW EXECUTION VERSION

Schedule 1 Definitions

1 Definitions

Agreement means this Vendor Loan Agreement;

BPA has the meaning set out in Recital (A);

Business Day means a day which is not a Saturday, a Sunday, or an official holiday in the Netherlands or the United Kingdom;

Event of Default means any event or circumstance specified as such in Clause 8.1 (Events of Default);

Fugro has the meaning set out in the introduction of this Agreement;

Fugro Group means Fugro and its direct and indirect Subsidiaries;

Global means Global Marine Holdings LLC;

GMSL has the meaning set out in the introduction of this Agreement;

Loan means the loan described in Clause 2 (*The loan*) or the principal amount outstanding for the time being of that loan (including any interest compounded pursuant to Clause 4.2;

Maturity Date means the date three hundred and sixty-three (363) days after the date of this Agreement (consequently, the term of this vendor loan is less than one (1) year);

Notice has the meaning set out in Clause 11.1.1;

Parties means Fugro and GMSL and Party means any one of them or the relevant one of them, as the context requires;

Subsidiary means a subsidiary (dochtermaatschappij) as defined in section 2:24a BW;

Tax Credit means a credit against, relief or remission for, or repayment of, any tax.

Tax Deduction has the meaning set out in Clause 7.3.2;

Transaction has the meaning set out in Recital (A);

Trencher means the Q1400-2 Trencher system (including launch and recovery system) with asset ID 1510996;

Trencher Lien Agreement has the meaning set out in Clause 5.1.1; and

W&I Agreement has the meaning ascribed thereto in the BPA.

2 Construction

- 2.1 In this agreement, unless a contrary indication appears:
 - (a) any reference to this "agreement" or any other document also refers to any amendment or supplement to it and any restatement or novation of it;
 - (b) any reference to a "Party" also refers to its successors in title, permitted assigns and permitted transferees;
 - (c) a "**person**" includes any natural person, legal entity or other person (whether under public or private law, whether or not having legal personality and wherever existing or registered);
 - (d) "the Netherlands" refers to the part of the Kingdom of the Netherlands located in Europe (and all derivate terms, including "Dutch", are to be construed accordingly).
- 2.2 In this agreement, unless a contrary indication appears:
 - (a) "as of" includes the day or moment referred to by it;
 - (b) "including" means including without limitation (and all derivate terms are to be construed accordingly);
 - (c) a "winding-up" includes, in relation to a person:
 - (i) any bankruptcy, suspension of payments, moratorium of indebtedness or other insolvency proceedings in relation to that person;
 - (ii) the appointment of a liquidator, trustee, receiver, administrator, administrative receiver, compulsory manager or other similar officer in respect of that person or any of its assets;
 - (iii) any composition, compromise, assignment or arrangement with that person's creditors generally or any class of them;
 - (iv) that person's death, dissolution or liquidation;
 - (v) any amalgamation, merger, demerger, transformation or corporate reconstruction of that person;
 - and any analogous procedure or step taken, or event occurring, in any jurisdiction, provided that "winding-up" shall not include:
 - (i) any solvent reorganisation;
 - (ii) any such procedure, step, event or circumstance which is frivolous or vexatious and it is being contested with due diligence, in good faith and

on reasonable grounds; and

(d) "financial indebtedness" includes:

- (i) moneys borrowed and debit balances at banks or other financial institutions;
- (ii) any note purchase facility or the issue of bonds, notes, debentures, loan stock or any similar instrument;
- (iii) the amount of any liability in respect of financial leases; and
- *(iv)* any amount raised under any other transaction having the commercial effect of a borrowing or otherwise classified as borrowings under the applicable accounting principles.

- **2.3** The Schedules are part of this agreement.
- 2.4 Headings do not affect the interpretation of this agreement.
- 2.5 Where the context permits, any reference to a gender includes all genders.

PROJECT FURROW EXECUTION VERSION

Schedule 2 Trencher Lien Agreement

FINAL DRAFT

5.1



Project Furrow | Information Package | February 2017

FINAL DRAFT

5.1 Q1400 trenching system (cont'd)

Trencher has a unique modular design comprising 3 main modules

- The Q1400 trencher is set up as a modular system consisting of three separate sub-parts (as shown below)
- The control module is installed on top of either the chain cutting skid or the jet trenching skid
- jet trencning sku Onboard switching between cutting and jetting facilitates work in changing soil conditions and different project environments I The trenching system can be mobilised an different wesels The intellectual property relating to the trenchers remains with SMD



- Continuous innovation To enhance cutting performance, Furrow is currently developing, together with Fugro and SMD teams, improvements to the Q1400 systems
- systems These improvements are being tried out an Rampion phase II The hybrid pitting cutter will allow for efficient transportation of sand while cutting, which removes the need to change from cutting to jetting tools
- Ideal for areas of mixed sands and clays The improved systems are expected to be ready for use in 2017
- Closed depressor design means the cable is placed in the trench and does not rely upon self weight to sink the cable
 The intellectual property remains with SMD
- Details of hybrid tool
- ✓ Jetting array of 28 Nozzles
 ✓ 5 out 20 nozzles compromised at any one time
 ✓ 375Kw power at 5bar flow of 1698m³/hr





20

Project Furrow | Information Package | February 2017

Charge over EQUIPMENT

<u>30 November</u> 2017 GLOBAL MARINE SYSTEMS LIMITED as Grantor

and

FUGRO FINANCIAL RESOURCES B.V. as Beneficiary

CONTENTS

Page

13

Clause

1	Definitions and Interpretation	3
2	Grant of security and release	4
3	Undertakings and restrictions	4
4	Notices and other formalities	5
5	Insurance	5
6	Default and Remedies	6
7	Powers of Receiver	7
8	Power of Attorney	7
9	Delegation and Consolidation	8
10	protection of security	8
11	Continuing security	8
12	Payments	8
13	Miscellaneous	9
14	Successors in Title, Assignment and Transfer	9
15	Notices	9
16	Counterparts	9
17	Governing Law and Jurisdiction	9
Schedu	le	

1.	Description of Charged Property	11
----	---------------------------------	----

Execution Page

THIS DEED (this Deed) is made on <u>30 November</u> 2017

BETWEEN:

- (1) **GLOBAL MARINE SYSTEMS LIMITED**, a company duly incorporated under the laws of England and Wales with registration number 01708481 and with its registered office at Ocean House, 1 Winsford Way, Boreham Interchange, Chelmsford, CM2 5PD (the **Grantor**); and
- (2) **FUGRO FINANCIAL RESOURCES B.V.,** a public limited liability company incorporated in the Netherlands, with corporate seat in Leidschendam, and registered address at Veurse Achterweg 10, 2264 SG Leidschendam, the Netherlands, registered with the trade register of the Dutch Chamber of Commerce with number 27120091 (the **Beneficiary**).

WHEREAS:

- (A) For the purpose of financing the acquisition of the Charged Property by the Grantor, the Beneficiary has agreed to make available to the Grantor a loan pursuant to the Loan Agreement.
- (B) It is a condition precedent under the Loan Agreement that the Grantor enter into this Deed as security for its obligations to the Beneficiary under the Loan Agreement.

THIS DEED WITNESSES as follows:

1. DEFINITIONS AND INTERPRETATION

(a) In this Deed (including in the recitals), except where the context otherwise requires, the words and expressions defined in the Loan Agreement shall bear the same respective meanings when used herein unless otherwise defined herein and the following words and expressions have the following meanings:

Charged Property means:

(a) the property identified in Schedule 1 (the "Trencher"); and

(b) all proceeds received, in whatever form they may be, under the Insurances and all other amounts payable to the Grantor under or in respect of the Insurances, including (without limitation) damages for breach and return of premium;

"Finance Documents" means the Loan Agreement, this Deed and any other document (whether creating any Security Interest or otherwise) which is executed at any time under or in connection with the Loan Agreement;

"**Insurances**" means any and all of the insurances taken out in respect of the Trencher from time to time, whether under a separate policy or otherwise, other than those in respect of liability cover;

Loan Agreement means the loan agreement dated <u>30 November</u> 2017 between the Grantor as borrower and the Beneficiary as lender;

Notice of Charge means a notice in the form set out in Schedule 2;

Receiver means any receiver or receiver and manager appointed by the Beneficiary hereunder or under any statutory power;

Relevant Currency means in relation to any payment obligation of the Grantor, the currency in which that obligation is denominated;

Secured Obligations means any and all moneys, liabilities and obligations from time to time expressed to be owing by the Grantor to the Beneficiary under or in connection with the Finance Documents; and

Security Interest means any charge, pledge, mortgage, lien, hypothecation and any other arrangement or other security interest of any kind in any jurisdiction having a similar effect.

- (b) Clause headings are for ease of reference only.
- (c) References in this Deed to Clauses or Schedules are, unless otherwise specified, references to clauses of, and schedules to, this Deed.
- (d) References in this Deed to any statute or other legislative provision shall be read to include any statutory or legislative modification or re-enactment thereof, or any substitution therefor.
- (e) References in this Deed to this Deed or any other agreement, instrument or document shall include a reference to this Deed or such agreement, instrument or document as it may from time to time be amended, restated, varied, substituted or supplemented from time to time.
- (f) Words importing the plural shall include the singular and vice versa.
- (g) Reference to the **Beneficiary** or the **Grantor** includes their respective successors and permitted assigns and transferees in accordance with their respective interests.
- (h) Any reference to this Deed in this Deed shall also include a reference to any other agreement entered into by the Grantor confirming or supplementing this Deed and the charge hereby granted or agreed to be granted.

2. GRANT OF SECURITY AND RELEASE

- (a) As a continuing security for the payment, performance and discharge of the Secured Obligations, the Grantor hereby charges to the Beneficiary all its right, title and interest, present and future, to and in the Charged Property by way of first priority fixed charge.
- (b) Upon payment, performance and discharge in full of all the Secured Obligations and the Beneficiary has no obligation to provide any further financial accommodation to the Grantor under the Finance Documents, the Beneficiary shall forthwith, at the request and cost of the Grantor, do such acts and execute such documents as may be necessary to release and discharge this Deed without recourse to, or any representation or warranty by the Beneficiary.
- (c) For the avoidance of doubt, this Deed does not constitute a charge or assignment of any policy or contract of insurances or reinsurances but includes all the benefit, rights, interest and proceeds thereunder insofar as the same relate to the Trencher.

4

3. UNDERTAKINGS AND RESTRICTIONS

- (a) The Grantor undertakes that until such time as the Secured Obligations have been discharged in full, it shall not:
 - (i) create or grant or agree to the creation or grant over any of its right, title and interest in and to the Charged Property by any person of any Security Interest;
 - (ii) sell, transfer or otherwise dispose of any of its right, title and interest in and to the Charged Property, or agree to any of the foregoing;

- (iii) allow, perform or consent to any act or any omission to act which would or might cause the Charged Property to be forfeited under any applicable law or which might prejudice the Security Interests of the Beneficiary in the Charged Property created by this Deed; and
- (iv) affix or allow the Trencher to be affixed to any vessel not owned or bareboat chartered by the Grantor in such a manner that the Trencher ceases to be separate and free of the right of any owner of such vessel;

without the prior written consent of the Beneficiary, which consent shall not be unreasonably withheld or delayed in the case of paragraph (iv) above.

(b) The Grantor undertakes to maintain (and to procure that any lessee of it as may be permitted by the Beneficiary, shall maintain) the Trencher in good repair and working condition until such time as the Secured Obligations have been discharged in full, provided that the Grantor shall not be in breach of this provision if the Trencher suffers any damage as long as the Grantor restores the condition of the Trencher in a reasonable period of time, taking account of timing for receipt for insurance proceeds and any other relevant factors.

4. NOTICES AND OTHER FORMALITIES

- (a) The Grantor shall deliver to (i) the Beneficiary upon the execution of this Deed, (ii) the Beneficiary promptly upon its request from time to time, and (iii) to each insurer of the Trencher from time to time, a Notice of Charge duly executed by the Grantor in the form attached as Schedule 2 to this Deed in respect of each policy of Insurance, and the Grantor shall provide to the Beneficiary evidence satisfactory to the Beneficiary of service of such Notice of Charge by the Grantor on each insurer arranging such Insurance.
- (b) The Grantor shall procure that until the release of the Security Interests created by this Deed in accordance with the terms of this Deed, a notice shall at all times remain fixed to the Trencher stating:

"Notice of Charge

This Q1400 high powered modular trencher system (including launch and recovery system) with asset identification number 1510996 has been charged by way of first priority fixed charge to Fugro Financial Resources B.V. The terms of that charge prohibits any person selling, disposing, or creating any security over this asset."

(c) The Grantor undertakes that it will arrange to register the prescribed particulars of the Security Interests contained in this Deed with the Registrar of Companies for England and Wales in accordance with the Companies Act 2006 (and, in any event, as soon as reasonably practicable after the date of this Deed), and the Grantor shall submit to the Beneficiary the original Certificate of Registration evidencing such registration.

5. INSURANCE

(a) The Grantor shall effect and maintain the Insurances in such amounts and on such terms and conditions as are customary for trencher systems of similar kind to the Trencher and/or as is considered good practice in this business.

- (b) The Grantor shall punctually pay when due all premiums payable in respect of the Insurances and deliver to the Beneficiary evidence satisfactory to the Beneficiary of payment of such premiums upon request by the Beneficiary.
- (c) After the occurrence of an Event of Default under the Loan Agreement, all monies received under all policies of Insurance shall be applied in discharge of the Secured Obligations. The Grantor waives any right it may have to require that any such monies are applied in reinstatement of any part of the Trencher.
- (d) At any time when the Trencher Insurances are the subject of a separate policy (rather than included in a policy together with the Grantor's other assets), the Grantor shall deposit all policies of Insurances with the Beneficiary upon request by the Beneficiary to do so.
- (e) If at any time the Trencher is not insured in accordance with the provisions of this Clause, or if the Grantor fails to produce evidence of that Insurance, the Beneficiary may at the expense of the Grantor insure the Trencher and the Grantor shall indemnify the Beneficiary on demand against any money expended by the Beneficiary for that purpose.

6. DEFAULT AND REMEDIES

- (a) The security constituted by this Deed shall become immediately enforceable if an Event of Default occurs and is continuing pursuant to the terms of the Loan Agreement. After the security constituted by this Deed has become enforceable, the Beneficiary may (without the restrictions contained in Section 93 and 103 of the Law of Property Act 1925):
 - (i) take possession of the Charged Property or any part thereof and for that purpose to enter upon any premises or any vessel owned or operated by the Grantor where the Charged Property or such part thereof may be located;
 - (ii) sell, call in, collect and convert into money the Charged Property or any part thereof by public or private contract;
 - (iii) settle, arrange, compromise or submit to arbitration any accounts, claims, questions or disputes whatsoever which may arise in connection with the Charged Property;
 - (iv) bring, take, defend, compromise, submit to arbitration or discontinue any actions, suits or proceedings whatsoever, civil or criminal, in relation to the Charged Property;
 - (v) manage or lease the Charged Property for such terms, at such rents and generally in such manner and upon such conditions and stipulations as the Beneficiary shall reasonably think fit;
 - (vi) execute and do all such acts, deeds and things which are reasonably necessary for or in relation to any of the purposes aforesaid; and
 - (vii) appoint a Receiver of all or any part of the Charged Property and from time to time remove any Receiver so appointed and appoint another in his stead. A Receiver will be deemed to be the agent of the Beneficiary for all purposes and accordingly will be deemed to be in the same position as a Receiver duly appointed by a beneficiary under the Law of Property Act 1925.
- (b) The proceeds of any enforcement of this Deed pursuant to this Clause 6 shall be applied in or towards payment of the Secured Obligations in accordance with the Loan Agreement.

- (c) Sections 109(6) and 109(8) of the Law of Property Act 1925 shall not apply in relation to any Receiver appointed under this Clause 6.
- (d) No Beneficiary or Receiver shall be liable for any loss, costs or expense incurred by the Grantor as a result of taking any action permitted by this Deed, any neglect or default in connection with the Charged Property or the taking possession of or realising any part of the Charged Property except in the case of gross negligence or wilful default.
- (e) The Beneficiary may enforce its rights under this Deed without taking any action or making any claim against any other person.

7. **POWERS OF RECEIVER**

- (a) Any Receiver appointed under Clause 6 shall have all the powers conferred on a receiver by law (save that Section 103 of the Law of Property Act 1925 shall not apply) and by way of addition and variation to but without limiting those powers:
 - (i) the Receiver shall have all the powers given to the Beneficiary hereunder, shall be entitled to the same protection and to exercise the same powers and discretion as are given to the Beneficiary hereunder and shall also have such other of the powers and discretion given to the Beneficiary hereunder as the Beneficiary may from time to time confer on him;
 - (ii) the remuneration of the Receiver may be fixed by the Beneficiary (acting reasonably), but shall be payable by the Grantor and shall, in any event, be secured on the Charged Property under this Deed; and
 - (iii) any expenses incurred by the Receiver in the exercise of any of its powers hereunder may be paid or retained by the Receiver out of any moneys coming into its hands as receiver or may be paid by the Beneficiary in which case they shall be treated as expenses properly incurred by the Beneficiary.
- (b) Other than as required by applicable law, neither the Beneficiary nor the Receiver shall be liable to account as a Beneficiary in possession of the Charged Property or on any other basis and neither the Beneficiary nor any Receiver shall be liable for any loss arising from or in connection with the realisation of the Charged Property or otherwise for any act, neglect, default or omission for which a Beneficiary in possession might be liable as such, unless, in each case, caused by the Beneficiary or the applicable Receiver's gross negligence or wilful misconduct.
- (c) The foregoing powers of appointment of a Receiver shall be in addition and variation to any statutory or other powers of the Beneficiary under the Law of Property Act 1925 and the Receiver shall in any event have and be entitled to exercise all the rights, powers and remedies conferred upon the Beneficiary by this Deed and by law with respect to the Charged Property.

7

8. POWER OF ATTORNEY

The Grantor hereby and by way of security appoints the Beneficiary and every Receiver as attorney of the Grantor and in its name and on its behalf and as its act and deed to execute, seal and deliver and otherwise perfect and do any act or thing which the Grantor has failed to execute and do under the provisions contained herein or which may be required or deemed proper in the exercise of any rights or powers hereunder or otherwise for any of the purposes of the security created hereby, provided that the power of attorney granted hereunder shall not be exercised by the Beneficiary or any Receiver prior to the occurrence of an Event of Default which is continuing.

9. DELEGATION AND CONSOLIDATION

The Beneficiary shall be entitled at any time and as often as it may deem expedient to delegate all or any of the powers and discretions vested in it by or in connection with this Deed in such manner, upon such terms and conditions (including power to sub-delegate) and to such persons as the Beneficiary in its absolute discretion may think fit. Section 93 of the Law of Property Act 1925 (and any other statutory provisions of equivalent effect) shall not apply to this Deed.

10. PROTECTION OF SECURITY

- (a) This Deed is in addition to any other rights or security, now and in the future, held by the Beneficiary from the Grantor or any other person for the Secured Obligations. This Deed shall not merge with or prejudice or be prejudiced by any such rights or security or any other contractual or legal rights of the Beneficiary. No action or inaction by the Beneficiary in relation to any other rights or Security Interest shall affect its rights under this Deed.
- (b) The Grantor shall at its own cost at the reasonable request of the Beneficiary or any Receiver execute any deed or document and take any action necessary or desirable to:
 - i. perfect or protect the Security Interests created by this Deed or their priority;
 - ii. facilitate the realisation of the Charged Property or the exercise of any rights or powers of the Beneficiary or any Receiver; or
 - iii. confer on the Beneficiary Security Interests over any Charged Property located outside England and Wales similar in effect to the Security Interests intended to be created by this Deed,

and the covenants contained in s.2(1)(b) of the Law of Property (Miscellaneous Provisions) Act 1994 shall extend to this obligation.

(c) The Beneficiary may open a new account upon receiving notice of any charge or interest affecting the Charged Property. No payment received by the Beneficiary after receiving such notice shall be appropriated towards or have the effect of discharging the Secured Obligations outstanding at the time of receiving such notice whether or not the Beneficiary opens any such accounts. The Beneficiary may redeem any prior Security Interest or procure the transfer of any prior Security Interest to itself. The Grantor shall reimburse the Beneficiary on demand for all costs and expenses incurred by the Beneficiary in connection with that redemption or transfer.

11. CONTINUING SECURITY

The security constituted by this Deed shall remain in full force and effect as a continuing security for the Secured Obligations, despite any settlement of account, or intermediate payment or otherwise, unless and until the Beneficiary discharges this Deed in writing.

12. PAYMENTS

The Grantor shall be liable to the Beneficiary for all reasonable costs, expenses and losses (including legal fees) sustained or incurred as a result of a failure by the Grantor to perform any of its obligations under this Deed.

13. MISCELLANEOUS

- (a) No failure to exercise, nor any delay in exercising, on the part of any party hereto, any right or remedy hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any right or remedy prevent any further or other exercise thereof or the exercise of any other right or remedy. The rights and remedies herein provided are cumulative and not exclusive of any rights or remedies provided by applicable law.
- (b) If at any time one or more of the provisions of this Deed becomes invalid, illegal or unenforceable in any respect under any law, the validity, legality and enforceability of the remaining provisions hereof shall not in any way be affected or impaired thereby.
- (c) The Grantor shall reimburse the Beneficiary on demand for all amounts (including legal fees and all value added tax and any other taxes) which the Beneficiary may from time to time pay in connection the enforcement and preservation of the rights for the Beneficiary under it.
- (d) A person who is not a party to this Deed shall have no right under the Contract (Rights of Third Parties) Act 1999 to enforce or to enjoy the benefit of any term of this Deed.

14. SUCCESSORS IN TITLE, ASSIGNMENT AND TRANSFER

- (a) This Deed shall be binding upon and enure to the benefit of each of the parties hereto and their respective and any subsequent successors, permitted assigns and permitted transferees.
- (b) The Grantor may not assign any of its rights or transfer or purport to transfer any of its obligations hereunder without the prior written consent of the Beneficiary.
- (c) The Beneficiary shall be entitled to assign its rights and/or transfer its rights and obligations hereunder to any person to whom it assigns its rights and/or transfers its rights and obligations under the Loan Agreement, but not otherwise.

15. NOTICES

All notices, requests, demands and other communications to or upon the respective parties hereto shall be given or made in English and in the manner and to the addresses and numbers and shall be effective as determined pursuant to the provisions of clause 11 (*Notices*) of the Loan Agreement.

16. COUNTERPARTS

This Deed may be executed in counterparts and by the parties hereto on separate counterparts, each of which when executed and delivered shall constitute an original but both counterparts shall together constitute both one and the same original.

17. GOVERNING LAW AND JURISDICTION

- (a) This Deed and any non-contractual obligations connected with it shall be governed by and construed in accordance with English Law.
- (b) The Grantor agrees that the English courts are to have exclusive jurisdiction to settle any dispute (a) arising from or in connection with this Deed or (b) relating to any non-contractual obligations arising from or in connection with this Deed and to entertain any suit, action or proceedings arising (the "**proceedings**").

- (c) The Beneficiary may take proceedings against the Grantor in any court of competent jurisdiction, and the taking of proceedings by the Beneficiary in one or more jurisdictions shall not preclude the taking of proceedings in any other jurisdiction, whether concurrently or not.
- (d) The Grantor waives any objection which it may now or in the future have to any court referred to in this Clause as a venue for any proceedings and shall not claim that any proceedings in such courts have been instituted in an inconvenient or inappropriate forum.

IN WITNESS whereof this Deed has been executed as a deed and is intended to be and is hereby delivered as a deed on the day and year first above written.

SCHEDULE 1

DESCRIPTION OF CHARGED PROPERTY

1. The Q1400 high powered modular trencher system (including launch and recovery system) as more particularly described below, consisting of 3 separate modules with asset identification number 1510996.

SCHEDULE 2

FORM OF NOTICE OF CHARGE

To: [Insurer]

We hereby notify you that we have granted a first fixed charge in favour of **Fugro Financial Resources B.V.**, a company incorporated in the Netherlands with trade register number 27163112 (the "**Lender**") pursuant to a first priority fixed charge created by us in favour of the Lender dated [] 2017 all our right, title and interest in and to the proceeds of [*insert details of insurance policy*] (the "**Policy**") [as it relates to a Q1400 high powered modular trencher system (including launch and recovery system) consisting of 3 separate modules with asset identification number 1510996, (the "**Trencher**")]*.

We hereby instruct you with effect from the date you receive this notice to:

- (a) Ensure that any time after you receive notice from the Lender that an Event of Default has occurred under and as defined in a loan agreement dated [] 2017 made between the Lender as lender and ourselves as borrower, make all payments under or in respect of the Policy [relating to the Trencher]* to the Lender or as the Lender may specify in writing from time to time:
- (b) note the interest of the Lender on the Policy [in relation to the Trencher]*;
- (c) disclose to the Lender without further approval from us such information regarding the Policy [in relation to the Trencher]* as the Lender may from time to time request; and
- (d) send a copy of all notices issued by you in respect of the Policy [in relation to the Trencher]* to the Lender.

All rights, interests and benefits accruing to us under the Policy [in relation to the Trencher]* belong to and are exercisable by the Lender following any notice of an Event of Default from the Lender referred to in paragraph (a) above

Yours faithfully

For and on behalf of Global Marine Systems Ltd.

Date:

2017

* this wording to be inserted when the Trencher is insured on a policy with other assets.

EXECUTION PAGE

13

The Grantor

EXECUTED and **DELIVERED** as a **DEED** by) **GLOBAL MARINE SYSTEMS LIMITED**) acting by:)

in the presence of:

Signature: Name: Title: Address:

The Beneficiary

EXECUTED and DELIVERED as a DEED by) FUGRO FINANCIAL RESOURCES B.V.) acting by:)

in the presence of:

Signature: Name: Title:

Address:

Fugro's details:

Fugro Financial Resources B.V.		
Address:	Veurse Achterweg 10 2264 SG Leidschendam The Netherlands	
Attention:	Paul Verhagen CFO Annabelle Vos General Counsel	
E-mail:	P.Verhagen@fugro.com a.vos@fugro.com	

GMSL's details:

Global Marine Systems Ltd.			
Address:	Ocean House 1 Winsford Way Boreham Interchange Chelmsford CM2 5PD United Kingdom		
Attention:	Ian Douglas CEO Richard Fraser-Smith CFO		
E-mail:	lan.Douglas@globalmarine.group Richard.Fraser-Smith@globalmarine.group		

TRANSITIONAL SERVICES AND FRAMEWORK SERVICES AGREEMENT

relating to	
	Project Furrow
Between	
	Fugro N.V.
And	
	Global Marine Systems Limited.

Dated 30 November 2017

Contents

Clau	se		Page
1	DEFINITIONS		6
2	TRAN	SITIONAL SERVICES	6
	2.1	Master Transitional Services Agreement	6
	2.2	Services	6
	2.3	Principles	7
	2.4	Standard of Transitional Services	8
	2.5	Relationship with Third Party Suppliers	8
	2.6	Third Party Consents	9
3	FRAM	EWORK SERVICES	10
	3.1	Master Framework Services Agreement	10
	3.2	Obligation to request an offer from Fugro	11
	3.3	Preferred services provider arrangement	11
	3.4	Framework Service Agreements	13
	3.5	Standard of Framework Services	13
4	VESSI	EL SERVICES	13
5	HUAW	EI JOINT VENTURE AND SBSS JOINT VENTURE	14
6	TERM		14
	6.1	Commencement	14
	6.2	Expiry	14
	6.3	Transitional Service Term extension	15
	6.4	Framework Term extension	15
7	TERM	S OF SERVICE	15
	7.1	Charges for the Services	15
	7.2	No withholding and gross-up	16
	7.3	VAT	16
	7.4	Invoicing and payment terms	16
	7.5	Validity of any Service Agreement	17
8	TERM	INATION	17
	8.1	Termination for cause	17
	8.2	Termination for breach	17
	8.3	Effect of termination	18
	8.4	Termination of Services for breach	18
	8.5	Termination of Transitional Services for Convenience	19
	8.6	Force Majeure	19
	8.7	Notice of termination	20
9	REMEDIES 2		20

2 of 30

	9.1	Affiliates	20
	9.2	Liquidated damages	20
10	WARF	RANTIES AND LIABILITY	21
	10.1	Warranties	21
	10.2	Liability regimes	21
11	RELA	TIONSHIP MANAGEMENT	21
	11.1	Relationship Manager	21
	11.2	Failed Bids	21
	11.3	Bid Dispute	22
12	INTEL	LECTUAL PROPERTY AND USE OF MATERIALS	22
	12.1	Intellectual Property	22
	12.2	Provider Materials	22
	12.3	IP Claim	23
13	IT SEC	CURITY AND DATA	23
	13.1	Unauthorised access	23
	13.2	No access to networks and data	24
	13.3	Security breach	24
14	SEPAF	RATION FROM SELLER'S NETWORKS	24
	14.1	Obligation to complete Network Separation	24
	14.2	Connection to Fugro's Networks	24
	14.3	Temporary disconnection	25
15	CONF	IDENTIALITY	25
16	MISCI	ELLANEOUS	25
	16.1	No assignment or subcontracting	25
	16.2	Waiver	26
	16.3	Amendment	26
	16.4	Third Party rights	26
	16.5	Employees	26
	16.7	Costs	27
	16.8	Rescission	27
	16.9	Notices	27
17	GOVE	RNING LAW AND DISPUTE RESOLUTION	27
	17.1	Governing law	27
	17.2	Jurisdiction and forum	27
	17.3	Other disputes	28

Schedules

- Schedule 1 Definitions and interpretation
- Schedule 2 Framework Services
- Schedule 3 Agreed Transitional Service Agreements
- Schedule 4 Agreed Framework Service Agreements
- Schedule 5 Parties' details for Notices

4 of 30

TRANSITIONAL AND FRAMEWORK SERVICES AGREEMENT

THIS AGREEMENT IS DATED 30 NOVEMBER 2017 AND MADE BETWEEN:

(1) **Fugro N.V.**, a public limited liability company incorporated in the Netherlands, with seat in Leidschendam, and registered address at Veurse Achterweg 10, 2264 SG Leidschendam, the Netherlands, registered with the trade register of the Dutch Chamber of Commerce with number 27120091 ("**Fugro**");

and

(2) Global Marine Systems Limited, a private limited liability company incorporated and registered in England and Wales with company number 01708481 whose registered address is at Ocean House, 1 Winsford Way, Boreham Interchange, Chelmsford CM2 5PD, United Kingdom ("GMSL"),

BACKGROUND:

- (A) The Parties and Global Marine Holdings LLC ("Global") entered into a business purchase agreement on 11 October 2017 ("BPA"), pursuant to which GMSL acquired Fugro's trenching business (the "Trenching Business") and Fugro obtained an (indirect) equity interest in Global (the "Transaction").
- (B) As part of the Transaction, GMSL agreed to enter into a service 'framework agreement' with Fugro under which Fugro would be the 'preferred provider' (at market pricing) for a range of survey, marine site investigation, asset integrity, ROV (operation) and other services needed by GMSL ("Framework Services"). GMSL agreed to commit to take on a pre-agreed set of these services upon completion of the Transaction and for the duration set out in this Agreement.
- (C) GMSL is also prepared to work with both its joint-venture partner in Huawei Marine Networks Co., Ltd. (a turnkey fiber-optics installation contractor operating on a world-wide basis owned forty-nine percent (49%) by GMSL) ("Huawei JV") and its joint-venture partner in SB Submarine Systems Co. Limited (China) ("SBSS JV") to explore the possibility of establishing a similar framework agreement.
- (D) GMSL and Fugro agree to cooperate on future work prospects in the "non-trenching", "non-cable laying" market in the event Fugro requires additional vessel capacity.
- (E) Finally, Parties agreed that GMSL needs certain transitional services in connection with the Transaction and Fugro is willing to provide these Transitional Services at cost plus 10% unless otherwise agreed within the relevant Transitional Service Agreement.

- (F) It is acknowledged that the Parties are each in good faith committed to maximising the opportunities to trade throughout the Term, which includes the intention to enter into Framework Services Agreements where that is commercially beneficial for both Parties.
- (G) If GMSL or any of its Affiliates request a Service from Fugro, and Fugro is able to provide such Services pursuant to the terms set forth in this Agreement, Fugro and the relevant party will enter into a separate Service Agreement in accordance with this Agreement.

THE PARTIES AGREE AS FOLLOWS:

1 DEFINITIONS

1 Certain terms used in this Agreement are defined in <u>Schedule 1</u> (*Definitions*).

2 TRANSITIONAL SERVICES

2.1 Master Transitional Services Agreement

This Agreement is a master agreement. Each Transitional Service Agreement shall be construed as a separate and independent agreement for each Transitional Service provided thereunder, which incorporates the terms of this Agreement, except as such terms may be expressly varied by the terms of a specific Transitional Service Agreement whereby the terms of the Transitional Service Agreement shall prevail.

2.2 Services

- 2.2.1 This Agreement governs the overall relationship of the Parties in relation to the Transitional Services.
- 2.2.2 <u>Schedule 1</u> (Agreed Transitional Services) includes the agreed form Transitional Service Agreements in respect of the Transitional Services that the Fugro Group will provide to the GMSL Group and Fugro and GMSL shall procure that these Transitional Service Agreements will be executed by the Provider and the relevant Recipient at or as soon as possible after the Effective Date.
- 2.2.3 This Agreement governs the overall relationship of the Parties in relation to the Transitional Services and until the last date of the Transition Period:

6 of 30

- (a) each relevant Affiliate identified as Provider or Recipient shall, and each of the Parties shall, provide or receive or procure each of those Affiliates to provide or receive, as applicable, the relevant Transitional Services in accordance with the terms and conditions as set forth in this Agreement, including the principles set forth in Clause 2.3 (*Principles*) and the relevant Transitional Service Agreement;
- (b) each Party shall, and shall procure that its relevant Affiliate identified as Provider or Recipient will, comply with the terms of this Agreement and the relevant Transitional Service Agreement; and
- (c) each Party shall be entitled to enjoy the benefit conferred on each of its relevant Affiliates identified as Provider under this Agreement and the relevant Transitional Service Agreement for and on behalf of such Provider.

2.3 Principles

- 2.3.1 The Transitional Services, with the exception of the service consisting of the sublease of the Aberdeen office facilities and parking or as specifically contemplated in any Transitional Service Agreement, will be carried out in accordance with the following principles:
 - (a) the term during which any Transitional Service will need to be rendered will be kept as short as reasonably possible and will not extend beyond the relevant Transitional Service Term, unless agreed in writing between the Parties; and
 - (b) GMSL shall procure that each Recipient will take all reasonable measures to be able to operate without the relevant Transitional Service provided to it as soon as reasonably possible, including making reasonable investments in respect thereof and entering into agreements on market terms with other service providers, when reasonably practicable.
- 2.3.2 The Parties shall apply the principles set out in Clause 2.3.1 in good faith and with a view to keep the Transition Period as short as possible (save as otherwise specifically contemplated in a Transitional Service Agreement). Such principles do not override any of the other provisions of this Agreement but shall serve as guiding principles to solve any unregulated matters and disagreements.

2.4 Standard of Transitional Services

- 2.4.1 Each Provider shall, and Fugro shall ensure that its Affiliates that act as Provider shall, ensure that the Transitional Services provided pursuant to this Agreement are provided:
 - (a) on a 'cost-plus 10%' basis (unless otherwise agreed within the relevant Transitional Service Agreement);
 - (b) with the scope, in the volumes, with the frequency, at a quality level and in a manner, that is in substantially the same form, to substantially the same extent, on substantially the same overall standards and quality and on substantially the same terms and conditions, in each case, as the Transitional Services it provided in the twelve (12) month period immediately prior to the Effective Date, except as otherwise set out in the relevant Transitional Service Agreements;
 - (c) with reasonable skill and care; and
- (d) in accordance with applicable Law.
- 2.4.2 Unless agreed otherwise in writing, no Provider shall be under any obligation to supply any extension of any Transitional Service beyond the standard and scope of such Transitional Service, nor any Transitional Service at a level of demand which exceeds the level of demand for that Transitional Service or similar service, during the twelve (12) month period immediately prior to the Effective Date. No Provider shall be required to provide any Transitional Service to the extent that doing so would be contrary to any applicable Law.
- 2.4.3 If any Provider changes the standard and scope of any service it provides in its own business or to any of its Affiliates which is of a similar nature to any Transitional Service, such Provider may make similar changes to the standard and scope of such Transitional Service, provided that (a) such Provider shall notify the relevant Recipient in writing regarding the change within a reasonable period, in advance, and, unless otherwise agreed in writing, (b) the standard and scope of such changed Transitional Service is not, when considered as a whole, materially lower than the standard and scope of the Service provided prior to such modifications.

2.5 Relationship with Third Party Suppliers

2.5.1 Subject to Clauses 2.5.2 and 2.5.3 to the extent it relates to the provision of a Transitional Service by a Provider to the relevant Recipient, the Provider may exclusively manage its relationship with Third Party Suppliers and a Recipient shall not discuss the provision of a Transitional Service with any Third Party Supplier, except (i) with prior written consent of the relevant Provider or (ii) to

8 of 30

the extent required to do so by a competent court, judicial or regulatory authority or to enable the Recipient to comply with its obligations under applicable Law.

- 2.5.2 Nothing in this Clause 2.5 (*Relationship with Third Party Suppliers*) will be interpreted as restricting any approach to a Third Party Supplier by a Recipient to discuss or agree service delivery after the Transition Period.
- 2.5.3 If a Third Party Supplier is in breach of a Third Party Agreement or does not rectify an issue relating to the provision of the Transitional Services for which it is responsible, the Parties shall cooperate and use their reasonable efforts to identify a resolution to such breach or issue and shall, subject to the terms and conditions of such Third Party agreement and the relevant Transitional Service Agreement, consult with each other to determine who will be responsible for any costs, expenses, fees, or charges incurred in relation to such resolution and who may bring any cause of action against the Third Party Supplier for such breach or failure to rectify, in each case insofar as it relates to the Recipient.
- 2.5.4 The Provider shall regularly disclose to the Recipient the amounts charged by Third Party Suppliers. Where Third Party Suppliers increase their charges in relation to services that are necessary for the provision of a Transitional Service, other than charges for Third Party Consents, such Provider may, subject to: (i) disclosure of the relevant terms to the Recipient; (ii) the Provider having used reasonable efforts to negotiate with the Third Party Supplier to minimise or eliminate the increase in charges; and (iii) the Provider having provided reasonable prior notice to and consultation with, the Recipient, charge the Recipient for such part of the increase as reasonably attributable to the Transitional Services (in addition to the Charges), as long as the relevant Services are provided. Charge increases of more than 10% require the Recipient's prior approval, which approval shall not be unreasonably withheld. If the Recipient does not approve, the Recipient may terminate the Transitional Service by giving one month's notice and after having reimbursed the Provider for the Recipient's equitable part of the costs incurred by the Provider in connection with such termination.

2.6 Third Party Consents

2.6.1 The Parties shall cooperate and use their best endeavours to obtain in writing all Third Party Consents, including extensions, necessary for the provision of the Services.

- 2.6.2 Each Recipient shall provide the relevant Provider with such assistance as such Provider may reasonably require to obtain any Third Party Consents.
- 2.6.3 A Recipient shall pay for any fees, charges or costs imposed by a Third Party Supplier relating to the provision or extension of any Third Party Consent in relation to a Service exclusively provided to such Recipient. For Services that are not exclusively provided to such Recipient, the Recipient shall bear such equitable portion of the cost as is attributable to the Service provided to such Recipient.

2.6.4 Where:

- (a) Third Party Consent is required in order for a Provider to provide any (part of a) Transitional Service; and
- (b) (i) such consent has not been obtained and the relevant Third Party Supplier has notified the Provider that it does not consent to the provision of that (part of the) Transitional Service; (ii) such consent expires before the end of the relevant Transitional Service Term; or (iii) the relevant Third Party Supplier has notified the Provider that it has terminated or revoked such consent,

the Provider shall be excused from providing the relevant (part of the) Transitional Service. In such event, the Provider shall notify the relevant Recipient as soon as reasonably practicable and the Parties shall use reasonable endeavours to put in place alternative arrangements to enable the Provider to provide services that are similar in all material respects to the Transitional Service for which the Third Party Consent was required. The cost of putting such alternative arrangements in place, where they are put in place exclusively for the Recipient, shall be borne by the Recipient. If the Parties cannot jointly solve the issue or agree on an alternative arrangement following good faith discussions, the Recipient will have the right to put in place alternative arrangements at its cost. If and to the extent the Provider ceases to provide the relevant (part of the) Service there shall be an equitable reduction in the Charges to reflect any cost savings to the Provider as a result of such cessation.

3 FRAMEWORK SERVICES

3.1 Master Framework Services Agreement

This Agreement is a master agreement. Each Framework Services Agreement shall be construed as a separate and independent agreement for each Framework Service provided thereunder, which incorporates the terms of this Agreement, except as such terms may be expressly varied or supplemented by the terms of a specific Framework Service Agreement, whereby the terms of such Framework Services Agreement shall prevail, provided that in case of a conflict between Clause 3 (*Framework Services*), 8.3.2 or 9.2 (*Liquidated damages*) (the "**Prevailing Clauses**") and a provisions of a Framework Service Agreement, the relevant Prevailing Clause shall apply. Such

10 of 30

supplemental terms shall include all terms and conditions specific to and customary for each such Framework Service. Parties shall procure that their relevant Affiliates act in accordance with the foregoing agreement.

3.2 Obligation to request an offer from Fugro

- 3.2.1 During the Initial Framework Term, and subject to Clause 3.2.4, the Fugro Group will hold preferred provider status for the Framework Services, with the rights and obligations set forth in this Agreement.
- 3.2.2 During the Initial Framework Term and to the extent a Framework Service cannot be provided under an existing Framework Service Agreement, the GMSL Group shall advise Fugro of the scope and specifications of any additional Framework Service(s) required by any GMSL Group Company as soon as reasonably possible but in any event at the same time as a request (for a proposal) for the relevant Framework Service(s) is made with any Third Party.
- 3.2.3 For the avoidance of doubt, the foregoing shall not restrict the GMSL Group from simultaneously requesting a proposal or offer for the relevant Framework Service(s) from a Third Party.
- 3.2.4 GMSL's obligations under this Clause 3.2, Clause 3.3 and Clause 9.2 shall cease to apply per relevant Framework Service if (a) Fugro is in persistent breach of Clause 3.5 and such breach is material to the provision of the relevant Framework Service, and (b) Fugro has been given written notice by GMSL of such breach containing details of the breach and has subsequently not remedied the relevant breach within thirty (30) Business Days after receipt of such notice.

3.3 Preferred services provider arrangement

- 3.3.1 Subject to Clauses 3.3.2 and 3.3.4, if the GMSL Group wishes to acquire a Framework Service, the GMSL Group shall acquire such Framework Service from the Fugro Group during the Initial Framework Term, as further identified in Clause 6.2, and shall enter into a Framework Service Agreement with the relevant member of the Fugro Group for such Framework Service, except if:
- (a) a Third Party made an offer to provide such Framework Service (or a similar service) on equivalent or more favourable terms and conditions, and at similar or better quality, at a price that is at least 5% (five percent) lower than offered by the relevant Fugro Group Company; or

- (b) Fugro has notified GMSL that no Fugro Group Company is able or willing to provide the requested Framework Service; or
- (c) neither Fugro nor any Fugro Group Company responds to the GMSL Group's request for a Framework Service within fifteen (15) Business Days after the GMSL Group's offer or request is issued; or
- (d) GMSL withdraws its request and elects not to proceed,

in which case no GMSL Group Company shall have any obligation to acquire such Framework Service from any Fugro Group Company.

- 3.3.2 If GMSL, acting reasonably and in good faith, determines that the pricing offered by a Fugro Group Company exceeds the price specified in Clause 3.3.1(a) for a Framework Service (or similar service) comparable in scope and standards of quality and with equivalent or more favourable terms and conditions, it will provide notice of this to Fugro with reasonable documented supporting evidence which allows Fugro to assess the details of the more favourable offer or proposal made by the relevant Third Party, provided that nothing in this Clause shall oblige GMSL to breach any obligation of confidentiality it owes to a Third Party.
- 3.3.3 If Fugro decides to revise its pricing and match the quality of work and pricing offered by the relevant Third Party in accordance with the foregoing sub-clause, the relevant GMSL Group Company shall accept such revised offer and acquire the relevant Framework Service from Fugro. In the latter case, no member of the GMSL Group shall request such Third Party for a revised offer in respect of the relevant Framework Service.
- 3.3.4 If the Fugro Group elects not to revise its pricing offered in accordance with Clause 3.3.2, the GMSL Group will be released from the obligation of Clause 3.3.1 in respect of the requested Framework Service. The foregoing release will not release the GMSL Group from the obligation to acquire any Framework Service required thereafter (including any Framework Service that in an earlier case was not carried out by Fugro) from the Fugro Group on a preferred basis during the term of the Agreement, as further identified in Clause 6.2.
- 3.3.5 Nothing in this Agreement shall bar or prohibit the GMSL Group from performing any of the services which are the same or similar to the relevant Framework Service within its organisation.

12 of 30

3.3.6 The Parties shall in good faith discuss Failed Bids (as defined below) in accordance with Clause 11.2.

3.4 Framework Service Agreements

- 3.4.1 <u>Schedule 4</u> (*Agreed Framework Service Agreements*) includes the agreed form Framework Service Agreements in respect of the Framework Services that the Fugro Group will in any event provide to the GMSL Group and Fugro and GMSL shall procure that these Framework Service Agreements will be executed by the Provider and the relevant Recipient at or as soon as possible after the Effective Date.
- 3.4.2 If any additional Framework Service cannot be provided under an existing Framework Agreement, Parties shall, or shall cause that their relevant Affiliates, enter into a new Framework Service Agreement to the extent possible in similar form as the Framework Service Agreements included in <u>Schedule 4</u> (Agreed Framework Service Agreements) or, to the extent Parties determine in good faith that the foregoing would not be appropriate in respect of the relevant Framework Service, a form based on the standard contracts for the UK offshore oil & gas industry as published by Logic or similar bodies.

3.5 Standard of Framework Services

- 3.5.1 Fugro shall, and shall ensure that each Provider shall, provide the Framework Services in accordance with:
 - (a) that level of skill, care, prudence, judgment, foresight, integrity and diligence that would be reasonably expected of a Third Party provider of services similar to the Framework Services; and
 - (b) applicable Law.

4 VESSEL SERVICES

Each Party agrees to inform the other Party in case it requires any additional vessel capacity and is or should reasonably be aware that the other Party has vessel capacity with the required specifications. The informed Party may, subject to availability, elect to offer the required vessel capacity at its terms and conditions to the informing Party. The informing Party may subsequently, in its sole discretion and without any obligation, elect to use the vessel offered by the informed Party on the offered terms and conditions so offered. For the avoidance of doubt and subject to the foregoing, each Party shall use reasonable best endeavours to make use of the other Party's vessel capacity if possible and required.

5 HUAWEI JOINT VENTURE AND SBSS JOINT VENTURE

GMSL shall use reasonable efforts to procure that the Huawei JV and the SBSS JV enter into a similar framework agreement with Fugro or another member of the Fugro Group pursuant to which the Fugro Group becomes the 'preferred provider' (at market pricing) of the Huawei JV and the SBSS JV for Framework Services. GMSL shall in any event use reasonable efforts to procure that Fugro (i) can as soon as reasonably possible after the Effective Date periodically showcase its services to the management of the Huawei JV and the SBSS JV, (ii) may tender for any service required by the Huawei JV and the SBSS JV that a member of the Fugro Group can provide and, in the case the relevant member of the Fugro Group's offer of services is substantially similar or better in scope, terms and conditions (including price) as any competing offer, GMSL shall use its decision and management powers in the Huawei JV or SBSS JV (as applicable) to procure, so far as it is able given such powers and subject to any fiduciary duties that any director appointed by GMSL to the boards of such companies may have, that the Huawei JV or SBSS JV (as applicable) award any such tender to the relevant member of the Fugro Group.

6 TERM

6.1 Commencement

This Agreement will become effective on the Effective Date. Application of section 7:408 of the Dutch Civil Code is excluded.

6.2 Expiry

- 6.2.1 Subject to the earlier termination of this Agreement in accordance with Clause 8 (Termination), this Agreement will expire:
 - (a) with regard to any obligation connected to the Transitional Services, including, but without limitation, any obligation stemming from Clause 2 (*Transitional Services*), and subject to Clause 6.3: on the expiry of the last Transitional Service Term ("**Transition Period**"); and
- (b) with regard to any obligation connected to the Framework Services, including, but without limitation, any obligation stemming from Clause 3 (*Framework Services*): on the last date of the Framework Term, subject to the provision of Clause 6.4.
- 6.2.2 Expiry of this Agreement will not cause the expiry or termination of any Framework Service Agreement in effect prior to the date of expiry of this Agreement. Any such Framework Agreement will expire on the date set out therein.

14 of 30

6.3 Transitional Service Term extension

If within twenty (20) Business Days prior to the expiry of a Transitional Service Term, GMSL notifies Fugro that the relevant Recipient is not, having used all reasonable efforts to do so, able to operate without the relevant Transitional Service and the Fugro Group can reasonably provide such Transition Service, the Parties shall in good faith agree an extension to that Transitional Service Term, which Fugro shall not unreasonably refuse.

6.4 Framework Term extension

The Parties may extend the Framework Term of this Agreement for successive terms consisting of twelve (12) months each (the "Framework Extension Term"), by mutual agreement at least twenty (20) Business Days before the end of the then-current term.

7 TERMS OF SERVICES

7.1 Charges for the Services

- 7.1.1 Unless explicitly agreed otherwise in the relevant Service Agreement, the monthly Charge is the aggregate amount monthly charged under each Service Agreement.
- 7.1.2 Charges will be fixed for the term stated within the relevant Service Agreement except to the extent the Parties agree to vary the Charges in writing or as provided for in Clause 2.5.4 and 7.1.3. Subject to Clause 2.5.4, any increase or decrease in Charges will not take effect until twenty (20) Business Days after the adjustment to the Charges has been agreed or otherwise taken to have occurred by the Parties.
- 7.1.3 Any Provider must promptly notify the relevant Recipient if it considers, acting reasonably, that any change to the Service required by a Recipient will materially increase or decrease the costs to the Provider of providing such Service.
- 7.1.4 A notice under Clause 7.1.3 must set out the Provider's reasonable calculation of any adjustment to the Charges to reflect the actual and proportionate increase or decrease in the cost in providing the Service as a result of the change on a pass-through basis and with any mark-up together with evidence to substantiate the increase or decrease.

7.2 No withholding and gross-up

All sums payable under this Agreement and the Service Agreements shall be paid gross, free and clear of all Tax Deductions, save only as may be required by Law. If a payor is required by Law to make a Tax Deduction under this Agreement or a Service

Agreement, the amount of the payment will be increased to an amount which, after making the required Tax Deduction, leaves the recipient in the same after-Tax position as it would have been, had no Tax Deduction been required. For the avoidance of doubt, this Clause 7.2 (*No withholding and gross-up*) does not apply to any income or trade Tax due by the Recipient in respect of any sum payable under or otherwise in connection with this Agreement or a Service Agreement. The payor and the recipient shall cooperate to minimise any Tax Deduction to the extent legally permissible.

7.3 VAT

- 7.3.1 All amounts stated in this Agreement are exclusive of VAT, if any. If any VAT (other than reverse charge VAT) is chargeable on any delivery of Services under this Agreement or a Service Agreement, the Recipient shall pay to the Provider, in addition to the Charge, an amount equal to the amount of the applicable VAT against delivery of an invoice that is appropriate and valid under the applicable VAT rules.
- 7.3.2 Each Party, Provider and Recipient shall cooperate to minimise any VAT due on invoices to the extent legally permitted.

7.4 Invoicing and payment terms

- 7.4.1 The Provider shall invoice the Recipient for each Service.
- 7.4.2 In case of recurring fixed charge services, the Provider shall invoice on a monthly basis at the beginning of the month in which the service is to be provided.
- 7.4.3 In case of variable cost charges, the Provider shall invoice every two (2) weeks, after each two (2) week period in which the relevant Service was delivered or performed.
- 7.4.4 The Recipient shall notify the relevant Provider as soon as reasonably possible if it considers the invoice incorrect, together with the reasons thereof, and may only withhold payment of any amounts that it disputes in good faith.
- 7.4.5 The Recipient shall pay all undisputed invoice amounts within twenty (20) Business Days after receipt of such invoice.

7.5 Validity of any Service Agreement

7.5.1 No Service Agreement has any legal effect until each party to the Service Agreement has validly signed such agreement.

16 of 30

7.5.2 If a Service Agreement is signed in counterparts, these counterparts will count as one agreement.

8 TERMINATION

8.1 Termination for cause

Without detracting from Clause 8.2 (*Termination for breach*), each Party may terminate this Agreement and each Provider and each Recipient may terminate any Service Agreement in accordance with Clause 8.7 (*Notice of termination*) in the event that:

- (a) the other Party has ceased to exist or has been dissolved other than for the sake of a business merger; or
- (b) the other Party has been declared bankrupt or has been granted suspension of payments by an irrevocable court decision, unless such Party gives reasonable assurance that it will and will be able to continue to perform its obligations under this Agreement.

8.2 Termination for breach

- 8.2.1 Without detracting from Clause 8.1 (*Termination for cause*), each Party may terminate this Agreement (in whole or in part) if the other Party is in material breach of any obligation under this Agreement in accordance with Clause 8.7 (*Notice of termination*), provided that:
- (a) the failure is not caused by a Force Majeure Event;
- (b) the other Party has been given written notice of default containing details of the default; and
- (c) the other Party has not performed such obligation within thirty (30) Business Days after receipt of the notice of default.
- 8.2.2 A Party may not terminate this Agreement (in whole or in part) in accordance with Clause 8.2.1 if the default of the other Party only relates to a default under any Service Agreement.

8.3 Effect of termination

8.3.1 Termination of this Agreement in accordance with Clause 8.1 (*Termination for cause*) or Clause 8.2 (*Termination for breach*) does not affect any accrued rights, liabilities or payments due or Clauses 9 (*Remedies*), 10 (*Liability*), 12 (*Intellectual Property and Use of Materials*), 13 (*IT Security and Data*), 16 (*Miscellaneous*) and 17 (*Governing Law and Dispute Resolution*).

8.3.2 Termination of this Agreement will not cause the termination of any Framework Service Agreement in effect prior to the date of termination of the Agreement.

8.4 Termination of Services for breach

- 8.4.1 Each Provider or Recipient may terminate a Service with immediate effect by a notice referred to in Clause 8.7 (*Notice of termination*) to the other Party (the "**Defaulting Party**") if that Defaulting Party commits in relation to such Service a material breach of its obligations under the relevant Service Agreement, and (where the breach is capable of being remedied) that breach has not been remedied within twenty (20) Business Days after receipt of a notice giving particulars of the breach and requiring the Defaulting Party to remedy it, provided that: (i) the breach is not caused by a Force Majeure Event; and (ii) in the case of non-payment of undisputed amounts by a Recipient, Fugro has served GMSL a further notice notifying GMSL that the Provider will terminate if payment is not made within five (5) Business Days after such further notice, and payment is not made within five (5) Business Days after such further notice.
- 8.4.2 Termination of a Service Agreement will not cause the termination of this Agreement.
- 8.4.3 Termination of a Service Agreement in accordance with Clause 8.4.1 does not affect any accrued rights, liabilities or payments due under the relevant Service Agreement or Clauses 9 (*Remedies*), 10 (*Liability*), 12 (*Intellectual Property and Use of Materials*), 13 (*IT Security and Data*), 16 (*Miscellaneous*) and 17 (*Governing Law and Dispute Resolution*) incorporated by reference in such Service Agreement.

8.5 Termination of Transitional Services for Convenience

A Recipient may terminate any Transitional Service for convenience with one (1) months' notice to the relevant Provider. Any committed and unavoidable costs incurred by the Provider as a result of the termination of a Transitional Service Agreement entered into exclusively for the performance of the Transitional Services, where such termination occurs before the expiry date set out in the relevant Transitional Service Agreement, shall be for the Recipient's account. The Recipient shall reimburse the Provider accordingly, provided that the Provider has used all reasonable endeavours to mitigate such committed and unavoidable costs.

18 of 30

8.6 Force Majeure

- 8.6.1 No Provider or Recipient is responsible for a failure or delay in the performance of this Agreement or any Service Agreement due to causes beyond its control, except for the obligation of a party to make a payment. These causes include, but without limitation, fire, storm, flood, earthquake, explosion, acts of any public enemy, war, rebellion, insurrection, sabotage, terrorism, epidemic, transportation embargoes, fuel or energy shortages, power interruptions, acts of God, acts, rules or orders of any Governmental Authority (each a "Force Majeure Event").
- 8.6.2 If a failure or delay in the performance of this Agreement or any Service Agreement is caused by a Force Majeure Event, the non-performing party shall give written notice to the other party of the reason for its failure or delay and the extent and expected duration of its failure or delay.
- 8.6.3 For the duration of any Force Majeure Event:
 - (a) any affected Provider or Recipient will be relieved of its obligations under this Agreement and/or the relevant Service Agreement in respect of such Service; and
 - (b) a relevant Recipient shall have the right to obtain elsewhere at its own risk and cost such quantities of a service similar to the Service in question to cover its requirements for the duration of such Force Majeure Event. The Provider will provide the Recipient all reasonable assistance required in this regard.
- 8.6.4 If a Provider or Recipient is excused under this Clause 8.6 (Force Majeure) from the performance of any material obligation under this Agreement or a Service Agreement for a continuous period of thirty (30) Business Days, then that party may at any time thereafter, and provided performance or punctual performance is still affected by the Force Majeure Event, terminate any Service affected by the Force Majeure Event by giving not less than ten (10) Business Days' notice referred to in Clause 8.7 (Notice of termination) to the Provider or Recipient (as the case may be) affected by the Force Majeure Event.

8.7 Notice of termination

If a Party terminates this Agreement and/or a Service Agreement pursuant to Clause 8.1 (*Termination for cause*) or Clause 8.2 (*Termination for breach*), the terminating party shall give written notice of termination to the other party specifying the reason for termination and the date on which the termination will be effected.

9 REMEDIES

9.1 Affiliates

Any breach of this Agreement:

- (a) by any GMSL Group Company will constitute a breach of this Agreement for which GMSL is principally responsible and liable; and
- (b) by any Fugro Group Company will constitute a breach of this Agreement for which Fugro is principally responsible and liable.

9.2 Liquidated damages

- 9.2.1 If the GMSL Group is in breach of an obligation under Clause 3.2 (*Obligation to request an offer for the provision of Framework Services from Fugro*) or Clause 3.3 (*Preferred services provider arrangement*), GMSL shall pay in cash to the relevant Fugro, as liquidated damages, 20% (twenty percent) of the fees owed for the relevant Framework Service to the relevant Third Party that provided the Framework Service without taking into account any rights of set-off, liquidated damage, volume discount or the like.
- 9.2.2 The remedies of this Clause 9.2 (*Liquidated damages*) are the sole and exclusive remedy with regard a breach of any obligation under Clauses 3.2 (*Obligation to request an offer for the provision of Services from Fugro*) and 3.3 (*Preferred services provider arrangement*).

10 WARRANTIES AND LIABILITY

10.1 Warranties

Save as expressly otherwise provided in a Transitional Service Agreement, Fugro nor any Fugro Group Company provides any warranty or representation, express or implied, with respect to any Transitional Service or other matters set forth in this Agreement.

10.2 Liability regimes

10.2.1 No Fugro Group Company is liable for any Losses incurred by the GMSL Group caused by a breach of its obligations under Clause 2 (*Transitional Services*) or a Transitional Service Agreement, including pursuant to a failure to supply a Transitional Service in accordance with the terms of this Agreement or the relevant Transitional Service Agreement, except where such Losses result from an intentional breach or breach caused by fraud, wilful misconduct or gross negligence of a member of the Fugro Group, in which case such Losses shall be compensated in full.

20 of 30

10.2.2No Fugro Group Company is liable for Losses incurred by GMSL Group resulting from a breach of a Third Party Supplier of an obligation arising from a Third Party Agreement except to the extent Fugro has been able to recover such Losses from the relevant Third Party Supplier.

11 RELATIONSHIP MANAGEMENT

11.1 Relationship Manager

Within five (5) Business Days following the date of this Agreement, each of the Parties shall nominate a project manager to manage and coordinate their respective obligations under this Agreement and to act as the primary contact between them for that purpose (a "**Relationship Manager**"). Neither Party may replace its Relationship Manager (except on a temporary and urgent basis) without obtaining the other's prior approval of the replacement, such approval not to be unreasonably withheld or delayed. The Relationship Managers shall meet (in person or by telephone) at a frequency to be agreed by Parties. Where a Relationship Manager is absent due to holiday or illness, the relevant party shall arrange for a suitable delegate to attend such meetings.

11.2 Failed Bids

The Relationship Managers shall meet once every three (3) months to review and discuss in good faith the bids offered by Fugro pursuant to Clause 3.3, but awarded by GMSL to a Third Party (such bids being the "Failed Bids"). During such review GMSL's Relationship Manager shall provide Fugro's Relationship Manager with:

- 11.2.1 an overview of the Failed Bids, which shall include an explanation of the reasons why the Failed Bids, taken together, were not awarded to Fugro, provided that GMSL shall have no obligation to breach any confidentiality obligation it owes to a Third Party in doing so; and
- 11.2.2 information about GMSL's bidding pipeline and forecasted scopes of work over the forthcoming six (6) to nine (9) months.

11.3 Bid Dispute

If a dispute arises as to whether a Failed Bid should have been awarded to Fugro, or a dispute in relation to any other matter in connection with the application of Clause 3.2 or Clause 3.3 arises (each such dispute being a "**Bid Dispute**"), the Relationship Managers shall attempt to resolve the Bid Dispute in the first instance. If, after a period of thirty (30) days from the date the Bid Dispute was first raised, the Relationship Managers are still unable to resolve the matter, either Relationship Manager may escalate the matter to the CEO of GMSL and the Division Director Marine of Fugro to resolve the Bid Dispute. If after a period of thirty (30) days from the date the CEO of GMSL and the Division Director Marine of Fugro, they are still unable to resolve the matter, either such person may escalate the matter to the Global board for resolution. Neither Party shall take any action pursuant to Clause 17.2

in respect of any Bid Dispute until such time as the Global board, having used reasonable efforts and acting promptly and in good faith, has notified the Parties that it is unable to resolve the Bid Dispute.

12 INTELLECTUAL PROPERTY AND USE OF MATERIALS

12.1 Intellectual Property

Other than as explicitly set out in this Agreement, nothing in this Agreement shall be construed to give any Party or any of its Affiliates any right, title or interest in Intellectual Property of the other Party or any of its Affiliates.

12.2 Provider Materials

If a Provider, in the performance of its obligations under this Agreement, makes available to a Recipient any Provider Materials, such Provider Materials shall, other than as explicitly set out in this Agreement, remain the sole property of the Provider or the relevant Affiliate of the Provider, and the Recipient will not be granted any right or title in such Provider's Materials except for a non-exclusive, non-transferable royalty free license to use the Provider Materials but only to the extent required in connection with the Services and for the relevant Service only.

12.3 IP Claim

12.3.1 If a (potential) claim pursuant to this Agreement relates to a claim or the commencement of an action or proceeding by a Third Party against a Provider or a Recipient and is arising out of or in connection with any claim alleging that the performance, or receipt or enjoyment by the Recipient of the Services under this Agreement infringes any intellectual property right belonging to that Third Party (an "IP Claim"), then the Provider shall have the right to defend and settle (without limiting its obligations vis-à-vis the Recipient to provide Services under this Agreement), at its own expense and by its own counsel, any such matter involving the asserted liability, provided, that the Provider shall keep the Recipient reasonably informed of the matter

22 of 30

12.3.2 Fugro shall indemnify each GMSL Group Company and their respective directors, officers, shareholders, employees, agents, representatives, successors and assigns against all liabilities, damages, costs, reasonable expenses and fees (including defence costs) arising out of an IP Claim in relation to a Service but only if and to the extent the provided Service was used in accordance with any and all provided instructions or materials in respect of uses or applicability of the given Service.

13 IT SECURITY AND DATA

13.1 Unauthorised access

Neither a Party nor any of its Affiliates shall use the network of the other Party and/or of its Affiliates for any improper purpose, including, but without limitation:

- (a) in breach of any Law;
- (b) for the purpose of gaining or attempting to gain access to data or to other networks other than exclusively for the purpose of receiving or providing any Service; or
- (c) for the purpose of carrying data for the benefit of a Third Party, consultant, agent, customer or vendor in connection with the conduct of their respective businesses or operations in the ordinary course.

13.2 No access to networks and data

- 13.2.1Neither a Party nor any of its Affiliates shall have access to the systems, networks, services or data of the other Party and/or of its Affiliates, except with (i) the prior written consent of the other Party and (ii) where necessary to provide or receive Services, and (iii) then only for the purpose of providing or receiving those Services.
- 13.2.2From the Effective Date, any Fugro Group Company must, and must procure that none of their Representatives, access or retrieve any email, data or information contained on any electronic devices, computers, storage facilities, servers or networks (including back-ups) of any GMSL Group Company.
- 13.2.3From the Effective Date, any GMSL Group Company must, and must procure that none of their Representatives, access or retrieve any email, data or information contained on any electronic devices, computers, storage facilities, servers or networks (including back-ups) of any Fugro Group Company.

13.3 Security breach

Each Party shall promptly notify the other Party of any security breach that affects or may affect such Party or its Affiliates or the provision of Services. The Provider may immediately suspend the provision of any Service if the nature of such security breach requires such suspension according to Laws, regulations, Governmental Authority instructions or internal Provider procedures, provided that such a breach is not caused, in whole or in part, by any failure or default in the performance of such Service to a Recipient by a Provider.

14 SEPARATION FROM SELLER'S NETWORKS

14.1 Obligation to complete Network Separation

GMSL shall use reasonable endeavours to complete Network Separation by no later than 1 November 2017.

14.2 Connection to Fugro's Networks

For so long as the GMSL Group remains connected to any part of the Fugro Networks: GMSL shall, and shall cause the relevant member of the GMSL Group to, procure that no connection is established from the Computer Systems to any other system except via a firewall approved by Fugro prior to implementation and complying with Fugro's information technologies security policies and such other standards as Fugro may reasonably require. Fugro shall have the right to audit the firewall, and may use automated tools to verify that the firewall is in accordance with Fugro's standards.

14.3 Temporary disconnection

If Fugro reasonably believes that actions by the GMSL Group or the GMSL Group's use of or access to the Computer Systems or the Fugro Network, in each case in breach of this Agreement, are causing a material adverse impact on the business operations of any member of the Fugro Group (by way of example only, an inability to conduct critical operations as a result of a computer virus), Fugro shall, after taking such other alternative actions as, in the circumstances, could reasonably have been expected of Fugro to consider and take, have the right to temporarily disconnect the relevant member(s) of the GMSL Group from the relevant parts of the Fugro Network. In this case, both Parties shall use their reasonable endeavours to take steps such that the problems in question are brought to an end. Fugro shall reconnect the relevant GMSL Group Company to the Fugro Network in question as soon as reasonably practicable after the issue is resolved.

24 of 30

15 CONFIDENTIALITY

During the term of this Agreement and thereafter, the Parties, any Provider and any Recipient shall, and shall instruct their respective representatives to, maintain in confidence and not disclose the other party's financial, technical, sales, marketing, development, personnel, and other information, records, or data, including, without limitation, customer lists, supplier lists, trade secrets, designs, product formulations, product specifications or any other proprietary or confidential information, however recorded or preserved, whether written or oral (any such information, the "**Confidential Information**"). Unless otherwise authorized in any other agreement between the Parties, any Party receiving any Confidential Information of the other Party (the "**Receiving Party**") is entitled to use Confidential Information only for the purposes of fulfilling its obligations under this Agreement. The Receiving Party shall be liable for any breach of these confidential Information is required to be disclosed by a Governmental Authority, in which case the Receiving Party shall promptly notify, to the extent possible, the disclosing Party.

16 MISCELLANEOUS

16.1 No assignment or subcontracting

- 16.1.1Except as otherwise expressly provided in this Agreement, no Party to this Agreement may, unless with the prior written consent (which shall not be unreasonably withheld) of the other Party, assign, grant any security interest over or otherwise transfer, in whole or in part, any of its rights and obligations under this Agreement.
- 16.1.2With the exception of the subcontractors and sub-processors in place as the Effective Date, Fugro shall not subcontract any of its obligations under this Agreement or appoint or replace any sub-processor or subcontractor, without the prior consent of GMSL, such consent not to be unreasonably withheld or delayed.

16.2 Waiver

No waiver of any provision of this Agreement shall be effective unless such waiver is in writing and signed by or on behalf of the Party entitled to such Waiver.

16.3 Amendment

No amendment of this Agreement shall be effective unless such amendment is in writing and signed by or on behalf of each of the Parties.

16.4 Third Party rights

Save as expressly otherwise stated, this Agreement does not contain any stipulation in favour of a Third Party (derdenbeding).

16.5 Employees

The employees of each Provider shall solely remain employees of the Provider and shall remain subject only to its supervision. Fugro shall be liable for (and shall reimburse GMSL for) any Losses sustained by any GMSL Group Company arising from any claim or action by employees or contractors of Fugro in connection with this Agreement or any Service Agreement, unless and to the extent such Losses are caused by unlawful acts or omissions carried out by a GMSL group Company or Representative thereof.

16.6 No agency or partnership

This Agreement does not constitute any Party, Provider or Recipient the agent of another Party, Provider or Recipient or imply that the Parties intend constituting a partnership, joint venture or other form of association in which any such party may be liable for the acts or omissions of the other party.

16.7 Costs

Unless this Agreement provides otherwise, all costs which a Party has incurred or must incur in preparing or concluding this Agreement are for its own account.

16.8 Rescission

Each Party waives its right to rescind (*ontbinden*) this Agreement on the basis of section 6:265 of the Dutch Civil Code. Furthermore a Party in error shall bear the risk of such error (*dwaling*) in making this Agreement.

16.9 Notices

- 16.9.1Any notice, request, consent, claim, demand and other communication between the Parties in connection with this Agreement (a "Notice") shall be in writing and shall be given and shall be deemed to have been duly given if written in the English language and:
- (a) delivered personally (Notice deemed given upon receipt);

- (b) delivered by registered post (Notice deemed given upon confirmation of receipt); or
- (c) sent by an internationally recognized overnight courier service such as Federal Express (Notice deemed given upon receipt),

in each case with a copy by email, which copy shall not constitute a Notice.

16.9.2A Notice to a Party shall be sent to such Party at the address set out in <u>Schedule 5</u> (*Parties' details for Notices*) or such other person or address as such Party may notify to the other Parties from time to time.

17 GOVERNING LAW AND DISPUTE RESOLUTION

17.1 Governing law

This Agreement and the documents to be entered into pursuant to it, save as expressly otherwise provided therein, shall be governed by and construed in accordance with the Laws of the Netherlands.

17.2 Jurisdiction and forum

The Parties irrevocably agree that all disputes which may arise out of or in connection with this Agreement and the documents to be entered into pursuant to it, save as expressly otherwise provided therein, including disputes concerning the existence and validity thereof, shall be finally and exclusively resolved in accordance with the arbitration rules of the Netherlands Arbitration Institute taking into account the following:

- (a) the arbitral tribunal shall be composed of three (3) arbitrators of which at least one (1) shall be a lawyer with a Netherlands law degree or admitted to practise Netherlands law, each of them appointed in accordance with the applicable arbitration rules;
- (b) the place of arbitration shall be Amsterdam, the Netherlands;
- (c) the proceedings shall be conducted in the English language;
- (d) the arbitral tribunal shall decide in accordance with the rules of law;
- (e) the right, if any, to discovery is excluded; and
- (f) neither the Parties nor the arbitration institute may have the arbitral award published.

17.3 Other disputes

This Clause 17 (*Governing Law and Dispute Resolution*) shall also apply to disputes arising in connection with Service Agreements that are connected with this Agreement, unless the relevant Service Agreement expressly provides otherwise.

[SIGNATURES TO FOLLOW ON THE NEXT PAGE]

28 of 30

THIS AGREEMENT HAS BEEN SIGNED ON THE DATE STATED AT THE BEGINNING OF THIS AGREEMENT BY:

<u>/s/ Annabelle Vos</u> **Fugro N.V.** By: Annabelle Vos Title: Power of Attorney

[Signature page to the Project Furrow Transitional and Framework Services Agreement]

THIS AGREEMENT HAS BEEN SIGNED ON THE DATE STATED AT THE BEGINNING OF THIS AGREEMENT BY:

<u>/s/ Ian Douglas</u> Global Marine Services Limited By: Ian Douglas Title: Chief Executive Officer

[Signature page to the Project Furrow Transitional and Framework Services Agreement]

Schedule 1 Definitions and interpretation

1 Definitions

Capitalised terms, including those used in the introduction and preamble of this Agreement, have the following meaning:

"Affiliate" means with respect to any Person, another Person directly or indirectly Controlling, Controlled by, or under common Control with the first Person as of the date on which, or at any time during the period for which, the determination of affiliation is being made;

"Agreement" means this Transitional and Framework Services Agreement;

"Bid Dispute" has the meaning given in Clause 11.3 (Bid Dispute);

"BPA" has the meaning as set out in recital (A) of this Agreement;

"Business Day" means a day which is not a Saturday, a Sunday or a public holiday in the Netherlands pursuant to section 3 of the Dutch General Time Limits Act (Algemene Termijnenwet) or a public holiday in the United Kingdom;

"Charges" means the charges payable by a Recipient to the relevant Provider in respect of the rendering of any Service by such Provider;

"Computer Systems" means the communications systems, hardware, software and networks infrastructure which are used by the Employees (as defined in the BPA) in connection with the Trenching Business;

"Confidential Information" is defined in Clause 15 (Confidentiality);

"Control" means, with respect to a Person, (i) the direct or indirect ownership of more than 50% (fifty per cent) of the outstanding voting securities (or comparable voting interest or financial participation) of such Person, (ii) the ability to appoint more than half of the directors of the board of directors or equivalent governing body of such Person, or (iii) the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person. The terms "Controlling", "Controlled by" and "under common Control" shall have the correlative meanings;

"Defaulting Party" has the meaning given in Clause 8.4.1;

"Effective Date" means the date of this Agreement;

"Failed Bids" has the meaning given in Clause 11.2 (Failed Bids);

"Force Majeure Event" is defined in Clause 8.6.1;

"Framework Extension Term" is defined in Clause 6.4 (Automatic extension);

"Framework Service Agreement" means a Service Agreement relating to a Framework Service;

"Framework Services" means the services listed in <u>Schedule 2</u> (Framework Services), and "Framework Service" means any of them or the relevant one of them, as the context may require;

"Framework Term" means the period between the Effective Date and the later of:

- (a) the date on which the Initial Framework Term ends; and
- (b) the date on which the Framework Extension Term ends;

"Fugro Group" or "Fugro Group Companies" means Fugro and all Persons Controlled by Fugro from time to time, and "Fugro Group Company" means any one of them or the relevant one of them, as the context requires;

"Fugro Network" means global enterprise network services of the Fugro Group;

"Global" has the meaning as set out in recital (A) of this Agreement;

"GMSL Group" means Global Marine Holdings LLC and GMSL and all Persons Controlled by any of them from time to time, and "GMSL Group Company" means any one of them or the relevant one of them, as the context requires;

"GMSL" has the meaning as set out in preamble of this Agreement;

"Governmental Authority" means, to the extent it has jurisdiction in respect of the relevant matter, any judicial, legislative, executive, regulatory or competition authority of the Netherlands or any other jurisdiction, including of the European Union and the United States of America, or in each case any political subdivision thereof;

"Huawei JV" has the meaning as set out in recital (C) of this Agreement;

"Initial Framework Term" means the period between the Effective Date and the earlier of:

(a) three (3) years after the Effective Date of this Agreement; and

Schedule 1 Definitions and interpretation

(b) the date on which Fugro no longer directly or indirectly holds ten percent (10%) or more of the aggregate number of outstanding Class A Units, Class A-1 Units and Class A-2 Units of Global Holdings LLC;

"Law" means any applicable statute, law, treaty, ordinance, order, rule, directive, regulation, code, executive order, injunction, judgement, decree or other requirement of any Governmental Authority;

"Losses" means all damage, losses, costs (including reasonable legal costs and reasonable experts' and consultants' fees), charges, expenses, claims and demands assessed in accordance with section 6:96 et seq. of the Dutch Civil Code;

"Network Separation" means the separation of the Computer Systems from the Fugro Network;

"Notice" has the meaning set out in Clause 16.9 (Notices);

"Parties" means Fugro and GMSL, and "Party" means any one of them or the relevant one of them, as the context requires;

"Person" means an individual, a company, or corporation, a partnership, a limited liability company, a trust, an association, a foundation, or other legal entity or unincorporated organisation, including a Governmental Authority;

"Prevailing Clauses" has the meaning set out in Clause 3.1 (Master Framework Services Agreement);

"Provider Materials" means any and all systems, equipment, programs, software (object code and, if available, source code), documents, drawings, plans, designs, models, documentation, texts, manuals, reports, schemes, algorithms, analyses, tools, methods, parameter settings and configurations and process descriptions and all other works that already exist before the date of this Agreement or come into existence and/or are created, designed, developed and/or produced by a Provider in connection with or in the framework of the provision of a given Service;

"Provider" means the relevant Fugro Group Company, as provider of a Service under a Service Agreement;

"Receiving Party" has the meaning set out in Clause 15 (Confidentiality);

"Recipient" means the relevant GMSL Group Company, as recipient of a Service under a Service Agreement;

"Relationship Manager" has the meaning set out in Clause 11.1 (Relationship Manager);

"SBSS JV" has the meaning as set out in recital (C) of this Agreement;

"Schedule" means a schedule to this agreement;

"Service Agreement" means a Framework Service Agreement or a Transitional Service Agreement;

"Services" means the Framework Services and the Transitional Services, and "Service" means any of them or the relevant one of them, as the context may require;

"Tax Deduction" means any deduction or withholding for or on account of Tax;

"Taxation" or "Tax" means all forms of taxation whether direct or indirect and whether levied by reference to income, profits, gains, net wealth, asset values, turnover, sales and/or consumption, added value or other reference and statutory, governmental, state, provincial, local governmental or municipal impositions, duties, contributions, rates and levies (including without limitation social security contributions and any other payroll taxes), whenever and wherever imposed (whether imposed by way of a withholding or deduction or otherwise) and in respect of any person as well as all penalties relating thereto;

"Third Party Agreement" means any agreement with a Third Party Supplier for the provision of a service, lease or licence relating to, or necessary for, the provision of the Services whether entered into before or after the date of this Agreement;

"Third Party Consent" means any permission, consent, agreement or authorisation required by a Third Party Supplier under a Third Party Agreement for the provision of the Services by a Provider, or the receipt of the Services by a Recipient;

"Third Party Supplier" means any Third Party, including subcontractors, providing a service, lease or licence under a Third Party Agreement;

"Third Party" means any Person other than each Fugro Group Company and each GMSL Group Company;

"Transaction" has the meaning as set out in recital (A) of this Agreement;

"Transition Period" is defined in Clause 6.2(a);

"Transitional Service Agreement" means a Service Agreement in respect of a Transitional Service;

"Transitional Service Term" means, in relation to a Transitional Service, the term set out in the relevant Service Agreement;

Schedule 1 Definitions and interpretation

"Transitional Services" means the services set out in the agreed transitional service agreements included in <u>1</u> (Agreed Transitional Service Agreements) and "Transitional Service" means any of them or the relevant one of them, as the context may require;

"Trenching Business" has the meaning as set out in recital (A) of this Agreement; and

"VAT" means within the European Union such Tax as may be levied in accordance with (but subject to derogations from) the Principal VAT Directive 2006/112/EC (or any such equivalent legislation which amends or succeeds it) and outside the European Union any Tax levied by reference to added value, sales, goods and services, and/or consumption.

2 References to persons

References to:

- 2.1 a person include any individual, company, partnership or unincorporated association (whether or not having separate legal personality); and
- **2.2** a company include any company, corporation or any body corporate, wherever incorporated.

3 Headings and references to Clauses, Schedules and Paragraphs

3.1 Headings have been inserted for convenience of reference only and do not affect the interpretation of any of the provisions of this Agreement.

3.2 A reference in this Agreement to:

- (a) a Clause or Schedule is to the relevant Clause of or Schedule to this Agreement; and
- (b) a Paragraph is to the relevant Paragraph of the relevant Schedule.
- **3.3** Where the context permits, any reference to a gender includes all genders.

4 Information

References to books, records or other information include books, records or other information stored in any form including paper, magnetic media, films, microfilms, electronic storage devices and any other data carriers.

5 Legal terms

In respect of any jurisdiction other than the Netherlands, a reference to any Netherlands legal term shall be construed as a reference to the term or concept which most nearly corresponds to it in that jurisdiction.

6 Other references

- 6.1 Whenever used in this Agreement, the words 'include', 'includes' and 'including' shall be deemed to be followed by the phrase 'without limitation'.
- 6.2 Whenever used in this Agreement, the words 'as of' shall be deemed to include the day or moment in time specified thereafter.
- 6.3 Any reference in this Agreement to any gender shall include all genders, and words importing the singular shall include the plural and vice versa.
- 6.4 Any reference to an obligation on a "Provider" shall also be construed as an obligation on Fugro to ensure that the Provider performs that obligation.
- 6.5 Any reference to an obligation on a "Recipient" shall also be construed as an obligation on GMSL to ensure that the Recipient performs that obligation.
- 6.6 Any reference to "GMSL" or to "Fugro" in a provision of this Agreement that is incorporated into a Transitional Services Agreement or a Framework Services Agreement, shall, in the context of that Service Agreement, be construed as a reference to the Recipient or the Provider (respectively).

Schedule 1 Definitions and interpretation

Schedule 2 Framework Services

1 Survey services

Survey services, including:

- (a) geological and geophysical survey;
- (b) surface and subsea positioning services;
- (c) field development survey support services;
- (d) pile jacket installation;
- (e) site and route surveys;
- (f) survey support for infield inspection and pipeline condition services;
- (g) AUV surveys;
- (h) cable route surveys;
- (i) burial depth surveys;
- (j) DP signal services;
- (k) hydrography surveys;
- (I) environmental studies;
- (m) spool piece metrology;
- (n) meteorology and oceanography services;
- (o) weather forecasting;
- (p) SEEP surveys;
- (q) above the waterline structural and FPSO monitoring services; and
- (r) GIS data solution services Marine site investigation services.

2 Asset integrity services

Asset integrity services, including:

(a) ROV support vessel services;

- (b) Platform and pipeline inspection;
- (c) Flooded member detection surveys;
- (d) CP/CD surveys of pipelines and structures;
- (e) Leak investigation and remediation;
- (f) Pipeline stabilisation and repair;
- (g) Anode clamp design and retrofit;
- (h) Structure repair and monitoring;
- (i) Subsea component change-out; and
- (j) Depth of burial surveys.

3 ROV operation services

Asset integrity services, including:

- (a) ROV services on-board GMSL vessels;
- (b) Provision of work-class ROV services;
- (c) Provision of inspection class ROV services; and
- (d) Provision of eyeball ROV services.

4 Other services

Other services, including:

- (a) Weather forecasting;
- (b) Metocean services;
- (c) Design and supply of bespoke subsea tooling solutions;
- (d) Project engineering and management;
- (e) Design engineering and drafting;
- (f) Structural engineering; and
- (g) Simulation engineering and animation.

Schedule 3 Agreed Transitional Service Agreements

Project Furrow Service Agreement		

General				
Service	Office accommodation and general office support, Fugro House, Denmore Road, Bridge of Don, Aberdeen, AB23 8JW.			
Service Number (if applicable)	210-0917			
Business / Functional Area	Construction and Installation			
Region(s) / Country(ies) (or 'Global') Europe and West Africa				
The relevant terms of the Transitional and framework services agreement entered into by Fugro N.V. and Global Marine Services Limited on 30 November 2017 ("Transitional and Framework Services Agreement"), including, but without limitation, clauses 7 (Terms of Services), 8.1 (Termination for cause), 8.4 (Termination of Services for breach), 8.5 (Termination of Transitional Services), 8.6 (Force Majeure), 8.7 (Notice of termination), 9 (Remedies), 10 (Warranties and Liability), 12 (Intellectual Property and Use of Materials)], 13 (IT Security and Data), 14 (Separation from Seller's networks), 15 (Confidentiality), are incorporated by reference into this service agreement. Clauses 16 (Miscellaneous) and 17 (Governing Law and Dispute Resolution) of the Transitional and Framework Services Agreement apply mutatis mutandis to this agreement				

Services provided by: Fugro Subsea Services Limited, a company incorporated under the laws of Scotland, registration number SC105684 and having its registered and principle place of business at Fugro House, Denmore Road, Bridge of Don, Aberdeen, AB23 8JW

Services provided to: Global Marine Systems Limited, a company incorporated under the laws of England and Wales, company registration number 10708481 and having its registered office at Ocean House, 1 Winsford Way, Boreham Interchange, Chelmsford, Essex, United Kingdom, CM2 5PD

Services Description & Charges				
Services description Lease of office accommodation at Fugro House, Denmore Road, Bridge of Don, Aberdeen, AB23 8JW, inclusive of Building Rates, Utilities, Security, Cleaning, Reception Support (administration of mail, direction of calls/visitors etc.), reasonable level of office furniture; Rent/Access to telephone system and internet communications within Fugro House (inclusive of ICT Support);	Charges (per mo Invoice Currency	onth, in advance) Amount		
	GBP (£)	As per below		

Charges Details						
Services item reference	Cost unit (m2, kg, FTE, etc.)	Number of cost units		Price per cost unit (in invoice currency)	(during contract	Total Cost Per Month
Lease of office accommodation at Fugro House, Denmore Road, Bridge of Don (inclusive of Building Rates, Utilities, Security, Cleaning, Reception Support, reasonable level of office furniture;						£1,450.00 per month
Access to office telephone and internet communications at Fugro House (inclusive of ICT Support)						£675.00 per month
Access to Fugro Symphony telephone and internet communications (estimated cost per month is £2,500.00)						Recharged at actual documented cost per month

Storage Facility in Montrose (Trencher 1 - part items and Trencher 2 - full system), estimated cost per month being £4,930.00			Recharged at actual documented cost per month

Date of Commencement of Service	01 December 2017		
	Twelve (12) months from Date of Commencement of Services		

Miscellaneous / Note

Contact Details				
Contact person Fugro Subsea Services Limited		Contact person Global Marine Systems Limited		
Name	Derek Cruickshank	Name Ian Bryan		
Department	Managing Director	Department	Corporate Development	
Address	Fugro House Denmore Road Bridge of Don Aberdeen AB23 8JW	Address	Ocean House, 1 Winsford Way, Boreham Interchange, Chelmsford, Essex, United Kingdom, CM2 5PD	
Telephone number	01224 257674	Telephone number	01245 70200	
E-mail address	d.cruickshank@fugro.com	E-mail address	lan.Bryan@globalmarine.group	

Signed for Fugro Subsea Services Limited		Signed for Global Marine Systems Limited		
Name		Name		
Title		Title		
Date	30 November 2017	Date	30 November 2017	
Signature		Signature		

Schedule 4 Agreed Framework Service Agreements



Contract for the Provision of ROV Services on board COMPANY provided Vessels

Contract No. 213-0917

FORM OF AGREEMENT

This CONTRACT is made between the following parties:

Global Marine Systems Limited a company incorporated under the laws of England and Wales, company registration number 10708481 and having its registered office at Ocean House, 1 Winsford Way, Boreham Interchange, Chelmsford, Essex, United Kingdom, CM2 5PD hereinafter called the COMPANY

and

Fugro Subsea Services Limited a company incorporated under the laws of Scotland, company registration number SC105684 and having its registered office at Fugro House, Denmore Road, Bridge of Don, Aberdeen, AB23 8JW hereinafter called the CONTRACTOR.

WHEREAS:

- 1. the COMPANY and its AFFILIATES (as defined in Clause 1.1) wishes that certain WORK shall be carried out, all as described in the CONTRACT and pursuant to a duly executed CALL-OFF ORDER; and
- 2. the CONTRACTOR wishes to carry out the WORK in accordance with the terms of this CONTRACT and pursuant to a duly executed CALL-OFF ORDER.

NOW:

The parties hereby agree as follows:

- 1. In this CONTRACT, all capitalised words and expressions shall have the meanings assigned to them in this FORM OF AGREEMENT or elsewhere in the CONTRACT.
- 2. The following Sections shall be deemed to form and be read and construed as part of the CONTRACT:
 - 1. Section I Form of Agreement including Appendix 1;
 - 2. Section II a) General Conditions of Contract for Services Edition 2 October 2003;

b) Special Conditions of Contract;

- 3. Section III Remuneration;
- 4. Section IV Scope of Work;
- 5. Section V Health, Safety and Environment;
- 6. Section VI Company's General Obligations;
- 7. Section VII Call-Off Order Template.

Page 1 of 30



The Sections shall be read as one document, the contents of which, in the event of ambiguity or contradiction between Sections, shall be given precedence in the order listed, with the exception that the Special Conditions of Contract shall take precedence over the General Conditions of Contract.

- 3. In accordance with the terms and conditions of the CONTRACT, the CONTRACTOR shall perform and complete the WORK and the COMPANY or its AFFILIATES shall pay the CALL-OFF ORDER price.
- 4. The terms and conditions of the CONTRACT shall apply from the date specified in Appendix I to this Section I Form of Agreement which date shall be the EFFECTIVE DATE OF COMMENCEMENT OF THE CONTRACT.
- 5. The duration of the CONTRACT shall be as set out in Appendix I to this Section I Form of Agreement.
- 6. The COMPANY or its AFFILIATE shall call-off WORK as and when required during the term of the CONTRACT by issuing a CALL-OFF ORDER. CONTRACTOR shall either: (i) notify COMPANY or its AFFILIATE of acceptance of the CALL-OFF ORDER, either by countersigning the CALL-OFF ORDER, in accordance with the notification provisions of the CONTRACT or by e-mail; or (ii) reject the CALL-OFF ORDER and/or propose modifications to the CALL-OFF ORDER. COMPANY or its AFFILATE shall consider any proposed modifications to the CALL-OFF ORDER, and if agreed, issue a revised CALL-OFF ORDER for acceptance by CONTRACTOR. A CALL-OFF ORDER will not become binding or form part of the CONTRACT until signed in writing by both COMPANY and CONTRACTOR.

The authorised representatives of the parties have executed the CONTRACT in duplicate upon the dates indicated below:

For: For:

Global Marine Systems Limited Fugro Subsea Services Limited

Name: Name:

Title: Title:

Date: Date:

Page 2 of 30



APPENDIX I TO SECTION I - FORM OF AGREEMENT

Reference

Section I

Clause 4 The EFFECTIVE DATE OF COMMENCEMENT OF THE CONTRACT is

1st December 2017;

Clause 5 The duration of the CONTRACT is twelve (12) months from the EFFECTIVE DATE OF COMMENCEMENT OF THE CONTRACT. The term of the CONTRACT may be extended for a further period of twelve (12) months by the mutual agreement of both COMPANY and CONTRACTOR no less than ninety (90) days prior to expiry of the initial CONTRACT period.

Section II

- Clause 3.1(a) The COMPANY REPRESENTATIVE is Paul Thomson or as identified within the applicable CALL-OFF ORDER. The CONTRACTOR REPRESENTATIVE is, as identified in the applicable CALL-OFF ORDER.
- Clause 5.1 The COMPANY designated heliport is Dyce, Aberdeen; The COMPANY designated supply base is Fugro House, Denmore Road, Bridge of Don, Aberdeen, AB23 8JW;
- Clause 10.2(b) The Defects Correction Period is: CONTRACTOR GROUP's Defects Correction (to reacquire data and/or rectify defective WORK) shall cease immediately upon CONTRACTOR GROUP's demobilisation and departure from the offshore WORKSITE as detailed in the relevant CALL-OFF ORDER.
- Clause 13.8 The period of suspension is seven (7) days.
- Clause 14.3 Latest time for receipt of invoices ninety (90) days commencing at the agreed date at which the WORK was completed.
- Clause 14.9 Interest rate per annum Base Rate plus three (3) per cent p.a.
- Clause 17.4 Rights shall vest in CONTRACTOR.
- Clause 19.2(d) This indemnity is given in respect of the following property and is subject to any exclusions or limitations specified below: (i) Property directly affected by the WORK, all as identified within Clause 19.2 d) of Section II B Special Conditions

of Contract.

- (ii) Other property, all as identified within Clause 19.2 (d) of Section II Special Conditions of Contract.
- (iii) For the purposes of Clause 19.2(d) only, consequential losses shall have the same meaning as in Clause 21
- Clause 20.2 Insurance by the CONTRACTOR, the amounts are Employers' Liability: UK Sterling fifteen million pounds sterling (£15,000,000) anyone occurrence.

Page 3 of 30



General Third Party: EURO ten million (EUR€10,000,000) per occurrence combined single limit for property damage and/or bodily injury, EURO twenty million (EUR€20,000,000) for products liability in the annual aggregate.

Clause 24.7(a) (iii) Special Conditions remaining in full force and effect shall be – All.

- Clause 28.6 The addresses for the service of notices are: (i) COMPANY (ii) CONTRACTOR (iii) CONTRACTOR (ii) CONTRACTOR
- Clause 29.1(a) Limitation of Liability before the date of completion of the WORK The sum is limited to 100% of the relevant CALL-OFF ORDER PRICE.
- Clause 29.1(b) Limitation of Liability after the date of completion of the WORK The sum is limited to 100% of the relevant CALL-OFF ORDER PRICE.
- Clause 29.2 The Limitation Period shall cease upon expiry of the Defects Correction Period identified in Clause 10 herein.
- Clause 30.1(b) Resolution of Disputes. The nominees are: (i) COMPANY XXX (iii) CONTRACTOR: Derek Cruickshank, Managing Director.

Page 4 of 30



SECTION II (A) LOGIC GENERAL CONDITIONS OF CONTRACT FOR SERVICES (ON- AND OFF-SHORE). EDITION 2 -OCTOBER 2003

Page 5 of 30



SECTION II (B) SPECIAL CONDITIONS

The following amendments shall be made to the LOGIC General Conditions of Contract for Services (On and Off-shore).

1. Definitions

1.2. Insert new definition as follows: "CLIENT shall mean any party (together with its CO-VENTURER's in any relevant hydrocarbon, exploration and/or production licence), with whom the COMPANY has contracted and for whose ultimate benefit the WORK hereunder is provided".

Renumber all existing clauses accordingly.

1.3. Definition of COMPANY GROUP: Add ", and the CLIENT from time to time," after "shall mean the COMPANY". Add "it's other contractors and subcontractors (of any tier)" after "CO-VENTURERS" in the definition of COMPANY GROUP.

1.16. Insert new definition as follows "CALL-OFF ORDER" shall mean the document in the form set out in Section VII Call-Off Order Template which incorporates the different parts of the CONTRACT and which when completed and signed will detail the specifics of the WORK to be carried out".

1.17. Insert new definition as follows "CALL-OFF ORDER PRICE" shall mean the sum(s) of money in the relevant CALL-OFF ORDER being payment for the satisfactory performance of the WORK by CONTRACTOR and as may be adjusted from time to time in accordance with the CONTRACT".

2. Interpretation

Insert a new clause 2.4 as follows:

2.4. Where approval or agreement is required by either Party under this CONTRACT, such approval shall not be unreasonably withheld, conditioned or delayed and where instructions are requested in writing from the COMPANY written instructions shall be provided within a reasonable time so as not to delay the WORK.

4. CONTRACTORS general obligations

In 4.2 add "But for the avoidance of doubt, the CONTRACTOR warrants that the results of the WORK will be fit for the purposes as defined in the applicable CALL-OFF ORDER and the respective specification".

In 4.4 add "However, notwithstanding anything to the contrary included elsewhere in the CONTRACT, breakdown of any equipment utilised by CONTRACTOR GROUP in performance of the WORK shall not constitute a breach of CONTRACT and COMPANY GROUP's remedy in respect thereof shall be limited to suspension of CONTRACTOR's relevant hire charges as further identified in Section III Remuneration. Provided always that the CONTRACTOR shall be allowed two (2) hours maintenance allowance per day up to a cumulative total of twelve (12) hours for maintenance and repair of its ROV spread during which time the CONTRACTOR's equipment and associated personnel shall be deemed on standby at the relevant hire charges as further identified in Section III Remuneration."

Further, CONTRACTOR'S ROV Supervisor shall have the authority to decline to deploy the ROV System if conditions are such that to do so may, in the CONTRACTOR'S ROV Supervisor's sole opinion, present an unacceptable risk to the safety of personnel or equipment and shall have the authority to decide if and when to recover the ROV System in the face of marginal weather with a worsening forecast. Additionally, the CONTRACTOR'S ROV Supervisor shall have the authority to decide if authority to decline to endeavour to access any area where, in the CONTRACTOR'S ROV Supervisor's opinion to do so may risk entrapment or loss of the

Page 6 of 30



vehicle. In any of the above cases, CONTRACTOR shall not be liable or penalised in any way and relevant day rates as further identified in Section III Remuneration shall continue to apply

6. CONTRACTOR to inform himself

6.1. Add at the beginning of the Clause "Subject to Clause 7.4"

7. CONTRACTOR to inform COMPANY / COMPANY to inform CONTRACTOR

7.3. Add at the end of the second paragraph "to the extent permitted by law".

7.4. Add: "Such information shall include but not be limited to: (i) the presence, type and precise location of facilities, obstructions and hazards on or above or below or within 500m of the WORKSITE that may affect the efficiency or safety of the WORK. Such facilities, obstructions, and hazards may include, but not be limited to: subterranean and sub aqueous structures, cables, pipelines, tunnels and utilities, explosive ordnance, fishing-gear, cavities, over- and under-pressurised groundwater, and shallow gas; (ii) the WORK Geodetic system for horizontal and vertical control.

8. ASSIGNMENT AND SUBCONTRACTING

8.2 a): Insert "except to an Affiliate" after "WORK" in second sentence.

9. CONTRACTOR PERSONNEL Insert new Clause 9.9 as follows:

9.9. from the EFFECTIVE DATE OF COMMENCEMENT OF THE CONTRACT until twelve (12) months after the CONTRACTOR completes the WORK, the COMPANY GROUP shall not directly or indirectly solicit the employment of, or interfere with or endeavour to entice away from the CONTRACTOR or endeavour to obtain the employment or engagement of, any person who has or had been allocated or assigned to the performance of the WORK by the CONTRACTOR provided that this prohibition shall not apply where any such person responds unsolicited to advertisements placed in the media by COMPANY.

10. Examination and defects correction

10.2. Add the word "material" before "defect" on each occasion it appears throughout the clause.

10.2 (b) At the end of sub-clause add "For the avoidance of doubt, CONTRACTOR's liability to reacquire data or rectify defective WORK shall cease immediately upon CONTRACTOR's departure from the offshore WORKSITE".

10.2.(c) Delete the first sentence and at the start of the second sentence insert "If the CONTRACTOR fails to remedy any defect within a reasonable period of time of being notified by COMPANY, such notification to be provided by COMPANY prior to CONTRACTOR's demobilisation and departure from the WORKSITE,

Add to Clause 10.2(d) "Notwithstanding anything to the contrary included elsewhere in this CONTRACT, CONTRACTOR shall not be liable for defects caused by CONTRACTOR's reliance on TECHNICAL INFORMATION or any other information, instruction or direction provided by the COMPANY".

Page 7 of 30



Add new Clause 10.2 (e) as follows:

10.2 (e)"Notwithstanding anything to the contrary included elsewhere in the CONTRACT, CONTRACTOR's liability for error or mistake in the performance of the WORK shall be limited only to CONTRACTOR's charges relating to the WORK executed erroneously in proportion to the extent of the error or mistake or, where feasible and reasonable, the re-execution of such offshore WORK at no further charge to COMPANY, subject to CONTRACTOR being notified of any error or mistake prior to CONTRACTOR's demobilisation and departure from the offshore WORKSITE following final scheduled WORK activities for that phase of the WORK.

Notwithstanding the above, if a component part of the CONTRACTOR's equipment spread is not functional but WORK can continue either with the scheduled task or other activities required to progress the WORK, this shall not be considered downtime or breakdown".

Insert new clause 10.3 as follows:

10.3 "Notwithstanding anything to the contrary the foregoing warranties shall be exclusive and in lieu of any other warranty whether express or implied or otherwise at law".

Insert new Clause 10.4 as follows:

10.4 "CONTRACTOR shall ensure regular quality checks are performed in order to confirm the accuracy and integrity of any data being acquired by CONTRACTOR. The CONTRACTOR shall raise any anomalies or inconsistencies in the data acquired by it to the COMPANY without delay".

11. Variations

At the end of Clause 11.2 add "by agreement between the parties".

Insert new clause 11.3 as follows:

"11.3 CONTRACTOR shall be entitled to a VARIATION and a consequent adjustment to the CONTRACT PRICE and SCHEDULE, if the COMPANY, through its instructions, actions or lack of actions, has varied the scope of the WORK or the time for performance of the WORK or if the COMPANY has failed to provide unobstructed access to the WORKSITE."

12. Force Majeure

12.2 Amend opening words to read "For the purposes of this CONTRACT, force majeure shall include but not be limited to:"

Insert new clause 12.6 as follows:

12.6 In the event that any Force Majeure occurrence lasts longer than seven (7) days, COMPANY may instruct CONTRACTOR to remain on standby at a safe location, in which case COMPANY shall bear the associated standby costs set forth in Section III Remuneration. If standby option is not exercised by COMPANY, either COMPANY or CONTRACTOR can terminate the PURCHASE ORDER. Having elected to retain CONTRACTOR on standby and the delay due to FORCE MAJEURE prolongs, COMPANY can terminate the PURCHASE ORDER. If the PURCHASE ORDER is terminated through such FORCE MAJEURE event, COMPANY shall pay CONTRACTOR all outstanding sums due for the WORK completed to date of FORCE MAJEUE occurrence. Upon cessation of any FORCE MAJEURE occurrence the Parties may agree a revised programme to include for rescheduling of the WORK so as to minimise the effects of the delay which will be subject to CONTRACTOR'S other contractual commitments. In such event COMPANY shall reimburse CONTRACTOR for the costs of re-mobilisation.

Page 8 of 30



13. Suspension

Insert the word 'material' before the word 'default' wherever appearing

13.5. Add at the end "Additional costs for which the CONTRACTOR is liable under this clause shall be documented and reasonable".

13.8: Change "fourteen (14) days" to "seven (7) days" on both occasions

Insert new clauses 13.9 - 13.10. as follows:

13.9. CONTRACTOR shall have the right, upon notice to the COMPANY, to suspend the WORK or any part thereof to the extent detailed in the notice, for failure of COMPANY to meet its obligations under this CONTRACT, including but not limited to failure to make payment of the amounts due within 30 days of notice from CONTRACTOR that amounts are overdue and non-fulfilment of the obligation to inform CONTRACTOR of hazards pursuant to clause 7.4;

13.10. COMPANY agrees that CONTRACTOR shall not be liable to COMPANY for, and indemnifies CONTRACTOR from and against, any loss or damage or other liability suffered or incurred by COMPANY arising out of or in connection with suspension by CONTRACTOR under clause 13.9. In addition, and for the avoidance of doubt, Clause 13.4 shall continue to apply in the event of suspension in accordance with Clause 13.9.

18. Laws and regulations

18.2: Delete and replace with; "CONTRACTOR shall, in due time, obtain and maintain such approvals and permits as are necessary for the performance of the WORK and which must be obtained in the name of CONTRACTOR. COMPANY shall provide any necessary assistance in this connection.

COMPANY shall, in due time, obtain and maintain all other approvals and permits, such as the right to enter into and exit from the WORKSITE.

COMPANY shall advise CONTRACTOR of any limitations or restrictions affecting the right of entry to the WORKSITE, and CONTRACTOR shall abide by such limitations or restrictions, but CONTRACTOR shall not be liable for failure to comply with the terms of any contract or permit of COMPANY of which CONTRACTOR is not aware and could not reasonably be expected to have known about. "

19. Indemnities

Add new Clause 19.1 (d) as follows:

Any such claims, losses, damages, costs (including legal costs) expenses and liabilities that may be brought by or in respect of any personnel employed by the CONTRACTOR against COMPANY pursuant to any loss of employment of such personnel with the CONTRACTOR.

Delete clause 19.2(d) and replace with "loss of, damage to or recovery of any third-party infrastructure within 500m of the WORKSITE including but not limited to oil and gas and telecommunications infrastructure and consequential losses arising therefrom where such loss, damage or recovery arise from or in connection with the performance or non-performance of the CONTRACT. This Clause 19.2(d) shall apply notwithstanding Clause 19.1(c)."

Page 9 of 30



20. Insurance by Contractor

20.1 Insert at the end of the clause the following "All insurance provided by COMPANY GROUP shall be endorsed to provide that underwriters waive any rights of recourse, including in particular subrogation rights against the CONTRACTOR GROUP in relation to the CONTRACT to the extent of the liabilities assumed by the COMPANY GROUP under the CONTRACT".

Insert New Clause 20.5 as follows:

CONTRACTOR's rates and pricing included within Section III Remuneration do not make allowance for additional insurance premiums required to extend insurance in relation to interaction with unexploded ordnances (UXOs). If any such operation is required, CONTRACTOR shall be entitled to a VARIATION to reflect additional insurance premium/s, if any. In the event that CONTRACTOR's insurer declines to extend insurance in respect of UXO interaction, CONTRACTOR and COMPANY shall discuss whether to proceed and if so, make an appropriate reapportionment of risk".

24. Termination

Insert the word "material" before "default" throughout this provision

24.4. Insert at the end of the Clause, the following "In addition to the above, in the event of such termination the COMPANY shall reimburse CONTRACTOR the applicable cancellation fee as further detailed within Section III Remuneration."

Insert a new clause 24.8 as follows:

24.8. The CONTRACTOR shall have the right by giving notice to terminate the CONTRACT and/or any CALL-OFF ORDER for any or all of the following reasons:

a) in the event of a material default on the part of the COMPANY. Failure by COMPANY to make payment of CONTRACTOR's correctly prepared and adequately supported invoices within forty-five (45) days of such invoice due date shall be deemed to be a material default hereunder; or

b) in the event of the COMPANY becoming bankrupt or making a composition or arrangement with its creditors or a winding-up order of the COMPANY being made or (except for the purposes of amalgamation or reconstruction) a resolution for its voluntary winding-up being passed or a provisional Liquidator, Receiver, Administrator or Manager of its business or undertaking being appointed or presenting a petition or having a petition presented applying for an administration order to be made pursuant to Section 9 Insolvency Act 1986, or possession being taken by or on behalf of the holders of any debenture secured by a Floating Charge of any property comprised in or subject to the Floating Charge, or any equivalent act or thing being done or suffered under any applicable law".

25. AUDIT AND STORAGE OF DOCUMENTS

25.4: Please change "six (6) years" to "two (2) years"

26. Liens

26.1. To be deleted.

Page 10 of 30



33. WAR RISK INSURANCE

"CONTRACTOR insurance coverage extends worldwide, but in the event of personnel, a vessel or craft insured sailing for, deviating towards, or being within the Territorial Waters of any of the Countries or places described in the Current Exclusions as set out in the attached notice (or current version thereof), CONTRACTOR must notify its insurers who will advise if an additional premium will be applicable, or, indeed, if insurance is withdrawn. Where this occurs during the currency of a contract, CONTRACTOR shall be reimbursed the amount of such additional premium at cost. In the event that insurance is withdrawn, clearly CONTRACTOR would have discussions with the COMPANY concerned as a decision would be required on whether to proceed".

34. LIQUIDATED DAMAGES

"Liquidated Damages shall be agreed on a case by case basis and detailed in the relevant CALL-OFF ORDER and shall apply only when COMPANY GROUP suffers loss as a consequence of later performance by CONTRACTOR GROUP. Notwithstanding anything included elsewhere in the CONTRACT to the contrary, the amount of such Liquidated Damages shall accrue at a rate of zero-point one percent (0.1%) of the CONTRACTOR's vessel day rate per day, as further identified in Section III Remuneration herein, and shall be subject to an aggregate and/or cumulative cap of a sum equal to ten percent (10%) of the relevant CALL-OFF ORDER value. Such Liquidated Damages shall be the COMPANY GROUP's sole and exclusive remedy and CONTRACTOR GROUP's sole and exclusive liability in respect of delay in the late commencement, performance or completion of the WORK under this CONTRACT".

Page 11 of 30



SECTION III REMUNERATION

1.0 GENERAL

1.1 The terms and conditions set forth in this Section III Remuneration specify the basis for compensating CONTRACTOR for WORK which CONTRACTOR shall provide to COMPANY or its AFFILIATE under this CONTRACT and an agreed CALL-OFF ORDER. The rates, prices and principles set out in this Section III Remuneration shall be applied to define the CALL-OFF ORDER PRICE and paid to CONTRACTOR as a result of the satisfactory execution of the WORK in accordance with the requirements of the CONTRACT and the agreed CALL-OFF ORDER.

1.2 The rates and prices specified herein shall be deemed as all-inclusive to allow for all costs incurred by the CONTRACTOR in complying with the terms and conditions of the CONTRACT and the agreed CALL-OFF ORDER, and shall represent the total remuneration due to the CONTRACTOR in full consideration of the satisfactory performance of the WORK performed.

2.0 BASIS OF RATES AND PRICES

2.1 The rates and prices set forth herein are fully inclusive of all costs (including both direct and indirect costs), management and administrative costs, overhead and profit and shall apply equally to performing the WORK in large or small quantities, in any sequence, or in difficult or easy situations and whether the WORK is performed by the CONTRACTOR or its SUBCONTRACTORS.

2.3 The currency of the agreement is pounds sterling (GBP (\pounds)). All lump sum amounts, unit rates, and other pricing shall be as detailed and all invoices prepared, submitted and paid in pounds sterling (GBP (\pounds)).

3.0 BASIS OF RATES

3.1 The rates set forth herein shall include but not be limited to, the cost of salaries and wages actually paid to personnel, cost of all CONTRACTOR's payroll burdens, benefits and contributions, all taxes (except VAT) and insurance requirements in relation to working at CONTRACTOR offices or any other worksite and in full compliance with all terms and conditions of the CONTRACT to carry out the WORK as described in Section IV Scope of Work and the agreed CALL-OFF ORDER.

3.2 Value Added Tax is excluded from the rates, sums and percentages herein. CONTRACTOR shall obtain for the benefit of COMPANY all available exemption from import duties, investment tax, Value Added Tax and all similar taxes and shall pass on to COMPANY any and all rebates.

4.0 COMPENSATION

4.1 CONTRACTOR shall be compensated for the satisfactory performance of the WORK in accordance with the rates, sums and prices contained within the rate tables included below. Any additional Associated Equipment or project specific equipment shall be priced on a case by case basis and identified within the applicable CALL-OFF ORDER:

Page 12 of 30



ROV Systems, Associated Equipment and Personnel

Project Management

Item	Price (GBP)
Project Management	TBC

Pricing Notes:

٠

CONTRACTOR'S Project Management Lump Sum shall be identified on a case by case basis and included within the applicable CALL-OFF ORDER;

Mobilisation and Demobilisation of CONTRACTOR ROV Systems and Associated Equipment.

Item	Price (GBP)
Transportation of ROV System, Associated Equipment and necessary materials from CONTRACTOR's base in Aberdeen to COMPANY's nominated Vessel mobilisation Quayside *, **	Cost + 10%
Transportation of ROV System, Associated Equipment and necessary materials from COMPANY's nominated Vessel demobilisation Quayside to CONTRACTOR's base in Aberdeen *, **	Cost + 10%
Installation of CONTRACTOR's ROV System and Associated Equipment on board COMPANY nominated Vessel ***	Cost + 10%
Demobilisation of CONTRACTOR'S ROV System and Associated Equipment from COMPANY nominated Vessel ***	Cost + 10%

Pricing Notes:

- * CONTRACTOR's transportation fee for the ROV System and Associated Equipment includes the following:
- Pre-transportation checks/re-delivery checks;
- Certifying authority release note in Aberdeen;
- Manifest and Loading;
- Delivery/redelivery of ROV System to/from CONTRACTOR's base in Aberdeen and COMPANY's nominated mobilisation/demobilisation
 Quayside;

** CONTRACTOR's transportation fee for the ROV System and Associated Equipment excludes the following:

- All port services;
- Quayside and Vessel craneage;
- Engineering deck calculations/sea-fastening;
- · Provision of welders plus load testing services and equipment;

Page 13 of 30



As the COMPANY Vessel and its port of Mobilisation/Demobilisation has not yet been identified at the time of writing, CONTRACTOR advises that the above rates and prices shall be compensated on a cost + 10% basis. Firm lump sum prices can be provided on request and subject to COMPANY providing CONTRACTOR with additional required information and in such event the applicable CALL-OFF ORDER will be updated to capture such position.

*** COMPANY shall free issue quayside services required for installation of the CONTRACTOR ROV System on board COMPANY'S Vessel, or as an alternative CONTRACTOR shall provide such installation services on a Cost + 10% basis whereby the applicable CALL-OFF ORDER shall be updated accordingly.

Mobilisation and Demobilisation of CONTRACTOR Personnel

ltem	Price (GBP)
Personnel Mobilisation, per person *	Cost + 10%
Personnel Demobilisation, per person *	Cost + 10%

Pricing Notes: * CONTRACTOR's Personnel Mobilisation and Demobilisation fee includes the following:
 Transport to/from COMPANY's nominated Vessel Mobilisation/Demobilisation Port from CONTRACTOR Personnel original point of origin.

Daily Hire Rates of CONTRACTOR ROV System and Associated Equipment

FCV 3000 WROV ROV System

Item	Day Rate (GBP)
FCV 3000 Operational Day Rate	£1,400
FCV 3000 Standby Day Rate	£1,400
FCV 3000 Unmanned Standby / Transit Day Rate	£900

FCV 600 WROV ROV System

Item	Day Rate (GBP)
FCV 600 Operational Day Rate	£800
FCV 600 Standby Day Rate	£
FCV 600 Unmanned Standby / Transit Day Rate	£

Page 14 of 30



Observation (ObsROV) ROV System

Item	Day Rate (GBP)
ObsROV Operational Day Rate	£
ObsROV Standby Day Rate	£
ObsROV Unmanned Standby / Transit Day Rate	£

Associated Equipment (WROV ROV System)

Item	Day Rate (GBP)
H4 Tree H4 Mandrel Type VX Gasket Profile Clean-Up Tool	£25
Tool Deployment Unit (TDU)	£
Linear Override Tool	£
Fluid Skid c/w Pump	£
Regan Latch Release Tool	£13
Water Jetting Unit	£40

Associated Equipment (ObsROV ROV System)

Item	Day Rate (GBP)
Inspection Suite (Computers, Laptops, Hard Drives, DVR System)	£100
Video Overlay	£20
Contact CP Probe	£20
Cygnus Wall Thickness Measurement Tool c/w Probe Handler (UT Probe)	£35
Cleaning Kit – Includes Wire / Nylon Brush Cleaning & Cheese-wire	£
Option – 1 x Spare DVR System – (1 x already included in inspection suite)	£90

Pricing Notes:

- For the avoidance of doubt, CONTRACTOR's hire Day Rates for the ROV System and Associated Equipment shall commence immediately upon departure from CONTRACTOR's base in Aberdeen and shall apply for each and every day until return to same and will not be prorated for part days. The ROV System and Associated Equipment shall remain on hire during periods of non-utilisation of the COMPANY Vessel;
 - For clarification purposes, the ROV System Transit Day Rate will apply from departure of the ROV System from CONTRACTOR's base in Aberdeen until commencement of installation of the ROV System on board COMPANY Vessel and from completion of demobilisation of the ROV System from COMPANY Vessel until return to CONTRACTOR's base in Aberdeen;

Page 15 of 30



- For clarification purposes, the ROV System Operational and Standby Day Rate will apply from commencement of installation of the ROV System on board the COMPANY Vessel, during all times the ROV System and CONTRACTOR Personnel are on board the COMPANY Vessel and during ROV System demobilisation from the COMPANY Vessel, ceasing immediately upon completion of ROV System demobilisation from COMPANY Vessel;
- For clarification purposes, the ROV System Unmanned Standby Day Rate shall apply during all periods when CONTRACTOR Personnel are not on-board COMPANY's Vessel;
- Should the ROV System mobilisation date be delayed by COMPANY prior to the ROV System departing CONTRACTOR's base in Aberdeen, the Unmanned Standby Day Rate shall apply until such time as the ROV System departs CONTRACTOR's base in Aberdeen. Any such delays exceeding a period of seven (7) days shall be subject to the mutual agreement of both COMPANY and CONTRACTOR;
- Any cable interconnection between CONTRACTOR'S ROV System and the COMPANY Vessel Survey Spread shall be installed by COMPANY's Surveyor/Survey Engineer at COMPANY cost;
- COMPANY to provide, at no charge/cost to CONTRACTOR, Subsea Navigation System, Nav screens, Round Robin and clear communications;

Daily Hire Rates of CONTRACTOR Personnel

Discipline	Day Rate (GBP)
ROV Supervisor, per person	£
ROV Senior Pilot Technician, per person	£
ROV Pilot Technician, per person	£
Inspection Engineer, per person	£

Pricing Notes:

- For the avoidance of doubt, CONTRACTOR's hire Day Rates for Personnel shall commence immediately upon CONTRACTOR's Personnel departure from their original point of origin and shall apply in full for everyday until return to same.
- Except where otherwise agreed within the applicable CALL-OFF ORDER, CONTRACTOR Personnel crew changes for the WORK shall be on twenty-eight (28) day rotation with overlap charges applicable on days where both CONTRACTOR Personnel are simultaneously on board the COMPANY Vessel, travelling offshore/returning or standing by at the CONTRACTOR crew change mobilisation point. A mobilisation/demobilisation fee will be charged per person per direction of travel;

Page 16 of 30



CONTRACTOR Reporting Lump Sums

Reporting Lump Sum	Day Rate (GBP)
Reporting Lump Sum	TBC

Pricing Notes:

• CONTRACTOR's Reporting Lump Sum shall be identified on a case by case basis and included within the applicable CALL-OFF ORDER;

Additional CONTRACTOR's pricing notes applicable to all of the above rates, sums and prices tables under this item 4.0 COMPENSATION of Section III Remuneration;

- The above rates, sums and prices shall remain valid up to and including 31st December 2018, thereafter the above rates, sums and prices shall be subject to escalation by the mutual agreement of both COMPANY and CONTRACTOR;
- The above rates, sums and prices are based strictly on WORK being performed within the UKCS only. Should COMPANY request WORK out with the UKCS, CONTRACTOR reserves the right to adjust the above rates, sums and prices accordingly;
- The above rates, sums and prices are exclusive of VAT;
- In the event of CONTRACTOR ROV System breakdown, subject to CONTRACTOR's ROV System maintenance allowance and should CONTRACTOR ROV System be unavailable to perform meaningful WORK then CONTRACTOR's ROV System, associated personnel and associated equipment shall be deemed off-hire until such time as the ROV System is again available to commence meaningful WORK;
- For the avoidance of doubt, CONTRACTOR's hire Day Rates apply per each twenty-four (24) hrs period, from midnight to midnight;

COMPANY PROVIDED ITEMS

The COMPANY shall provide the following, at no charge/cost to CONTRACTOR:

For CONTRACTOR ROV System and Associated Equipment;

- Provision of a Vessel with acceptable position keeping and deck area of suitable size and position to allow safe ROV System deployment/recovery. Actual
 arrangement to be established;
- Facilities and services necessary for installation/removal of CONTRACTOR's ROV System and Associated Equipment to/from COMPANY supplied Vessel
 including, where relevant, structural calculations & sea-fastening design, quayside cranes, dock labour, welding, MPI and load-testing services plus any
 engineering fabrication & materials that may be required for any vessel-specific installation requirements such as load spreading beams, mezzanine deck,
 crane pedestal, grillage, etc;

Page 17 of 30



- Suitable, clean, stable motor rated power supplies for the operation of CONTRACTOR'S ROV System and Associated Equipment, meeting the
 requirements specified in the attached applicable Equipment Spec Appendices and with access to deck power within twenty (20) meters of the
 CONTRACTOR'S ROV System. If this is not available or cannot be guaranteed, then the CONTRACTOR will provide a suitable diesel power generator at
 documented cost + 10%.
- · Cooling water, fresh water and compressed air supplied to the CONTRACTOR's ROV System and Associated Equipment;
- Waste disposal;
- Hardware and wiring for connection of CONTRACTOR Equipment to COMPANY's Vessel internal communications;
- Acoustic Positioning system for ROV System tracking, including display's to be installed in the ROV System control container and any associated cabling;
- Remote Video Monitors from ROV System to Bridge, Control Room, On-line Area, offices, etc. including associated cabling;
- Interfacing to/from any Survey/Positioning logging systems, including associated cabling;
- All support flights to/from COMPANY's Vessel for all CONTRACTOR ROV System, Associated Equipment and Spares, when required, along with supply vessels for such equipment shipment;

For CONTRACTOR Personnel;

- All support flights to/from COMPANY's Vessel for all CONTRACTOR Personnel;
- Accommodation subsistence / victualling for CONTRACTOR Personnel while on-board COMPANY Vessel's (including periods of installation of ROV System on board COMPANY Vessel);
- Reasonable business communications from COMPANY Vessel to shore and UK office for CONTRACTOR Personnel. (Access to internet & email system from ROV control container);
- Emergency Medical Treatment and Medivac to approved Hospitals;
- In the event that offshore transportation for CONTRACTOR Personnel is delayed or cancelled six (6) hrs or less prior to the arrival of CONTRACTOR's Personnel at COMPANY nominated quayside or heliport, a cancellation fee of £250.00 per person and the applicable Personnel hire Day Rates shall apply;
- Should the delay of transportation mean that additional overnight accommodation is required for CONTRACTOR Personnel, all such costs shall be recharged to COMPANY at cost + 10% and shall apply over and above the mobilisation lump sum and Personnel hire Day Rates;
- If a twelve (12) hr ROV team is required by COMPANY note that only twelve (12)hrs will be worked per day by CONTRACTOR Personnel including dive
 operations and maintenance;

CONTRACTOR STANDARD/TYPICAL ROV EQUIPMENT SPECIFICATION

Unless otherwise agreed within the applicable CALL-OFF ORDER, CONTRACTOR's standard/typical ROV System specification shall be as follows:

Page 18 of 30



FCV 3000C Key Specifications:

Depth Rating	3000msw
Hydraulic HPU	150hp (225bar/217lmp
Tooling HPU	210bar/70lpm
Speed	3.2knots
Manipulators	Schilling Rigmaster, Schilling Orion
Through Frame Lift	3000kg + 400kg payload.
Sensors Capabilities	12 Camera channels, 30+ communication channels, 10 dimmable LED channels.

• For full FCV 3000 WROV ROV System specification please refer to attached Appendix 1 – FCV 3000 (150hp)_Rev2 2016_LR;

FCV 600 Key Specifications:

Depth Rating	600msw
Hydraulic HPU	125 hp (210 bar/217 lmp)
Tooling HPU	210 bar/90 lpm
Speed	3.2 knots
Manipulators	Schilling Rigmaster, Schilling Orion
Through Frame Lift	1,000 kg + 250 kg payload.
Sensors Capabilities	12 Camera channels, 30+ communication channels, 10 dimmable LED channels.

• For full FCV 600 WROV ROV System specification, please refer to attached Appendix 2 – FCV 600_Rev2 2016_LR;

Observation ROV Key Specifications:

<u>Seaeye Lynx</u>

Depth Rating	1500msw	
Hydraulic HPU	N/A	
Tooling HPU	N/A	
Speed	3.0 knots	
Tooling Options	Tooling Skid Options comprising, 5F Manipulator Skid, FMD Orientation Skid, Cleaning Brush Skid, Cable Cutting Skid and High-Pressure Water Jetting Skid	
Pay Load	34kg	
Sensors Capabilities	3 simultaneous Video Channels + 1 switchable, 1 x lowlight monochrome, 1 x colour, 1 x colour tooling Electrical Tilt unit +/- 90 degrees, 4 x dimmable LED (2 mounted under tilt unit)	

Page 19 of 30



• For full Seaeye Lynx ROV System specification, please refer to attached Appendix 3 – Seaeye Lynx_Rev1_LR;

Seaeye Tiger

Depth Rating	600msw or 1000msw
Hydraulic HPU	N/A
Tooling HPU	N/A
Speed	Fore/aft 3.0 knots / Lat 2.0 knots
Tooling Options	Tooling Skid Options comprising, 4F Manipulator Skid, FMD Orientation Skid, Cleaning Brush Skid and Cable Cutting Skid
Pay Load	32kg
Sensors Capabilities	2 simultaneous Video Channels & 1 switchable, 1 x lowlight monochrome, 1 x colour, Electrical Tilt unit +/- 90 degrees, 2 x dimmable LED (2 mounted under tilt unit)

• For full Seaeye Tiger ROV System specification, please refer to attached Appendix 4 – Seaeye Tiger_Rev1_LR;

CONTRACTOR STANDARD/TYPICAL ROV EQUIPMENT MANNING LEVELS

Unless otherwise agreed within the applicable CALL-OFF ORDER, CONTRACTOR's standard/typical ROV Equipment manning levels shall be as follows:

• Work Class ROV System (WROV), per system, per twelve (12) hours shift;

1 x ROV Supervisor, per WROV System; 2 x ROV Pilot Technician, per WROV System;

• Work Class ROV System (WROV), per system, per twenty-four (24) hour shift;

2 x ROV Supervisor, per WROV System; 4 x ROV Pilot Technician, per WROV System;

Observation Class ROV System (ObsROV), per system, per twelve (12) hour shift;

1 x ROV Supervisor, per WROV System; 1 x ROV Pilot Technician, per WROV System;

• Observation Class ROV System (ObsROV), per system, per twenty-four (24) hour shift;

2 x ROV Supervisor, per WROV System;

2 x ROV Pilot Technician, per WROV System

In this Section III the words and expressions established as definitions in both the Conditions of Contract, the Scope of Services and Appendices, shall have the same meanings as assigned to them in the said Conditions of Contract, Scope of Services and Appendices except when otherwise specifically stated.

Page 20 of 30



5.0 VARIATIONS

5.1 Any Variation in the WORK shall be valued in strict accordance with the rates and prices contained within Clause 4 COMPENSATION.

5.2 If the Variation WORK is of a differing nature and cannot be readily evaluated in accordance with the foregoing rates and prices the COMPANY may request the CONTRACTOR to propose a lump sum for COMPANY's consideration. The CONTRACTOR shall be requested to demonstrate the build up to any lump sum in order to demonstrate that it is priced on the same basis as the rates and prices contained herein and is fair and reasonable.

5.3 Any work that COMPANY instructs CONTRACTOR to undertake on a reimbursable basis shall be remunerated to CONTRACTOR at the cost of the invoice (net) plus ten percent (10%).

6.0 INVOICING

Lump Sum and Unit Rate Invoices shall be invoiced in accordance with the pricing schedule detailed in this section, 4.0 COMPENSATION.

- Reimbursable items shall be fully supported.
- The invoice shall clearly make reference to the payment stage, Contract No., item Nos., description and quantities.
- Where the WORK is expected to continue for more than one-month duration or extend from one month in to the following month, monthly invoices shall be
 raised, otherwise CONTRACTOR shall be entitled to submit an invoice on successful completion of the WORK.
- Invoices shall be submitted marked for the attention of the COMPANY Representative in the CALL-OFF ORDER.
- The following reference shall be clearly quoted on the invoice:
- CONTRACT No:213-0917;
- PROJECT Name: XXXX;
- In order to facilitate payment of invoices, CONTRACTOR is required within two weeks of receiving this CONTRACT to furnish COMPANY with details of its bank account to which payment will be made. This should include the Bank's name and address, sort code and CONTRACTOR's account number

Page 21 of 30



SECTION IV SCOPE OF WORK

1.0 INTRODUCTION

1.1 The CONTRACTOR shall be responsible for providing COMPANY with the Provision of ROV Services on board COMPANY provided Vessels (hereinafter called the "WORK"),

1.2 The specific scope of services to be performed by CONTRACTOR under the CONTRACT is further identified below:

The CONTRACTOR shall provide the WORK (Provision of ROV Services on board COMPANY provided Vessels) according to the specific requirements as
agreed between the Parties and as identified within the applicable CALL-OFF ORDER, inclusive of the required equipment and manning levels and the
specifics in relation to any required deliverables (final Report) and the timeframe for delivery of such deliverables;

WORK COMMENCEMENT, Mobilisation Window:

The WORK shall commence during a Mobilisation Window as further identified below unless otherwise agreed within the applicable CALL-OFF ORDER:

The COMPANY agrees to notify the CONTRACTOR of the mobilisation date in accordance with the following:

- 60 days' notice of a 30-day Mobilisation Window;
- 35 days' notice of a 14-day Mobilisation Window;
- 21 days' notice of a 7-day Mobilisation Window;
- 10 days' notice of a 3-day Mobilisation Window;
- 3 days' notice of actual mobilisation day in agreed port of mobilisation;

Each notified Mobilisation Window and actual mobilisation date following wholly within the previously notified window;

OUTLINE SCOPE OF SERVICES/SUPPLY

The CONTRACTOR shall provide the WORK (Provision of ROV Services on board COMPANY provided Vessels) as outlined below, however the below list is not exhaustive and the WORK shall instead be limited to the safe capabilities and resources of the CONTRACTOR personnel and equipment. As above, the specific WORK to be provided shall be identified within the applicable CALL-OFF ORDER.

- Supply of ROV System, Associated Equipment and project specific equipment as stated in the applicable CALL-OFF ORDER (mobilisation, commissioning and operation);
- Supply of Offshore ROV Personnel (including ROV Supervisors, ROV Pilot Technicians and Inspection Engineers);
- Supply of additional ROV Tooling/equipment (as specified in the applicable CALL-OFF ORDER);
- Onshore Management of ROV System (including technical support, spares supply and logistics);
- CONTRACTOR shall perform the WORK in compliance with applicable IMCA standards;
- Standards of reporting deliverables (as specified in the applicable CALL-OFF ORDER);

1.3 Additional services not listed herein may from time-to-time by requested under the provisions of the CONTRACT and identified within the applicable CALL-OFF ORDER.

Page 22 of 30



2.0 SCOPE OF SERVICES

2.1 The CONTRACTOR has been selected for the performance of the WORK on the understanding that it is an expert and professional in the class of work involved and that it will use and show all reasonable skill and judgement of such an expert and professional in the performance of the WORK.

The CONTRACTOR shall perform the WORK in accordance with the terms and conditions of the CONTRACT, all applicable legislation, regulations, statutory instruments, codes of practice, guidance notes and recommendations of the CONTRACT extracts, specifications and drawings attached hereto, which are now or may in the future become applicable to the CONTRACT.

• The WORK to be performed by CONTRACTOR will include but not be limited to the provision of all management, engineering, supervision, equipment, materials, consumables and where requested, transport, necessary to perform the WORK.

• CONTRACTOR is responsible for providing personnel, resources and equipment required for the proper and expeditious performance of the agreed scope of services in accordance with the CONTRACT.

• COMPANY is committed to preserving the health and safety of people and the protection of the environment throughout its organisation, and the organisations with which it places contracts for the performance of the WORK. Accordingly, the CONTRACTOR shall promote health, safety and environmental protection throughout its organisation and those of its SUBCONTRACTOR's to ensure that the CONTRACT is performed without harm to persons, property or the environment.

2.2 ERRORS AND OMISSIONS

• The review and approval by COMPANY of any procedures, documents or drawings shall only indicate a general requirement and shall not relieve CONTRACTOR of his obligation to comply with the requirements of the CONTRACT.

Any errors or omissions noted by CONTRACTOR shall be immediately brought to the attention of COMPANY.

2.3 DEVIATIONS

• All deviations to this WORK and other referenced specifications or attachments listed in the CONTRACT shall be made in writing and shall require written approval by COMPANY prior to execution of the WORK.

2.4 PERSONNEL (GENERAL)

• CONTRACTOR guarantees that sufficient experienced and competent personnel shall be made available to carry out the WORK, in accordance with the CONTRACT. CONTRACTOR personnel shall be able to maintain a 24 hour a day operation, if required.

2.5 EQUIPMENT (GENERAL)

CONTRACTOR shall supply all equipment required for the WORK in accordance with the CONTRACT.

Page 23 of 30



• Equipment spread and material will be provided by CONTRACTOR in quantity and quality, such as to carry out the WORK in the CONTRACT in a continuous and optimised fashion.

The Equipment Spread selected by CONTRACTOR shall be capable of performing the WORK within the requirements of the CONTRACT.

• CONTRACTOR will supply enough spares in order to avoid any interruptions in the operation. CONTRACTOR shall provide evidence of regular planned maintenance of equipment.

• CONTRACTOR will supply PUWER and LOLER Assessments for all equipment provided for use at CLIENT, COMPANY, COMPANY's other contractors or Third Party managed WORKSITES.

• CONTRACTOR shall ensure that as a minimum all items provided for provision of the WORK comply with industry best practice, regulations, and legislative requirements for the location where WORK are to be performed.

3.0 COMPANY PROVIDED ITEMS

As further identified in Section III Remuneration, COMPANY shall be responsible for providing vessels, transportation to and from the mobilisation and demobilisation points specified in the CONTRACT and/or agreed CALL-OFF ORDER to the offshore WORKSITE, accommodation, victualing and reasonable access to communications for business purposes while CONTRACTOR and/or SUBCONTRACTOR personnel are at WORKSITES managed by COMPANY or CLIENT.

4.0 MEETINGS

• The COMPANY REPRESENTATIVE may hold regular meetings with the CONTRACTOR to review progress as necessary. These meetings may be attended by additional COMPANY or CLIENT personnel, and shall normally be chaired by the COMPANY REPRSENTATIVE. The attendance of CONTRACTOR or its SUBCONTRACTOR's at any meetings arranged by COMPANY shall be included within the rates and prices identified within Section III Reimbursement of this CONTRACT.

Page 24 of 30



SECTION V HEALTH, SAFETY AND ENVIRONMENTAL

1.0 SAFETY

CONTRACTOR and its SUBCONTRACTORs shall perform the WORK in a manner such that risk to the health and safety of people or to the environment has been reduced to a level which is as low as reasonably practicable. The WORK shall be performed in accordance with all relevant Statutes, regulations and byelaws, codes of practice concerning safety, health and welfare and environmental protection matters whether current or promulgated at the date of signature of this CONTRACT.

CONTRACTOR shall implement an on-going programme of health, safety and environmental reviews of all stages of the WORK. Any adverse findings shall be addressed immediately by the CONTRACTOR.

CONTRACTOR's organisation may be subject to on-going health, safety or environmental audits at various stages of the WORK. Such audits shall be to measure conformity to relevant statutory requirements and CONTRATOR's health, safety or environmental policy.

2.0 HSE PROCEDURES

Notwithstanding anything to the contrary included elsewhere in the CONTRACT, CONTRACTOR shall comply with and shall perform the WORK in accordance with the COMPANY's/CLIENTS Worksite HSE Management system (Appendix XX attached herein) and with the following basic principles with respect to health, safety and environmental management:

• To be responsible for their own and SUBCONTRACTOR's personnel safety and prevent unsafe behaviour, or infringements against statutory health, safety or environmental rules.

• To co-operate fully with managers and supervisors of the COMPANY at the WORKSITE regarding all work matters including HSE management.

• To perform their WORK scope in accordance with instructions and procedures approved by COMPANY and with the general requirements laid out in COMPANY's specific Worksite HSE Management system.

- To provide equipment which is fit for purpose and with all safety features in place in accordance with statutory health, safety or environmental rules.
- To report all injury, damage (including pollution or spill), occupational illness, or (near miss) incident events as soon as practicable after the event.

It is mandatory that all people working at any WORKSITE wear the correct PPE. All personnel provided by CONTRACTOR shall be provided (by CONTRACTOR) as a minimum, full overall, safety boots / shoes, safety glasses and hard hat for all working areas. For outside work, suitable weather protective clothing and lifejackets if required by working area rules.

Where appropriate, CONTRACTOR's offshore personnel are expected to have documentary evidence to show that they have trained in basic firefighting and survival at sea, either through externally supplied firefighting / survival courses or by appropriate induction / training prior their personnel travelling offshore.

The CONTRACTOR shall ensure that all CONTRACTOR GROUP personnel have an acceptable current medical certificate(s) to prove their fitness to work offshore and must provide on request the certificate(s) for COMPANY's review.

Page 25 of 30



Alcoholic drinks or illegal drugs / controlled substances are prohibited at all WORKSITEs under any circumstances. COMPANY reserves the right to search CONTRACTOR GROUP or any other related personnel and /or equipment. Any person breaking this rule may be dismissed from the WORKSITE. If offshore, they may be returned to shore on the next available transport.

CONTRACTOR GROUP's personnel will be provided with the COMPANY's golden rules of safety when they arrive at WORKSITE. CONTRACTOR GROUP's personnel shall participate in the specific WORKSITE induction, which will cover all worksite safety features, HSE rules and procedures that must be complied with.

CONTRACTOR GROUP shall comply with any local WORKSITE rules.

3.0 COMPANY HSE POLICY

Notwithstanding anything to the contrary included elsewhere in the CONTRACT, CONTRACTOR GROUP shall comply with and perform the WORK in accordance with the COMPANY's/CLIENT's HSE Policy which is attached hereto (Appendix XX).

4.0 QUALITY CONTROL AND QUALITY ASSURANCE

• CONTRACTOR GROUP shall operate a quality system in accordance with ISO 9001, ISO 14001 and OHSAS 18001 series of standards, which shall be maintained by a group of persons sufficient in number and competence to ensure its effective implementation.

• CONTRACTOR GROUP shall carry out the WORK using equipment, processes, procedures and personnel fully qualified to, and compliant with the CONTRACT and of best industry practice.

• CONTRACTOR GROUP shall ensure that all plant, equipment, materials and consumables required for the WORK undergo quality control checks within a sufficient time prior to despatch of the items from premises of CONTRACTOR or its SUBCONTRACTORs.

• CONTRACTOR GROUP's quality control department, and those of its SUBCONTRACTORs, shall be responsible for all inspection, testing and certification, and shall be independent of design and production activities.

• The qualifications of inspection and testing personnel shall be subject to approval and the complete adherence to approved testing and inspection procedures shall be subject to regular monitoring/surveillance and periodic audit.

• CONTRACTOR may be requested to provide information, advice and expertise to be expected of a professional contractor for incorporation into COMPANY's project quality plans.

• CONTRACTOR shall allow COMPANY or CLIENTS QA auditor's full access to CONTRACTOR's and or any SUBCONTRACTOR's premises, personnel records and documentation, as permitted by law, for the purpose of conducting quality audits.

• Where non-conformances are identified during audits and/or inspections by COMPANY or its CLIENT, CONTRACTOR shall undertake the corrective actions as required by the COMPANY, within agreed specified time limits and at no cost to the COMPANY.

Notwithstanding anything to the contrary included elsewhere in this CONTRACT, COMPANY shall perform the WORK all in accordance with the quality systems identified within this CONTRACT.

Page 26 of 30



SECTION VI ADMINISTRATION INSTRUCTIONS

1. GENERAL

1.1. This Section of the CONTRACT details the procedures that shall be used by COMPANY and CONTRACTOR to administer the CONTRACT.

2. COMMUNICATIONS

- 2.1. Correspondence shall be in English and by letter, electronic mail, or fax.
- 2.2. All correspondence and communications in relation to the CONTRACT shall be addressed and directed between the COMPANY REPRESENTATIVE and the CONTRACTOR REPRESENTATIVE or their nominated deputies.
- 2.3. All mailing shall be by first class mail/airmail, courier or other equally prompt means.
- 2.4. Oral communication of instructions or information in connection with the CONTRACT shall be confirmed in writing using minutes of meetings or formal correspondence as appropriate.

3. VARIATION PROCEDURE

- 3.1. The rights of COMPANY to issue VARIATIONS are set out in the General Conditions of CONTRACT, the Clause headed VARIATIONS. The COMPANY shall issue an instruction to vary the WORK using a VARIATION order or form.
- 3.2. The rights of the CONTRACTOR to issue a VARIATION are set out in the General Conditions of CONTRACT, the Clause headed VARIATIONS. If CONTRACTOR considers that an occurrence, which shall include the COMPANY instruction, interpretation, decision or act has taken place that should give rise to a VARIATION, then the CONTRACTOR shall so inform the COMPANY immediately detailing each of the following:
 - a. The specific occurrence.
 - b. The date of the occurrence.
 - c. The part of the relevant section of the CONTRACT under which the CONTRACTOR considers that the occurrence should give rise to a VARIATION.
 - d. The impact of such occurrence on the CONTRACT duration and the effect on the CONTRACT PRICE.
- 3.3. CONTRACTOR shall quantify the request for a VARIATION by giving the COMPANY detailed supporting data within a period of fourteen (14) days of occurrence or such longer time as COMPANY may agree for any specific VARIATION.
- 3.4. If CONTRACTOR fails to inform COMPANY of the occurrence, CONTRACTOR shall forfeit any entitlement to adjustment to the CONTRACT PRICE and/or completion date.

Page 27 of 30



However, to the extent that COMPANY judges that non-submission of said notice by the CONTRACTOR was unavoidable, the COMPANY may, at its sole discretion, elect to extend the time for the CONTRACTOR to so inform the COMPANY.

- 3.5. Upon receipt by the COMPANY of information relating to each occurrence and advice relating to the effect of such occurrence, the COMPANY shall investigate the circumstances of the occurrence and the CONTRACTOR shall be advised in writing of the acceptance or otherwise of any request for a VARIATION within fourteen (14) days of receipt of the CONTRACTOR's request or such longer time as CONTRACTOR may agree for any specific VARIATION.
- 3.6. In the event that the CONTRACTOR's request is accepted in part or in whole by the COMPANY, the CONTRACTOR shall be so advised and the COMPANY shall authorise a VARIATION.
- 3.7. COMPANY shall generally not consider any CONTRACTOR's requests made on the basis of matters not included within the CONTRACT, but should the CONTRACTOR consider that any occurrence arises on the basis of such matters, and that such an occurrence entitles the CONTRACTOR to a VARIATION, then the CONTRACTOR shall immediately so inform the COMPANY in writing in the same as detailed in Clause 3.2 above. The CONTRACTOR shall continue to keep the COMPANY informed to any effects of such occurrence with fully quantified and detailed supporting data provided on a weekly basis, summarising all effects to date.

4. PROJECT MANAGEMENT

- 4.1. As a minimum, the CONTRACTOR is responsible for providing the following project management functions:
 - a. Overall management of the CONTRACT;
 - b. Sufficient numbers of suitably qualified and skilled personnel with relevant experience;
 - c. Suitable facilities;
 - d. All necessary systems, procedures and methods;
 - e. All other resources whatsoever required for the proper performance and completion of the CONTRACT;
 - f. Management of the CONTRACT from a single location by one integrated organisation.

Generally, the CONTRACTOR's project management responsibilities include but are not limited to:

- a. Overall management of the CONTRACT;
- b. Quality management;
- c. Health and safety management;
- d. Environmental management;
- e. Engineering management;

Page 28 of 30



- f. Procurement and Materials management;
- g. Fabrication/manufacturing management;
- h. Information management;
- i. Regulations, certification, and permitting support to the limit of permits, licences that are normally required to be maintained by CONTRACTOR with respect to the CONTRACTOR Material, equipment and personnel irrespective of actual Work location.
- j. Project control and Contract Program;
- k. Documentation Management.

The CONTRACTOR is responsible for the proper management and execution of the CONTRACT, wherever performed, and must have appropriate representation at all sites to complete all elements of the CONTRACT in a safe, environmentally responsible and timely manner in accordance with the CONTRACT. The CONTRACTOR must keep COMPANY fully informed at all times of progress and areas of concern.

COMPANY shall assist and advise the CONTRACTOR when COMPANY, in its sole judgement, considers it appropriate in relation to the CONTRACTOR's performance of the CONTRACT, without this relieving the CONTRACTOR in any way of its responsibilities, duties and obligations under the CONTRACT.

5. WEEKLY REPORTING

5.1. CONTRACTOR shall issue fortnightly reports for the duration of the CONTRACT. Within thirty (30) days of CONTRACT signature CONTRACTOR shall submit a fortnightly report template for COMPANY approval.

The fortnightly Report shall cover as a minimum:

- a. Work Achieved and Meetings held in the Week;
- b. Work Planned and Meetings arranged for Next Week;
- c. Areas of Concern / Critical Items;
- d. Health, Safety, Environment and Quality Assurance & Control Summary;
- e. Planning and Scheduling;
- f. Subcontracts / Procurement;
- g. Technical Queries;
- h. Interface Status.

Page 29 of 30



SECTION VII CALL-OFF ORDER TEMPLATE

твс

Page 30 of 30

THIS SECONDMENT AGREEMENT (the "Agreement") is made as of the <u>day of</u>, 20 (the "Effective Date") by and between **Fugro Subsea Services Limited**, a company incorporated under the laws of Scotland, company registration number SC105684 and having its registered office at Fugro House, Denmore Road, Bridge of Don, Aberdeen, AB23 8JW (the "Employer"), and **Global Marine Systems Limited**, a company incorporated under the laws of England and Wales, company registration number 10708481 and having its registered office at Ocean House, 1 Winsford Way, Boreham Interchange, Chelmsford, Essex, United Kingdom, CM2 5PD (the "Host"). The Employer and the Host are referred to herein individually as a "Party" and collectively as the "Parties".

WHEREAS, the Host is engaged in the business of Offshore Trenching Services (the "Business");

WHEREAS, the Employer employs personnel possessing skills that could assist the Host in the performance of its Business;

WHEREAS, the Host desires to utilize on a full time, temporary basis personnel of the Employer identified in Exhibit 1 to this Agreement (each a "Secondee") to perform the duties described in Exhibit 1 to this Agreement; and

WHEREAS, the Employer and the Host desire to set forth the terms and conditions by which a Secondee is seconded to the Host by the Employer.

NOW THEREFORE, in consideration of the premises and the mutual covenants and agreements contained herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Employer and the Host hereby agree as follows:

1. SECONDMENT

The Employer shall second the Secondee to the Host in accordance with the terms and conditions of this Agreement (the "Secondment"). The Secondment shall be on a full time basis and commence on the date set forth in Exhibit 1 and continue thereafter for the period of time set forth in Exhibit 1 to this Agreement (the "Secondment Period"). If applicable, the Host shall provide, at its cost without reimbursement from the Employer, roundtrip transportation for the Secondee from the Secondee's point of origin to the location specified by the Host in Exhibit 1 where the Secondee shall perform his or her duties for the Host (the Secondment Location"). During the Secondment Period, the Secondee shall perform the duties for the Host described in Exhibit 1 under the supervision and control of the Host. The Secondee shall at all times be and remain an employee of the Employer, but shall, during the Secondment Period, report to and act upon the instructions of the Host. Accordingly, the Host assumes full responsibility for the acts of the Secondee during the Secondment Period. The Secondee shall resume the performance of his or her duties for the Employer when the Secondment Period ends.

2. PAYMENT

2.1 Payroll and Benefits.

During the Secondment Period, (a) the Secondee shall remain on the payroll of the Employer and continue to be eligible for and accrue benefits under those employment benefits plans of the Employer in which the Secondee participated immediately prior to the Secondment (the "Plans"); and (b) the Employer shall continue to compensate the Secondee in the same amount and manner as before the Secondment including the withholding and payment of taxes and, to the extent permitted by the Plans and by applicable law, provide the benefits to which the Secondee is entitled pursuant to the Plans, in each case, in a timely manner.

2.2 Reimbursement of Secondee Costs.

The Host agrees to pay to the Employer the agreed daily hire rate as specified within Exhibit 1 herein and any actual out-of-pocket expenses (documented cost plus ten percent (+10%)) that are incurred by the Secondee or the Employer in the course of the Secondee's performance of his or her duties and paid for or reimbursed by the Employer all as reasonably determined by the Employer using its usual methods of cost accounting.

2.3 Invoicing.

On the first business day of each month, the Employer shall submit to the Host an invoice setting out the daily hire charges and any out of pocket expenses incurred as a result of the performance of the Secondee during the preceding month and all in accordance with Clause 2.2 above. Subject to the following sentence, the Host shall pay each invoice no later than thirty (30) days after its receipt thereof by wire transferring to the Employer's bank account designated in the invoice immediately available funds equal to the amount set forth in such invoice. The Host may deduct from any invoiced amount any amounts to be reimbursed by the Employer to the Host pursuant to Section 5 whereupon the Host shall pay the revised invoiced amount in the manner provided for in this subsection.

3. RESPONSIBILITIES OF THE EMPLOYER AND THE HOST

3.1 Employer Rights and Obligations.

The Employer shall perform all obligations which may be imposed on it by law or otherwise in its capacity as employer of the Secondee. The Employer shall determine in its sole discretion the compensation and benefits that are provided to the Secondee during the Secondment Period. The Host shall have no right to terminate the employment of the Secondee with the Employer, but may, for just cause, request that the Employer replace the Secondee with another employee of the Employer. The Employer may, at any time and in its discretion, terminate the employment of the Secondee during the Secondment Period.

3.2 Compliance with Laws.

The Host shall comply with all applicable laws and regulations to permit the Secondee to legally work for the Host during the Secondment Period including, if required, obtaining visas, work permits, taxpayer identification numbers, driver's licenses or other documentation. All costs incurred for such compliance shall be for the account of the Host.

2

4. INDEMNIFICATION

4.1 Injury to Secondee.

The Employer agrees to protect, defend, indemnify and hold harmless the Host and its directors, officers and employees (the "Host Indemnitees") from and against any and all damages, claims, losses, charges, actions, suits, proceedings, deficiencies, taxes, interest, penalties and reasonable costs and expenses (including without limitation reasonable attorney's fees and disbursements) arising out of or resulting from injury to or death of the Secondee during the Secondment Period regardless of the cause thereof including the simple, active, passive or contributory negligence of the Host Indemnitees and/or the unseaworthiness of any vessel being used by the Host Indemnities.

4.2 Acts or Omissions of Secondee.

The Host agrees to protect, defend, indemnify and hold harmless the Employer and its directors, officers and employees (the "Employer Indemnitees") from and against any and all damages, claims, losses, charges, actions, suits, proceedings, deficiencies, taxes, interest, penalties and reasonable costs and expenses (including without limitation reasonable attorney's fees and disbursements) arising out of or resulting from, the acts or omissions of the Secondee while under Secondment Period regardless of the cause thereof including the simple, active, passive or contributory negligence of the Employer Indemnitees and/or the unseaworthiness of any vessel being used by the Host Indemnities.

5. INSURANCE

Employer shall maintain during the Secondment Period the same insurance coverage in place for the Secondee as was maintained immediately prior to the Secondment. If at any time during the Secondment, the Secondee is required to perform his or her duties in a country where such insurance provides no coverage or is insufficient the Employer shall so notify the Host who shall then be obligated to provide sufficient insurance coverage for the Secondee during the Period of Secondment and all costs therefor shall be for the account of the Employer. If the Secondee requires urgent medical care during the Secondment, the Host shall provide to the Secondee the same emergency transportation services to a medical facility as it provides to its employees. The Host shall do all things necessary to insure that its insurance coverage is in compliance with the provisions of this Section.

6. TERMINATION

6.1 Without Cause.

Either Party may terminate this Agreement by giving ten (10) days' notice in writing of its intent to terminate to the other Party.

6.2 For Cause.

In the event that a Party fails to comply with the terms and condition of this Agreement, the other Party may terminate this Agreement immediately upon written notice to the other Party.

<u>3</u>

6.3 Termination of Secondee Employment.

This Agreement shall terminate no later than ten (10) days after the Employer notifies the Host of its intent to terminate the employment of the Secondee with the Employer

6.4 Expiration of Secondment.

This Agreement shall terminate on the expiration of the Secondment Period unless extended by mutual agreement of the Parties in writing.

6.4 Resumption of Employment.

On the termination of the Secondment for whatever reason, the Host shall have no further responsibility for the Secondee thereafter with the exception of making payment to the Employer for the work performed by the Secondee up to the date of termination.

7. NOTICES.

All notices required under this Agreement shall be made in writing and either delivered personally (effective upon delivery), by facsimile transmission (effective on the next day after transmission), by recognized overnight delivery service (effective on the next day after delivery to the service), or by registered or certified mail, postage prepaid and return receipt requested (effective on the fifth day after being so mailed). The address for notices of each Party is provided in Exhibit 1.

8. AMENDMENT; WAIVER

Any provision of this Agreement may be amended or waived if, and only if, such amendment or waiver is in writing and signed, in the case of an amendment, by both Parties hereto, or in the case of a waiver, by the Party or Parties against whom the waiver is to be effective.

No failure or delay by any Party in exercising any right, power or privilege under this Agreement shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege.

9. ENTIRE AGREEMENT

This Agreement contains the entire agreement between the Parties concerning their respective obligations with respect to the Secondment of the Secondee and supersedes all prior agreements and understandings, oral or written, with respect to such matters.

10. SUCCESSORS AND ASSIGNS

The provisions of this Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors. This Agreement is for the sole benefit of the Parties and, except as otherwise contemplated herein, nothing herein expressed or implied shall give or be construed to give any person or entity, other than the Parties, any legal or equitable rights hereunder.

<u>4</u>

11. GOVERNING LAW AND DISPUTE RESOLUTION

This Agreement shall be governed by and construed in accordance with the laws set forth in Exhibit 1 to this Agreement without giving effect to its conflict of law rules which would refer to the laws of another jurisdiction.

In the event there is a dispute between the Parties regarding this Agreement or the performance by a Party of its obligations hereunder, the disputing Party shall send a written notice to the other Party describing the dispute or issue in reasonably sufficient detail to allow the other Party to investigate and, if applicable, take the actions contemplated by the dispute notice, and the Parties shall promptly meet and attempt to resolve the dispute in good faith. If the Parties are unable to resolve the dispute pursuant to the foregoing, each Party shall have all rights and remedies available to it at law or in equity with respect to such dispute. The Parties agree to submit to jurisdiction of the English courts and comply with English Law for the resolution of disputes.

12. <u>SEVERABILITY</u>

In case any one or more of the provisions or part of a provision contained in this Agreement shall for any reason be held to be invalid, illegal or unenforceable, such invalidity, illegality or unenforceability shall be deemed not to affect any other provision or part of a provision of this Agreement and the Agreement shall be construed as if such provision or part of a provision held to be invalid, illegal or unenforceable had never been contained herein.

13. CONTINUING OBLIGATIONS

Notwithstanding the termination of this Agreement, the Parties shall continue to be bound by the provisions of this Agreement that reasonably require some action or forbearance after such termination including Sections 4 and 11.

14. CONSEQUENTIAL LOSS

For the purposes of this Agreement the expression "Consequential Loss" shall mean:

- (i) Consequential or indirect loss under English Law; and
- (ii) Loss and/or deferral of production, loss of product, loss of use, loss of revenue, profit or anticipated profit (if any), in each case whether direct or indirect to the extent that these are not included in (i), and whether or not foreseeable at the Effective Date of this Agreement.

Notwithstanding any provision to the contrary elsewhere in the Agreement, the Host shall save, indemnify, defend and hold harmless the Employer Indemnities from the Host Indemnities own Consequential Loss and the Employer shall save, indemnify, defend and hold harmless the Host Indemnities from the Employer Indemnities own Consequential Loss, arising from, relating to or in connection with the performance or non-performance of the Agreement.

All exclusions and indemnities given under this Clause 14 shall apply irrespective of cause and notwithstanding the negligence or breach of duty (whether statutory or otherwise) of the indemnified party or any other entity or party and shall apply irrespective of any claim in tort, under contract or otherwise at law.

<u>5</u>

For the purposes of this Clause 14 the expression "Host Indemnities" and "Employer Indemnities" shall have the same meaning as provided in Clause 4 above.

IN WITNESS WHEREOF, the Parties have executed this Agreement on the day and year first above written.

FOR THE EMPLOYER: FUGRO SUBSEA SERVICES LIN	 GLOBAL MARINE SYSTEMS LIMITED
Ву:	 Ву:
Name:	 Name:
Title:	 Title:
	<u>6</u>

EXHIBIT 1 TO SECONDMENT AGREEMENT THE SECONDEES

Following is the name and other information describing the individual who shall be seconded by the Employer to the Host for the Secondment Period.

NAME:

PASSPORT NUMBER:

DATE OF BIRTH:

PERMANENT HOME ADDRESS:

EMERGENCY CONTACT INFORMATION:

ADDRESS FOR NOTICES:

COMMENCEMENT DATE OF SECONDMENT:

DURATION OF SECONDMENT:

DESCRIPTION OF THE DUTIES TO BE PERFORMED BY THE SECONDEE DURING THE SECONDMENT (Secondee Discipline):

POINT OF ORIGIN OF SECONDEE:

GOVERNING LAW: English Law.

LOCATION OF SECONDMENT: Fugro House, Denmore Road, Bridge of Don, Aberdeen, AB23 8JW.

ADDITIONAL INSURANCE TO BE OBTAINED BY THE HOST:

DAILY HIRE RATE TO BE PAID BY HOST TO EMPLOYER: Daily Hire Rate of GBP £XXX.XX to commence immediately upon commencement date of Secondment and shall cease immediately upon completion of Secondment.

REIMBURSABLE EXPENSES: To be reimbursed at documented cost plus ten percent (10%).

7



Contract for the Provision of Survey Personnel & Equipment Services

Contract No. 214-0917

FORM OF AGREEMENT

This CONTRACT is made between the following parties:

Global Marine Systems Limited a company incorporated under the laws of England and Wales, company registration number 10708481 and having its registered office at Ocean House, 1 Winsford Way, Boreham Interchange, Chelmsford, Essex, United Kingdom, CM2 5PD hereinafter called the "COMPANY".

and

Fugro Subsea Services Limited a company incorporated under the laws Scotland, company registration number SC105684 and having its registered office at Fugro House, Denmore Road, Bridge of Don, Aberdeen, AB23 8JW hereinafter called the "CONTRACTOR".

WHEREAS:

- 1. the COMPANY and its AFFILIATES (as defined in Clause 1.1) wishes that certain WORK shall be carried out, all as described in the CONTRACT and pursuant to a duly executed CALL-OFF ORDER; and
- 2. the CONTRACTOR wishes to carry out the WORK in accordance with the terms of this CONTRACT and pursuant to a duly executed CALL-OFF ORDER.

NOW:

The parties hereby agree as follows:

- 1. In this CONTRACT, all capitalised words and expressions shall have the meanings assigned to them in this FORM OF AGREEMENT or elsewhere in the CONTRACT.
- 2. The following Sections shall be deemed to form and be read and construed as part of the CONTRACT:
 - 1. Section I Form of Agreement including Appendix 1;
 - 2. Section II a) General Conditions of Contract for Services Edition 2 October 2003;

b) Special Conditions of Contract;

- 3. Section III Remuneration;
- 4. Section IV Scope of Work;
- 5. Section V Health, Safety and Environment;
- 6. Section VI Company's General Obligations;
- 7. Section VII Call-Off Order Template.

Page 1 of 33



The Sections shall be read as one document, the contents of which, in the event of ambiguity or contradiction between Sections, shall be given precedence in the order listed, with the exception that the Special Conditions of Contract shall take precedence over the General Conditions of Contract.

- 3. In accordance with the terms and conditions of the CONTRACT, the CONTRACTOR shall perform and complete the WORK and the COMPANY or its AFFILIATE shall pay the CALL-OFF ORDER price.
- 4. The terms and conditions of the CONTRACT shall apply from the date specified in Appendix I to this Section I Form of Agreement which date shall be the EFFECTIVE DATE OF COMMENCEMENT OF THE CONTRACT.
- 5. The duration of the CONTRACT shall be as set out in Appendix I to this Section I Form of Agreement.
- 6. The COMPANY or its AFFILIATE shall call-off WORK as and when required during the term of the CONTRACT by issuing a CALL-OFF ORDER. CONTRACTOR shall either: (i) notify COMPANY or its AFFILIATE of acceptance of the CALL-OFF ORDER, either by countersigning the CALL-OFF ORDER, in accordance with the notification provisions of the CONTRACT or by e-mail; or (ii) reject the CALL-OFF ORDER and/or propose modifications to the CALL-OFF ORDER. COMPANY or its AFFILIATE shall consider any proposed modifications to the CALL-OFF ORDER, and if agreed, issue a revised CALL-OFF ORDER for acceptance by CONTRACTOR. A CALL-OFF ORDER will not become binding or form part of the CONTRACT until signed in writing by COMPANY or its AFFILIATE and CONTRACTOR

The authorised representatives of the parties have executed the CONTRACT in duplicate upon the dates indicated below:

For: For:

Global Marine Systems Limited Fugro Subsea Services Limited

Name: Name:

Title: Title:

Date: Date:

Page 2 of 33



APPENDIX I TO SECTION I - FORM OF AGREEMENT

Reference

Section I

Clause 4 The EFFECTIVE DATE OF COMMENCEMENT OF THE CONTRACT is

1st December 2017;

Clause 5 The duration of the CONTRACT is twelve (12) months from the EEFECTIVE DATE OF COMMENCEMENT OF THE CONTRACT. The term of the CONTRACT may be extended for a further period of twelve (12) months by the mutual agreement of both COMPANY and CONTRACTOR no less than ninety (90) days prior to expiry of the initial CONTRACT period.

Section II

- Clause 3.1(a) The COMPANY REPRESENTATIVE is Paul Thomas or as identified in the applicable CALL-OFF ORDER. The CONTRACTOR REPRESENTATIVE is as identified within the applicable CALL-OFF ORDER.
- Clause 5.1 The COMPANY designated heliport is Dyce, Aberdeen; The COMPANY designated supply base is Fugro House, Denmore Road, Bridge of Don, Aberdeen, AB23 8JW;
- Clause 10.2(b) The Defects Correction Period is: CONTRACTOR GROUP's Defects Correction (to reacquire data and/or rectify defective WORK) shall cease immediately upon CONTRACTOR GROUP's demobilisation and departure from the offshore WORKSITE (as detailed in the relevant CALL-OFF ORDER).
- Clause 13.8 The period of suspension is seven (7) days.
- Clause 14.3 Latest time for receipt of invoices ninety (90) days commencing at the agreed date at which the WORK was completed.
- Clause 14.9 Interest rate per annum Base Rate plus three (3) per cent p.a.
- Clause 17.4 Rights shall vest in CONTRACTOR.
- Clause 19.2(d) This indemnity is given in respect of the following property and is subject to any exclusions or limitations specified below:
 - (i) Property directly affected by the WORK, all as identified within Clause 19.2 d) of Section II B Special Conditions of Contract.
 - (ii) Other property, all as identified within Clause 19.2 (d) of Section II Special Conditions of Contract.
 - (iii) For the purposes of Clause 19.2(d) only, consequential losses shall have the same meaning as in Clause 21
- Clause 20.2 Insurance by the CONTRACTOR, the amounts are Employers' Liability: UK Sterling fifteen million pounds sterling (£15,000,000) anyone occurrence.

Page 3 of 33



General Third Party: EURO ten million (EUR€10,000,000) per occurrence combined single limit for property damage and/or bodily injury, EURO twenty million (EUR€20,000,000) for products liability in the annual aggregate.

Clause 24.7(a) (iii) Special Conditions remaining in full force and effect shall be - All.

- Clause 28.6 The addresses for the service of notices are: (i) COMPANY (ii) CONTRACTOR (iii) CONTRACTOR (ii) CONTRACTOR
- Clause 29.1(a) Limitation of Liability before the date of completion of the WORK The sum is limited to 100% of the relevant CALL-OFF ORDER PRICE.
- Clause 29.1(b) Limitation of Liability after the date of completion of the WORK The sum is limited to 100% of the relevant CALL-OFF ORDER PRICE.
- Clause 29.2 The Limitation Period shall cease upon expiry of the Defects Correction Period identified in Clause 10 herein.
- Clause 30.1(b) Resolution of Disputes. The nominees are: (i) COMPANY Ian Bryan, Director Corporate Development. (iii) CONTRACTOR: Derek Cruickshank, Managing Director.

Page 4 of 33



SECTION II (A) LOGIC GENERAL CONDITIONS OF CONTRACT FOR SERVICES (ON- AND OFF-SHORE). EDITION 2 -OCTOBER 2003

Page 5 of 33



SECTION II (B) SPECIAL CONDITIONS

The following amendments shall be made to the LOGIC General Conditions of Contract for Services (On and Off-shore).

1. Definitions

1.2. Insert new definition as follows: "CLIENT shall mean any party (together with its CO-VENTURER's in any relevant hydrocarbon, exploration and/or production licence), with whom the COMPANY has contracted and for whose ultimate benefit the WORK hereunder is provided".

Renumber all existing clauses accordingly.

1.3. Definition of COMPANY GROUP: Add ", and the CLIENT from time to time," after "shall mean the COMPANY". Add "it's other contractors and subcontractors (of any tier)" after "CO-VENTURERS" in the definition of COMPANY GROUP

1.16. Insert new definition as follows "CALL-OFF ORDER" shall mean the document in the form set out in Section VII Call-Off Order Template which incorporates the different parts of the CONTRACT and which when completed and signed will detail the specifics of the WORK to be carried out.

1.17. Insert new definition as follows "CALL-OFF ORDER PRICE" shall mean the sum(s) of money in the relevant CALL-OFF ORDER being payment for the satisfactory performance of the WORK by CONTRACTOR and as may be adjusted from time to time in accordance with the CONTRACT.

2. Interpretation

Insert a new clause 2.4 as follows:

2.4. Where approval or agreement is required by either Party under this CONTRACT, such approval shall not be unreasonably withheld, conditioned or delayed and where instructions are requested in writing from the COMPANY written instructions shall be provided within a reasonable time so as not to delay the WORK.

4. CONTRACTORS general obligations

In 4.2 add "But for the avoidance of doubt, the CONTRACTOR warrants that the results of the WORK will be fit for the purposes as defined in the applicable CALL-OFF ORDER and the respective specifications".

In 4.4 add "However, notwithstanding anything to the contrary included elsewhere in the CONTRACT, breakdown of any equipment provided by CONTRACTOR GROUP in performance of the WORK shall not constitute a breach of CONTRACT and COMPANY GROUP's remedy in respect thereof shall be limited to suspension of CONTRACTOR's relevant hire charges as further identified in Section III Remuneration."

6. CONTRACTOR to inform himself

6.1. Add at the beginning of the Clause "Subject to Clause 7.4"

7. CONTRACTOR to inform COMPANY / COMPANY to inform CONTRACTOR

7.3. Add at the end of the second paragraph "to the extent permitted by law".

Page 6 of 33



7.4. Add: "Such information shall include but not be limited to: (i) the presence, type and precise location of facilities, obstructions and hazards on or above or below or within 500m of the WORKSITE that may affect the efficiency or safety of the WORK. Such facilities, obstructions, and hazards may include, but not be limited to: subterranean and sub aqueous structures, cables, pipelines, tunnels and utilities, explosive ordnance, fishing-gear, cavities, over- and under-pressurised groundwater, and shallow gas; (ii) the WORK Geodetic system for horizontal and vertical control.

8. ASSIGNMENT AND SUBCONTRACTING

8.2 a): Insert "except to an Affiliate" after "WORK" in second sentence.

9. CONTRACTOR PERSONNEL

Insert a new Clause 9.9 as follows:

9.9. from the EFFECTIVE DATE OF COMMENCMENT OF THE CONTRACT until twelve (12) months after the CONTRACTOR completes the WORK, the COMPANY GROUP shall not directly or indirectly solicit the employment of, or interfere with or endeavour to entice away from the CONTRACTOR or endeavour to obtain the employment or engagement of, any person who has or had been allocated or assigned to the performance of the WORK by the CONTRACTOR provided that this prohibition shall not apply where any such person responds unsolicited to advertisements placed in the media by COMPANY.

10. Examination and defects correction

10.2. Add the word "material" before "defect" on each occasion it appears throughout the clause.

10.2 (b) At the end of sub-clause add "For the avoidance of doubt, CONTRACTOR's liability to reacquire data or rectify defective WORK shall cease immediately upon CONTRACTOR's departure from the offshore WORKSITE. Following CONTRACTOR's departure from the offshore WORKSITE, CONTRACTOR's liability to rectify defects will be limited to the re-analysis of data and re-issuing of reports for a maximum of two (2) separate occasions expiring twelve (12) months from receipt by COMPANY of CONTRACTOR's final report.

10.2.(c) Delete the first sentence and at the start of the second sentence insert "If the CONTRACTOR fails to remedy any defect within a reasonable period of time of being notified by COMPANY, such notification to be provided by COMPANY prior to CONTRACTOR's demobilisation and departure from the WORKSITE,

Add to Clause 10.2(d) "Notwithstanding anything to the contrary included elsewhere in this CONTRACT, CONTRACTOR shall not be liable for defects caused by CONTRACTOR's reliance on TECHNICAL INFORMATION, any other information, instruction or direction provided by the COMPANY GROUP:

Add new Clause 10.2 (e) as follows:

10.2 (e)"Notwithstanding anything to the contrary included elsewhere in the CONTRACT, CONTRACTOR's liability for error or mistake in the performance of the WORK shall be limited only to CONTRACTOR's charges relating to the WORK executed erroneously in proportion to the extent of the error or mistake or, where feasible and reasonable, the re-execution of such offshore WORK at no further charge to COMPANY, subject to CONTRACTOR being notified of any error or mistake prior to CONTRACTOR's demobilisation and departure from the offshore WORKSITE following final scheduled WORK activities for that phase of the WORK.

Page 7 of 33



Notwithstanding the above, if a component part of the CONTRACTOR's equipment spread is not functional but WORK can continue either with the scheduled task or other activities required to progress the WORK, this shall not be considered downtime or breakdown".

Insert new clause 10.3 as follows:

10.3 "Notwithstanding anything to the contrary the foregoing warranties shall be exclusive and in lieu of any other warranty whether express or implied or otherwise at law".

Insert new clause 10.4 as follows:

10.4 "CONTRACTOR shall ensure regular quality checks are performed in order to confirm the accuracy and integrity of the survey data being acquired by CONTRACTOR. The CONTRACTOR shall raise any anomalies or inconsistencies in the survey data acquired by it to the COMPANY without delay".

11. Variations

At the end of Clause 11.2 add "by agreement between the parties".

Insert new clause 11.3 as follows:

"11.3 CONTRACTOR shall be entitled to a VARIATION and a consequent adjustment to the CONTRACT PRICE and SCHEDULE, if the COMPANY, through its instructions, actions or lack of actions, has varied the scope of the WORK or the time for performance of the WORK or if the COMPANY has failed to provide unobstructed access to the WORKSITE."

12. Force Majeure

12.2 Amend opening words to read "For the purposes of this CONTRACT, force majeure shall include but not be limited to:"

Insert new clause 12.6 as follows:

12.6 In the event that any Force Majeure occurrence lasts longer than seven (7) days, COMPANY may instruct CONTRACTOR to remain on standby at a safe location, in which case COMPANY shall bear the associated standby costs set forth in Section III Remuneration. If standby option is not exercised by COMPANY, either COMPANY or CONTRACTOR can terminate the PURCHASE ORDER. Having elected to retain CONTRACTOR on standby and the delay due to FORCE MAJEURE prolongs, COMPANY can terminate the PURCHASE ORDER. If the PURCHASE ORDER is terminated through such FORCE MAJEURE event, COMPANY shall pay CONTRACTOR all outstanding sums due for the WORK completed to date of FORCE MAJEUE occurrence. Upon cessation of any FORCE MAJEURE occurrence the Parties may agree a revised programme to include for rescheduling of the WORK so as to minimise the effects of the delay which will be subject to CONTRACTOR'S other contractual commitments. In such event COMPANY shall reimburse CONTRACTOR for the costs of re-mobilisation.

13. Suspension

Insert the word 'material' before the word 'default' wherever appearing

13.5. Add at the end "Additional costs for which the CONTRACTOR is liable under this clause shall be documented and reasonable".

13.8: Change "fourteen (14) days" to "seven (7) days" on both occasions

Page 8 of 33



Insert new clauses 13.9 - 13.10. as follows:

13.9. CONTRACTOR shall have the right, upon notice to the COMPANY, to suspend the WORK or any part thereof to the extent detailed in the notice, for failure of COMPANY to meet its obligations under this CONTRACT, including but not limited to failure to make payment of the amounts due within 30 days of notice from CONTRACTOR that amounts are overdue and non-fulfilment of the obligation to inform CONTRACTOR of hazards pursuant to clause 7.4;

13.10. COMPANY agrees that CONTRACTOR shall not be liable to COMPANY for, and indemnifies CONTRACTOR from and against, any loss or damage or other liability suffered or incurred by COMPANY arising out of or in connection with suspension by CONTRACTOR under clause 13.9. In addition, and for the avoidance of doubt, Clause 13.4 shall continue to apply in the event of suspension in accordance with Clause 13.9.

18. Laws and regulations

18.2: Delete and replace with; "CONTRACTOR shall, in due time, obtain and maintain such approvals and permits as are necessary for the performance of the WORK and which must be obtained in the name of CONTRACTOR. COMPANY shall provide any necessary assistance in this connection.

COMPANY shall, in due time, obtain and maintain all other approvals and permits, such as the right to enter into and exit from the WORKSITE.

COMPANY shall advise CONTRACTOR of any limitations or restrictions affecting the right of entry to the WORKSITE, and CONTRACTOR shall abide by such limitations or restrictions, but CONTRACTOR shall not be liable for failure to comply with the terms of any contract or permit of COMPANY of which CONTRACTOR is not aware and could not reasonably be expected to have known about. "

19. Indemnities

Add new Clause 19.1 (e) as follows:

Survey operations that require intrusion of equipment or goods into the seabed (ground) will cause a change in the physical, chemical and hydraulic properties of the seabed (ground). In such instances, the COMPANY shall be responsible for, and shall indemnify CONTRACTOR GROUP against all consequences of such change notwithstanding the negligence or breach of duty (statutory or otherwise) of CONTRACTOR.

Add new Clause 19.1 (f) as follows:

Any such claims, losses, damages, costs (including legal costs) expenses and liabilities that may be brought by or in respect of any personnel employed by the CONTRACTOR against COMPANY pursuant to any loss of employment of such personnel with the CONTRACTOR.

Delete clause 19.2(d) and replace with "loss of, damage to or recovery of any third-party infrastructure within 500m of the WORKSITE including but not limited to oil and gas and telecommunications infrastructure and consequential losses arising therefrom where such loss, damage or recovery arise from or in connection with the performance or non-performance of the CONTRACT. This Clause 19.2(d) shall apply notwithstanding Clause 19.1(c)."

Page 9 of 33



20. Insurance by Contractor

20.1 Insert at the end of the clause the following "All insurance provided by COMPANY GROUP shall be endorsed to provide that underwriters waive any rights of recourse, including in particular subrogation rights against the CONTRACTOR GROUP in relation to the CONTRACT to the extent of the liabilities assumed by the COMPANY GROUP under the CONTRACT".

24. Termination

Insert the word "material" before "default" throughout this provision.

24.4. Insert at the end of the Clause, the following "In addition to the above, in the event of such termination the COMPANY shall reimburse CONTRACTOR the applicable cancellation fee as further detailed within Section III Remuneration."

Insert a new clause 24.8 as follows:

24.8. The CONTRACTOR shall have the right by giving notice to terminate the CONTRACT and/or any CALL-OFF ORDER for any or all of the following reasons:

a) in the event of a material default on the part of the COMPANY. Failure by COMPANY to make payment of CONTRACTOR's correctly prepared and adequately supported invoices within forty-five (45) days of such invoice due date shall be deemed to be a material default hereunder; or

b) in the event of the COMPANY becoming bankrupt or making a composition or arrangement with its creditors or a winding-up order of the COMPANY being made or (except for the purposes of amalgamation or reconstruction) a resolution for its voluntary winding-up being passed or a provisional Liquidator, Receiver, Administrator or Manager of its business or undertaking being appointed or presenting a petition or having a petition presented applying for an administration order to be made pursuant to Section 9 Insolvency Act 1986, or possession being taken by or on behalf of the holders of any debenture secured by a Floating Charge of any property comprised in or subject to the Floating Charge, or any equivalent act or thing being done or suffered under any applicable law.

25. AUDIT AND STORAGE OF DOCUMENTS

25.4: Please change "six (6) years" to "two (2) years"

26. Liens

26.1. To be deleted.

33. WAR RISK INSURANCE

Insert new Clause 33 as follows:

"CONTRACTOR insurance coverage extends worldwide, but in the event of personnel, a vessel or craft insured sailing for, deviating towards, or being within the Territorial Waters of any of the Countries or places described in the Current Exclusions as set out in the attached notice (or current version thereof), CONTRACTOR must notify its insurers who will advise if an additional premium will be applicable, or, indeed, if insurance is withdrawn. Where this occurs during the currency of a contract, CONTRACTOR shall be reimbursed the amount of such additional premium at cost. In the event that insurance is withdrawn, clearly CONTRACTOR would have discussions with the COMPANY concerned as a decision would be required on whether to proceed".

Page 10 of 33



34. LIQUIDATED DAMAGES

Insert new Clause 34 as follows:

"Liquidated Damages shall be agreed on a case by case basis and detailed in the relevant CALL-OFF ORDER and shall apply only when COMPANY GROUP suffers loss as a consequence of later performance by CONTRACTOR GROUP. Notwithstanding anything included elsewhere in the CONTRACT to the contrary, the amount of such Liquidated Damages shall accrue at a rate of zero-point three percent (0.3%) of the CONTRACTOR's personnel and equipment spread day rate per day, as further identified in Section III Remuneration herein, and shall be subject to an aggregate and/or cumulative cap of a sum equal to ten percent (10%) of the relevant CALL-OFF ORDER PRICE. Such Liquidated Damages shall be the COMPANY GROUP's sole and exclusive remedy and CONTRACTOR GROUP's sole and exclusive liability in respect of delay in the late commencement, performance or completion of the WORK under this CONTRACT".

35. GLOBAL NAVIGATION SATELLITE SYSTEM

Insert new Clause 35 as follows:

35.1. The CONTRACTOR has no control over any part and shall not be liable to the COMPANY GROUP in respect of the WORK to the extent that such failure of performance is due to the failure of the Global Navigation Satellite Systems space and ground segments ("GNSS"), as described below. For the avoidance of doubt, any warranty provided herein shall specifically exclude the GNSS.

35.2. COMPANY acknowledges as follows:

(a) GNSS has been developed and is operated by the United States Department of Defence and the Russian Federal Space Agency ("the GNSS Operators") and if the level of performance of the GNSS is impaired or ceases operation due to any act or omission of the GNSS Operators and/or any other authority and/or regulatory body which may now or later have responsibility for any aspects of the GNSS, then CONTRACTOR shall not be held responsible for any such act or omission, regardless of whether a complete or partial withdrawal or modification of the GNSS results; and

(b) Differential GNSS ("DGNSS") relies upon the availability of normal uninterrupted satellite and telecommunications services; power supplies (including electrical supplies); equipment which may be owned or controlled by COMPANY or third parties (other than members of CONTRACTOR GROUP), and to the extent that the unavailability of such services, supplies or equipment affects the performance of the WORK, the CONTRACTOR shall not be liable therefor.

35.3. COMPANY agrees that the CONTRACTOR shall not be liable to COMPANY for, and indemnifies CONTRACTOR GROUP from and against, any loss or damage or other liability suffered or incurred by COMPANY GROUP arising out of or in connection with the provision of or failure to provide DGNSS; or the unavailability, malfunctioning, interruption or other defect in such services, supplies or equipment referred to in this clause hereof for which third parties or COMPANY are responsible pursuant to Clause 35.2 above. No such unavailability, malfunctioning, interruption, non-operation or other defect shall relieve COMPANY of any of its obligations in the CONTRACT.

Page 11 of 33



SECTION III REMUNERATION

1.0 GENERAL

1.1 The terms and conditions set forth in this Section III Remuneration specify the basis for compensating CONTRACTOR for WORK which CONTRACTOR shall provide to COMPANY or its AFFILIATE under this CONTRACT and an agreed CALL-OFF ORDER. The rates, prices and principles set out in this Section III Remuneration shall be applied to define the CALL-OFF ORDER PRICE and paid to CONTRACTOR as a result of the satisfactory execution of the WORK in accordance with the requirements of the CONTRACT and the agreed CALL-OFF ORDER.

1.2 The rates and prices specified herein shall be deemed as all-inclusive to allow for all costs incurred by the CONTRACTOR in complying with the terms and conditions of the CONTRACT and the agreed CALL-OFF ORDER, and shall represent the total remuneration due to the CONTRACTOR in full consideration of the satisfactory performance of the WORK performed.

2.0 BASIS OF RATES AND PRICES

2.1 The rates and prices set forth herein are fully inclusive of all costs (including both direct and indirect costs), management and administrative costs, overhead and profit and shall apply equally to performing the WORK in large or small quantities, in any sequence, or in difficult or easy situations and whether the WORK is performed by the CONTRACTOR or its SUBCONTRACTORS.

2.3 The currency of the agreement is pounds sterling (GBP (£)). All lump sum amounts, unit rates, and other pricing shall be as detailed and all invoices prepared, submitted and paid in pounds sterling (GBP (£)).

3.0 BASIS OF RATES

3.1 The rates set forth herein shall include but not be limited to, the cost of salaries and wages actually paid to personnel, cost of all CONTRACTOR's payroll burdens, benefits and contributions, all taxes (except VAT) and insurance requirements in relation to working at CONTRACTOR offices or any other worksite and in full compliance with all terms and conditions of the CONTRACT and to carry out the WORK as described in Section IV Scope of Work and the agreed CALL-OFF ORDER.

3.2 Value Added Tax is excluded from the rates, sums and percentages herein. CONTRACTOR shall obtain for the benefit of COMPANY all available exemption from import duties, investment tax, Value Added Tax and all similar taxes and shall pass on to COMPANY any and all rebates.

4.0 COMPENSATION

4.1 CONTRACTOR shall be compensated for the satisfactory performance of the WORK in accordance with the rates, sums and prices contained within the rates tables included below. Any additional or project specific equipment shall be priced on a case by case basis and identified within the applicable CALL-OFF ORDER:

Page 12 of 33



	SECTION	Day Rate / LS	Hire of CONTRACTOR Resources (GBP)
1.0	Survey Services for COMPANY provided Vessels requiring Survey Services on long term commitment basis (typically in excess of three (3) months duration.		
1.1	 Day Rates for Survey Personnel (Europe) a) Party Chief, per person b) All other disciplines, per person c) Mobilisation/Demobilisation of CONTRACTOR Personnel, Per Person 	Day Rate Day Rate Lump Sum	510 480 Cost + 10%
1.2	Starfix DGNSS Signals (Primary and Secondary) and Hardware, worldwide – per vessel Usage	P/year P/Month P/Day	48,000 4,800 240
	Standard spread, hardware and software – per vessel (see Standard spread as further identified under heading "CONTRACTOR STANDARD SURVEY EQUIPMENT SPREAD, ROV SYSTEM MANNING LEVELS AND SURVEY EQUIPMENT MANNING LEVELS") Package Hardware		
1.3	Online Survey System (when used) Offline Processing System (when used) Digital Video System (when used)	Per Month Per Month Per Month Per/Month	1,900 2,200
1.4	Starfix Processing suite per onshore suite		Included in item 1.3
1.1.5	In-house Processing/Reporting Services – per man/day Per man/hour	Day Rate Hourly Rate	565 40
2.0	TRANSPONDERS		
2.1	Mobilisation of MF/EHF LBL Transponder		
	a) N.E Scottish Port	Lump Sum	79
	b) Norway	Lump Sum	247
2.2	Demobilisation of MF/EHF LBL Transponder		

Page 13 of 33



	a) N.E Scottish Port	Lump Sum	49
	b) Norway	Lump Sum	185
2.3	Mobilisation of USBL Transponder to NE Scottish Port	Lump Sum	79
2.4	Demobilisation of USBL Transponder from NE Scottish Port	Lump Sum	49
2.5	LBL Transponder Day rates		
2.5.1	MKV Compatt MF	Day Rate	35
2.5.2	MKV Compatt MF c/w SVS Option	Day Rate	40
2.5.3	MKV Compatt MF c/w Digiquartz/Inclinometer Option	Day Rate	48
2.5.4	MKV Compatt MF c/w SVS and High Precision Strain Gauge Option	Day Rate	52
2.5.5	6G Compatt MF	Day Rate	50
2.5.6	6G Compatt MF c/w SVS Option	Day Rate	60
2.5.7	6G Compatt MF c/w Digiquartz Option	Day Rate	65
2.5.8	6G Compatt MF c/w SVS and High Precision Strain Gauge Option	Day Rate	60
2.5.9	Flotation Collar	Day Rate	3
2.5.10	Floatation Collar Upgrade for 400+ meters	Day Rate	3
2.6	USBL Transponder Day Rates		
2.6.1	Mini HPR USBL Transponder	Day Rate	21
2.6.2	Simrad RPT319 Mini Beacon	Day Rate	21
2.6.3	Standard HPR USBL Transponder	Day Rate	21
2.6.4	SPT HiPAP Transponder	Day Rate	21
2.6.5	Battery charger	Day Rate	6
2.7	Transponder Moorings		
2.7.1	Sandbag Mooring assembly / Seabed stand per set / per deployment (Preparation)	Lump Sum	53
2.7.2	Quadroped seabed stand (each)	Day Rate	11
		1	



2.7.4	Sonar Reflector	Lump Sum	25
2.7.5	Strop	Lump Sum	6
2.7.6	Bow Shackle	Lump Sum	5
2.7.8	Sandbag	Lump Sum	7
3.0	Survey Services for COMPANY provided Ad-Hoc Vessels and projects requiring Survey Services on short term commitment basis (typically less than three (3) months duration).		
3.1	Survey Personnel		
3.1.1	Day Rates for Survey Personnel (Europe)		
	a) Party Chief, per person	Day rate	510
	b) All other disciplines, per person	Day rate	480
3.1.2	Mobilisation / Demobilisation of Personnel		
	a) Mobilisation/demobilisation of CONTRACTOR Personnel, per person, per direction	Lump Sum	Cost + 10%
3.2	On-line Computer & peripherals		
	a) Basic online computer system c/w printer, Colour Monitor: Working rate, incl consumables	Day Rate	87
	b) Additional colour Monitor	Day Rate	5
	c) Remote computer (Drone)	Day Rate	15
	d) UPS to cover navigation equipment (Typical spread 2 x online + 2 x drone + UPS)	Day Rate	7
3.3	Off-Line Computer package c/w peripherals		
	a) Logging/Processing comprising:	Day Rate	254
	- 3 No Workstations, High capacity disk drive & Software licences		
	Working rate, incl Consumables		
	b) A0 Plotter	Day Rate	43
	c) Additional colour Monitor	Day Rate	5
I	1	l	



	d) Additional Processing or Inspection Computer c/w 2 monitors	Day Rate	25
3.3.1	Mobilisation / Demobilisation		
3.3.1.1	Mob / Demob Aberdeen / Peterhead		
	a) Preparation and delivery of On-Line package, typical:	Lump Sum	639
	2 x Computers		
	2 x Drone computers		
	2 x Monitors		
	2 x Interfacing		
	1 x UPS		
	b) Preparation and delivery of Off-Line package, typical:	Lump Sum	941
	3 x Computers		
	1 x Plotter		
	Data storage		
	Video Review		
	c) Demobilisation of computer spreads	Lump Sum	642
3.3.1.2	Delivery charges		
	a) Bergen	Lump Sum	995
	b) Stavanger	Lump Sum	800
	c) Kristiansund	Lump Sum	1,043
	d) Rotterdam	Lump Sum	995
	e) Den Helder	Lump Sum	1,082
	f) Leith	Lump Sum	379
	g) Teeside	Lump Sum	584
	h) Hull	Lump Sum	639
	l	l	



	i) South Shields	Lump Sum	444
	j) Le Trait	Lump Sum	1,265
	k) Dundee	Lump Sum	254
	Unless otherwise stated 1.25 x Contract Day Rate shall be applied for preparation and delivery to Aberdeen / Peterhead		
	Delivery charges are in addition to Lump sums in 4.3.1.1 and cover freight charges for main planned Mobilisation / Demobilisation to ROVSV's per 1 No. 8' x 10' container only.		
3.4	LBL Acoustics Topsides Equipment		
3.4.1	Day rates: Sonardyne LUSBL system		
	a) MKV NCU with Starfix Acoustics Dongle	Day Rate	95
	b) MKV ROVNAV c/w MF Transducer	Day Rate	65
	c) 6G Starfix Acoustics Dongle	Day Rate	80
	d) 6G ROVNAV c/w Transducer	Day Rate	100
	e) 6G Dunker Transducer with 100m deck cable	Day Rate	105
	f) 6G IWAND (Deck Control Unit)	Day Rate	20
3.4.2	Day rates: TDS Probe		
	a) Basic: Valeport CTD600 or equivalent	Day Rate	38
	b) Self-recording: Seabird SBE-19, or equivalent	Day Rate	61
3.5	Video Equipment		
	a) 3 x Channel High Quality Digital Video recording/review	Day Rate	175
	b) Single Channel Basic Digital Video recording/review	Day Rate	77
	c) Delivery Media	Recharge	Cost + 10%
3.6	Subsea Gyro / Inclinometer as fitted to structures, (not ROV Gyro's)		
	a) Standard Gyro c/w 250m Umbilical	Day Rate	438
	b) Spare 250m Umbilical	Day Rate	33
	c) 400m rated system c/w 400m Umbilical	Day Rate	458
	d) Telemetry package c/w Topside receiver, Subsea Transmitter and Subsea battery pack	Day Rate	288
1	1	1	

Page 17 of 33



	e) Rental of Subsea Gyro table	Day Rate	19
	f) Standalone Gyro c/w underwater display, battery, logging laptop.	Day Rate	551
3.6.1	Mob/Demob to/from quayside in Aberdeen / Peterhead		
	a) Purchase of Subsea Gyro stab plate	Lump Sum	881
	b) Preparation and delivery of Subsea Gyro package	Lump Sum	350
	c) Preparation and delivery of Telemetry package	Lump Sum	410
3.6.2	Additional Equipment		
	a) Data telemetry link between vessels (per system)	Day Rate	44
	b) Video Telemetry link between vessels (per system)	Day Rate	58
	c) Transit / Storage Container	Day Rate	7
4.0	Acoustic Metrology/Structures Installations		
	Management, Liaison, Engineering and Preparation of Project Procedures		5,163
	a) Acoustic Metrology, per project	Lump Sum	4,130
4.1	b) Structure Installation, per project	Lump Sum	
	 Preparation of Yard visit Report and Final Installation/Metrology Report 	Lump Sum	2,868
	d) Preparation of Yard Visit report	Lump Sum	
			1,930
	Transponder Brackets		
4.2	a) Fabrication of Transponder Brackets		Cost + 5%
	b) Delivery of Transponder Mounting Brackets to Fabrication Yards where sent in advance of personnel		Cost + 5%
	Fabrication Yard Visits		
4.3	(To carry out Dimensional Survey of transponder Brackets and Gyro Plates pre-installation on structures relative to tie-in points)		
101	Mob and Demob of Calibration Equipment and 3 No. Personnel to and from:		2,244
4.3.1	a) Scottish fabrication Yard	Lump Sum	
	b) Other UK Fabrication Yard	Lump Sum	2,630



4.3.2	Day Rate for Site Calibration Equipment, 2 No. Personnel, transport, accommodation and subsistence, per full day or part thereof	Day Rate	1,315
4.3.3	Day Rate for Scanning Laser Equipment, 2 No. Personnel, transport, accommodation and subsistence, per full day or part thereof	Day Rate	1,665
5.0	PROJECT DOCUMENTATION		
5.1	Project Management	Per Project	TBC
5.2	Liason, Engineering and preparation of Project Survey Procedures	Per Project	TBC
5.3	Preparation of 2 draft and 4 final copies of report + CD's		
	Mob and Cal Report	Lump Sum	4,200
	Operational Reports	Lump Sum	4,725
	As Laid, As Trenched, Backfill Reports	Lump Sum	3,150
			+23/km
5.4	Charting		
	Charts A0, A1	Per Chart	472
	Charts A2, A3	Per Chart	420
5.5	Data Listings		
	Client Specific	Lump Sum	210
			+17/km
	Р5	Lump Sum	210
			+15/km
5.6	Office processing Tasks	Per Day	595
		Per Hour	60
6.0	ADDITIONAL EQUIPMENT		



6.1	Additional Services/Equipment		
6.1.1	CDL MiniRLG2 c/w raw IMU option	Day Rate	140
6.1.2	OCTANS 3000 FOG c/w raw IMU option	Day Rate	190
6.1.3	ROVINS	Day Rate	260
6.1.4	RDI Workhorse Navigator DVL (600 or 1200Khz)	Day Rate	87
6.1.5	Reson Seabat 8125 MBES	Day Rate	300
6.2	Vessel Gyro	Day Rate	102
	Mob/demob to/from quayside Aberdeen/Peterhead	Lump Sum	250
6.3	GNSS Systems		
	Starpack DGNSS Receiver (HP/XP/G2) Installation	Day Rate	30
	HF DGPS Receiver Installation	Day Rate	25
	PC with Multifix or Starpack QC software, inc dongle	Day Rate	15
	Mob/demob to/from quayside Aberdeen/Peterhead	Lump Sum	250
6.3.1	GNSS Correction Signals		
	HF Delivery (Subject to local coverage)		
	North Sea	Day Rate	92
	Egypt and South Persian Gulf	Day Rate	152
	West Africa	Day Rate	304
	Starfix HP or Skyfix XP or Starfix G2		
	NWECs and Mediterranean	Day Rate	150
	Americas- North and South	Day Rate	250
	West Africa	Day Rate	250
	SE Asia, Australasia & Middle East	Day Rate	150
	E Asia (China, Taiwan, Japan, Korea, Sakhalin)		
	Applies quayside to quayside		
6.4	Calibration Equipment		
	Total Station	Day Rate	100
	Trimble R6 GNSS Receiver	Day Rate	175



Pricing Notes

- Please note that the above item 1.3 "Standard spread, hardware and software" Charges per month shall be charged on a when used/pro-rata basis, however each 1) minimum hire period shall be for a period of no less than seven (7) continuous days;
- 2) The above rates, sums and prices shall remain valid up to and including 31st December 2018, thereafter the above rates, sums and prices shall be subject to escalation by the mutual agreement of both COMPANY and CONTRACTOR;
- The above rates, sums and prices are based strictly on WORK being performed within the UKCS only. Should COMPANY request WORK out of the UKCS, 3) CONTRACTOR reserves the right to adjust the above rates, sums and prices accordingly;
- The above rates, sums and prices are exclusive of VAT; 4)
- 5) In the event of CONTRACTOR Survey Equipment breakdown and should the CONTRACTOR Survey Equipment be unavailable to perform meaningful WORK then CONTRACTOR's Survey Equipment, associated personnel and associated equipment shall be deemed off-hire until such time as the Survey Equipment is again available to commence meaningful WORK:
- 6)
- For the avoidance of doubt, CONTRACTOR's hire Day Rates apply per each twenty-four (24) hrs period from midnight to midnight; CONTRACTOR's Personnel hire Day Rates apply immediately from departure of CONTRACTOR's Personnel from CONTRACTOR Personnel's original point of 7) origin until return to same. Costs of CONTRACTOR's Personnel travel etc to be recharged at cost+ 10%;
- 8) CONTRACTOR's Equipment hire Day Rates apply immediately upon departure from CONTRACTOR's base in Aberdeen and shall cease upon return to same,
- CONTRACTOR's monthly Equipment and DGNSS hire Day Rates are applicable irrespective of COMPANY's Vessel utilization or not; 9)
- 10) Any equipment cross hired from third parties by CONTRACTOR shall be provided at Cost+ 10%;
- Except where stated otherwise, a mobilisation/demobilisation lump sum of 1 x th applicable hire Day Rate will be applicable for all CONTRACTOR Equipment to 11) cover preparation, maintenance and consumable costs:
- Except where otherwise agreed, CONTRACTOR Personnel crew changes for the WORK shall be on twenty-eight (28) days rotation with overlap charges 12) applicable on days where both CONTRACTOR Personnel are simultaneously on board the COMPANY Vessel, travelling offshore/returning or standing by at the CONTRACTOR Personnel crew change mobilization point. A mobilization/demobilization fee will be charged per person per direction of travel;

COMPANY PROVIDED ITEMS

The COMPANY shall provide the following, at no charge/cost to CONTRACTOR:

- All support flights to/from COMPANY Vessels for all CONTRACTOR Survey Equipment and associated Spares, when required, along with supply vessels for such 1) CONTRACTOR equipment shipping;
- All support flights to/from COMPANY Vessels for all CONTRACTOR Personnel;
- 3) Provision of a Vessel with acceptable area of suitable size and position to allow safe operation of the CONTRACTOR Survey Equipment. Actual arrangement to be established;
- Suitable, clean, stable motor rated power supplies for the operation of CONTRACTOR's Survey Equipment; 4)
- 5) Waste Disposal
- 6) Hardware and wiring for connection of CONTRACTOR Survey Equipment to COMPANY's Vessel internal communications;
- All freight costs of CONTRACTOR's Survey Equipment to be for COMPANY's account, mobilisation/demobilisation included above are for equipment preparation and 7) return acceptance only;
- 8) Accommodation subsistence/victualling for CONTRACTOR Personnel while on-board COMPANY's Vessel (including periods of installation of CONTARCTOR Survey Equipment on-board COMPANY Vessel):
- Reasonable business communications from COMPANY Vessel to shore and UK office for CONTRACTOR personnel (Access to internet & e-mail systems); 9)
- 10) Emergency Medical Treatment and Medivac to approved Hospitals;

Page 21 of 33



- 11) In the event that offshore transportation for CONTRACTOR Personnel is delayed or cancelled six (6) hrs or less prior to the arrival of CONTRACTOR's Personnel at quayside or heliport, a cancellation fee of £250.00 per person and the applicable Personnel hire Day Rates shall apply;
- 12) Should the delay of transportation mean that additional overnight accommodation is required for CONTRACTOR Personnel, all such costs shall be recharged to COMPANY at cost + 10% and shall apply over and above the mobilisation lump sumand Personnel hire Day Rates;
- If a twelve (12) hrs Survey Team is required by COMPANY note that only twelve (12) hrs will be worked per day by CONTRACTOR personnel;

CONTRACTOR STANDARD SPREAD, HARDWARE AND SOFTWARE, PER VESSEL, FOR LONG TERM COMMITMENTS

Unless otherwise agreed within the applicable CALL-OFF ORDER, CONTRACTOR's standard Survey Equipment Spread shall be as follows:

NOTE:

The survey spread provided on the Symphony is a full construction support survey acquisition and processing system, based on Fugro Starfix software.

This includes a large number of specific acquisition modules and solutions such as:

- MBE Acquisition
 INS (Inertial Navigation engine
- INS (Inertial Navigation engine)
 Eventing module
- Eventing module
 Digital Video acq
- Digital Video acquisition and archive solution (4 Channel)
 Back2Base data compression and transmission solution.
- Unlimited software instances around the vessel (subject to Network constraints)

The survey package also include three offline processing licences including

- Multibeam Processing and DTM engine
- INS Post Processing solution
- Digital Video Playback
- Offline Eventing Module
- Automated Charting

The survey package also includes external software for 1x Microsoft SQL Server

2x Autocad Map software

The survey spread includes the annual calibration cost for the CTD/SVP Probes (2) and the Digital Protractor.

Hardware	Comment	Quantity
Starpak Novael	Primary GNSS receiver + spare	2
Starpack Trimble	Secondary GNSS receiver	1
AD492	Standard GNSS and Lband antenna	3
AD493	High latitude Lband antenna	1
Processing PC	Tower PC	3
21" Monitor	For Processing PC	6
Online PC	Rack Mount PC	5
20" Monitor	For Online PC's	10

Page 22 of 33



Starfix Smart Remotes	Small Form Factor PC	8
19" Monitor	For Smart Remotes	7
15" Monitor	For Crane	1
Starport	Serial to Network Converter	5
PPS Distribution	PPS	2
Starfix DV System (including spares)	DV	1
Quad Splitter	DV	1
LT04	DV	2
CTD	CTD	2
A0 Plotter	Plotter	1
Digital Protractor	Calibration	1
Disto	Calibration	2
Starfix Suite – as required for above PC's	Software	
AutoCad Map	Software	2

CONTRACTOR STANDARD SPREAD, HARDWARE AND SOFTWARE, PER VESSEL, FOR SHORT TERM COMMITMENTS

Survey Equipment Spread's for short term commitments shall be identified on a case by case basis and included within the applicable CALL-OFF ORDER.

CONTRACTOR STANDARD SURVEY EQUIPMENT MANNING LEVELS

Survey Equipment Manning Levels is dependant of the specifics of the WORK and will be agreed on a case by case basis and identified within the applicable CALL-OFF ORDER.

5.0 VARIATIONS

5.1 Any Variation in the WORK shall be valued in strict accordance with the rates and prices contained within Clause 4 COMPENSATION.

5.2 If the Variation WORK is of a differing nature and cannot be readily evaluated in accordance with the foregoing rates and prices the COMPANY may request the CONTRACTOR to propose a lump sum for COMPANY's consideration. The CONTRACTOR shall be requested to demonstrate the build up to any lump sum in order to demonstrate that it is priced on the same basis as the rates and prices contained herein and is fair and reasonable.

5.3 Any work that COMPANY instructs CONTRACTOR to undertake on a reimbursable basis shall be remunerated to CONTRACTOR at the cost of the invoice (net) plus ten percent (10%).

6.0 INVOICING

Lump Sum and Unit Rate Invoices shall be invoiced in accordance with the pricing schedule detailed in this section, 4.0 COMPENSATION.

- · Reimbursable items shall be fully supported.
- The invoice shall clearly make reference to the payment stage, Contract No., item Nos., description and quantities.

Page 23 of 33



- Where the WORK is expected to continue for more than one-month duration or extend from one month in to the following month, monthly invoices shall be • raised, otherwise CONTRACTOR shall be entitled to submit an invoice on successful completion of the WORK.
- Invoices shall be submitted marked for the attention of the COMPANY Representative in the CALL-OFF ORDER.
- The following reference shall be clearly quoted on the invoice:
 - CONTRACT No:214-0917: CALL-OFF ORDER No. XXXX;

PROJECT Name: Provision of Survey Personnel and Equipment Services. In order to facilitate payment of invoices, CONTRACTOR is required within two weeks of receiving this CONTRACT to furnish COMPANY with details of its • bank account to which payment will be made. This should include the Bank's name and address, sort code and CONTRACTOR's account number

Page 24 of 33



SECTION IV SCOPE OF WORK

1.0 INTRODUCTION

1.1 The CONTRACTOR shall be responsible for providing COMPANY with the Provision of Survey Personnel & Equipment Services (hereinafter called the "WORK"),

1.2 The specific scope of services to be performed by CONTRACTOR under the CONTRACT is further identified below:

 The CONTRACTOR shall provide the WORK (Provision of Survey Personnel & Equipment Services) according to the specific requirements as agreed between the Parties and as identified within the applicable CALL-OFF ORDER, inclusive of the required equipment and manning levels and the specifics in relation to any required deliverables (final Report) and the timelines for delivery of such deliverables;

WORK COMMENCEMENT, Mobilisation Window:

The WORK shall commence during a Mobilisation Window as further identified below unless otherwise agreed within the applicable CALL-OFF ORDER;

The COMPANY agrees to notify the CONTRACTOR of the mobilisation date in accordance with the following:

- 60 days' notice of a 30-day Mobilisation Window;
- 35 days' notice of a 14-day Mobilisation Window;
- 21 days' notice of a 7-day Mobilisation Window;
- 10 days' notice of a 3-day Mobilisation Window;
- 3 days' notice of actual mobilisation day in agreed port of mobilisation;

Each notified Mobilisation Window and actual mobilisation date following wholly within the previously notified window;

OUTLINE SCOPE OF SERVICES/SUPPLY

The CONTRACTOR shall provide the WORK (Provision of Survey Personnel & Equipment Services) as outlined below, however the below list is not exhaustive and the WORK shall instead be limited to the safe capabilities and resources of the CONTRACTOR Personnel and Equipment. As above, the specific WORK to be provided shall be identified within the applicable CALL-OFF ORDER.

• Provision of Survey Services on board COMPANY provided Vessels;

1.3 Additional services not listed herein may from time-to-time by requested under the provisions of the CONTRACT.

2.0 SCOPE OF SERVICES

2.1 The CONTRACTOR has been selected for the performance of the WORK on the understanding that it is an expert and professional in the class of work involved and that it will use and show all reasonable skill and judgement of such an expert and professional in the performance of the WORK.

The CONTRACTOR shall perform the WORK in accordance with the terms and conditions of the CONTRACT, all applicable legislation, regulations, statutory instruments, codes of practice, guidance notes and recommendations of the CONTRACT extracts, specifications and drawings attached hereto, which are now or may in the future become applicable to the CONTRACT.

Page 25 of 33



• The WORK to be performed by CONTRACTOR will include but not be limited to the provision of all management, supervision, equipment, materials, consumables and where requested, transport, necessary to perform the WORK.

• CONTRACTOR is responsible for providing personnel, resources and equipment required for the proper and expeditious performance of the agreed scope of services in accordance with the CONTRACT.

• COMPANY is committed to preserving the health and safety of people and the protection of the environment throughout its organisation, and the organisations with which it places contracts for the performance of the WORK. Accordingly, the CONTRACTOR shall promote health, safety and environmental protection throughout its organisation and those of its SUBCONTRACTOR's to ensure that the CONTRACT is performed without harm to persons, property or the environment.

2.2 ERRORS AND OMISSIONS

• The review and approval by COMPANY of any procedures, documents or drawings shall only indicate a general requirement and shall not relieve CONTRACTOR of his obligation to comply with the requirements of the CONTRACT.

· Any errors or omissions noted by CONTRACTOR shall be immediately brought to the attention of COMPANY.

2.3 DEVIATIONS

• All deviations to this WORK and other referenced specifications or attachments listed in the CONTRACT shall be made in writing and shall require written approval by COMPANY prior to execution of the WORK.

2.4 PERSONNEL (GENERAL)

• CONTRACTOR guarantees that sufficient experienced and competent personnel shall be made available to carry out the WORK, in accordance with the CONTRACT. CONTRACTOR personnel shall be able to maintain a 24 hour a day operation, if required.

2.5 EQUIPMENT (GENERAL)

CONTRACTOR shall supply all equipment required for the WORK in accordance with the CONTRACT.

• Equipment spread and material will be provided by CONTRACTOR in quantity and quality, such as to carry out the WORK in the CONTRACT in a continuous and optimised fashion.

The Equipment Spread selected by CONTRACTOR shall be capable of performing the WORK within the requirements of the CONTRACT.

• CONTRACTOR will supply enough spares in order to avoid any interruptions in the operation. CONTRACTOR shall provide evidence of regular planned maintenance of equipment.

• CONTRACTOR will supply PUWER and LOLER Assessments for all equipment provided for use at COMPANY or COMPANY's other contractors or Third Party managed WORKSITES.

Page 26 of 33



• CONTRACTOR shall ensure that as a minimum all items provided for provision of the WORK comply with industry best practice, regulations, and legislative requirements for the location where WORK is to be performed.

3.0 COMPANY PROVIDED ITEMS

COMPANY shall be responsible for providing vessels, transportation to and from the mobilisation and demobilisation points specified in the CONTRACT and/or agreed CALL-OFF ORDER to the offshore WORKSITE, accommodation, victualing and reasonable access to communications for business purposes while CONTRACTOR and/or SUBCONTRACTOR personnel are at WORKSITES managed by COMPANY or CLIENT.

4.0 MEETINGS

• The COMPANY REPRESENTATIVE may hold regular meetings with the CONTRACTOR to review progress as necessary. These meetings may be attended by additional COMPANY or CLIENT personnel, and shall normally be chaired by the COMPANY REPRSENTATIVE. The attendance of CONTRACTOR or its SUBCONTRACTOR's at any meetings arranged by COMPANY shall be included within the rates and prices identified within Section III Reimbursement of this CONTRACT.

Page 27 of 33



SECTION V HEALTH, SAFETY AND ENVIRONMENTAL

1.0 SAFETY

CONTRACTOR and its SUBCONTRACTORs shall perform the WORK in a manner such that risk to the health and safety of people or to the environment has been reduced to a level which is as low as reasonably practicable. The WORK shall be performed in accordance with all relevant Statutes, regulations and byelaws, codes of practice concerning safety, health and welfare and environmental protection matters whether current or promulgated at the date of signature of this CONTRACT.

CONTRACTOR shall implement an on-going programme of health, safety and environmental reviews of all stages of the WORK. Any adverse findings shall be addressed immediately by the CONTRACTOR.

CONTRACTOR's organisation may be subject to on-going health, safety or environmental audits at various stages of the WORK. Such audits shall be to measure conformity to relevant statutory requirements and CONTRATOR's health, safety or environmental policy.

2.0 HSE PROCEDURES

Notwithstanding anything to the contrary included elsewhere in the CONTRACT, CONTRACTOR shall comply with and shall perform the WORK in accordance with the COMPANY's/CLIENTS Worksite HSE Management system (Appendix XX attached herein) and with the following basic principles with respect to health, safety and environmental management:

• To be responsible for their own and SUBCONTRACTOR's personnel safety and prevent unsafe behaviour, or infringements against statutory health, safety or environmental rules.

• To co-operate fully with managers and supervisors of the COMPANY at the WORKSITE regarding all work matters including HSE management.

• To perform their WORK scope in accordance with instructions and procedures approved by COMPANY and with the general requirements laid out in COMPANY's specific Worksite HSE Management system.

- To provide equipment which is fit for purpose and with all safety features in place in accordance with statutory health, safety or environmental rules.
- To report all injury, damage (including pollution or spill), occupational illness, or (near miss) incident events as soon as practicable after the event.

It is mandatory that all people working at any WORKSITE wear the correct PPE. All personnel provided by CONTRACTOR shall be provided (by CONTRACTOR) as a minimum, full overall, safety boots / shoes, safety glasses and hard hat for all working areas. For outside work, suitable weather protective clothing and lifejackets if required by working area rules.

Where appropriate, CONTRACTOR's offshore personnel are expected to have documentary evidence to show that they have trained in basic firefighting and survival at sea, either through externally supplied firefighting / survival courses or by appropriate induction / training prior their personnel travelling offshore.

The CONTRACTOR shall ensure that all CONTRACTOR GROUP personnel have an acceptable current medical certificate(s) to prove their fitness to work offshore and must provide on request the certificate(s) for COMPANY's review.

Page 28 of 33



Alcoholic drinks or illegal drugs / controlled substances are prohibited at all WORKSITEs under any circumstances. COMPANY reserves the right to search CONTRACTOR GROUP or any other related personnel and /or equipment. Any person breaking this rule may be dismissed from the WORKSITE. If offshore, they may be returned to shore on the next available transport.

CONTRACTOR GROUP's personnel will be provided with the COMPANY's golden rules of safety when they arrive at WORKSITE. CONTRACTOR GROUP's personnel shall participate in the specific WORKSITE induction, which will cover all worksite safety features, HSE rules and procedures that must be complied with.

CONTRACTOR GROUP shall comply with any local WORKSITE rules.

3.0 COMPANY HSE POLICY

Notwithstanding anything to the contrary included elsewhere in the CONTRACT, CONTRACTOR GROUP shall comply with and perform the WORK in accordance with the COMPANY's/CLIENT's HSE Policy which is attached hereto (Appendix XX).

4.0 QUALITY CONTROL AND QUALITY ASSURANCE

• CONTRACTOR GROUP shall operate a quality system in accordance with ISO 9001, ISO 14001 and OHSAS 18001 series of standards, which shall be maintained by a group of persons sufficient in number and competence to ensure its effective implementation.

• CONTRACTOR GROUP shall carry out the WORK using equipment, processes, procedures and personnel fully qualified to, and compliant with the CONTRACT and of best industry practice.

• CONTRACTOR GROUP shall ensure that all plant, equipment, materials and consumables required for the WORK undergo quality control checks within a sufficient time prior to despatch of the items from premises of CONTRACTOR or its SUBCONTRACTORS.

• CONTRACTOR GROUP's quality control department, and those of its SUBCONTRACTORs, shall be responsible for all inspection, testing and certification, and shall be independent of design and production activities.

• The qualifications of inspection and testing personnel shall be subject to approval and the complete adherence to approved testing and inspection procedures shall be subject to regular monitoring/surveillance and periodic audit.

• CONTRACTOR may be requested to provide information, advice and expertise to be expected of a professional contractor for incorporation into COMPANY's project quality plans.

• CONTRACTOR shall allow COMPANY or CLIENTS QA auditor's full access to CONTRACTOR's and or any SUBCONTRACTOR's premises, personnel records and documentation, as permitted by law, for the purpose of conducting quality audits.

• Where non-conformances are identified during audits and/or inspections by COMPANY or its CLIENT, CONTRACTOR shall undertake the corrective actions as required by the COMPANY, within agreed specified time limits and at no cost to the COMPANY.

Notwithstanding anything to the contrary included elsewhere in this CONTRACT, COMPANY shall perform the WORK all in accordance with the quality systems identified within this CONTRACT,

Page 29 of 33



SECTION VI ADMINISTRATION INSTRUCTIONS

1. GENERAL

1.1. This Section of the CONTRACT details the procedures that shall be used by COMPANY and CONTRACTOR to administer the CONTRACT.

2. COMMUNICATIONS

- 2.1. Correspondence shall be in English and by letter, electronic mail, or fax.
- 2.2. All correspondence and communications in relation to the CONTRACT shall be addressed and directed between the COMPANY REPRESENTATIVE and the CONTRACTOR REPRESENTATIVE or their nominated deputies.
- 2.3. All mailing shall be by first class mail/airmail, courier or other equally prompt means.
- 2.4. Oral communication of instructions or information in connection with the CONTRACT shall be confirmed in writing using minutes of meetings or formal correspondence as appropriate.

3. VARIATION PROCEDURE

- 3.1. The rights of COMPANY to issue VARIATIONS are set out in the General Conditions of CONTRACT, the Clause headed VARIATIONS. The COMPANY shall issue an instruction to vary the WORK using a VARIATION order or form.
- 3.2. The rights of the CONTRACTOR to issue a VARIATION are set out in the General Conditions of CONTRACT, the Clause headed VARIATIONS. If CONTRACTOR considers that an occurrence, which shall include the COMPANY instruction, interpretation, decision or act has taken place that should give rise to a VARIATION, then the CONTRACTOR shall so inform the COMPANY immediately detailing each of the following:
 - a. The specific occurrence.
 - b. The date of the occurrence.
 - c. The part of the relevant section of the CONTRACT under which the CONTRACTOR considers that the occurrence should give rise to a VARIATION.
 - d. The impact of such occurrence on the CONTRACT duration and the effect on the CONTRACT PRICE.
- 3.3. CONTRACTOR shall quantify the request for a VARIATION by giving the COMPANY detailed supporting data within a period of fourteen (14) days of occurrence or such longer time as COMPANY may agree for any specific VARIATION.
- 3.4. If CONTRACTOR fails to inform COMPANY of the occurrence, CONTRACTOR shall forfeit any entitlement to adjustment to the CONTRACT PRICE and/or completion date.

Page 30 of 33



However, to the extent that COMPANY judges that non-submission of said notice by the CONTRACTOR was unavoidable, the COMPANY may, at its sole discretion, elect to extend the time for the CONTRACTOR to so inform the COMPANY.

- 3.5. Upon receipt by the COMPANY of information relating to each occurrence and advice relating to the effect of such occurrence, the COMPANY shall investigate the circumstances of the occurrence and the CONTRACTOR shall be advised in writing of the acceptance or otherwise of any request for a VARIATION within fourteen (14) days of receipt of the CONTRACTOR's request or such longer time as CONTRACTOR may agree for any specific VARIATION.
- 3.6. In the event that the CONTRACTOR's request is accepted in part or in whole by the COMPANY, the CONTRACTOR shall be so advised and the COMPANY shall authorise a VARIATION.
- 3.7. COMPANY shall generally not consider any CONTRACTOR's requests made on the basis of matters not included within the CONTRACT, but should the CONTRACTOR consider that any occurrence arises on the basis of such matters, and that such an occurrence entitles the CONTRACTOR to a VARIATION, then the CONTRACTOR shall immediately so inform the COMPANY in writing in the same as detailed in Clause 3.2 above. The CONTRACTOR shall continue to keep the COMPANY informed to any effects of such occurrence with fully quantified and detailed supporting data provided on a weekly basis, summarising all effects to date.

4. PROJECT MANAGEMENT

- 4.1. As a minimum, the CONTRACTOR is responsible for providing the following project management functions:
 - a. Overall management of the CONTRACT;
 - b. Sufficient numbers of suitably qualified and skilled personnel with relevant experience;
 - c. Suitable facilities;
 - d. All necessary systems, procedures and methods;
 - e. All other resources whatsoever required for the proper performance and completion of the CONTRACT;
 - f. Management of the CONTRACT from a single location by one integrated organisation.

Generally, the CONTRACTOR's project management responsibilities include but are not limited to:

- a. Overall management of the CONTRACT;
- b. Quality management;
- c. Health and safety management;
- d. Environmental management;
- e. Engineering management;

Page 31 of 33



- f. Procurement and Materials management;
- g. Fabrication/manufacturing management;
- h. Information management;
- i. Regulations, certification, and permitting support to the limit of permits, licences that are normally required to be maintained by CONTRACTOR with respect to the CONTRACTOR Material, equipment and personnel irrespective of actual Work location.
- j. Project control and Contract Program;
- k. Documentation Management.

The CONTRACTOR is responsible for the proper management and execution of the CONTRACT, wherever performed, and must have appropriate representation at all sites to complete all elements of the CONTRACT in a safe, environmentally responsible and timely manner in accordance with the CONTRACT. The CONTRACTOR must keep COMPANY fully informed at all times of progress and areas of concern.

COMPANY shall assist and advise the CONTRACTOR when COMPANY, in its sole judgement, considers it appropriate in relation to the CONTRACTOR's performance of the CONTRACT, without this relieving the CONTRACTOR in any way of its responsibilities, duties and obligations under the CONTRACT.

5. WEEKLY REPORTING

5.1. CONTRACTOR shall issue fortnightly reports for the duration of the CONTRACT. Within thirty (30) days of CONTRACT signature CONTRACTOR shall submit a fortnightly report template for COMPANY approval.

The fortnightly Report shall cover as a minimum:

- a. Work Achieved and Meetings held in the Week;
- b. Work Planned and Meetings arranged for Next Week;
- c. Areas of Concern / Critical Items;
- d. Health, Safety, Environment and Quality Assurance & Control Summary;
- e. Planning and Scheduling;
- f. Subcontracts / Procurement;
- g. Technical Queries;
- h. Interface Status.

Page 32 of 33



SECTION VII CALL-OFF ORDER TEMPLATE

твс

Page 33 of 33



Contract for the Provision of Management of COMPANY owned Q1400 Trencher Systems & Management and Operation of COMPANY owned FCV3000 ROV Systems and the supply of Survey Services on board the Fugro Symphony

Contract No. 212-0917

FORM OF AGREEMENT

This CONTRACT is made between the following parties:

Global Marine Systems Limited a company incorporated under the laws of England and Wales, company registration number 10708481 and having its registered office at Ocean House, 1 Winsford Way, Boreham Interchange, Chelmsford, Essex, United Kingdom, CM2 5PD hereinafter called the COMPANY

and

Fugro Subsea Services Limited a company incorporated under the laws of Scotland, company registration number SC105684 and having its registered office at Fugro House, Denmore Road, Bridge of Don, Aberdeen, AB23 8JW hereinafter called the CONTRACTOR.

WHEREAS:

- 1. the COMPANY and its AFFILIATES (as defined in Clause 1.1) wishes that certain WORK shall be carried out, all as described in the CONTRACT and pursuant to a duly executed CALL-OFF ORDER; and
- 2. the CONTRACTOR wishes to carry out the WORK in accordance with the terms of this CONTRACT and pursuant to a duly executed CALL-OFF ORDER.

NOW:

The parties hereby agree as follows:

- 1. In this CONTRACT, all capitalised words and expressions shall have the meanings assigned to them in this FORM OF AGREEMENT or elsewhere in the CONTRACT.
- 2. The following Sections shall be deemed to form and be read and construed as part of the CONTRACT:
 - 1. Section I Form of Agreement including Appendix 1;
 - 2. Section II a) General Conditions of Contract for Services Edition 2 October 2003;

b) Special Conditions of Contract;

- 3. Section III Remuneration;
- 4. Section IV Scope of Work;
- 5. Section V Health, Safety and Environment;
- 6. Section VI Company's General Obligations;

Page 1 of 34



7. Section VII Call-Off Order Template.

The Sections shall be read as one document, the contents of which, in the event of ambiguity or contradiction between Sections, shall be given precedence in the order listed, with the exception that the Special Conditions of Contract shall take precedence over the General Conditions of Contract.

- 3. In accordance with the terms and conditions of the CONTRACT, the CONTRACTOR shall perform and complete the WORK and the COMPANY or its AFFILIATES shall pay the CALL-OFF ORDER price.
- 4. The terms and conditions of the CONTRACT shall apply from the date specified in Appendix I to this Section I Form of Agreement which date shall be the EFFECTIVE DATE OF COMMENCEMENT OF THE CONTRACT.
- 5. The duration of the CONTRACT shall be as set out in Appendix I to this Section I Form of Agreement.
- 6. The COMPANY or its AFFILIATE shall call-off WORK as and when required during the term of the CONTRACT by issuing a CALL-OFF ORDER. CONTRACTOR shall either: (i) notify COMPANY or its AFFILIATE of acceptance of the CALL-OFF ORDER, either by countersigning the CALL-OFF ORDER, in accordance with the notification provisions of the CONTRACT or by e-mail; or (ii) reject the CALL-OFF ORDER and/or propose modifications to the CALL-OFF ORDER. COMPANY or its AFFILATE shall consider any proposed modifications to the CALL-OFF ORDER, and if agreed, issue a revised CALL-OFF ORDER for acceptance by CONTRACTOR. A CALL-OFF ORDER will not become binding or form part of the CONTRACT until signed in writing by both COMPANY and CONTRACTOR.

The authorised representatives of the parties have executed the CONTRACT in duplicate upon the dates indicated below:

For: For:

Global Marine Systems Limited Fugro Subsea Services Limited

Name: Name:

Title: Title:

Date: Date:

Page 2 of 34



APPENDIX I TO SECTION I - FORM OF AGREEMENT

Reference

Section I

Clause 4 The EFFECTIVE DATE OF COMMENCEMENT OF THE CONTRACT is

1st December 2017;

Clause 5 The duration of the CONTRACT is twelve (12) months from the EFFECTIVE DATE OF COMMENCEMENT OF THE CONTRACT. The term of the CONTRACT may be extended for a further period of twelve (12) months by the mutual agreement of both COMPANY and CONTRACTOR no less than ninety (90) days prior to expiry of the initial CONTRACT period.

Section II

- Clause 3.1(a) The COMPANY REPRESENTATIVE is Paul Thomas or as identified within the applicable CALL-OFF ORDER. The CONTRACTOR REPRESENTATIVE is, as identified in the applicable CALL-OFF ORDER.
- Clause 5.1 The COMPANY designated heliport is Dyce, Aberdeen; The COMPANY designated supply base is Fugro House, Denmore Road, Bridge of Don, Aberdeen, AB23 8JW;
- Clause 10.2(b) The Defects Correction Period is: CONTRACTOR GROUP's Defects Correction (to reacquire data and/or rectify defective WORK) shall cease immediately upon CONTRACTOR GROUP's demobilisation and departure from the offshore WORKSITE as detailed in the relevant CALL-OFF ORDER.
- Clause 13.8 The period of suspension is seven (7) days.
- Clause 14.3 Latest time for receipt of invoices ninety (90) days commencing at the agreed date at which the WORK was completed.
- Clause 14.9 Interest rate per annum Base Rate plus three (3) per cent p.a.
- Clause 17.4 Rights shall vest in CONTRACTOR.
- Clause 19.2(d) This indemnity is given in respect of the following property and is subject to any exclusions or limitations specified below: (i) Property directly affected by the WORK, all as identified within Clause 19.2 d) of Section II B Special Conditions

of Contract.

- (ii) Other property, all as identified within Clause 19.2 (d) of Section II Special Conditions of Contract.
- (iii) For the purposes of Clause 19.2(d) only, consequential losses shall have the same meaning as in Clause 21
- Clause 20.2 Insurance by the CONTRACTOR, the amounts are Employers' Liability: UK Sterling fifteen million pounds sterling (£15,000,000) anyone occurrence.

Page 3 of 34



General Third Party: EURO ten million (EUR€10,000,000) per occurrence combined single limit for property damage and/or bodily injury, EURO twenty million (EUR€20,000,000) for products liability in the annual aggregate.

Clause 24.7(a) (iii) Special Conditions remaining in full force and effect shall be – All.

- Clause 28.6 The addresses for the service of notices are: (i) COMPANY (ii) CONTRACTOR (iii) CONTRACTOR (ii) CONTRACTOR
- Clause 29.1(a) Limitation of Liability before the date of completion of the WORK The sum is limited to 100% of the relevant CALL-OFF ORDER PRICE.
- Clause 29.1(b) Limitation of Liability after the date of completion of the WORK The sum is limited to 100% of the relevant CALL-OFF ORDER PRICE.
- Clause 29.2 The Limitation Period shall cease upon expiry of the Defects Correction Period identified in Clause 10 herein.
- Clause 30.1(b) Resolution of Disputes. The nominees are: (i) COMPANY Ian Bryan, Director Corporate Development. (iii) CONTRACTOR: Derek Cruickshank, Managing Director.

Page 4 of 34



SECTION II (A) LOGIC GENERAL CONDITIONS OF CONTRACT FOR SERVICES (ON- AND OFF-SHORE). EDITION 2 -OCTOBER 2003

Page 5 of 34



SECTION II (B) SPECIAL CONDITIONS

The following amendments shall be made to the LOGIC General Conditions of Contract for Services (On and Off-shore).

1. Definitions

1.2. Insert new definition as follows: "CLIENT shall mean any party (together with its CO-VENTURER's in any relevant hydrocarbon, exploration and/or production licence), with whom the COMPANY has contracted and for whose ultimate benefit the WORK hereunder is provided".

Renumber all existing clauses accordingly.

1.3. Definition of COMPANY GROUP: Add ", and the CLIENT from time to time," after "shall mean the COMPANY". Add "it's other contractors and subcontractors (of any tier)" after "CO-VENTURERS" in the definition of COMPANY GROUP.

1.16. Insert new definition as follows "CALL-OFF ORDER" shall mean the document in the form set out in Section VII Call-Off Order Template which incorporates the different parts of the CONTRACT and which when completed and signed will detail the specifics of the WORK to be carried out".

1.17. Insert new definition as follows "CALL-OFF ORDER PRICE" shall mean the sum(s) of money in the relevant CALL-OFF ORDER being payment for the satisfactory performance of the WORK by CONTRACTOR and as may be adjusted from time to time in accordance with the CONTRACT".

2. Interpretation

Insert a new clause 2.4 as follows:

2.4. Where approval or agreement is required by either Party under this CONTRACT, such approval shall not be unreasonably withheld, conditioned or delayed and where instructions are requested in writing from the COMPANY written instructions shall be provided within a reasonable time so as not to delay the WORK.

4. CONTRACTORS general obligations

In 4.2 add "But for the avoidance of doubt, the CONTRACTOR warrants that the results of the WORK will be fit for the purpose as identified in the applicable CALL-OFF ORDER and the respective specification".

In 4.4 add "However, notwithstanding anything to the contrary included elsewhere in the CONTRACT, breakdown of any equipment utilised by CONTRACTOR GROUP in performance of the WORK shall not constitute a breach of CONTRACT and COMPANY GROUP's remedy in respect thereof shall be limited to suspension of CONTRACTOR's relevant hire charges as further identified in Section III Remuneration. Provided always that the CONTRACTOR shall be allowed two (2) hours maintenance allowance per day up to a cumulative total of twelve (12) hours for maintenance and repair of its ROV spread during which time the CONTRACTOR's equipment and associated personnel shall be deemed on standby at the relevant hire charges as further identified in Section III Remuneration."

Further, CONTRACTOR'S ROV Supervisor shall have the authority to decline to deploy the ROV System if conditions are such that to do so may, in the CONTRACTOR'S ROV Supervisor's sole opinion, present an unacceptable risk to the safety of personnel or equipment and shall have the authority to decide if and when to recover the ROV System in the face of marginal weather with a worsening forecast. Additionally, the CONTRACTOR'S

Page 6 of 34



ROV Supervisor shall have the authority to decline to endeavour to access any area where, in the CONTRACTOR'S ROV Supervisor's opinion to do so may risk entrapment or loss of the vehicle. In any of the above cases, CONTRACTOR shall not be liable or penalised in any way and relevant day rates as further identified in Section III Remuneration shall continue to apply

6. CONTRACTOR to inform himself

6.1. Add at the beginning of the Clause "Subject to Clause 7.4"

7. CONTRACTOR to inform COMPANY / COMPANY to inform CONTRACTOR

7.3. Add at the end of the second paragraph "to the extent permitted by law".

7.4. Add: "Such information shall include but not be limited to: (i) the presence, type and precise location of facilities, obstructions and hazards on or above or below or within 500m of the WORKSITE that may affect the efficiency or safety of the WORK. Such facilities, obstructions, and hazards may include, but not be limited to: subterranean and sub aqueous structures, cables, pipelines, tunnels and utilities, explosive ordnance, fishing-gear, cavities, over- and under-pressurised groundwater, and shallow gas; (ii) the WORK Geodetic system for horizontal and vertical control.

8. ASSIGNMENT AND SUBCONTRACTING

8.2 a): Insert "except to an Affiliate" after "WORK" in second sentence.

9. CONTRACTOR PERSONNEL

Insert new Clause 9.9 as follows:

9.9. from the EFFECTIVE DATE OF COMMENCEMENT OF THE CONTRACT until twelve (12) months after the CONTRACTOR completes the WORK, the COMPANY GROUP shall not directly or indirectly solicit the employment of, or interfere with or endeavour to entice away from the CONTRACTOR or endeavour to obtain the employment or engagement of, any person who has or had been allocated or assigned to the performance of the WORK by the CONTRACTOR provided that this prohibition shall not apply where any such person responds unsolicited to advertisements placed in the media by COMPANY.

10. Examination and defects correction

10.2. Add the word "material" before "defect" on each occasion it appears throughout the clause.

10.2 (b) At the end of sub-clause add "For the avoidance of doubt, CONTRACTOR's liability to reacquire data or rectify defective WORK shall cease immediately upon CONTRACTOR's departure from the offshore WORKSITE. Following CONTRACTOR's departure from the offshore WORKSITE, CONTRACTOR's liability to rectify defects will be limited to the re-analysis of data and re-issuing of reports for a maximum of two (2) separate occasions expiring twelve (12) months from receipt by COMPANY of CONTRACTOR's final report".

10.2.(c) Delete the first sentence and at the start of the second sentence insert "If the CONTRACTOR fails to remedy any defect within a reasonable period of time of being notified by COMPANY, such notification to be provided by COMPANY prior to CONTRACTOR's demobilisation and departure from the WORKSITE,

Add to Clause 10.2(d) "Notwithstanding anything to the contrary included elsewhere in this CONTRACT, CONTRACTOR shall not be liable for defects caused by CONTRACTOR's reliance on TECHNICAL INFORMATION or any other information, instruction or direction provided by the COMPANY".

Page 7 of 34



Add new Clause 10.2 (e) as follows:

10.2 (e)"Notwithstanding anything to the contrary included elsewhere in the CONTRACT, CONTRACTOR's liability for error or mistake in the performance of the WORK shall be limited only to CONTRACTOR's charges relating to the WORK executed erroneously in proportion to the extent of the error or mistake or, where feasible and reasonable, the re-execution of such offshore WORK at no further charge to COMPANY, subject to CONTRACTOR being notified of any error or mistake prior to CONTRACTOR's demobilisation and departure from the offshore WORKSITE following final scheduled WORK activities for that phase of the WORK.

Notwithstanding the above, if a component part of the CONTRACTOR's equipment spread is not functional but WORK can continue either with the scheduled task or other activities required to progress the WORK, this shall not be considered downtime or breakdown".

Insert new clause 10.3 as follows:

10.3 "Notwithstanding anything to the contrary the foregoing warranties shall be exclusive and in lieu of any other warranty whether express or implied or otherwise at law".

Insert new clause 10.4 as follows:

10.4 "CONTRACTOR shall ensure regular quality checks are performed in order to confirm the accuracy and integrity of the survey data being acquired by CONTRACTOR. The CONTRACTOR shall raise any anomalies or inconsistencies in the survey data acquired by it to the COMPANY without delay".

11. Variations

At the end of Clause 11.2 add "by agreement between the parties".

Insert new clause 11.3 as follows:

"11.3 CONTRACTOR shall be entitled to a VARIATION and a consequent adjustment to the CONTRACT PRICE and SCHEDULE, if the COMPANY, through its instructions, actions or lack of actions, has varied the scope of the WORK or the time for performance of the WORK or if the COMPANY has failed to provide unobstructed access to the WORKSITE."

12. Force Majeure

12.2 Amend opening words to read "For the purposes of this CONTRACT, force majeure shall include but not be limited to:"

Insert new clause 12.6 as follows:

12.6 In the event that any Force Majeure occurrence lasts longer than seven (7) days, COMPANY may instruct CONTRACTOR to remain on standby at a safe location, in which case COMPANY shall bear the associated standby costs set forth in Section III Remuneration. If standby option is not exercised by COMPANY, either COMPANY or CONTRACTOR can terminate the PURCHASE ORDER. Having elected to retain CONTRACTOR on standby and the delay due to FORCE MAJEURE prolongs, COMPANY can terminate the PURCHASE ORDER. If the PURCHASE ORDER is terminated through such FORCE MAJEURE event, COMPANY shall pay CONTRACTOR all outstanding sums due for the WORK completed to date of FORCE MAJEUE occurrence. Upon cessation of any FORCE MAJEURE occurrence the Parties may agree a revised programme to include for rescheduling of the WORK so as to minimise the effects of the delay which will be subject to CONTRACTOR'S other contractual commitments. In such event COMPANY shall reimburse CONTRACTOR for the costs of re-mobilisation.



13. Suspension

Insert the word 'material' before the word 'default' wherever appearing

13.5. Add at the end "Additional costs for which the CONTRACTOR is liable under this clause shall be documented and reasonable".

13.8: Change "fourteen (14) days" to "seven (7) days" on both occasions

Insert new clauses 13.9 - 13.10. as follows:

13.9. CONTRACTOR shall have the right, upon notice to the COMPANY, to suspend the WORK or any part thereof to the extent detailed in the notice, for failure of COMPANY to meet its obligations under this CONTRACT, including but not limited to failure to make payment of the amounts due within 30 days of notice from CONTRACTOR that amounts are overdue and non-fulfilment of the obligation to inform CONTRACTOR of hazards pursuant to clause 7.4;

13.10. COMPANY agrees that CONTRACTOR shall not be liable to COMPANY for, and indemnifies CONTRACTOR from and against, any loss or damage or other liability suffered or incurred by COMPANY arising out of or in connection with suspension by CONTRACTOR under clause 13.9. In addition, and for the avoidance of doubt, Clause 13.4 shall continue to apply in the event of suspension in accordance with Clause 13.9.

18. Laws and regulations

18.2: Delete and replace with; "CONTRACTOR shall, in due time, obtain and maintain such approvals and permits as are necessary for the performance of the WORK and which must be obtained in the name of CONTRACTOR. COMPANY shall provide any necessary assistance in this connection.

COMPANY shall, in due time, obtain and maintain all other approvals and permits, such as the right to enter into and exit from the WORKSITE.

COMPANY shall advise CONTRACTOR of any limitations or restrictions affecting the right of entry to the WORKSITE, and CONTRACTOR shall abide by such limitations or restrictions, but CONTRACTOR shall not be liable for failure to comply with the terms of any contract or permit of COMPANY of which CONTRACTOR is not aware and could not reasonably be expected to have known about. "

19. Indemnities

Add new Clause 19.1(e) as follows:

Survey operations that require intrusion of equipment or goods into the seabed (ground) will cause a change in the physical, chemical and hydraulic properties of the seabed (ground). In such instances, the COMPANY shall be responsible for, and shall indemnify CONTRACTOR GROUP against all consequences of such change notwithstanding he negligence or breach of duty (statutory or otherwise) of CONTRACTOR.

Add new Clause 19.1 (f) as follows:

Any such claims, losses, damages, costs (including legal costs) expenses and liabilities that may be brought by or in respect of any personnel employed by the CONTRACTOR against the COMPANY pursuant to any loss of employment of such personnel with the CONTRACTOR.

Page 9 of 34



Delete clause 19.2(d) and replace with "loss of, damage to or recovery of any third-party infrastructure within 500m of the WORKSITE including but not limited to oil and gas and telecommunications infrastructure and consequential losses arising therefrom where such loss, damage or recovery arise from or in connection with the performance or non-performance of the CONTRACT. This Clause 19.2(d) shall apply notwithstanding Clause 19.1(c)."

20. Insurance by Contractor

20.1 Insert at the end of the clause the following "All insurance provided by COMPANY GROUP shall be endorsed to provide that underwriters waive any rights of recourse, including in particular subrogation rights against the CONTRACTOR GROUP in relation to the CONTRACT to the extent of the liabilities assumed by the COMPANY GROUP under the CONTRACT".

Insert New Clause 20.5 as follows:

CONTRACTOR's rates and pricing included within Section III Remuneration do not make allowance for additional insurance premiums required to extend insurance in relation to interaction with unexploded ordnances (UXOs). If any such operation is required, CONTRACTOR shall be entitled to a VARIATION to reflect additional insurance premium/s, if any. In the event that CONTRACTOR's insurer declines to extend insurance in respect of UXO interaction, CONTRACTOR and COMPANY shall discuss whether to proceed and if so, make an appropriate reapportionment of risk".

24. Termination

Insert the word "material" before "default" throughout this provision

24.4. Insert at the end of the Clause, the following "In addition to the above, in the event of such termination the COMPANY shall reimburse CONTRACTOR the applicable cancellation fee as further detailed within Section III Remuneration."

Insert a new clause 24.8 as follows:

24.8. The CONTRACTOR shall have the right by giving notice to terminate the CONTRACT and/or any CALL-OFF ORDER for any or all of the following reasons:

a) in the event of a material default on the part of the COMPANY. Failure by COMPANY to make payment of CONTRACTOR's correctly prepared and adequately supported invoices within forty-five (45) days of such invoice due date shall be deemed to be a material default hereunder; or

b) in the event of the COMPANY becoming bankrupt or making a composition or arrangement with its creditors or a winding-up order of the COMPANY being made or (except for the purposes of amalgamation or reconstruction) a resolution for its voluntary winding-up being passed or a provisional Liquidator, Receiver, Administrator or Manager of its business or undertaking being appointed or presenting a petition or having a petition presented applying for an administration order to be made pursuant to Section 9 Insolvency Act 1986, or possession being taken by or on behalf of the holders of any debenture secured by a Floating Charge of any property comprised in or subject to the Floating Charge, or any equivalent act or thing being done or suffered under any applicable law".

Page 10 of 34

25. AUDIT AND STORAGE OF DOCUMENTS



25.4: Please change "six (6) years" to "two (2) years"

26. Liens

26.1. To be deleted.

33. WAR RISK INSURANCE

"CONTRACTOR insurance coverage extends worldwide, but in the event of personnel, a vessel or craft insured sailing for, deviating towards, or being within the Territorial Waters of any of the Countries or places described in the Current Exclusions as set out in the attached notice (or current version thereof), CONTRACTOR must notify its insurers who will advise if an additional premium will be applicable, or, indeed, if insurance is withdrawn. Where this occurs during the currency of a contract, CONTRACTOR shall be reimbursed the amount of such additional premium at cost. In the event that insurance is withdrawn, clearly CONTRACTOR would have discussions with the COMPANY concerned as a decision would be required on whether to proceed".

34. LIQUIDATED DAMAGES

"Liquidated Damages shall be agreed on a case by case basis and detailed in the relevant CALL-OFF ORDER and shall apply only when COMPANY GROUP suffers loss as a consequence of later performance by CONTRACTOR GROUP. Notwithstanding anything included elsewhere in the CONTRACT to the contrary, the amount of such Liquidated Damages shall accrue at a rate of zero-point three percent (0.3%) of the CONTRACTOR's vessel day rate per day, as further identified in Section III Remuneration herein, and shall be subject to an aggregate and/or cumulative cap of a sum equal to ten percent (10%) of the relevant CALL-OFF ORDER value. Such Liquidated Damages shall be the COMPANY GROUP's sole and exclusive remedy and CONTRACTOR GROUP's sole and exclusive liability in respect of delay in the late commencement, performance or completion of the WORK under this CONTRACT".

35. GLOBAL NAVIGATION SATELLITE SYSTEM

Insert new Clause 35 as follows:

35.1. The CONTRACTOR has no control over any part and shall not be liable to the COMPANY GROUP in respect of the WORK to the extent that such failure of performance is due to the failure of the Global Navigation Satellite Systems space and ground segments ("GNSS"), as described below. For the avoidance of doubt, any warranty provided herein shall specifically exclude the GNSS.

35.2. COMPANY acknowledges as follows:

(a) GNSS has been developed and is operated by the United States Department of Defence and the Russian Federal Space Agency ("the GNSS Operators") and if the level of performance of the GNSS is impaired or ceases operation due to any act or omission of the GNSS Operators and/or any other authority and/or regulatory body which may now or later have responsibility for any aspects of the GNSS, then CONTRACTOR shall not be held responsible for any such act or omission, regardless of whether a complete or partial withdrawal or modification of the GNSS results; and

(b) Differential GNSS ("DGNSS") relies upon the availability of normal uninterrupted satellite and telecommunications services; power supplies (including electrical supplies); equipment which may be owned or controlled by COMPANY or third parties (other than members of

Page 11 of 34



CONTRACTOR GROUP), and to the extent that the unavailability of such services, supplies or equipment affects the performance of the WORK, the CONTRACTOR shall not be liable therefor.

35.3. COMPANY agrees that the CONTRACTOR shall not be liable to COMPANY for, and indemnifies CONTRACTOR GROUP from and against, any loss or damage or other liability suffered or incurred by COMPANY GROUP arising out of or in connection with the provision of or failure to provide DGNSS; or the unavailability, malfunctioning, interruption or other defect in such services, supplies or equipment referred to in this clause hereof for which third parties or COMPANY are responsible pursuant to Clause 35.2 above. No such unavailability, malfunctioning, interruption, non-operation or other defect shall relieve COMPANY of any of its obligations in the CONTRACT.

Page 12 of 34



SECTION III REMUNERATION

1.0 GENERAL

1.1 The terms and conditions set forth in this Section III Remuneration specify the basis for compensating CONTRACTOR for WORK which CONTRACTOR shall provide to COMPANY or its AFFILIATE under this CONTRACT and an agreed CALL-OFF ORDER. The rates, prices and principles set out in this Section III Remuneration shall be applied to define the CALL-OFF ORDER PRICE and paid to CONTRACTOR as a result of the satisfactory execution of the WORK in accordance with the requirements of the CONTRACT and the agreed CALL-OFF ORDER.

1.2 The rates and prices specified herein shall be deemed as all-inclusive to allow for all costs incurred by the CONTRACTOR in complying with the terms and conditions of the CONTRACT and the agreed CALL-OFF ORDER, and shall represent the total remuneration due to the CONTRACTOR in full consideration of the satisfactory performance of the WORK performed.

2.0 BASIS OF RATES AND PRICES

2.1 The rates and prices set forth herein are fully inclusive of all costs (including both direct and indirect costs), management and administrative costs, overhead and profit and shall apply equally to performing the WORK in large or small quantities, in any sequence, or in difficult or easy situations and whether the WORK is performed by the CONTRACTOR or its SUBCONTRACTORS.

2.3 The currency of the agreement is pounds sterling (GBP (£)). All lump sum amounts, unit rates, and other pricing shall be as detailed and all invoices prepared, submitted and paid in pounds sterling (GBP (£)).

3.0 BASIS OF RATES

3.1 The rates set forth herein shall include but not be limited to, the cost of salaries and wages actually paid to personnel, cost of all CONTRACTOR's payroll burdens, benefits and contributions, all taxes (except VAT) and insurance requirements in relation to working at CONTRACTOR offices or any other worksite and in full compliance with all terms and conditions of the CONTRACT to carry out the WORK as described in Section IV Scope of Work and the agreed CALL-OFF ORDER.

3.2 Value Added Tax is excluded from the rates, sums and percentages herein. CONTRACTOR shall obtain for the benefit of COMPANY all available exemption from import duties, investment tax, Value Added Tax and all similar taxes and shall pass on to COMPANY any and all rebates.

4.0 COMPENSATION

4.1 CONTRACTOR shall be compensated for the satisfactory performance of the WORK in accordance with the rates, sums and prices contained within the rate tables included below. Any additional or project specific equipment shall be priced on a case by case basis and identified within the applicable CALL-OFF ORDER:

Management of COMPANY owned Q1400 Trencher Systems (2x) & Management and Operation of COMPANY owned & FCV 3000 WROV ROV Systems (2x) and the supply of Survey services onboard the Fugro Symphony.

Page 13 of 34



Management of COMPANY owned Q1400 Trencher Systems (2x):

Item	Price (GBP)
Monthly Lump Sum charge for the provision of Management of COMPANY owned Q1400 Trencher Systems *	£

Pricing Notes:

- * CONTRACTOR'S Monthly Lump Sum charge for the provision of Management of COMPANY owned Q1400 Trencher Systems (2x) includes the following:
- In-house Project Management;
- Access to onshore/in-house Technical Support;
- Management of the maintenance of Q1400 Trencher Systems (2x);
- Management of the procurement and supply and inventory management of suitable level of spares for Q1400 Trencher Systems(2x);

Purchase of Spares for COMPANY owned Q1400 Trencher Systems (2x)

Item	Price (GBP)
Purchase of Spares for COMPANY owned Q1400 Trencher Systems (2x), subject to COMPANY's prior approval *	Cost + 10%
Packaging and Transportation of Spares for COMPANY owned Q1400 Trencher Systems (2x)	Cost + 10%

Pricing Notes: * Process for the approval by COMPANY to proceed with the purchase by CONTRACTOR of identified Spares for COMPANY owned Q1400 Trencher Systems (2x) to be identified. However, and notwithstanding anything to the contrary included elsewhere in the CONTRACT, CONTRACTOR accepts no liability or responsibility of any kind for any losses, costs or claims associated with COMPANY owned Q1400 Trencher Systems breakdown or failure as a result of COMPANY's failure to approve the purchase of Spares equipment identified and recommended by CONTRACTOR and in such event CONTRACTOR's personnel and equipment Daily Hire Rates shall continue to apply.

Management of COMPANY owned FCV 3000 WROV ROV Systems (2x) onboard the Fugro Symphony;

Item	Price (GBP)	
Monthly Lump Sum charge for the provision of Management of COMPANY owned FCV 3000 (2x) WROV ROV System's on board the Fugro Symphony *	t	

Page 14 of 34



Pricing Notes:

- * CONTRACTOR's Monthly Lump Sum charge for the provision of Management of COMPANY owned FCV 3000 WROV ROV Systems (2x) on board the Fugro Symphony includes the following:
- In-house Project Management;
- Access to onshore/in-house Technical Support;
- Management of the maintenance of FCV 3000 WROV ROV Systems (2x);
- Management of the procurement and supply and inventory management of suitable level of spares for FCV 3000 WROV ROV Systems (2x);

Purchase of Spares for COMPANY owned FCV 3000 WROV ROV System (2x)

Item	Price (GBP)		
Purchase of Spares for COMPANY owned FCV 3000 WROV ROV System (2x), subject to COMPANY's prior approval *	Cost + 10%		
Packaging and Transportation of Spares for COMPANY owned FCV 3000 WROV ROV System (2x)	Cost + 10%		

Pricing Notes: * Process for the approval by COMPANY to proceed with the purchase by CONTRACTOR of identified Spares for COMPANY owned FCV 3000 WROV ROV System (2x) to be identified. However, and notwithstanding anything to the contrary included elsewhere in the CONTRACT, CONTRACTOR accepts no liability or responsibility of any kind for any losses, costs or claims associated with COMPANY owned FCV 3000 WROV ROV System breakdown or failure as a result of COMPANY's failure to approve the purchase of Spares equipment identified and recommended by CONTRACTOR and in such event CONTRACTOR's personnel and equipment Daily Hire Rates shall continue to apply;

CONTRACTOR owned Associated ROV Equipment (FCV 3000 WROV ROV System compatible)

Page 15 of 34



Mobilisation and Demobilisation of CONTRACTOR owned Associated ROV Equipment

Item	Price (GBP)
Transportation of CONTRACTOR Associated ROV Equipment and necessary materials from CONTRACTOR's base in Aberdeen to COMPANY's nominated Vessel mobilisation Quayside *, **	Cost + 10%
Transportation of CONTRACTOR Associated ROV Equipment and necessary materials from COMPANY's nominated Vessel demobilisation Quayside to CONTRACTOR's base in Aberdeen *, **	Cost + 10%

Pricing Notes: * CONTRACTOR's transportation fee for CONTRACTOR's Associated ROV Equipment includes the following:

- Pre-transportation checks/re-delivery checks;
- Certifying authority release note in Aberdeen;
- Manifest and loading;
- Delivery/re-delivery of CONTRACTOR Associated ROV Equipment to/from CONTRACTOR's base in Aberdeen and COMPANY's nominated mobilisation/demobilisation Quayside;

** CONTRACTOR's transportation fee for CONTRACTOR's Associated ROV Equipment excludes the following:

- All port services;
- Quayside and Vessel craneage;
- Engineering deck calculations/sea-fastening;
- Provision of welders plus load testing services and equipment;

Daily Hire Rates of CONTRACTOR owned Associated ROV Equipment

ltem	Day Rate (GBP)
H4 Tree H4 Mandrel Type VX Gasket Profile Clean-Up Tool	£25
Tool Deployment Unit (TDU)	£
Linear Override Tool	£
Fluid Skid c/w Pump	£
Regan Latch Release Tool	£13
Water Jetting Unit	£40

Page 16 of 34



Pricing Notes:

٠

 COMPANY to provide, at no charge/cost to CONTRACTOR, Subsea Navigation System, Nav screens, Round Robin and clear communications;

Mobilisation and Dempbilisation of CONTRACTOR Personnel for Operation of COMPANY owned FCV 3000 WROV ROV System (2x)

Item	Price (GBP)	
Personnel Mobilisation, per person *	Cost + 10%	
Personnel Demobilisation, per person *	Cost + 10%	

Pricing Notes: * CONTRACTOR's Personnel Mobilisation and Demobilisation fee includes the following:

Transportation to/from COMPANY nominated Vessel mobilisation/demobilisation Quayside from CONTRACTOR's Personnel original point of origin.

Daily Hire Rates of CONTRACTOR Personnel for Operation of COMPANY owned FCV 3000 WROV ROV System (2x)

Discipline	Day Rate (GBP)	
ROV Supervisor, per person	£	
ROV Senior Pilot Technician, per person	£	
ROV Pilot Technician, per person	£	

Daily Hire Rates and Lump Sum for CONTRACTOR provided Survey Services on-board the Fugro Symphony

	SECTION	Day Rate / LS	Hire of CONTRACTOR Resources (GBP)
1.0	Survey Services for COMPANY provided Vessels requiring Survey Services on long term commitment basis		
	Day Rates for Survey Personnel (Europe)		
	a) Party Chief, per person	D/Rate	510
1.1	b) All other disciplines, per person	D/Rate	480
	c) Mobilisation / Demobilisation of CONTRACTOR Personnel, per person	L/Sum	Cost + 10%

Page 17 of 34



1.2	Starfix DGNSS Signals (Primary and Secondary) and Hardware, worldwide – per vessel Usage	P/year P/month P/day	48,000 4,800 240
1.3	Standard spread, hardware and software – per vessel (see Standard spread as further identified under heading " <u>CONTRACTOR STANDARD SURVEY EQUIPMENT</u> <u>SPREAD, ROV SYSTEM MANNING LEVELS AND</u> <u>SURVEY EQUIPMENT MANNING LEVELS"</u>) Package Hardware Online Survey System (when used) Offline Processing System (when used) Digital Video System (when used)	/Month /Month /Month /Month	3,200 1,900 2,200 3,000
1.4	Starfix Processing suite per onshore suite		included in item 1.3
1.1.5	In-house Processing/Reporting Services – per man/day	D/Rate	565
1.1.0	Per man/hour	Hourly Rate	40
2.0	TRANSPONDERS		
2.1	Mobilisation of MF/EHF LBL Transponder		
	a) N.E Scottish Port	Lump Sum	79
	b) Norway	Lump Sum	247
2.2	Demobilisation of MF/EHF LBL Transponder		ļ
	a) N.E Scottish Port	Lump Sum	49
	b) Norway	Lump Sum	185
2.3	Mobilisation of USBL Transponder to NE Scottish Port	Lump Sum	79
2.4	Demobilisation of USBL Transponder from NE Scottish	Lump	49
2.4	Port	Sum	

Page 18 of 34



0.5			
2.5	LBL Transponder Day rates		
2.5.1	MKV Compatt MF	Day Rate	35
2.5.1	MKV Compatt MF c/w SVS Option	Day	40
		Rate	
2.5.3	MKV Compatt MF c/w Digiquartz/Inclinometer Option	Day Rate	48
2.5.4	MKV Compatt MF c/w SVS and High Precision Strain Gauge Option	Day Rate	52
2.5.5	6G Compatt MF	Day Rate	50
2.5.6	6G Compatt MF c/w SVS Option	Day Rate	60
2.5.7	6G Compatt MF c/w Digiquartz Option	Day Rate	65
2.5.8	6G Compatt MF c/w SVS and High Precision Strain Gauge Option	Day Rate	60
2.5.9	Flotation Collar	Day Rate	3
2.5.10	Floatation Collar Upgrade for 400+ meters	Day Rate	3
2.6	USBL Transponder Day Rates		
2.6.1	Mini HPR USBL Transponder	Day Rate	21
2.6.2	Simrad RPT319 Mini Beacon	Day Rate	21
2.6.3	Standard HPR USBL Transponder	Day Rate	21
2.6.4	SPT HiPAP Transponder	Day Rate	21
2.6.5	Battery charger	Day Rate	6
2.7	Transponder Moorings		
2.7.1	Sandbag Mooring assembly / Seabed stand per set / per deployment (Preparation)	Lump Sum	53
2.7.2	Quadroped seabed stand (each)	Day Rate	11
2.7.3	Grimsby Bouy	Lump Sum	26
2.7.4	Sonar Reflector	Lump	25

		Sum	
2.7.5	Strop	Lump Sum	6
2.7.6	Bow Shackle	Lump Sum	5
2.7.8	Sandbag	Lump Sum	7

Page 19 of 34



Pricing Notes:

- Please note that the above item 1.3 "Standard spread, hardware and software" Charges per month shall be charged on a when used/prorata basis, however each minimum hire period shall be for a period of no less than seven (7) continuous days;
- In the event of CONTRACTOR Survey Equipment breakdown and should the CONTRACTOR Survey Equipment be unavailable to
 perform meaningful WORK then CONTRACTOR's Survey Equipment, associated personnel and associated equipment shall be deemed
 off-hire until such time as the CONTRACTOR's Survey Equipment is again available to commence meaningful WORK;
- For the avoidance of doubt, CONTRACTOR's hire Day Rates apply per each twenty-four (24) hrs period, from midnight to midnight;
- Any Survey Equipment cross hired from third parties by CONTRACTOR shall be provided at Cost+ 10%;
- Except where stated otherwise, a mobilisation/demobilisation lump sum of 1 x the applicable Daily Hire Rate will be applicable for all CONTRACTOR Survey Equipment to cover preparation, maintenance and consumable costs;
- All freight costs for CONTRACTOR's Survey Equipment to be for COMPANY's account, mobilization/demobilization included above are for equipment preparation and return acceptance only;
- CONTRACTOR's Personnel mobilisation/demobilisation fee includes for transportation to/from COMPANY nominated Vessel
 mobilisation/demobilisation Quayside from CONTARCTOR Personnel original point of origin;

Additional CONTRACTOR's pricing notes applicable to all of the above rates, sums and price tables under this Section 4.0 COMPENSATION of Section III Remuneration;

• The above rates, sums and prices shall remain valid up to and including 31st December 2018, thereafter the above rates, sums and prices shall be subject to escalation by the mutual agreement of both COMPANY and CONTRACTOR;

Page 20 of 34



- The above rates, sums and prices are based strictly on WORK being performed within the UKCS only. Should COMPANY request WORK out
 with the UKCS, CONTRACTOR reserves the right to adjust the above rates, sums and prices accordingly;
- The above rates, sums and prices are exclusive of VAT;
- For the avoidance of doubt, CONTRACTOR's hire Day Rates for the above listed Associated ROV Equipment and Survey Equipment shall commence immediately upon departure from CONTRACTOR's base in Aberdeen and shall apply for each and every day until return to same and will not be pro-rated for part days. The Associated ROV Equipment and Survey Equipment shall remain on hire during periods of nonutilisation of the COMPANY Vessel;
- For the avoidance of doubt, CONTRACTOR's hire Day Rates for CONTRACTOR Personnel shall commence immediately upon CONTRACTOR's Personnel departure from their original point of origin and shall apply for each and every day until return to same;
- Except where otherwise agreed, CONTRACTOR's Personnel crew changes for the WORK shall be on twenty-eight (28) day rotation with
 overlap charges applicable on days where both CONTRACTOR Personnel are simultaneously on board the COMPANY Vessel, travelling
 offshore/returning or standing by at the crew change mobilisation point. A mobilisation/demobilisation fee will be charged per person per
 direction of travel;

COMPANY PROVIDED ITEMS

The COMPANY shall provide the following, at no charge/cost to CONTRACTOR:

- All support flights to/from Fugro Symphony for all CONTRACTOR Associated ROV Equipment, Survey Equipment and all associated Spares, when required, along with supply vessels for such CONTRACTOR equipment shipment;
- All support flights to/from Fugro Symphony for all CONTRACTOR Personnel;
- Suitable, clean, stable motor rated power supplies for the operation of COMPANY's owned FCV 3000 WROV ROV Systems, CONTRACTOR's Associated ROV Equipment and CONTRACTOR's Survey Equipment and with access to deck power within twenty (20) meters of the COMPANY owned FCV 3000 WROV ROV System. If this is not available or cannot be guaranteed, then the CONTRACTOR will provide a suitable diesel power generator at documented cost + 10%.
- · Cooling water and fresh water supplied to the COMPANY owned FCV 3000 WROV ROV System and CONTRACTOR Associated ROV Equipment;
- Accommodation subsistence / victualling for all CONTRACTOR Personnel while on-board Fugro Symphony;
- Reasonable business communications from Fugro Symphony to shore and UK office for CONTRACTOR Personnel. (Access to internet & email system from ROV control container);
- In the event that offshore transportation for CONTRACTOR Personnel is delayed or cancelled six (6) hrs or less prior to the arrival of CONTRACTOR's
 Personnel at COMPANY nominated quayside or heliport, a cancellation fee of £250 per person and the applicable Personnel hire Day Rates will apply;

Page 21 of 34



- Should the delay of transportation mean that additional overnight accommodation is required for CONTRACTOR Personnel, all such costs shall be recharged to COMPANY at cost + 10% and shall apply over and above the mobilisation lump sum and Personnel hire Day Rates;
- If a twelve (12) hr ROV team is required by COMPANY note that only twelve (12) hrs will be worked per day by CONTRACTOR Personnel including dive
 operations and maintenance;

CONTRACTOR STANDARD SURVEY EQUIPMENT SPREAD & FCV 3000 ROV SYSTEM MANNING LEVELS AND SURVEY EQUIPMENT MANNING LEVELS

Unless otherwise agreed within the applicable CALL-OFF ORDER, CONTRACTOR's standard Survey Equipment Spread on board the Fugro Symphony shall be as follows:

NOTE:

The survey spread provided on the Symphony is a full construction support survey acquisition and processing system, based on Fugro Starfix software.

This includes a large number of specific acquisition modules and solutions such as:

- MBE Acquisition
- INS (Inertial Navigation engine)
- Eventing module
- Digital Video acquisition and archive solution (4 Channel)
- Back2Base data compression and transmission solution.
- Unlimited software instances around the vessel (subject to Network constraints)

The survey package also include three offline processing licences including

- Multibeam Processing and DTM engine
- INS Post Processing solution
- Digital Video Playback
- Offline Eventing Module
- Automated Charting

The survey package also includes external software for 1x Microsoft SQL Server 2x Autocad Map software

The survey spread includes the annual calibration cost for the CTD/SVP Probes (2) and the Digital Protractor.

Hardware	Comment	Quantity
Starpak Novael	Primary GNSS receiver + spare	2
Starpack Trimble	Secondary GNSS receiver	1
AD492	Standard GNSS and Lband antenna	3
AD493	High latitude Lband antenna	1
Processing PC	Tower PC	3
21" Monitor	For Processing PC	6
Online PC	Rack Mount PC	5
20" Monitor	For Online PC's	10
Starfix Smart Remotes	Small Form Factor PC	8

Page 22 of 34

f	UGRO
E.	\approx
Ľ	\rightarrow

19" Monitor	For Smart Remotes	7
15" Monitor	For Crane	1
Starport	Serial to Network Converter	5
PPS Distribution	PPS	2
Starfix DV System (including spares)	DV	1
Quad Splitter	DV	1
LT04	DV	2
СТD	CTD	2
A0 Plotter	Plotter	1
Digital Protractor	Calibration	1
Disto	Calibration	2
Starfix Suite – as required for above PC's	Software	
AutoCad Map	Software	2

CONTRACTOR standard manning levels for operation of COMPANY owned FCV 3000 WROV ROV Systems

Unless otherwise agreed within the applicable CALL-OFF ORDER, CONTRACTOR's standard manning levels for operation of COMPANY owned FCV 3000 WROV ROV Systems shall be as follows:

• FCV 3000 WROV ROV System; per system, per twelve (12) hours shift;

1 x ROV Supervisor, per WROV System;

2 x ROV Pilot Technician, per WROV System;

• FCV 3000 WROV ROV System, per system, per twenty-four (24) hour shift;

2 x ROV Supervisor, per WROV System;

4 x ROV Pilot Technician, per WROV System;

CONTRACTOR standard Survey Equipment manning levels

Survey Equipment Manning Levels is dependant of the specifics of the WORK and will be agreed on a case by case basis and identified within the applicable CALL-OFF ORDER;

In this Section III the words and expressions established as definitions in both the Conditions of Contract, the Scope of Services and Appendices, shall have the same meanings as assigned to them in the said Conditions of Contract, Scope of Services and Appendices except when otherwise specifically stated.

5.0 VARIATIONS

5.1 Any Variation in the WORK shall be valued in strict accordance with the rates and prices contained within Clause 4 COMPENSATION.

5.2 If the Variation WORK is of a differing nature and cannot be readily evaluated in accordance with the foregoing rates and prices the COMPANY may request the CONTRACTOR to propose a lump sum for COMPANY's consideration. The CONTRACTOR shall be requested to demonstrate the build up to any lump sum in order to demonstrate that it is priced on the same basis as the rates and prices contained herein and is fair and reasonable.

Page 23 of 34



5.3 Any work that COMPANY instructs CONTRACTOR to undertake on a reimbursable basis shall be remunerated to CONTRACTOR at the cost of the invoice (net) plus ten percent (10%).

6.0 INVOICING

Lump Sum and Unit Rate Invoices shall be invoiced in accordance with the pricing schedule detailed in this section, 4.0 COMPENSATION.

- Reimbursable items shall be fully supported.
- · The invoice shall clearly make reference to the payment stage, Contract No., item Nos., description and quantities.
- Where the WORK is expected to continue for more than one-month duration or extend from one month in to the following month, monthly invoices shall be
 raised, otherwise CONTRACTOR shall be entitled to submit an invoice on successful completion of the WORK.
- Invoices shall be submitted marked for the attention of the COMPANY Representative in the CALL-OFF ORDER.
- The following reference shall be clearly quoted on the invoice:
- CONTRACT No:212-0917; PROJECT Name: XXXX;
- In order to facilitate payment of invoices, CONTRACTOR is required within two weeks of receiving this CONTRACT to furnish COMPANY with details of its bank account to which payment will be made. This should include the Bank's name and address, sort code and CONTRACTOR's account number

Page 24 of 34



SECTION IV SCOPE OF WORK

1.0 INTRODUCTION

1.1 The CONTRACTOR shall be responsible for providing COMPANY with the Provision of Management of COMPANY owned Q1400 Trencher Systems (2x) & Management and Operation of COMPANY owned FCV 3000 ROV Systems (2x) and the supply of Survey Services on board the Fugro Symphony (hereinafter called the "WORK"),

1.2 The specific scope of services to be performed by CONTRACTOR under the CONTRACT is further identified below:

 The CONTRACTOR shall provide the WORK (Provision of Management of COMPANY owned Q1400 Trencher Systems (2x) & Management and Operation of the COMPANY owned FCV 3000 ROV Systems (2x) and the supply of Survey Services on board the Fugro Symphony) according to the specific requirements as agreed between the Parties and as identified within the relevant CALL-OFF ORDER, inclusive of the required equipment and manning levels, estimated durations of such WORK and the specifics in relation to any required deliverables (final Report) and the timeframe for delivery of such deliverables;

WORK COMMENCEMENT, Mobilisation Window:

The WORK shall commence during a Mobilisation Window as further identified below unless otherwise agreed within the applicable CALL-OFF ORDER:

The COMPANY agrees to notify the CONTRACTOR of the mobilisation date in accordance with the following:

- 60 days' notice of a 30-day Mobilisation Window;
- 35 days' notice of a 14-day Mobilisation Window;
- 21 days' notice of a 7-day Mobilisation Window;
- 10 days' notice of a 3-day Mobilisation Window;
- 3 days' notice of actual mobilisation day in agreed port of mobilisation;

Each notified Mobilisation Window and actual mobilisation date following wholly within the previously notified window;

OUTLINE SCOPE OF SERVICES/SUPPLY

The CONTRACTOR shall provide the WORK (Provision of Management of COMPANY owned Q1400 Trencher Systems (2x) & Management and Operation of the COMPANY owned FCV 3000 ROV Systems (2x) and the supply of Survey Services on board the Fugro Symphony) as outlined below, however the below list is not exhaustive and the WORK shall instead be limited to the safe capabilities and resources of the CONTRACTOR personnel and equipment. As above, the specific WORK to be provided shall be identified within the relevant CALL-OFF ORDER.

- Management of COMPANY owned Q1400 Trencher Systems (2x);
- Management of the procurement, supply and inventory management of suitable level of spares for Q1400 Trencher Systems (2x);
- Provision of technical support for Q1400 Trencher Systems (2x);
- Management and Operation of COMPANY owned FCV 3000 WROV ROV Systems on board the Fugro Symphony (2x);
- Management of the procurement, supply and inventory management of suitable level of spares for FCV 3000 WROV ROV System (2x);
- Provision of technical support for FCV 3000 WROV ROV Systems (2x);

Page 25 of 34



- Supply of Offshore ROV Personnel (including ROV Supervisors, ROV Pilot Technicians and ROV Subsea Engineers);
- Supply of additional ROV Tooling/equipment (as specified in the applicable CALL-OFF ORDER);
- Supply of Survey Services on board Fugro Symphony inclusive of equipment spread and associated personnel and data capture, analysis and reporting for operation thereof;
- WORK to be performed in compliance with applicable IMCA standards;
- Standards of reporting deliverables (as specified in the applicable CALL-OFF ORDER);

1.3 Additional services not listed herein may from time-to-time by requested under the provisions of the CONTRACT.

2.0 SCOPE OF SERVICES

2.1 The CONTRACTOR has been selected for the performance of the WORK on the understanding that it is an expert and professional in the class of work involved and that it will use and show all reasonable skill and judgement of such an expert and professional in the performance of the WORK.

The CONTRACTOR shall perform the WORK in accordance with the terms and conditions of the CONTRACT, all applicable legislation, regulations, statutory instruments, codes of practice, guidance notes and recommendations of the CONTRACT extracts, specifications and drawings attached hereto, which are now or may in the future become applicable to the CONTRACT.

• The WORK to be performed by CONTRACTOR will include but not be limited to the provision of all management, engineering, supervision, equipment, materials, consumables and where requested, transport, necessary to perform the WORK.

CONTRACTOR is responsible for providing personnel, resources and equipment required for the proper and expeditious performance of the agreed scope of services in accordance with the CONTRACT.

• COMPANY is committed to preserving the health and safety of people and the protection of the environment throughout its organisation, and the organisations with which it places contracts for the performance of the WORK. Accordingly, the CONTRACTOR shall promote health, safety and environmental protection throughout its organisation and those of its SUBCONTRACTOR's to ensure that the CONTRACT is performed without harm to persons, property or the environment.

2.2 ERRORS AND OMISSIONS

• The review and approval by COMPANY of any procedures, documents or drawings shall only indicate a general requirement and shall not relieve CONTRACTOR of his obligation to comply with the requirements of the CONTRACT.

Any errors or omissions noted by CONTRACTOR shall be immediately brought to the attention of COMPANY.

2.3 DEVIATIONS

• All deviations to this WORK and other referenced specifications or attachments listed in the CONTRACT shall be made in writing and shall require written approval by COMPANY prior to execution of the WORK.

Page 26 of 34



2.4 PERSONNEL (GENERAL)

• CONTRACTOR guarantees that sufficient experienced and competent personnel shall be made available to carry out the WORK, in accordance with the CONTRACT. CONTRACTOR personnel shall be able to maintain a 24 hour a day operation, if required.

2.5 EQUIPMENT (GENERAL)

CONTRACTOR shall supply all equipment required for the WORK in accordance with the CONTRACT.

• Equipment spread and material will be provided by CONTRACTOR in quantity and quality, such as to carry out the WORK in the CONTRACT in a continuous and optimised fashion.

The Equipment Spread selected by CONTRACTOR shall be capable of performing the WORK within the requirements of the CONTRACT.

• CONTRACTOR will supply enough spares in order to avoid any interruptions in the operation. CONTRACTOR shall provide evidence of regular planned maintenance of equipment.

 CONTRACTOR will supply PUWER and LOLER Assessments for all equipment provided for use at CLIENT, COMPANY, COMPANY's other contractors or Third Party managed WORKSITES.

• CONTRACTOR shall ensure that as a minimum all items provided for provision of the WORK comply with industry best practice, regulations, and legislative requirements for the location where WORK is to be performed.

3.0 COMPANY PROVIDED ITEMS

COMPANY shall be responsible for providing vessels, transportation to and from the mobilisation and demobilisation points specified in the CONTRACT and/or agreed CALL-OFF ORDER to the offshore WORKSITE, accommodation, victualing and reasonable access to communications for business purposes while CONTRACTOR and/or SUBCONTRACTOR personnel are at WORKSITES managed by COMPANY or CLIENT.

4.0 MEETINGS

• The COMPANY REPRESENTATIVE may hold regular meetings with the CONTRACTOR to review progress as necessary. These meetings may be attended by additional COMPANY or CLIENT personnel, and shall normally be chaired by the COMPANY REPRSENTATIVE. The attendance of CONTRACTOR or its SUBCONTRACTOR's at any meetings arranged by COMPANY shall be included within the rates and prices identified within Section III Reimbursement of this CONTRACT.

5.0 SERVICE LEVEL AGREEMENT

CONTRACTOR will provide technical support twenty-four (24) hours per day, seven (7) days a week, three hundred and sixty-five (365) days per year, according to the below table:

Page 27 of 34



Description	Response Time
Critical – A failure rendering the asset inoperable	Response within two (2) hours of being notified by On- call Project Manager;
High – Major functionality disabled	Response within four (4) hours of being notified by On-call Project Manager;
Medium – Malfunction of non- essential part	Response within twenty-four (24) hours of being notified by On-call Project Manager;
Low – Minimal impact malfunction	Response within forty-eight (48) hours of being notified by On-call Project Manager;

Note: Timeframes for the resolution of any fault notified to CONTRACTOR in accordance with the above shall be determined on a case by case basis considering the nature and specifics of the actual fault notified.

Page 28 of 34



SECTION V HEALTH, SAFETY AND ENVIRONMENTAL

1.0 SAFETY

CONTRACTOR and its SUBCONTRACTORs shall perform the WORK in a manner such that risk to the health and safety of people or to the environment has been reduced to a level which is as low as reasonably practicable. The WORK shall be performed in accordance with all relevant Statutes, regulations and byelaws, codes of practice concerning safety, health and welfare and environmental protection matters whether current or promulgated at the date of signature of this CONTRACT.

CONTRACTOR shall implement an on-going programme of health, safety and environmental reviews of all stages of the WORK. Any adverse findings shall be addressed immediately by the CONTRACTOR.

CONTRACTOR's organisation may be subject to on-going health, safety or environmental audits at various stages of the WORK. Such audits shall be to measure conformity to relevant statutory requirements and CONTRATOR's health, safety or environmental policy.

2.0 HSE PROCEDURES

Notwithstanding anything to the contrary included elsewhere in the CONTRACT, CONTRACTOR shall comply with and shall perform the WORK in accordance with the COMPANY's/CLIENTS Worksite HSE Management system (Appendix XX attached herein) and with the following basic principles with respect to health, safety and environmental management:

• To be responsible for their own and SUBCONTRACTOR's personnel safety and prevent unsafe behaviour, or infringements against statutory health, safety or environmental rules.

• To co-operate fully with managers and supervisors of the COMPANY at the WORKSITE regarding all work matters including HSE management.

• To perform their WORK scope in accordance with instructions and procedures approved by COMPANY and with the general requirements laid out in COMPANY's specific Worksite HSE Management system.

• To provide equipment which is fit for purpose and with all safety features in place in accordance with statutory health, safety or environmental rules.

• To report all injury, damage (including pollution or spill), occupational illness, or (near miss) incident events as soon as practicable after the event.

It is mandatory that all people working at any WORKSITE wear the correct PPE. All personnel provided by CONTRACTOR shall be provided (by CONTRACTOR) as a minimum, full overall, safety boots / shoes, safety glasses and hard hat for all working areas. For outside work, suitable weather protective clothing and lifejackets if required by working area rules.

Where appropriate, CONTRACTOR's offshore personnel are expected to have documentary evidence to show that they have trained in basic firefighting and survival at sea, either through externally supplied firefighting / survival courses or by appropriate induction / training prior their personnel travelling offshore.

The CONTRACTOR shall ensure that all CONTRACTOR GROUP personnel have an acceptable current medical certificate(s) to prove their fitness to work offshore and must provide on request the certificate(s) for COMPANY's review.

Page 29 of 34



Alcoholic drinks or illegal drugs / controlled substances are prohibited at all WORKSITEs under any circumstances. COMPANY reserves the right to search CONTRACTOR GROUP or any other related personnel and /or equipment. Any person breaking this rule may be dismissed from the WORKSITE. If offshore, they may be returned to shore on the next available transport.

CONTRACTOR GROUP's personnel will be provided with the COMPANY's golden rules of safety when they arrive at WORKSITE. CONTRACTOR GROUP's personnel shall participate in the specific WORKSITE induction, which will cover all worksite safety features, HSE rules and procedures that must be complied with.

CONTRACTOR GROUP shall comply with any local WORKSITE rules.

3.0 COMPANY HSE POLICY

Notwithstanding anything to the contrary included elsewhere in the CONTRACT, CONTRACTOR GROUP shall comply with and perform the WORK in accordance with the COMPANY's/CLIENT's HSE Policy which is attached hereto (Appendix XX).

4.0 QUALITY CONTROL AND QUALITY ASSURANCE

• CONTRACTOR GROUP shall operate a quality system in accordance with ISO 9001, ISO 14001 and OHSAS 18001 series of standards, which shall be maintained by a group of persons sufficient in number and competence to ensure its effective implementation.

• CONTRACTOR GROUP shall carry out the WORK using equipment, processes, procedures and personnel fully qualified to, and compliant with the CONTRACT and of best industry practice.

• CONTRACTOR GROUP shall ensure that all plant, equipment, materials and consumables required for the WORK undergo quality control checks within a sufficient time prior to despatch of the items from premises of CONTRACTOR or its SUBCONTRACTORs.

• CONTRACTOR GROUP's quality control department, and those of its SUBCONTRACTORs, shall be responsible for all inspection, testing and certification, and shall be independent of design and production activities.

• The qualifications of inspection and testing personnel shall be subject to approval and the complete adherence to approved testing and inspection procedures shall be subject to regular monitoring/surveillance and periodic audit.

• CONTRACTOR may be requested to provide information, advice and expertise to be expected of a professional contractor for incorporation into COMPANY's project quality plans.

• CONTRACTOR shall allow COMPANY or CLIENTS QA auditor's full access to CONTRACTOR's and or any SUBCONTRACTOR's premises, personnel records and documentation, as permitted by law, for the purpose of conducting quality audits.

• Where non-conformances are identified during audits and/or inspections by COMPANY or its CLIENT, CONTRACTOR shall undertake the corrective actions as required by the COMPANY, within agreed specified time limits and at no cost to the COMPANY.

Notwithstanding anything to the contrary included elsewhere in this CONTRACT, COMPANY shall perform the WORK all in accordance with the quality systems identified within this CONTRACT.

Page 30 of 34



SECTION VI ADMINISTRATION INSTRUCTIONS

1. GENERAL

1.1. This Section of the CONTRACT details the procedures that shall be used by COMPANY and CONTRACTOR to administer the CONTRACT.

2. COMMUNICATIONS

- 2.1. Correspondence shall be in English and by letter, electronic mail, or fax.
- 2.2. All correspondence and communications in relation to the CONTRACT shall be addressed and directed between the COMPANY REPRESENTATIVE and the CONTRACTOR REPRESENTATIVE or their nominated deputies.
- 2.3. All mailing shall be by first class mail/airmail, courier or other equally prompt means.
- 2.4. Oral communication of instructions or information in connection with the CONTRACT shall be confirmed in writing using minutes of meetings or formal correspondence as appropriate.

3. VARIATION PROCEDURE

- 3.1. The rights of COMPANY to issue VARIATIONS are set out in the General Conditions of CONTRACT, the Clause headed VARIATIONS. The COMPANY shall issue an instruction to vary the WORK using a VARIATION order or form.
- 3.2. The rights of the CONTRACTOR to issue a VARIATION are set out in the General Conditions of CONTRACT, the Clause headed VARIATIONS. If CONTRACTOR considers that an occurrence, which shall include the COMPANY instruction, interpretation, decision or act has taken place that should give rise to a VARIATION, then the CONTRACTOR shall so inform the COMPANY immediately detailing each of the following:
 - a. The specific occurrence.
 - b. The date of the occurrence.
 - c. The part of the relevant section of the CONTRACT under which the CONTRACTOR considers that the occurrence should give rise to a VARIATION.
 - d. The impact of such occurrence on the CONTRACT duration and the effect on the CONTRACT PRICE.
- 3.3. CONTRACTOR shall quantify the request for a VARIATION by giving the COMPANY detailed supporting data within a period of fourteen (14) days of occurrence or such longer time as COMPANY may agree for any specific VARIATION.
- 3.4. If CONTRACTOR fails to inform COMPANY of the occurrence, CONTRACTOR shall forfeit any entitlement to adjustment to the CONTRACT PRICE and/or completion date.

However, to the extent that COMPANY judges that non-submission of said notice by the CONTRACTOR was unavoidable, the COMPANY may, at its sole discretion, elect to extend the time for the CONTRACTOR to so inform the COMPANY.

Page 31 of 34



- 3.5. Upon receipt by the COMPANY of information relating to each occurrence and advice relating to the effect of such occurrence, the COMPANY shall investigate the circumstances of the occurrence and the CONTRACTOR shall be advised in writing of the acceptance or otherwise of any request for a VARIATION within fourteen (14) days of receipt of the CONTRACTOR's request or such longer time as CONTRACTOR may agree for any specific VARIATION.
- 3.6. In the event that the CONTRACTOR's request is accepted in part or in whole by the COMPANY, the CONTRACTOR shall be so advised and the COMPANY shall authorise a VARIATION.
- 3.7. COMPANY shall generally not consider any CONTRACTOR's requests made on the basis of matters not included within the CONTRACT, but should the CONTRACTOR consider that any occurrence arises on the basis of such matters, and that such an occurrence entitles the CONTRACTOR to a VARIATION, then the CONTRACTOR shall immediately so inform the COMPANY in writing in the same as detailed in Clause 3.2 above. The CONTRACTOR shall continue to keep the COMPANY informed to any effects of such occurrence with fully quantified and detailed supporting data provided on a weekly basis, summarising all effects to date.

4. PROJECT MANAGEMENT

- 4.1. As a minimum, the CONTRACTOR is responsible for providing the following project management functions:
 - a. Overall management of the CONTRACT;
 - b. Sufficient numbers of suitably qualified and skilled personnel with relevant experience;
 - c. Suitable facilities;
 - d. All necessary systems, procedures and methods;
 - e. All other resources whatsoever required for the proper performance and completion of the CONTRACT;
 - f. Management of the CONTRACT from a single location by one integrated organisation.

Generally, the CONTRACTOR's project management responsibilities include but are not limited to:

- a. Overall management of the CONTRACT;
- b. Quality management;
- c. Health and safety management;
- d. Environmental management;
- e. Engineering management;

Page 32 of 34



- f. Procurement and Materials management;
- g. Fabrication/manufacturing management;
- h. Information management;
- i. Regulations, certification, and permitting support to the limit of permits, licences that are normally required to be maintained by CONTRACTOR with respect to the CONTRACTOR Material, equipment and personnel irrespective of actual Work location.
- j. Project control and Contract Program;
- k. Documentation Management.

The CONTRACTOR is responsible for the proper management and execution of the CONTRACT, wherever performed, and must have appropriate representation at all sites to complete all elements of the CONTRACT in a safe, environmentally responsible and timely manner in accordance with the CONTRACT. The CONTRACTOR must keep COMPANY fully informed at all times of progress and areas of concern.

COMPANY shall assist and advise the CONTRACTOR when COMPANY, in its sole judgement, considers it appropriate in relation to the CONTRACTOR's performance of the CONTRACT, without this relieving the CONTRACTOR in any way of its responsibilities, duties and obligations under the CONTRACT.

5. WEEKLY REPORTING

5.1. CONTRACTOR shall issue fortnightly reports for the duration of the CONTRACT. Within thirty (30) days of CONTRACT signature CONTRACTOR shall submit a fortnightly report template for COMPANY approval.

The fortnightly Report shall cover as a minimum:

- a. Work Achieved and Meetings held in the Week;
- b. Work Planned and Meetings arranged for Next Week;
- c. Areas of Concern / Critical Items;
- d. Health, Safety, Environment and Quality Assurance & Control Summary;
- e. Planning and Scheduling;
- f. Subcontracts / Procurement;
- g. Technical Queries;
- h. Interface Status.

Page 33 of 34



SECTION VII CALL-OFF ORDER TEMPLATE

твс

Schedule 5 Parties' details for Notices

Fugro's details:

Fugro N.V.	
Address:	Veurse Achterweg 10 2264 SG Leidschendam The Netherlands
Attention:	Mark Heine Director Marine Division Annabelle Vos General Counsel
E-mail:	<u>M.Heine@fugro.com</u> <u>a.vos@fugro.com</u>

GMSL's details:

Global Marine Systems Limited		
Address:	Global Marine Group Ocean House 1 Winsford Way Boreham Interchange Chelmsford Essex CM2 5PD United Kingdom	
Attention:	lan Douglas	
E-mail:	ian.douglas@globamarine.group	

Page 34 of 34